### UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF MINNESOTA

In re:

Diocese of Winona-Rochester,

Case No. 18-33707

Debtor.

Chapter 11

# JOINT DISCLOSURE STATEMENT FOR FIRST AMENDED JOINT CHAPTER 11 PLAN OF REORGANIZATION PROPOSED BY THE DIOCESE OF WINONA-ROCHESTER AND THE OFFICIAL COMMITTEE OF UNSECURED CREDITORS

#### **RESTOVICH BRAUN & ASSOCIATES**

Thomas R. Braun (#350631) Christopher W. Coon (#390083) 117 East Center Street Rochester, MN 55904 (507) 216-8652 thomas@restovichlaw.com christopher@restovichlaw.com

#### **BODMAN PLC**

Robert J. Diehl, Jr. (P31264)
Brian R. Trumbauer (P57747)
Jaimee L. Witten (P70068)
6th Floor at Ford Field 1901 St. Antoine Street
Detroit, Michigan 48226
(313) 259-7777
rdiehl@bodmanlaw.com
btrumbauer@bodmanlaw.com
jwitten@bodmanlaw.com

#### STINSON, LLP

Robert T. Kugler (#194116) Edwin H. Caldie (#388930) Andrew J. Glasnovich (#398366) 50 South Sixth Street, Suite 2600 Minneapolis, MN 55402 robert.kugler@stinson.com ed.caldie@stinson.com drew.glasnovich@stinson.com

Telephone: 612-335-1500 Facsimile: 612-335-1657

ATTORNEYS FOR THE OFFICIAL COMMITTEE OF UNSECURED CREDITORS FOR THE DIOCESE OF WINONA-ROCHESTER

# ATTORNEYS FOR THE DIOCESE OF WINONA-ROCHESTER

#### **TABLE OF CONTENTS**

		<u>Page</u>
I.	INTRODUCTION	5
II.	NOTICE TO HOLDERS OF CLAIMS	6
III.	EXPLANATION OF CHAPTER 11	8
	A. OVERVIEW OF CHAPTER 11  B. CHAPTER 11 PLAN  C. CONFIRMATION OF A CHAPTER 11 PLAN  D. SUMMARY OF CLASSIFICATION AND TREATMENT OF CLAIMS	8 9
IV.	QUESTIONS AND ANSWERS REGARDING THIS DISCLOSURE STATEMEN AND THE PLAN	lΤ
	A. MANNER OF VOTING ON PLAN. B. CLAIM HOLDERS ENTITLED TO VOTE. C. CLASSES IMPAIRED AND ENTITLED TO VOTE ON THE PLAN. D. VOTE REQUIRED FOR CLASS ACCEPTANCE.	15 16 16
V.	THE DEBTOR AND ITS OPERATIONS  A. PREPETITION  B. NEED FOR REORGANIZATION  C. RESPONSE TO SEXUAL ABUSE.	16 17
VI.	THE CHAPTER 11 CASE	19
	A. THE CHAPTER 11 FILING.  B. RETENTION OF COUNSEL	19 20 20 20 21 22 22
VII.	SUMMARY OF THE PLAN	
	A. GENERAL  B. CLASSIFICATION AND TREATMENT OF CLAIMS UNDER THE PLAN  C. PRIORITY CLAIMS (CLASS 1)  D. GOVERNMENTAL UNIT CLAIMS (CLASS 2)  E. TORT CLAIMS OTHER THAN IMPAIRED UNKNOWN TORT CLAIMS (CLASS 3)  F. UNIMPAIRED UNKNOWN TORT CLAIMS (CLASS 4A)  G. IMPAIRED UNKNOWN TORT CLAIMS (CLASS 4B)  H. GENERAL UNSECURED CLAIMS (CLASS 5)  I. UNIMPAIRED ABUSE RELATED CONTINGENT CLAIMS (CLASS 6)	27 29 29 31 31

	J.	MINNESOTA DEPARTMENT OF COMMERCE (CLASS 7)	. 33
	K.	COMMUNITY BANK OF MANKATO (CLASS 8)	. 33
VIII.	MEA	NS OF IMPLEMENTATION OF THE PLAN	. 34
	A.	Trust Formation and Funding	. 34
	B.	PAYMENT OF PROFESSIONAL FEES	. 35
	C.	PAYMENTS EFFECTIVE UPON TENDER	. 36
IX.	TRUS	ST	. 36
	A.	ESTABLISHMENT OF TRUST	. 36
	B.	PURPOSE, FORMATION AND ASSETS	. 36
	C.	ALLOCATIONS WITHIN AND DISTRIBUTIONS AND PAYMENTS FROM THE TRUST	. 36
	D.	TAX MATTERS	. 37
	E.	APPOINTMENT OF THE TRUSTEE	. 37
	F.	RIGHTS AND RESPONSIBILITIES OF TRUSTEE	. 38
	G.	TRANSFERRED INSURANCE INTERESTS	. 38
	H.	SPECIAL DISTRIBUTION CONDITIONS	. 42
	I.	INVESTMENT POWERS; PERMITTED CASH EXPENDITURES	. 42
	J.	REGISTRY OF BENEFICIAL INTERESTS	. 42
	K.	NON-TRANSFERABILITY OF INTERESTS	. 42
	L.	TERMINATION	. 42
	M.	IMMUNITY; LIABILITY; INDEMNIFICATION	. 42
	N.	TREATMENT OF TORT CLAIMS	. 44
	O.	OBJECTIONS AND LITIGATION AFTER THE EFFECTIVE DATE	. 49
	P.	CLAIM WITHDRAWAL	. 49
X.	SETT	LING INSURERS	. 49
	A.	SETTLING INSURER SETTLEMENT AGREEMENTError! Bookmark not defin	red.
	B.	SALE FREE AND CLEAR OF INTERESTS OF SETTLING INSURER POLICIES	. 49
	C.	RESOLUTION OF CLAIMS INVOLVING SETTLING INSURERS	. 50
	D.	THE SETTLING OF INSURERS' PAYMENTS	. 50
	E.	JUDGMENT REDUCTION	. 50
	F.	FURTHER ASSURANCES; NON-MATERIAL MODIFICATIONS	. 51
	G.	INDEMNIFICATION OBLIGATIONS	. 51
	H.	WAIVER/ CONSENT	
	I.	DEBTOR WAIVER AND RELEASE OF CLAIMS	. 52
	J.	SUPPLEMENTAL SETTLING INSURER INJUNCTION	. 52
XI.	NON-	SETTLING INSURERS	. 53
	A.	PRESERVATION OF RIGHTS AND OBLIGATIONS	. 53
	B.	ESTIMATIONS/ASSESSMENTS ARE NOT BINDING	. 54
	C.	RIGHTS UNDER LMI/INTERSTATE SETTLEMENT AGREEMENT	
	D.	THE PLAN IS NEUTRAL AS TO NON-SETTLING INSURER POLICIES	
	E.	THE DIOCESE'S OBLIGATIONS SURVIVE	
	F.	TRUST POWERS WITH RESPECT TO TORT CLAIMS AND NON-SETTLING INSURERS	55
XII	INSII	RANCE POLICIES	56

	A.	CONTINUATION OF INSURANCE POLICIES	56
	B.	THE PLAN IS NEUTRAL AS TTO NON-IMPLICATED INSURANCE POLICIES	56
XIII.	PROC	EDURES FOR GENERAL CLAIMS ADMINISTRATION	57
	A.	RESERVATION OF RIGHTS TO OBJECT TO NON-TORT CLAIMS	57
	B.	OBJECTIONS TO NON-TORT CLAIMS	57
	C.	DETERMINATION OF CLAIMS	
	D.	NO DISTRIBUTIONS PENDING ALLOWANCE	58
	E.	CLAIM ESTIMATION	58
XIV.	DISTR	IBUTIONS UNDER THE PLAN	58
	A.	PAYMENT DATE	58
	B.	UNDELIVERABLE DISTRIBUTIONS	58
	C.	Setoffs	
	D.	No Interest On Claims	
	E.	WITHHOLDING TAXES	59
XV.	COND	ITIONS TO EFFECTIVE DATE	59
	A.	CONDITIONS TO OCCURRENCE OF EFFECTIVE DATE	59
	B.	NOTICE OF EFFECTIVE DATE	59
	C.	EFFECT OF NON-OCCURRENCE OF CONDITIONS	60
XVI.	EFFEC	CT OF PLAN CONFIRMATION	60
	A.	DISSOLUTION OF COMMITTEE	60
	B.	DISCHARGE AND INJUNCTION	60
	C.	CHANNELING INJUNCTION	61
	D.	SUPPLEMENTAL SETTLING INSURER INJUNCTION	
	E.	EXCULPATION AND LIMITATION OF LIABILITY OF EXCULPATED PARTIES	
	F.	TIMING	
	G.	No Bar on Certain Claims	64
XVII.	CHILI	PROTECTION PROTOCOLS	64
XVIII.	THE R	EORGANIZED DEBTOR	64
	A.	CONTINUED CORPORATE EXISTENCE	64
	B.	VESTING OF ASSETS	64
	C.	IDENTIFY OF OFFICERS OF REORGANIZED DEBTOR	65
	D.	FURTHER AUTHORIZATION	65
XIX.	CERT	AIN OTHER GENERAL PROVISIONS OF THE PLAN	65
	A.	CAUSES OF ACTION/AVOIDANCE ACTIONS	65
	B.	RETENTION OF JURISDICTION BY THE BANKRUPTCY COURT	
	C.	MODIFICATION OF THE PLAN	
	D.	SEVERABILITY	67
	E.	SECTION 1146 EXEMPTION	67
	F	MANAGEMENT OF THE REORGANIZED DEBTOR	68

# Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 5 of 222

XX.	CONFIRMATION PROCEDURES		
	A. B.	SOLICITATION OF VOTES; ACCEPTANCE	
	C.	BEST INTERESTS OF CREDITORS TEST	
	D.	Feasibility	
	E.	CRAM DOWN	71
XXI.	ALT	ERNATIVES TO CONFIRMATION AND	
		SUMMATION OF THE PLAN	71
	A.	ALTERNATIVE CHAPTER 11 PLANS	71
	B.	DISMISSAL OF THE CHAPTER 11 CASE	72
	C.	CONVERSION TO CHAPTER 7 NOT A VIABLE ALTERNATIVE	72
XXII.	CER'	TAIN FEDERAL INCOME TAX CONSIDERATIONS	72
	A.	THE TRUST	73
	B.	FEDERAL INCOME TAX CONSEQUENCES TO HOLDERS OF CLAIMS	
XXIII.	. VOT	ING INSTRUCTIONS	74
XXIV	. CON	CLUSION	75
Exhil	oit A:	First Amended Joint Chapter 11 Plan of Reorganization of th of Winona-Rochester (Filed as a Separate Document)	e Diocese
Exhil	oit B:	Order Approving the Disclosure Statement (Proposed)	
Exhil	oit C:	Liquidation Analysis	
Exhil	oit D:	LMI/Interstate Settlement Agreement	

#### DISCLOSURE STATEMENT<sup>1</sup>

On November 30, 2018 (the "Petition Date"), the Diocese of Winona-Rochester (the "Debtor" or the "Diocese") filed a voluntary Chapter 11 petition with the United States Bankruptcy Court for the District of Minnesota (the "Bankruptcy Court"). Since the Petition Date, the Debtor has remained in possession of its assets and has continued to own, operate and manage its affairs pending the approval of a plan of reorganization in accordance with the provisions of Title 11 of the United States Code (as amended, the "Bankruptcy Code"). The Debtor and the Official Committee of Unsecured Creditors (the "Committee" and together with the Debtor, the "Plan Proponents") seek confirmation of the First Amended Joint Chapter 11 Plan of Reorganization (as it may be amended or modified, the "Plan").

Under § 1125 of the Bankruptcy Code, the Plan Proponents now submit this Disclosure Statement (the "**Disclosure Statement**") in connection with the Plan.

# I. INTRODUCTION

The Debtor and the Committee provide this Disclosure Statement to all of the Debtor's known creditors and other parties in interest in order to provide adequate information to enable them to make an informed decision on whether to accept or reject the Plan. All holders of Claims are hereby advised and encouraged to read this Disclosure Statement and the Plan in their entirety before voting to accept or reject the Plan.

The Plan summary and statements made in this Disclosure Statement are qualified in their entirety by reference to the Plan (a copy of which accompanies this Disclosure Statement as **Exhibit A**).

THE DEBTOR AND THE COMMITTEE BELIEVE THAT THE PLAN IS IN THE BEST INTERESTS OF, AND PROVIDES THE HIGHEST AND MOST EXPEDITIOUS RECOVERIES TO HOLDERS OF, ALL CLAIMS AGAINST THE DEBTOR. ALL HOLDERS

<sup>&</sup>lt;sup>1</sup> Capitalized terms not otherwise defined in this Disclosure Statement have the meanings and definitions assigned to them in the Plan.

Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 7 of 222

OF CLAIMS ENTITLED TO VOTE TO ACCEPT OR REJECT THE PLAN ARE URGED TO VOTE IN FAVOR OF THE PLAN.

VOTING INSTRUCTIONS ARE CONTAINED IN THE ATTACHED DISCLOSURE STATEMENT ORDER. IN ADDITION, THE SOLICITATION PACKAGE ACCOMPANYING EACH OF THE BALLOTS CONTAINS APPLICABLE VOTING INSTRUCTIONS. TO BE COUNTED, YOUR BALLOT MUST BE PROPERLY COMPLETED, EXECUTED AND ACTUALLY RECEIVED BY THE DEBTOR BY 5:00 P.M. (PREVAILING CENTRAL TIME), ON \_\_\_\_\_\_, 2021 (THE "VOTING DEADLINE").

The Bankruptcy Court's approval of the Disclosure Statement does not constitute a recommendation by the Court either for or against the Plan. No statements or information concerning the Plan and the transactions contemplated thereby have been authorized, other than the statements and information set forth in this Disclosure Statement. All other statements regarding the Plan and the transactions contemplated, whether written or oral, are unauthorized.

The Bankruptcy Court has scheduled a hearing to consider confirmation of the Plan for 2021 at 9:30 A.M. (Prevailing Central Time) (the "Confirmation Hearing") at the United States Bankruptcy Court, 300 South Fourth Street, Minneapolis, MN. This hearing may be adjourned from time to time, including without further notice other than by announcement in the Bankruptcy Court on the scheduled date of such hearing. At that hearing, the Bankruptcy Court will consider whether the Plan satisfies the various requirements of the Bankruptcy Code. The Bankruptcy Court will then also receive and consider a ballot report prepared by the Debtor concerning the votes for acceptance or rejection of the Plan by the parties entitled to vote.

### II. NOTICE TO HOLDERS OF CLAIMS

The purpose of this Disclosure Statement is to enable you, as a creditor whose Claim is impaired under the Plan, to make an informed decision in exercising your right to accept or reject the Plan.

THIS DISCLOSURE STATEMENT CONTAINS IMPORTANT INFORMATION THAT MAY BEAR ON YOUR DECISION TO VOTE TO ACCEPT OR REJECT THE PLAN. PLEASE READ THIS DOCUMENT AND ALL EXHIBITS WITH CARE.

PLAN SUMMARIES AND STATEMENTS MADE IN THIS DISCLOSURE STATEMENT ARE QUALIFIED IN THEIR ENTIRETY BY REFERENCE TO THE PLAN. IN THE EVENT OF ANY CONFLICT BETWEEN THE DESCRIPTIONS SET FORTH IN THIS DISCLOSURE STATEMENT AND THE TERMS OF THE PLAN OR ANY OTHER APPLICABLE DOCUMENT, THE TERMS OF THE PLAN OR ANY SUCH APPLICABLE DOCUMENT SHALL GOVERN. UNLESS OTHERWISE SPECIFIED, THE STATEMENTS CONTAINED IN THIS DISCLOSURE STATEMENT ARE MADE ONLY AS OF THE DATE HEREOF AND THERE CAN BE NO ASSURANCE THAT THE STATEMENTS CONTAINED HEREIN WILL BE CORRECT AT ANY LATER DATE. THIS DISCLOSURE STATEMENT DOES NOT REFLECT EVENTS THAT MAY

OCCUR AFTER THAT DATE AND MAY HAVE A MATERIAL IMPACT ON THE INFORMATION CONTAINED IN THIS DISCLOSURE STATEMENT.

NO REPRESENTATIONS CONCERNING THE DEBTOR, THE ESTIMATED VALUE OF THE DEBTOR'S PROPERTY AND/OR THE ESTIMATED ASSETS TO BE GENERATED FROM THE LIQUIDATION OF THE DEBTOR'S ASSETS, ARE AUTHORIZED BY THE PLAN PROPONENTS OTHER THAN AS SET FORTH IN THIS DISCLOSURE STATEMENT. ANY REPRESENTATIONS OR INDUCEMENTS MADE TO SECURE YOUR ACCEPTANCE THAT ARE NOT CONTAINED IN THIS DISCLOSURE STATEMENT, SHOULD NOT BE RELIED UPON BY YOU IN CASTING YOUR VOTE WITH RESPECT TO THE PLAN.

THIS DOCUMENT WAS COMPILED FROM INFORMATION OBTAINED BY THE PLAN PROPONENTS FROM NUMEROUS SOURCES BELIEVED TO BE ACCURATE TO THE BEST OF THE PLAN PROPONENTS' KNOWLEDGE, INFORMATION AND BELIEF. HOWEVER, THE PLAN PROPONENTS AND THEIR ADVISORS DO NOT REPRESENT OR WARRANT THAT THIS DISCLOSURE STATEMENT IS COMPLETE OR THAT THE INFORMATION CONTAINED HEREIN IS FREE FROM ANY INACCURACY OR OMISSION.

THIS DISCLOSURE STATEMENT HAS BEEN PREPARED IN ACCORDANCE WITH SECTION 1125 OF THE BANKRUPTCY CODE, AND RULE 3016(b) OF THE FEDERAL RULES OF BANKRUPTCY PROCEDURE (THE "BANKRUPTCY RULES") AND NOT NECESSARILY IN ACCORDANCE WITH FEDERAL OR STATE SECURITIES LAW OR OTHER NON-BANKRUPTCY LAW.

NOTHING IN THIS DISCLOSURE STATEMENT IS OR SHALL BE DEEMED TO BE AN ADMISSION OR A DECLARATION AGAINST INTEREST BY THE DEBTOR FOR PURPOSES OF ANY EXISTING OR FUTURE LITIGATION. AS TO CONTESTED MATTERS, ADVERSARY PROCEEDINGS AND OTHER ACTIONS OR THREATENED ACTIONS, THIS DISCLOSURE STATEMENT SHALL NOT CONSTITUTE OR BE CONSTRUED AS AN ADMISSION OF ANY FACT OR LIABILITY, STIPULATION, OR WAIVER, BUT RATHER AS A STATEMENT MADE IN SETTLEMENT NEGOTIATIONS. THIS DISCLOSURE STATEMENT SHALL NOT BE ADMISSIBLE IN ANY NON-BANKRUPTCY PROCEEDING NOR SHALL IT BE CONSTRUED TO BE CONCLUSIVE ADVICE ON THE TAX, SECURITIES, OR OTHER LEGAL EFFECTS OF THE PLAN AS TO HOLDERS OF CLAIMS AGAINST THE DEBTOR IN THIS CASE.

ALTHOUGH THE DEBTOR'S PROFESSIONALS AND THE COMMITTEE'S PROFESSIONALS HAVE ASSISTED IN THE PREPARATION OF THIS DISCLOSURE STATEMENT BASED ON THE FACTUAL INFORMATION AND ASSUMPTIONS RESPECTING THE FINANCIAL, BUSINESS AND ACCOUNTING DATA PROVIDED BY THE DEBTOR, THE DEBTOR'S PROFESSIONALS AND THE COMMITTEE'S PROFESSIONALS HAVE NOT INDEPENDENTLY VERIFIED THE INFORMATION SET FORTH IN THIS DISCLOSURE STATEMENT AND MAKE NO REPRESENTATIONS OR WARRANTIES AS TO SUCH INFORMATION. SUCH

# PROFESSIONALS DO NOT REPRESENT OR WARRANT THAT THIS DISCLOSURE STATEMENT IS COMPLETE OR IS FREE FROM ANY INACCURACY OR OMISSION.

Each holder of a Claim entitled to vote to accept or reject the Plan should read this Disclosure Statement and the Plan (including all Exhibits and Schedules to the Plan and Disclosure Statement) in their entirety before voting. No solicitation of votes to accept or reject the Plan may be made except pursuant to this Disclosure Statement and Section 1125 of the Bankruptcy Code. Except for the Debtor and certain of the Professionals it has retained, no person has been authorized to use or promulgate any information concerning the Debtor, its business, or the Plan other than the information contained in this Disclosure Statement and if given or made, such information may not be relied upon as having been authorized by the Debtor. You should not rely on any information relating to the Debtor, its business or the Plan other than that contained in this Disclosure Statement and the Exhibits hereto.

After carefully reviewing this Disclosure Statement, including the attached Exhibits, please indicate your acceptance or rejection of the Plan by voting in favor of or against the Plan pursuant to the procedures set forth in the Solicitation Package.

You will be bound by the Plan if it is confirmed by the Bankruptcy Court, even if you do not vote to accept the Plan, or if you are the holder of an unimpaired Claim.

# THE DEBTOR AND COMMITTEE URGE ALL HOLDERS OF IMPAIRED CLAIMS TO ACCEPT THE PLAN.

# III. EXPLANATION OF CHAPTER 11

**A.** OVERVIEW OF CHAPTER 11. Chapter 11 is the principal chapter of the Bankruptcy Code pursuant to which a debtor may reorganize its operations in an orderly fashion for the benefit of its creditors, stockholders, and other parties in interest.

The commencement of a Chapter 11 case creates an estate comprising all the legal and equitable interests of the Debtor as of the date the petition is filed. Sections 1101, 1107, and 1108 of the Bankruptcy Code provide that a debtor may continue to operate and remain in possession of its property as a "Debtor-in-Possession" unless the Bankruptcy Court orders the appointment of a trustee. In this case, the Debtor remains as the Debtor-in-Possession.

The filing of a petition under the Bankruptcy Code triggers the automatic stay provisions of the Bankruptcy Code. Section 362 of the Bankruptcy Code provides, among other things, for an automatic stay of all attempts by creditors or other third parties to collect on prepetition claims against a debtor or otherwise interfere with its property or operations. Exempted from the automatic stay are governmental authorities seeking to exercise regulatory or policing powers. Except as otherwise ordered by the Bankruptcy Court, the automatic stay remains in full force and effect until the effective date of a confirmed Chapter 11 plan.

**B.** <u>CHAPTER 11 PLAN</u>. The formulation of a Chapter 11 plan is the principal purpose of a Chapter 11 case. A Chapter 11 plan sets forth the means for satisfying the holders of claims against and interests in a debtor's estate. A Chapter 11 plan may provide anything from a complex

restructuring of a debtor's operations and its related obligations to a simple liquidation of a debtor's assets. In either event, upon confirmation of a plan, it becomes binding on a debtor and all of its creditors, and the prior obligations owed by a debtor to such parties are compromised and exchanged for the obligations specified in the plan. The Plan incorporates a compromise reached among the Debtor and the Committee that the Debtor and the Committee believe provides a fair and equitable allocation of the Debtor's assets that will be distributed to creditors and treatment of all Claims against the Debtor.

After a Chapter 11 plan has been filed, the holders of impaired claims against a debtor are permitted to vote to accept or reject the plan. Before soliciting acceptances of a proposed plan, Section 1125 of the Bankruptcy Code requires the Debtor to prepare and file a disclosure statement containing adequate information of a kind, and in sufficient detail, to enable a hypothetical reasonable investor to make an informed judgment about the Plan. This Disclosure Statement is presented to holders of Claims against the Debtor to satisfy the requirements of Section 1125 of the Bankruptcy Code in connection with the Debtor's solicitation of votes on the Plan.

C. <u>CONFIRMATION OF A CHAPTER 11 PLAN</u>. If all classes of Claims accept the Plan, the Bankruptcy Court may confirm the Plan if the Bankruptcy Court independently determines that the requirements of Section 1129(a) of the Bankruptcy Code have been satisfied. **The Debtor and the Committee believe that the Plan satisfies all the applicable requirements of Section 1129(a) of the Bankruptcy Code**.

Chapter 11 of the Bankruptcy Code does not require that each holder of a claim or, where applicable, interest in a particular class vote in favor of a plan for the Bankruptcy Court to determine that such class has accepted the plan. Rather, a class of claims or interests will have accepted a plan if the Bankruptcy Court determines that the plan has been accepted by more than a majority in number and at least two-thirds in amount of those claims actually voting in such class. Only the holders of allowed Claims who actually vote will be counted as either accepting or rejecting the Plan.

Furthermore, classes that are to receive no distribution under a plan are conclusively deemed to have rejected such plan. Accordingly, acceptances of a plan will generally be solicited only from those persons who hold claims or equity interests in an impaired class.

Classes 3 (Tort Claims other than Impaired Unknown Tort Claims) and 4B (Impaired Unknown Tort Claims) are impaired under the Plan and entitled to vote on the Plan.

Classes 1 (Priority Claims), 2 (Governmental Unit Claims), 4A (Unimpaired Unknown Tort Claims), 5 (General Unsecured Claims), 6 (Unimpaired Abuse Related Contingent Claims), 7 (Secured Claim of Minnesota Department of Commerce) and 8 (Secured Claim of Community Bank of Mankato), are deemed unimpaired under the Plan and are deemed to accept the Plan.

In general, a Bankruptcy Court also may confirm a Chapter 11 plan even though fewer than all the classes of impaired claims against and equity interests in a debtor accept such plan. For a Chapter 11 plan to be confirmed, despite its rejection by a class of impaired claims or equity interests, a plan must be accepted by at least one class of impaired claims (determined without

counting the vote of insiders) and the proponent of the plan must show, among other things, that the plan does not "discriminate unfairly" and that the plan is "fair and equitable" with respect to each impaired class of claims or equity interests that have not accepted the plan.

Under Section 1129(b) of the Bankruptcy Code, a plan is "fair and equitable" as to a rejecting class of claims or equity interests if, among other things, the plan provides: (a) with respect to secured claims, that each such holder will receive or retain on account of its claim property that has a value, as of the effective date of the plan, in an amount equal to the allowed amount of such claim or such other treatment as accepted by the holder of such claim; and (b) with respect to unsecured claims and equity interests, that the holder of any claim or equity interest that is junior to the claims or equity interests of such class will not receive on account of such junior claim or equity interest any property at all unless the senior class is paid in full.

A plan does not "discriminate unfairly" against a rejecting class of claims or equity interests if the relative value of the recovery of such class under the plan does not differ materially from that of any class (or classes) of similarly situated claims or equity interests. The Debtor and the Committee believe that the Plan will satisfy the foregoing requirements as to any rejecting Class of Claims, and can therefore be confirmed despite any such rejection by any Class.

D. <u>SUMMARY OF CLASSIFICATION AND TREATMENT OF CLAIMS</u>. Detailed elsewhere in this Disclosure Statement is a description of the technical aspects of the classification of Claims, the relative allocations of distributions to holders of such Claims, the methodology as to how such distributions are to be distributed, the risks inherent in the proposed Plan, and the applicable bankruptcy and tax consequences of the Plan. However, the Plan Proponents believe that a broad overview of what, in their opinion, the Debtor and creditors are likely to receive under the Plan, will be helpful for your consideration of whether you wish to accept or reject the Plan.

The following is a summary of the classification of all Claims under the Plan and the proposed treatment of each such Class under the Plan. This summary is qualified in its entirety by reference to the Plan:

CLASS	DESCRIPTION	IMPAIRMENT	VOTING
1	Priority Claims	Unimpaired	No
2	Governmental Unit Claims	Unimpaired	No
3	Tort Claims Other Than Impaired Unknown Tort Claims	Impaired	Yes
4A	Unimpaired Unknown Tort Claims	Unimpaired	No
4B	Impaired Unknown Tort Claims	Impaired	Yes
5	General Unsecured Claims	Unimpaired	No
6	Unimpaired Abuse Related Contingent Claims	Unimpaired	No
7	Secured Claim of Minnesota Department of Commerce	Unimpaired	No
8	Secured Claim of Community Bank of Mankato	Unimpaired	No

As discussed in the Liquidation Analysis attached as **Exhibit C**, the Debtor estimates that recoveries for holders of Classes 3 and 4B will be greater than in liquidation under Chapter 7 of the Bankruptcy Code because the total amount of property available for distribution is greater under the Plan than in liquidation under Chapter 7. In addition, the Debtor believes that theoretical distributions under a Chapter 7 case would likely be delayed due to the time it will take a Chapter 7 trustee to assess the Debtor's assets, review and analyze claims, and evaluate and litigate claims against third parties. Holders of allowed Claims entitled to vote to accept or reject the Plan should review the Liquidation Analysis (including all footnotes thereto and documents referenced therein) in assessing whether to vote to accept or reject the Plan.

# IV. <u>QUESTIONS AND ANSWERS REGARDING THIS</u> <u>DISCLOSURE STATEMENT AND THE PLAN</u>

#### Why is the Debtor sending me this Disclosure Statement?

The Debtor and the Committee are seeking to obtain Bankruptcy Court approval of the Plan. Prior to soliciting acceptances of the Plan, Section 1125 of the Bankruptcy Code requires the preparation and approval of a disclosure statement containing adequate information of a kind, and in sufficient detail, to enable a hypothetical reasonable investor to make an informed judgment regarding acceptance of the Plan. This Disclosure Statement is being submitted in accordance with such requirements.

The Plan is based on two settlements. One settlement is among the Diocese, the Catholic Entities, and certain Settling Insurers and amounts to \$6,500,000.00. This settlement is evidenced by the LMI/Interstate Settlement Agreement, which is subject to Bankruptcy Court approval. In general terms, the LMI/Interstate Settlement Agreement provides for (a) the buy back by certain Underwriters at Lloyd's, London, and certain London Market Companies (as defined in the LMI/Interstate Settlement Agreement) (collectively, "LMI") and Interstate Fire & Casualty Company ("Interstate" and together with LMI, "LMI/Interstate") of their policies from the Catholic Entities and (b) injunctions which prohibit, among others, Tort Claimants from suing LMI/Interstate. LMI/Interstate constitute "Settling Insurers" under the Plan.

The second settlement is among the Diocese, certain Catholic Entities, and the Committee and amounts to \$13,560,000.00 (minus (i) amounts paid by the Debtor for certain administrative expenses paid after February 29, 2020 and (ii) counseling expenses for Tort Claimants paid by the Debtor) to be paid within five days after the Effective Date of the Plan (which payment shall constitute substantial consummation of the Plan), plus an additional \$7,552,500 to be paid as soon as practical but in no event more than 12 months after the Effective Date. \$112,500 of the \$7,552,500 payment will be contributed by the following Catholic Entities: Lourdes High School of Rochester, Inc. a/k/a Lourdes Catholic High School and Rochester Catholic Schools. All of the settlement amounts will be payable to the Trust set up through the Plan and Disclosure Statement process. The funds will be allocated pursuant to the Trust Distribution Protocols attached as **Exhibit D** to the Plan.

Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 13 of 222

The Plan also contains provisions related to the rights of Tort Claimants to pursue litigation with respect to Claims insured by, and for the Trust to pursue litigation against, Non-Settling Insurers to collect additional money. The procedures, protocols, and limitations related to Non-Settling Insurer litigation is provided in Section XI below.

#### What happens to my recovery if the Plan is not confirmed, or does not go effective?

If the Plan is not confirmed, the Debtor believes it is unlikely that the Debtor will be able to reorganize. The Plan memorializes a comprehensive settlement between the Debtor and the Committee, which allocates a substantial portion of the Debtor's assets to the Trust that will be established for the benefit of Tort Claimants. In addition, confirmation of the Plan is necessary to effectuate the settlement with LMI/Interstate that will be used to fund the Plan and the Trust created pursuant to the Plan for the benefit of Tort Claimants. If the Plan is not confirmed in a timely manner, it is unclear whether the transactions contemplated in the Plan could be implemented and what holders of Claims would ultimately receive on account of their Claims. It is possible that any alternative may provide holders of Claims with less than they would have received pursuant to the Plan on account of, among other things, the cost of negotiating, drafting and potentially litigating an alternative Plan, as well as complex litigation regarding Tort Claims and the potential loss of funds from LMI/Interstate. Moreover, non-confirmation of the Plan may result in dismissal of this case in its entirety. For a more detailed description of the consequences of this scenario, see "Best Interests of Creditors Test," which is in Section XX.C below, and the Liquidation Analysis attached as Exhibit C to this Disclosure Statement.

If the Plan provides that I get a distribution, do I get it upon confirmation or when the Plan goes effective, and what do you mean when you refer to ''Confirmation,'' ''Effective Date'' and ''Consummation?''<sup>2</sup>

"Confirmation" of the Plan refers to the approval of the Plan by the Bankruptcy Court. Confirmation of the Plan does not guarantee that you will receive the distribution contemplated under the Plan. After Confirmation of the Plan by the Bankruptcy Court, there are conditions that need to be satisfied or waived so that the Plan can be consummated and become effective. References to the "Effective Date" means the date that all conditions to the effectiveness of the Plan have been satisfied or waived. Distributions will only be made on the Effective Date or as soon as practicable thereafter, based on, among other things, the amount of cash available to satisfy Claims, the time that the Tort Claims Reviewer requires to complete his analysis of certain Tort Claims, the amount of Claims outstanding against the Debtor, and the Trustee's business judgment. The Tort Claims Reviewer analyzes the Tort Claims pursuant to the Trust Distribution Protocols. The Trust will distribute the funds after review of the Tort Claims is complete pursuant to the terms of the Plan, the Confirmation Order, the Trust Distribution Protocols, the Trust Agreement, the applicable releases, and any other applicable Plan documents.

#### Where is the cash required to fund the Plan coming from?

The cash required to fund the Trust that will pay holders of Class 3 and 4B Claims, will come from (i) \$13,560,000.00 of cash from the Debtor (minus (A) amounts paid by the Debtor for

<sup>&</sup>lt;sup>2</sup> The descriptions' capitalized terms in response to this question are qualified in their entirety by reference to the definitions in the Plan.

certain administrative expenses paid after February 29, 2020 and (B) counseling expenses for Tort Claimants paid by the Debtor) to be paid within five days after the Effective Date of the Plan (which payment shall constitute substantial consummation of the Plan), plus an additional \$7,552,500 of cash from the Debtor and certain Catholic Entities to be paid within 12 months after the Effective Date, (ii) \$6,500,000.00 of cash from LMI/Interstate, and (iii) the Transferred Insurance Interests. The Debtor currently has sufficient funds on hand to make the initial cash payment required of it by the Plan, and the Plan provides the Debtor up to 12 months to make the second cash payment of \$7,552,500, which the Debtor believes is sufficient time to raise such funds via the sale or financing of certain assets and contributions from certain non-Diocesan entity resources.

In consideration for these contributions LMI/Interstate and the Protected Parties (including the foregoing Catholic Entities) will be beneficiaries of the Channeling Injunction. In addition, LMI/Interstate will be beneficiaries of the Supplemental Settling Insurer Injunction contained in the Plan. Tort Claimants must sign a release of the Settling Insurers as a prerequisite to receiving any distributions through the Plan and from the Trust. In addition, Tort Claimants must sign a release of any Tort Claims that do not implicate coverage under Non-Settling Insurer Policies with respect to the Protected Parties and the Settling Insurers, as a prerequisite to receiving any distributions through the Plan and from the Trust.

#### Will there be any releases granted to parties other than the Debtor as part of the Plan?

Yes. *See* "Exculpation and Limitation of Liability," which is in section XVI.E below. A Tort Claimant's distribution is conditioned upon the Tort Claimant signing a release of the Settling Insurers, and signing a release of any Tort Claims that do not implicate coverage under Non-Settling Insurer Policies with respect to the Protected Parties. The Ballots for the Tort Claimants contain the releases.

#### Will there be any injunctions entered pursuant to the Plan?

Yes. The Protected Parties obtain the "Discharge and Injunction," which is in section XVI.B below and the "Channeling Injunction," which is in section XVI.C below. The Settling Insurers receive the benefit of the "Channeling Injunction" and also the "Supplemental Settling Insurer Injunction," which is in section XVI.D below. These injunctions will require holders of Tort Claims to seek an exclusive remedy provided in the Plan and preclude claimants from pursuing Claims against the parties protected by such injunctions, to the extent of such injunctions, whether or not such claimants receive a distribution under the Plan. The Tort Claimants will specifically reserve, and do not release, any Tort Claims against the Protected Parties that implicate insurance coverage under any Non-Settling Insurer Policy, but recourse is limited to the proceeds of Non-Settling Insurer Policies and all other damages (including extra-contractual damages), awards, judgments in excess of policy limits, penalties, punitive damages and attorney's fees and costs that may be recoverable against any Non-Settling Insurers because of their conduct concerning insurance coverage for, or defense or settlement of, any Tort Claim, and any such judgments or awards will be turned over to the Trust for handling in accordance with Sections 6.14(i) and (j) of the Plan. The Class 3 and Class 4B Claims will not be released or enjoined as against the Diocese, the Reorganized Debtor, or any other Protected Party for any Abuse that may be covered under Non-Settling Insurer Policies until such claims are settled with the Diocese, the

Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 15 of 222

Reorganized Debtor, or any other Protected Party, as applicable, and such Non-Settling Insurer or are fully adjudicated, resolved and subject to Final Order, but recourse is limited as described above.

#### How do I vote for or against the Plan?

This Disclosure Statement, accompanied by a Ballot or Ballots to be used for voting on the Plan, is being distributed to the holders of Claims entitled to vote on the Plan. If you are a holder of a Claim in Classes 3 or 4B (collectively, the "Voting Classes"), you may vote for or against the Plan by completing the Ballot and returning it as set forth in the Solicitation Packages which will be mailed out. See "Voting Instructions," which is in section XXIII below.

#### What is the deadline to vote on the Plan?

All Ballots must be actually received by the Debtor no later than 5:00 p.m. (Prevailing Central Time) on [•], 2021 (the "Voting Deadline").

#### Why is the Bankruptcy Court holding a Confirmation Hearing?

Section 1128(a) of the Bankruptcy Code requires the Bankruptcy Court to hold a hearing on Confirmation of the Plan. Section 1128(b) of the Bankruptcy Code provides that any party in interest may object to Confirmation of the Plan.

#### When is the Confirmation Hearing scheduled to occur?

The Bankruptcy Court has scheduled the Confirmation Hearing for [•], 2021 to take place at 9:30 a.m. (Prevailing Central Time) before the Honorable Judge Robert J. Kressel, United States Bankruptcy Judge, in the United States Bankruptcy Court, Courtroom [•], 300 South Fourth Street, Minneapolis, MN. The Confirmation Hearing may be adjourned from time to time, including without further notice except for an announcement of the adjourned date made at the Confirmation Hearing or any adjournment thereof. Objections to Confirmation of the Plan must be filed and served on the Debtor and certain other parties, by no later than [•], 2021 at 5:00 p.m. (Prevailing Central Time) in accordance with the notice of the Confirmation Hearing that accompanies this Disclosure Statement. Unless an objection to Confirmation of the Plan is timely served and filed in compliance with the Disclosure Statement Order, which is attached to this Disclosure Statement as **Exhibit B**, it might not be considered by the Bankruptcy Court.

#### What is the purpose of the Confirmation Hearing?

The confirmation of a plan of reorganization is the principal objective of a Chapter 11 case. The confirmation of a plan of reorganization by the Bankruptcy Court binds a debtor, any person acquiring property under the plan of reorganization, any creditor or equity interest holder of a debtor and any other person or entity as may be ordered by the Bankruptcy Court in accordance with the applicable provisions of the Bankruptcy Code.

#### What role does the Bankruptcy Court play after the Confirmation Hearing?

After the Plan is confirmed, the Bankruptcy Court will still have exclusive jurisdiction over all matters arising out of, or related to, the case and the Plan. A detailed description of the Bankruptcy Court's post-confirmation jurisdiction is provided in Section 16.1 of the Plan.

#### Do the Debtor and the Committee recommend voting in favor of the Plan?

Yes. The Debtor and the Committee recommend voting for the Plan because the Plan provides for a larger distribution, at the estimated Effective Date of the Plan, to the Debtor's unsecured creditors, including holders of Tort Claims, than would otherwise result from liquidation or any other reasonably available alternative. Accordingly, the Debtor and the Committee recommend that holders of Claims in Voting Classes support Confirmation of the Plan and vote to accept the Plan.

**A.** MANNER OF VOTING ON PLAN. Before voting, this Disclosure Statement, as well as the Plan, should be read in their entirety. You should only use the Ballot sent to you in the Solicitation Package to cast your vote for or against the Plan.

Ballots must be completed, dated, signed and returned under the procedures set forth in the Solicitation Package.

**B.** CLAIM HOLDERS ENTITLED TO VOTE. Under the Bankruptcy Code, any holder of a claim in a class that is "impaired" under a plan is entitled to vote to accept or reject the plan, unless such class of claims neither receives nor retains any property under the plan (in which case such class is deemed to have rejected the plan). Bankruptcy Code §1124 provides generally that a Claim is impaired if the legal, equitable or contractual rights of the claim are altered.

Subject to the exceptions provided below, any holder whose Claim is impaired under the Plan is entitled to vote if either (i) its Claim has been scheduled by the Debtor and such Claim is not scheduled as disputed, contingent or unliquidated, or (ii) such Claim holder has filed a Proof of Claim with respect to a Disputed Claim. Under Bankruptcy Rule 3018(a), Class 3 Claims (Tort Claims Other Than Impaired Unknown Tort Claims) will be estimated at \$1.00 for voting purposes only. The actual amount payable on account of Class 3 Claims will be determined under the Trust Distribution Plan.

A holder of a Disputed Claim is not entitled to vote on the Plan unless such Claim is temporarily allowed by the Debtor or by an order of the Bankruptcy Court in an estimated amount that it deems proper for the purpose of voting to accept or reject the Plan. In other words, only holders of allowed Class 3 Claims (Tort Claims Other Than Impaired Unknown Tort Claims which are estimated for voting purposes only at \$1.00 for each Claim) and Class 4B Claims (Impaired Unknown Tort Claims, which are estimated for voting purposes only at \$1.00), may vote to accept or reject the Plan. A Claim to which an objection has been filed by the Debtor or any other party-in-interest no later than the date of the Confirmation Hearing, or a Claim (i) that is listed on the Debtor's schedules as disputed, unliquidated or contingent, and (ii) with respect to which a superseding Proof of Claim has not been filed is not an allowed Claim for voting purposes, unless the Claim is settled by agreement or the Bankruptcy Court allows the Claim (in whole or in part) by a Non-Appealable Order. Upon request of a party-in-interest, the Bankruptcy Court may

temporarily allow or estimate a Disputed Claim for purpose of voting on the Plan. Ballots cast in respect of Claims other than allowed Claims, Class 3 Claims and Class 4B Claims will not be counted. In addition, a vote may be disregarded if the Bankruptcy Court determines that the acceptance or rejection of the Plan by the claimant was not solicited or procured in good faith or in accordance with the provisions of the Bankruptcy Code.

- C. <u>CLASSES IMPAIRED AND ENTITLED TO VOTE ON THE PLAN</u>. Claim holders in Classes 3 and 4B are impaired under the Plan and are eligible, subject to the limitations set forth above, to vote to accept or reject the Plan. Any controversy as to whether any Claim or class of Claims is impaired under the Plan will, after notice of any hearing, be determined by the Bankruptcy Court.
- **D.** <u>VOTE REQUIRED FOR CLASS ACCEPTANCE</u>. Section 1126(c) of the Bankruptcy Code defines acceptance of a plan by a class of impaired Claims as acceptance by at least two-thirds (2/3) in dollar amount and more than one-half (1/2) in number of holders of allowed claims in that class who cast ballots. Class 3 Claims will be estimated at \$1.00 each for voting purposes only. As such, if at least two thirds (2/3) of the Class 3 Claimants who vote cast Ballots in favor of the Plan, Class 3 will be deemed to have accepted the Plan. The vote for Class 4B will be cast by the Unknown Claims Representative.

# V. THE DEBTOR AND ITS OPERATIONS

**A.** PREPETITION. Every Catholic entity, including the Debtor, is subject to the laws of the Catholic Church known as Canon Law, originally codified in 1917 and subsequently amended in 1983. Under Canon Law the Debtor is a "juridic person" or legal entity established by competent authority, classified as public or private. The Diocese is the ecclesiastical entity subject to the Bishop of the Diocese, currently the Most Reverend John M. Quinn, who was appointed and made Bishop of the Diocese on May 7, 2009. Bishop Quinn is responsible to govern the Diocese, following the precepts of Canon Law.

The Diocese and every parish within the geographical territory of the Diocese (the "**Region**") is a separate, public, juridic person. The Diocese was originally established as a juridic person by the Vatican in 1889. However, the Debtor is also a civil legal entity, incorporated April 21, 1911, as a religious Diocesan Corporation under Minn. Stat. §315.16. As set forth in its Articles of Incorporation, the Debtor's general purpose is to manage the temporal affairs of the Roman Catholic Church in the region, and to promote spiritual, educational and other interests, including charitable, benevolent and missionary work.

The Diocese serves a geographical area covering more than 12,000 square miles and encompassing the 20 southernmost counties of the state of Minnesota. There are 107 parishes and approximately 130,000 Catholic individuals within the Region. These individuals and parishes are served by approximately 108 priests, 354 sisters, 19 brothers and 29 deacons. The Diocese currently employs approximately 41 individuals, which includes clergy and laity.

In addition to the parishes, there are 23 Catholic schools in the Region, including four Catholic high schools, with a total enrollment of over 3,700 students, as well as a Catholic

university for women and men, St. Mary's University of Minnesota, with an undergraduate enrollment of approximately 1,000 students (collectively, the "Schools"). Various other Catholic-based social and community service organizations operate in the Region, including six Catholic nursing homes, one Catholic hospital, 106 catholic cemeteries, and the Immaculate Heart of Mary Seminary. The Region also includes nine religious communities that are maintained by separate religious institutions. Parishes, the Schools, and other separately incorporated Catholic entities within the Diocese's Region are not under the fiscal or operating control of the Diocese.

As a religious corporation, the Diocese has no significant, ongoing for-profit business activities or business income. Gross revenue for the fiscal year ending on June 30, 2020 was approximately \$5,590,000 (exclusive of restricted funds and amounts received for self-insurance premiums). The Diocese's revenue primarily derives from donations and parish assessments.

The Diocese maintains a number of departments, including the ministerial offices of Divine Worship; Faith Formation; Lay Formation & RCIA; Life, Marriage & Family; Hispanic Ministry; Youth and Young Adults (responsible for evangelization and missionary discipleship of the people they serve); Tribunal; and the Office of Catholic Schools. Other offices include the Office of Vocations, the Office of the Permanent Diaconate and the Office of Vicar for Clergy (responsible for the education and ministerial standards for seminarians, diaconate candidates and clergy). Additionally, the Diocese maintains offices of (a) Safe Environment (whose responsibilities include ensuring the safety of children and young people who have been entrusted to the Diocese's care in parishes, schools, religious education classes, and other programs); (b) Chancellor and Archives (whose responsibilities include maintaining the diocesan records in the archives); (c) Communications (whose responsibilities include communicating the mission and values of the Diocese); (d) Finance (whose responsibilities include overseeing the accounting, budgeting, and financial administration for the diocesan administrative offices); and (e) Human Resources (whose responsibilities include the administration of employment and benefit services for institutions of the Diocese).

The Diocese does not operate under the corporation sole model. Each parish ("Parish Corporation") has been separately organized and incorporated as a religious corporation under Minn. Stat. § 315.15. Each Parish Corporation owns the separate property used in the operation of the parish, normally including the church itself, administrative offices, and, in most cases, a rectory that serves as the pastor's residence. Each Parish Corporation maintains its own tax identification number and each corporation owns and manages its own property and assets, and bears responsibility for its own corporate actions. The Parish Corporations are not debtors in this Chapter 11 case and have not otherwise sought bankruptcy relief. This is also true with respect to other non-debtor Catholic entities in the Region, which have no corporate relationship with the Diocese and are incorporated as separate non-profit corporations under Chapter 317A of the Minnesota Statutes.

**B.** <u>NEED FOR REORGANIZATION</u>. Over the last several decades, some clergy members in the Catholic Church (the "Church") have violated the sacred trust placed in them by children and their families and the Church by committing acts of sexual abuse. This conduct runs contrary to the teaching and traditions of the Church. As such, the Diocese worked for more than two decades to meet the needs of alleged survivors of sexual abuse without filing for Chapter 11 reorganization. Since the 1980s, the Diocese has directed substantial resources towards providing

financial, psychological, pastoral and spiritual support to survivors. However, in May 2013, Minnesota enacted the Minnesota Child Victims' Act, Minn. Stat. § 541.073 (the "CVA"), which altered, expanded, and in some circumstances eliminated the statute of limitations applicable to civil causes of action for damages based on sexual abuse. This window expired in May 2016.

On May 29, 2013, an individual identified as *Doe 1* initiated litigation proceedings against the Diocese and the Archdiocese of Minneapolis and St. Paul in the matter of *Doe 1 v. Archdiocese* of St. Paul and Minneapolis and Diocese of Winona, Court File No. 62-CV-13-4075 (Ramsey County District Court). The Diocese and the Archdiocese eventually entered into a confidential settlement agreement with Doe 1. However, the Diocese had been served with an additional 122 lawsuits under the CVA alleging minor sexual abuse by the close of the CVA window.

Verdicts in sexual abuse cases across the country have ranged into the millions of dollars. The Diocese has insurance coverage available for some of the amounts sought by the claimants in the CVA cases, but, for some of the CVA cases pending against the Diocese, the insurance carriers have denied or otherwise sought to limit the available insurance coverage. Faced with the likely prospect of substantial adverse judgments in the remaining lawsuits, the high costs associated with litigating the remaining lawsuits, the uncertainty of insurance coverage for some claims, and the difficulty of equitably settling all the cases simultaneously, all of which would jeopardize the financial viability of the Diocese and its ability to carry on its mission, a Chapter 11 bankruptcy petition was filed on November 30, 2018.

C. RESPONSE TO SEXUAL ABUSE. In 1987, the Diocese drafted and implemented its Policies & Procedures Regarding Sexual Abuse of Minors by Priests, Deacons and Other Church Personnel, one of the first such policies in the nation designed to protect minors from clergy sexual abuse. In implementing this policy, the Diocese held workshops and training seminars aimed at ensuring that the protocols of this policy were understood and implemented in the parishes and schools within the geographical boundaries of the Diocese. This policy has been refined and updated over the years to reflect best practices and the most recent developments in order to mitigate and reduce the risk of minor sexual abuse. This policy was revised and updated in 1993, 2001, 2003, and again in 2011.

Notably, the 2003 update incorporated the findings and recommendations of the nationwide study entitled The Nature and Scope of the Problem of Sexual Abuse of Minors by Catholic Priests and Deacons in the United States, commissioned by the United States Conference of Catholic Bishops (the "USCCB"). As a response to the findings of the study, the USCCB promulgated a comprehensive national set of policies and procedures for addressing the sexual abuse of minors by clergy known as the Charter for the Protection of Children and Young People (the "Charter"). The Diocese incorporated and implemented all aspects and recommendations of the Charter in its policies and practices at that time.

Moreover, since 2002, background checks have been a requirement for all clergy, employees, and volunteers in the Diocese. In 2002, the Diocese also established the Office of Safe Environment, to ensure the safety of children and young people who have been entrusted to the Diocese's care in parishes, schools, religious education classes, and other programs and to ensure the protocols and processes detailed in the Charter are properly implemented and observed.

In 2002, the Diocese also created the Ministerial Standards Board, which is composed primarily of lay persons not in the employ of the Diocese, and which includes experts in medicine, therapy, law, canon law, and victim advocacy. This board functions as a confidential consultative body to the Bishop in discharging his responsibilities to ensure a safe environment for all minors and individuals in the Diocese.

Also in 2002, the Diocese created a new position entitled Victim Assistance Coordinator. This individual serves as a point-person who interacts with survivors and other claimants in obtaining financial and pastoral assistance and healing from the Diocese.

In October of 2014, in connection with the confidential settlement agreement reached in the Doe 1 matter, the Diocese and Plaintiff's counsel, Jeff Anderson and Associates, worked handin-hand to develop unprecedented child protection protocols and procedures (the "Child **Protection Protocols**") to ensure the protection of youth and young people going forward. Key measures included the implementation of a whistleblower protection policy for reporters of abuse, specific training sessions for mandated reporters, public disclosure of the names of clergy that are the subject of substantiated claims of minor sexual abuse, and the names of clergy deemed unsuitable for ministry under circumstances that arise, in whole or in part, out of accusations or risk of sexual abuse of a minor, policies prohibiting being alone and unsupervised with minors but for common-sense and emergency situations, and the mandatory disclosure of accusations of sexual abuse to other religious or secular institutions that may inquire for employment reasons. The Diocese also agreed to engage in on-going review of the allegations of minor sexual abuse by clergy and supplement its publication and disclosure where necessary. Since the initial publication of the names and files of priests credibly accused of minor sexual abuse in December 2013, the Diocese has supplemented its disclosure with public disclosure of an additional five clergy names and files.

The Diocese continues to provide outreach through services that address survivors' emotional, psychological and spiritual well-being, such as counseling and spiritual direction. Survivors are invited to talk with the Bishop if and when they wish to do so as part of their healing process. Furthermore, the Debtor has implemented Diocese-wide policies requiring that any reports of sexual abuse of minor by clergy are promptly referred to law enforcement and placing the accused clergy on administrative leave while the matter is investigated or reviewed.

#### VI. THE CHAPTER 11 CASE

- A. <u>THE CHAPTER 11 FILING</u>. The Debtor commenced its Chapter 11 case on November 30, 2018 (the "Petition Date"). The Debtor's case was assigned to the Honorable Robert Kressel, United States Bankruptcy Judge. The Bankruptcy Court has entered several orders in this Chapter 11 case, each of which is available from the clerk of the Bankruptcy Court or may be viewed at the Bankruptcy Court's website: www.mnb.uscourts.gov.
- **B.** <u>RETENTION OF COUNSEL</u>. Subsequent to the Petition Date, the Debtor remained in possession of its assets and property and continued to operate as the Debtor-in-Possession under §§ 1107 and 1108 of the Bankruptcy Code. By order of the Court, Bodman PLC was authorized to act as bankruptcy counsel and Restovich Braun & Associates as general corporate and litigation

Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 21 of 222

counsel for the Debtor for this case. Later, Burns Bowen Bair LLP was authorized to act as special insurance litigation counsel for the Debtor.

**C.** <u>APPOINTMENT OF CREDITORS' COMMITTEE</u>. Under §§ 1102(a) and 1102(b) of the Bankruptcy Code, the United States Trustee appointed the Committee to serve in the Debtor's case. The Committee consists of five (5) individuals who hold Tort Claims against the Debtor.

The Committee retained the law firm of Stinson, LLP to represent it throughout this case. Since its appointment, the Committee has taken an active role in the Debtor's case and been involved in virtually every major event that transpired during the Chapter 11 process, including negotiating with the Debtor the LMI/Interstate Settlement Agreement (defined below), jointly proposing the Plan with the Debtor, and negotiating the terms of the Plan and the procedures for allocating funds to Tort Claimants.

The Committee has also performed its investigatory function by reviewing information supplied by the Debtor and third parties, as well as conducting its investigation to determine if any other assets could be made available to pay Claims of Tort Claimants or other creditors.

D. UNKNOWN CLAIMS REPRESENTATIVE. The Debtor intends to file a motion to approve the appointment of U.S. District Judge Michael R. Hogan, retired, as Unknown Claims Representative for the Debtor. If the motion is granted, the Unknown Claims Representative will be the legal representative for holders of Impaired Unknown Tort Claims (as defined in the Plan), which include certain Tort Claims that arose prior to July 1, 2002 for which a Claim was not timely filed by the Claims Filing Date (i.e., April 8, 2019). If appointed, the Unknown Claims Representative's responsibilities and duties will include: (i) undertaking an investigation and analysis regarding the estimated number of Impaired Unknown Tort Claims and the estimated value of Impaired Unknown Tort Claims; (ii) casting a ballot on the Plan on behalf of Impaired Unknown Tort Claimants; (iii) advocating the legal position of the Impaired Unknown Tort Claimants before the Bankruptcy Court, and if necessary, filing pleadings and presenting evidence on any issue affecting the claims of the Impaired Unknown Tort Claimants; (iv) taking all other legal actions reasonably necessary to represent the interests of the Impaired Unknown Tort Claimants; and (v) serving as an independent fiduciary acting on behalf of all Impaired Unknown Tort Claimants.

Once the Unknown Claims Representative has completed his analysis, he will issue his report and recommendation. The number of Impaired Unknown Tort Claimants will be estimated for purposes of Plan confirmation. The treatment of Impaired Unknown Tort Claims is set forth below in the discussion of the treatment of Class 4B Claims.

- **E.** ASSET SALES AND OTHER DISPOSITIONS. As detailed on Schedule AB of the Debtor's Schedules, as of the Petition Date, the Debtor owned several parcels of real property. The Debtor intends to sell or otherwise monetize certain property to raise a portion of the funds that the Debtor will be contributing to fund the Debtor's share of the settlement with Tort Claimants set forth in the Plan.
- **F.** BAR DATES AND OBJECTIONS TO CLAIMS. By Order dated December 7, 2018 (the "Bar Order"), the Bankruptcy Court set April 8, 2019 (the "Claims Filing Date") as the last day

for creditors, including Tort Claimants, to file a proof of claim. A notice of the Claims Filing Date was sent to all known Tort Claimants (either directly or through their counsel), as well to over 500 licensed therapists and over 375 licensed alcohol and addiction treatment centers in the State of Minnesota, as identified by the Committee's counsel, with a request that it be posted in a conspicuous place within their respective offices and shared with any potential Tort Claimant of whom they were aware. A notice of the Claims Filing Date was also sent to (1) the Minnesota Attorney General, (2) the county attorney, county administrator and sheriff's department for each county in the Diocese, (3) the Minnesota Department of Health's locations within the Diocese and (4) each hospital located within the Diocese, with a request that the notice be posted in a conspicuous public place within each of their facilities. In addition, publication notice of the Claims Filing Date was made in the USA Today, and in over 40 additional regional and local newspapers on multiple separate occasions, and was made available for posting at various radio and TV stations, and at parishes within the Diocese.

The Debtor and the Committee's Professionals have reviewed the Claims filed by creditors. A total of 147 Class 3 Claims were filed in the Debtor's Chapter 11 case. (Four of the 147 Class 3 Claims were filed after the Claims Filing Date).

Various parishes and schools, as well as the Catholic Foundation of Southern Minnesota, also filed contingent claims for contribution or indemnification to the extent they are or may be named as defendants along with the Debtor on various abuse allegations. Those claims also included claims for potential contribution or indemnification with respect to various non-abuse related matters such as health insurance claims, retirement plan claims and the like. The Debtor expects all of those claims will be withdrawn.

In addition, approximately a dozen non-Tort Claims were scheduled, and two Proofs of Claim were timely filed for non-Tort Claims. One of those Proofs of Claim was for a postpetition vendor who has since been paid in full. The other Proof of Claim was an IRS claim filed as an estimate of FICA withholding for the December 31, 2018 period, which Proof of Claim was subsequently amended to \$0.00. All payroll taxes and FICA withholding with respect to amounts accrued prior to the Petition Date have been paid in full, per authority of an order of the Bankruptcy Court.

To the extent the Debtor has the right to object to Claims and deems it prudent and/or cost effective to do so, the Plan provides that any objections to non-Tort Claims must be filed and served not later than thirty (30) days after the later of (i) the Effective Date, or (ii) the date such Claim is filed. If the Debtor fails to object to a properly filed Claim on or before such date, then such Claim will be deemed allowed if such Claim is a non-Tort Claim and will be entitled to the Distribution under the Plan applicable to the particular class in which such Claim is classified. Notwithstanding the foregoing, there will be no deadline for the Reorganized Debtor to object to Class 4A Claims (Unimpaired Unknown Tort Claims) or Class 6 Claims (Unimpaired Abuse Related Contingent Claims).

**G.** PLAN EXCLUSIVITY. Under §1121 of the Bankruptcy Code, a debtor-in-possession is granted a 120-day exclusive period from the Chapter 11 filing date to file a plan of reorganization. During such time, only the Debtor can file a plan of reorganization. However, the Bankruptcy Code provides that the court can increase a debtor's exclusive period to file and

Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 23 of 222

confirm a plan of reorganization for cause shown up to eighteen (18) months after commencement of the case. As of June 1, 2020, the Debtor's exclusive right to file a Plan expired.

**H.** LITIGATION – ADVERSARY PROCEEDING 18-03094 REGARDING INSURANCE COVERAGE. Prior to the Petition Date, Debtor filed a complaint in the State of Minnesota, County of Winona, Third Judicial District Court against United States Fire Insurance Company ("U.S. Fire") and various other insurers, seeking a declaratory judgment to determine the extent of the rights of the Debtor under certain insurance policies and certificates. Shortly after the commencement of its Chapter 11 case, the Debtor removed that action to the United States District Court and the matter was referred to the Bankruptcy Court as Adversary Proceeding No. 18-03094. In that action, the Debtor sought to determine the extent of coverage as to the CVA claims raised against the Debtor.

On December 20, 2018, a number of the insurance company defendants filed in the Adversary Proceeding a motion to transfer the action back to the United States District Court. The Debtor objected to that motion and on January 24, 2019, the Bankruptcy Court entered an order denying the motion. Several of the insurance company defendants then filed a motion to withdraw the reference, which, if granted would have resulted in the Adversary Proceeding being transferred back to the United States District Court. The Debtor also objected to that motion and on April 18, 2019, the United States District Court denied that motion as well. On March 5, 2021, U.S. Fire filed a renewed motion to withdraw the reference. On March 18, 2021, the Debtor filed an objection to U.S. Fire's renewed motion to withdraw the reference remains pending.

The Debtor, with consent of the Committee, agreed to voluntarily dismiss certain defendants -- St. Paul Surplus Lines Insurance Company, St. Paul Fire and Marine Insurance Company, and Assicurazioni Generali SPA -- from the Adversary Proceeding without prejudice because the policies issued by each of those insurers were excess, claims-made policies and there was no evidence that any of the underlying sexual abuse claims and lawsuits for which the Debtor sought insurance coverage were first made while those policies were in effect.

The Adversary Proceeding was stayed while the parties attempted to reach a negotiated settlement with the assistance of a professional mediator. By stipulation of the Diocese, the Committee, U.S. Fire, 21<sup>st</sup> Century Centennial Insurance Company (f/k/a Colonial Penn Insurance Company) ("Colonial Penn") and counsel for the vast majority of Tort Claimants, the Bankruptcy Court recently entered an order lifting the stay on the Adversary Proceeding as it relates to U.S. Fire and Colonial Penn, to allow litigation to proceed against those Non-Settling Insurers under the Plan.

I. <u>SETTLEMENT NEGOTIATIONS AND MEDIATION</u>. On May 21, 2019, the court appointed John E. Vukelich as mediator. The parties engaged in several mediated settlement negotiation sessions under the supervision of Mr. Vukelich. Extensive negotiations took place among the Debtor, the Committee, counsel representing Tort Claimants, and the insurers' representatives and their counsel. The negotiations and mediation were extensive and intensive. As a result of the mediation sessions and further negotiations, an agreement in principle was reached among the Debtor, the Committee, the attorneys representing the vast majority of the sexual abuse survivors who have asserted claims against the Debtor (Jeff Anderson and Associates,

Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 24 of 222

P.A.), and LMI/Interstate (the "LMI/Interstate Settlement"). As part of the LMI/Interstate Settlement, LMI will pay the sum of \$3,250,000 and Interstate will also pay the sum of \$3,250,000 to a Trust established for the benefit of Tort Claimants.

The Debtor, the Committee, counsel for the survivors and U.S. Fire then stipulated to the appointment of a second mediator, Paul J. Van Osselaer, whom the Bankruptcy Court appointed to co-mediate with Mr. Vukelich. The Debtor had hoped that the second mediator would facilitate the parties' ability to reach a negotiated resolution that also included U.S. Fire and Colonial Penn. Unfortunately, the efforts to reach a mediated settlement with U.S. Fire and Colonial Penn were not successful.

The Debtor and the Committee then turned their efforts to documenting the LMI/Interstate Settlement and crafting a plan of reorganization that incorporated the terms of the LMI/Interstate Settlement. As discussed in more detail below, the Plan also provides for the transfer of certain rights and interests of the Debtor and other insureds in respect of coverage for Class 3 and Class 4B Claims under insurance policies issued by U.S. Fire, Colonial Penn and the other Non-Settling Insurers to a Trust established for the benefit of survivors.

In addition to U.S. Fire and Colonial Penn, the following additional insurance companies, each of which held a percentage share of the risk on policies insuring the Diocese, are also Non-Settling Insurers: (i) Northwestern National Insurance Company of Milwaukee Wisconsin (f/k/a Bellefonte Insurance Co.) ("Northwestern National"); (ii) Stronghold Insurance Co. Ltd. ("Stronghold"); and (iii) CX Reinsurance Company Ltd. ("CX").<sup>3</sup> Northwestern National is the subject of a state court insolvency proceeding in Wisconsin. Stronghold and CX are each the subject of insolvency proceedings in the United Kingdom. Northwestern National, Stronghold and CX are identified collectively as the "Insolvent Insurers." In any event, under the Plan, the Debtor's rights and interests in respect of actual or potential coverage under policies issued by Non-Settling Insurers for any Class 3 Claim or Class 4B Claim will be transferred to the Trust. The Non-Settling Insurers also include any other Persons that are not "Settling Insurers" and that have, or are alleged to have, extended insurance coverage for Tort Claims to the Diocese or any Diocesan Parish, or have issued, subscribed in any interest in, assumed any liability for, or underwritten any risk in, any Non-Settling Insurer Policy. Non-Settling Insurers of which the Diocese is aware that have extended insurance coverage for Tort Claims to certain Diocesan Parishes include State Farm Fire and Casualty Company and Insurance Company of North America.

In short, the Debtor and the Committee, through mediation, have now successfully negotiated both the monetary compensation to be provided to sexual abuse Tort Claimants from the Debtor and LMI/Interstate, and the non-monetary undertakings by the Diocese which will assist with the healing of sexual abuse Tort Claimants and mitigating the risk of any such abuse in the future, all of which is incorporated into the Plan. The Debtor and the Committee have also agreed upon the mechanism for addressing the rights of the Debtor and Tort Claimants with respect to Non-Settling Insurers, which is further detailed at Article VIII of the Plan.

<sup>&</sup>lt;sup>3</sup> CX acquired responsibility for certain policies issued to the Diocese by CNA Reinsurance of London, Ltd. and CNA International Reinsurance Co. Ltd.

- **J.** <u>THE INSURANCE SETTLEMENT</u>. Principal terms of the agreement documenting the insurance settlement include:
  - i. Settlement Amounts. LMI will pay the sum of \$3,250,000 and Interstate will also pay the sum of \$3,250,000 to purchase back their respective policies and to obtain the benefits of the Channeling Injunction and Supplemental Settling Insurer Injunction. The proceeds will be paid to a Trust established for the benefit of Tort Claimants as set forth in and in accordance with the terms of the Settlement Agreement with LMI/Interstate (the "LMI/Interstate Settlement Agreement"). The LMI/Interstate Settlement constitutes an "Insurance Settlement Agreement" under the Plan.
  - Releases. The Debtor and other releasing Protected Parties, on the one hand, and LMI/Interstate, on the other, will grant complete mutual releases as to, among other things, any and all past, present, or future Claims in connection with, relating to, or arising out of, in any manner or fashion, the Tort Claims, the policies or certificates of insurance issued by LMI/Interstate, and the reorganization case, as set forth in the LMI/Interstate Settlement Agreement and the Plan. All Tort Claimants will be required to sign a release of LMI/Interstate as to their claims prior to any distribution. All Tort Claimants will also be required to sign a release of the Diocese, the Reorganized Debtor, and any other Protected Party for any claims that do not implicate coverage under any Non-Settling Insurer Policy. The Tort Claimants specifically reserve, and do not release, any and all Tort Claims that implicate coverage under any Non-Settling Insurer Policy, but recourse is limited to the proceeds of Non-Settling Insurer Policies and all other damages (including extra-contractual damages), awards, judgments in excess of policy limits, penalties, punitive damages and attorney's fees and costs that may be recoverable against any Non-Settling Insurers because of their conduct concerning insurance coverage for, or defense or settlement of, any Tort Claim, and any such judgments or awards will be turned over to the Trust for handling in accordance with Sections 6.14(i) and (j) of the Plan. The Class 3 and Class 4B Claims will not be released or enjoined as against the Diocese, the Reorganized Debtor, or any other Protected Party for any Abuse that may be covered under Non-Settling Insurer Policies until such claims are settled with the Diocese, the Reorganized Debtor, any other Protected Party and such Non-Settling Insurer or are fully adjudicated, resolved, and subject to Final Order, but recourse is limited as described above. The Unimpaired Unknown Tort Claimants also reserve their rights against the Reorganized Debtor pursuant to the terms of the Plan.
  - iii. Sale and Buyback of Policies and Certificates. The Debtor and other releasing Protected Parties will sell all of their interests in the policies of insurance issued by LMI/Interstate, free and clear of all liens, claims, encumbrances and other interests under 11 U.S.C. § 363.
  - iv. Channeling Injunction. LMI/Interstate, the Debtor and the other Protected Parties will be entitled to receive the benefit of a Channeling Injunction under the Plan and order confirming the Plan. Any and all Channeled Claims are channeled to the Trust and will be treated, administered, determined, and resolved under the procedures and protocols and in the amounts as established under the Plan and Trust Agreement as their sole and exclusive remedy for all holders of Channeled Claims. Tort Claimants and the Trust shall

be permitted, however, to name the Diocese and any other Protected Party in any proceeding to resolve whether the Diocese or such other Protected Party has liability for a Tort Claim, and the amount of any such liability, for the purpose of obtaining insurance coverage from Non-Settling Insurers under the Non-Settling Insurer Policies, but recourse is limited as described above. In addition, the Unimpaired Unknown Tort Claimants are not releasing their claims against the Reorganized Debtor, and the Reorganized Debtor assumes the Protected Parties' and LMI/Interstate's liability for the obligation to pay, if any, Unimpaired Unknown Tort Claims.

- v. Supplemental Settling Insurer Injunction. LMI/Interstate will be entitled to receive the benefit of a Supplemental Settling Insurer Injunction under the Plan and order confirming such Plan under 11 U.S.C. §§ 105(a) and 363. Any and all Persons who have held, now hold or who may in the future hold any Interests (including all debt holders, all equity holders, governmental, tax and regulatory authorities, lenders, trade and other creditors, Tort Claimants, perpetrators, and all others holding Interests of any kind or nature whatsoever, including those Claims released or to be released under the LMI/Interstate Settlement Agreement) against any of LMI/Interstate, or any other Person covered or allegedly covered under the Settling Insurer Policies, will be permanently stayed, enjoined, barred, and restrained from taking any action, directly or indirectly, to assert, enforce or attempt to assert or enforce any such Interest against LMI/Interstate or the Settling Insurer Policies. The Supplemental Settling Insurer Injunction bars these actions only against LMI/Interstate or the Settling Insurer Policies issued by LMI/Interstate.
- vi. *Insolvent Insurers*. The Channeling Injunction and Supplemental Insurer Injunction will not be for the benefit of Non-Settling Insurers, including the Insolvent Insurers. Nothing in the Plan will be construed to authorize a violation of any stay that is applicable in light of the pending insolvency proceedings with respect to such Insolvent Insurers.
- vii. *Indemnification of Settling Insurers*. The Trust indemnifies LMI/Interstate against certain claims, including Tort Claims, Medicare Claims and Contribution Claims.
- viii. *Conditions to Settling Insurers' Payments*. LMI/Interstate's payments are conditioned on, among other things, entry of a Final Order approving the LMI/Interstate Settlement Agreement, entry of the Confirmation Order, and such Order becoming a Final Order.

# VII. SUMMARY OF THE PLAN

The Debtor and the Committee submit that the treatment of creditors under the Plan is more favorable than the treatment creditors would receive if the case were converted to Chapter 7. Therefore, the Debtor and the Committee submit that the Plan is in the best interests of the creditors and the Debtor and the Committee recommend acceptance of the Plan by holders of Claims in Classes 3 and 4B.

Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 27 of 222

THE SUMMARY OF THE PLAN SET FORTH BELOW IS QUALIFIED IN ITS ENTIRETY BY REFERENCE TO THE PROVISIONS SET FORTH IN THE PLAN, THE TERMS OF WHICH CONTROL.

#### A. GENERAL

Brief Explanation of Chapter 11. Chapter 11 is the principal business or operations reorganization chapter of the Bankruptcy Code. Under Chapter 11, a debtor is authorized to reorganize its business or operations for the benefit of itself and its creditors. Upon the filing of a petition for reorganization under Chapter 11 and during the pendency of the case, the Bankruptcy Code imposes an automatic stay of all attempts to collect claims or enforce liens against the Debtor, including the commencement of any sexual abuse litigation.

Confirmation and consummation of a plan of reorganization is the principal objective of a Chapter 11 case. In general, a plan divides the claims against a debtor into separate classes and allocates plan distributions among those classes. If the legal, equitable and contractual rights of a class are unaffected by a plan, such class is considered "unimpaired". All unimpaired classes are deemed to have accepted a plan and therefore are not entitled to vote thereon. Bankruptcy Code §1126(g), on the other hand, provides that all classes of claims that do not receive or retain any property under a plan on account of such claims are deemed to have rejected such plan. All classes of claims that are considered "impaired" are entitled to vote on a plan.

Under the Bankruptcy Code, acceptance of a plan is determined by class; therefore, it is not required that each holder of a claim in an impaired class vote in favor of a plan in order for the Bankruptcy Court to confirm a plan. Generally, each impaired class must vote to accept a plan; however, the Bankruptcy Court may confirm a plan in certain circumstances without the acceptance of all impaired classes if at least one (1) impaired class votes to accept a plan and certain other statutory tests are satisfied. Many of these tests are designed to protect the interests of creditors who either do not vote or vote to reject such plan but who will nonetheless be bound by such plan if it is confirmed by the Bankruptcy Court.

1. Acceptance of the Plan. As a condition to confirmation, Bankruptcy Code §1129(a) requires that: (a) each impaired class of claims votes to accept the plan; and (b) the plan meets the other requirements of §1129(a). As explained above, classes that are unimpaired are deemed to have accepted the plan and therefore are not entitled to vote thereon, and classes that do not receive or retain any property under the plan are deemed to have rejected the Plan and likewise are not entitled to vote thereon. Accordingly, acceptances of the Plan are being solicited only from those parties who hold Claims classified in impaired Classes that are to receive distributions under the Plan.

An impaired class of claims will be deemed to have accepted a plan if holders of at least two-thirds in dollar amount and a majority in number of claims in such class who cast timely ballots vote to accept the plan.

2. Classification of Claims, Generally. Bankruptcy Code Section 101(5) defines a claim as: (a) a "right to payment, whether or not such right is reduced to judgment, liquidated, unliquidated, fixed, contingent, matured, unmatured, disputed, undisputed, legal,

Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 28 of 222

equitable, secured or unsecured"; or (b) a "right to an equitable remedy for breach of performance if such breach gives rise to a right to payment, whether or not such right to an equitable remedy is reduced to judgment, fixed, contingent, matured, unmatured, disputed, undisputed, secured or unsecured."

Bankruptcy Code Section 1123 provides that a plan of reorganization will designate classes of claims against a debtor. Bankruptcy Code Section 1122 further requires that each class of claims contain only claims that are "substantially similar" to each other. The Debtor believes that they have classified all Claims in compliance with the requirements of Sections 1122 and 1123. However, it is possible that a holder of a Claim may challenge such classification and that the Bankruptcy Court may find that a different classification is required for the Plan to be confirmed. In such event, the Debtor would, to the extent permitted by the Bankruptcy Court, modify the classifications in the Plan as required and use the acceptances received in this solicitation for the purpose of obtaining the approval of a Class or Classes of which the accepting holder is ultimately deemed to be a member. Any such reclassification could adversely affect the Class of which such holder was initially a member, or any other Class under the Plan, by changing the composition of such Class and the vote required of that Class for approval of the Plan. Furthermore, a reclassification of Claims may necessitate a re-solicitation.

- **B.** CLASSIFICATION AND TREATMENT OF CLAIMS UNDER THE PLAN. The following describes the classification of Claims under the Plan and the treatment that holders of Claims, whether Tort Claims or otherwise, are to receive if the Plan is confirmed and becomes effective. A Claim is classified in a particular Class only to the extent that the Claim fits within the description of that Class and is classified in a different Class to the extent that any remainder of the Claim fits within the description of such different Class.
- 1. Unclassified Claims. The Plan does not classify Administrative Claims, statutory fees due to the United States Trustee, or Priority Tax Claims, but does provide for the following treatment of such Claims:
  - i. *United States Trustee Fees*. All fees due and payable pursuant to 28 U.S.C. § 1930 and not paid prior to the Effective Date will be paid by the Reorganized Debtor as soon as practicable after the Effective Date. After the Effective Date, the Trust will pay quarterly fees to the U.S. Trustee until the Chapter 11 case is closed, but in no event will the payments made to the Trust made pursuant to the Plan by any Person other than the Debtor be considered "disbursements" under 28 U.S.C. § 1930, nor will any payment made by the Trust to any Person be considered a disbursement under 28 U.S.C. § 1930. The Reorganized Debtor will file post-Confirmation Date quarterly reports in conformance with the U.S. Trustee guidelines. The U.S. Trustee will not be required to file a request for payment of its quarterly fees, which will be deemed Administrative Claims against the Debtor and its Estate. The Reorganized Debtor will remain responsible for any reporting and pre-Effective Date unpaid fees.
  - ii. Administrative Claims. An "Administrative Claim" is a Claim for payment of an administrative expense of a kind specified in Bankruptcy Code Section 503(b) and referred to in Bankruptcy Code Section 507(a)(2), including the actual and necessary costs and expenses of preserving the Estate or operating the Debtor's businesses

after the commencement of a Chapter 11 case, loans and advances made to the Debtor after the Petition Date, compensation for legal and other services and reimbursement of expenses awarded or allowed under Bankruptcy Code Sections 330(a), 331 or 503, certain retiree benefits, certain reclamation Claims, and all fees and charges against the Estate pursuant to Chapter 123 of Title 28 of the United States Code.

iii. *Priority Tax Claims*. A "**Priority Tax Claim**" is an Unsecured Claim of a governmental unit entitled to priority in payment pursuant to any provision of the Bankruptcy Code §507(a)(8). The Debtor has no pre-petition tax claims entitled to priority.

The Plan provides that each holder of an Allowed Professional Claim (i.e., the professionals' fees and expenses incurred by the Professionals and allowed in a Final Order of the Bankruptcy Court) will be paid in full, in Cash, by the Reorganized Debtor (i) within seven (7) days after the later to occur of the Effective Date or the date the Order allowing such Administrative Claim becomes a Final Order; or (ii) upon such terms as may exist pursuant to Order of the Bankruptcy Court or an agreement between such holder of an Allowed Administrative Claim and the Debtor.

To date, the Debtor has paid all professional fees and expenses incurred through February 28, 2021, including the professional fees of Bodman, PLC, bankruptcy counsel for the Debtor, Restovich Braun & Associates, general corporate and litigation counsel for the Debtor, Burns Bowen and Bair LLP, special insurance counsel for the Debtor, Alliance Management, LLC, financial consultant for the Debtor, and Stinson, LLP, counsel for the Committee. Any allowed unpaid professional fees outstanding as of the Effective Date will be paid as set forth in the Plan. As to other allowed Administrative Claims, except as otherwise provided in the Plan, the Plan provides that each holder of an allowed Administrative Claim will receive, on account of and in full satisfaction of such allowed Administrative Claim, cash equal to the amount thereof, unless the holder agrees to less favorable treatment of such allowed Administrative Claim.

The Plan further provides that Professional Persons with Claims for services rendered on or before the Effective Date, must file requests for payment within thirty (30) days after notice of the Effective Date is filed with the Bankruptcy Court.

Administrative Claims representing obligations incurred by the Debtor after the Effective Date (including, without limitation, Claims for professionals' fees and expenses) will not be subject to application to the Bankruptcy Court and may be paid by the Reorganized Debtor in the ordinary course of business and without Bankruptcy Court approval. In addition, holders of Administrative Claims representing trade debt incurred after the Petition Date in the ordinary course of Debtor's operations are not required to file requests for allowance of an Administrative Claim and will be paid by the Debtor in the ordinary course. The Reorganized Debtor will, in the ordinary course of business and without the necessity for any approval by the Court, pay the reasonable fees and expenses incurred after the Effective Date of the Professional Persons employed by the Debtor in connection with the implementation and consummation of the Plan, the claims reconciliation process and any other matters as to which such Professionals may be engaged.

Allowed Administrative Claims representing fees and expenses of Professionals that are employed by the Committee, which are incurred on or prior to the Effective Date of the Plan in

connection with the implementation and consummation of the Plan may be paid by the Debtor or the Reorganized Debtor, after notice and a hearing, or by the Trust from contributions by the Debtor or the Reorganized Debtor.

Administrative Claims representing fees and expenses of Professionals that are employed by the Debtor prior to the Effective Date of the Plan will not be paid by the Trust.

#### C. PRIORITY CLAIMS (CLASS 1)

- 1. Definition. A Class 1 Claim means an allowed Claim described in, and entitled to priority under Sections 507(a) and 503(b)(9) of the Bankruptcy Code other than an Administrative Claim or a Priority Tax Claim.
- 2. Treatment. Unless the holder of an allowed Class 1 Claim and the Diocese or the Reorganized Debtor (as applicable) agree to a different treatment, the Reorganized Debtor will pay each such allowed Class 1 Claim in full, in cash, without interest, from ongoing operations on the later of the Effective Date (or as soon thereafter as is practicable) and the date a Class 1 Claim becomes an allowed Claim (or as soon thereafter as is practicable).

#### D. GOVERNMENTAL UNIT CLAIMS (CLASS 2)

- 1. Definition. A "Class 2 Claim" means an allowed Claim of Governmental Units not otherwise included in Article II of the Plan or Section 4.1 of the Plan.
- 2. Treatment. Unless the holder of an allowed Class 2 Claim and the Diocese or the Reorganized Debtor (as applicable) agree to a different treatment, the Reorganized Debtor will pay each such allowed Class 2 Claim in full, in cash, without interest, from ongoing operations on the later of the Effective Date (or as soon thereafter as is practicable) and the date a Class 2 Claim becomes an allowed Claim (or as soon thereafter as is practicable).

#### E. TORT CLAIMS OTHER THAN IMPAIRED UNKNOWN TORT CLAIMS (CLASS 3)

- 1. Definition. A Class 3 Claim means a Tort Claim other than an Impaired Unknown Tort Claim ("Class 3 Claim"). A "Class 3 Claimant" will mean a holder of a Class 3 Claim.
- 2. Summary. The Plan creates a Trust to fund payments to Class 3 Claimants entitled to such payments under the Plan, Trust Agreement, and Trust Distribution Plan. The Trust will be funded as provided in Articles IV, V, and VI of the Plan, including by contributions from the Diocese and others and the assignment of the Transferred Insurance Interests. The Trust will make distributions to the Class 3 Claimants, as provided by the Plan, the Trust Agreement, and the Trust Distribution Plan, which will represent the sole recovery available to Class 3 Claimants in respect to any obligation owed by the Settling Insurers. Distribution from the Trust, however, does not preclude or affect claims or recoveries by Class 3 Claimants against the Non-Settling Insurers.

No Class 3 Claimant will receive any payment on any award unless and until such Class 3 Claimant has executed the Release attached as **Exhibit E** to the Plan. Each Class 3 Claimant must execute a release of all claims against the Settling Insurers and must release all claims against the

Diocese, the Reorganized Debtor, and any other Protected Party that do not implicate insurance coverage under Non-Settling Insurer Policies. To preserve coverage under Non-Settling Insurer Policies, Class 3 Claimants specifically reserve, and do not release, any and all claims that they may have against the Diocese, the Reorganized Debtor, or any other Protected Party that implicate coverage under Non-Settling Insurer Policies, but recourse is limited to the proceeds of Non-Settling Insurer Policies and all other damages (including extra-contractual damages), awards, judgments in excess of policy limits, penalties, punitive damages and attorney's fees and costs that may be recoverable against any Non-Settling Insurers because of their conduct concerning insurance coverage for, or defense or settlement of, any Tort Claim, and any such judgments or awards will be turned over to the Trust for handling in accordance with Sections 6.14(i) and (j) of the Plan. The Class 3 Claims will not be released or enjoined as against the Diocese, the Reorganized Debtor, or any other Protected Party for any Abuse that may be covered under Non-Settling Insurer Policies until such claims are settled with the Diocese, the Reorganized Debtor, any other Protected Party and such Non-Settling Insurer or are fully adjudicated, resolved, and subject to Final Order, but recourse is limited as described above. Any release of Class 3 Claims, in whole or in part, will be pursuant to the principles set forth in *Pierringer v. Hoger*, 124 N.W.2d 106 (Wis. 1963) and Frey v. Snelgrove, 269 N.W.2d 918 (Minn. 1978). The Claimants will expressly reserve their rights against other Persons, including joint tortfeasors, who will remain severally liable on any Class 3 Claims. Any Person that is or was alleged to be a joint tortfeasor with any of the Protected Parties in connection with the Abuse that forms the basis of a Class 3 Claim will not be liable for any Protected Party's share of causal liability or fault. In no event may a Class 3 Claimant collect on that portion of any judgment or obtain any reallocation of any judgment based on the causal fault or share of liability of any Protected Parties. Any Person that is or was alleged to be a joint tortfeasor with any of the Protected Parties in connection with the Abuse that forms the basis of a Class 3 Claim will be provided by the Trustee with a copy of the executed Release upon reasonable request and provision of an appropriate, executed confidentiality agreement and will not be liable for any Protected Parties' share of liability or fault. The Trust will be obligated to provide copies of the Class 3 Claimants' releases and certifications to any of the Protected Parties or the Settling Insurers upon request provided that such Protected Parties or Settling Insurer have signed a confidentiality agreement encompassing such information.

The Trust will fund the defense of the Diocese, the Reorganized Debtor, and any other Protected Party as against any Litigation Claims brought by Class 3 Claimants, but only to the extent that the Diocese, the Reorganized Debtor, or any other Protected Party, as applicable, is not defended or otherwise reimbursed for its defense expenses on an advance basis by any Insurer. The Trust will advance funding to the Diocese, the Reorganized Debtor, or any other Protected Party, as applicable, with respect to any judgments or settlements of any Litigation Claims brought by Class 3 Claimants, but only to the extent that such judgments or settlements are not funded by any Insurer. The Trust will pursue recoveries against any Non-Settling Insurers in respect of the Transferred Insurance Interests.

The Non-Settling Insurers remain fully liable for their obligations related in any way to the Class 3 Claims, and their obligations are not reduced by the fact that the Diocese is in bankruptcy or by the amount of distributions Class 3 Claimants receive, or are entitled to receive, based on the Trust Distribution Plan. For the avoidance of doubt, determinations by the Tort Claims Reviewer and/or any distributions entitled to be received from the Trust will not constitute a determination of any Protected Party's liability or damages for Class 3 Claims. The Trust may continue efforts to

obtain recoveries from Non-Settling Insurers related to the Class 3 Claims. Any such recoveries by the Trust from Non-Settling Insurers will likewise become Trust Assets to be distributed pursuant to the Trust Distribution Plan. To bar any argument by the Non-Settling Insurers that any provision of the Plan, including the assignment and transfer of the Transferred Insurance Interests to the Trust, results in a forfeiture of coverage, the Plan preserves the Non-Settling Insurers' rights to the extent required under their respective Non-Settling Insurer Policies and applicable law.

- 3. Treatment. Responsibility for preserving and managing Trust Assets and distributing Trust Assets to Class 3 Claimants will be assigned to, assumed and treated by the Trust as further provided in Article VI of the Plan, the Trust Agreement, and the Trust Distribution Plan. Class 3 Claims will be paid in accordance with the provisions of the Trust and Trust Distribution Plan.
- 4. Stipulated Judgments. Certain Class 3 Claimants have entered into, or may enter into, agreements with the Diocese, the Reorganized Debtor, or any other Protected Party for settlement of a Tort Claim allocated by applicable non-bankruptcy law, including but not limited to settlements consistent with *Miller v. Shugart*, 316 N.W.2d 729 (Minn. 1982) or *Drake v. Ryan*, 514 N.W.2d 785 (Minn. 1994). If a Class 3 Claimant enters into such an agreement with the Diocese, the Reorganized Debtor, or any other Protected Party, the Trust will pursue any judgment against the Non-Settling Insurer on behalf of the Class 3 Claimant. Any recoveries by the Trust from Non-Settling Insurers will become Trust Assets to be distributed pursuant to the Trust Distribution Plan.

#### F. UNIMPAIRED UNKNOWN TORT CLAIMS (CLASS 4A)

- 1. Definition. A Class 4A Claim means an Unimpaired Unknown Tort Claim ("Class 4A Claim"). A "Class 4A Claimant" will mean a holder of a Class 4A Claim.
- 2. Treatment. As of the date of the Plan, the Debtor is not aware of the existence of any Class 4A Claims. The Reorganized Debtor will be responsible for defending any Class 4A Claim that is asserted and for payment of any amount determined to be owed on a Class 4A Claim upon settlement or adjudication of such Class 4A Claim. The Trust will not be responsible for paying Class 4A Claims and will not have a role in the settlement or adjudication of Class 4A Claims.

#### G. <u>IMPAIRED UNKNOWN TORT CLAIMS (CLASS 4B)</u>

- 1. Definition. A Class 4B Claim means an Impaired Unknown Tort Claim ("Class 4B Claim"). A "Class 4B Claimant" will mean a holder of a Class 4B Claim.
- 2. Treatment. The Plan creates a Trust to administer payments to Class 4B Claimants entitled to such payments under the Plan, Trust Agreement, and Trust Distribution Plan. The Trust will be funded as provided in Articles IV, V, and VI of the Plan, including from post-confirmation payments from the Reorganized Debtor and third parties. The Trust will make distributions to the Class 4B Claimants, as provided by the Plan, the Trust Agreement, and the Trust Distribution Plan, which will represent the sole recovery available to Class 4B Claimants in respect to any obligation owed by the Settling Insurers. Distribution from the Trust, however, does

Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 33 of 222

not preclude or affect claims or recoveries by Class 4B Claimants against the Non-Settling Insurers.

No Class 4B Claimant will receive any payment on any award unless and until such Class 4B Claimant has executed the Release attached as Exhibit F to the Plan. Each Class 4B Claimant must execute a release of all claims against the Settling Insurers and must release all claims against the Diocese, the Reorganized Debtor, and any other Protected Party that do not implicate insurance coverage under Non-Settling Insurer Policies. To preserve coverage under Non-Settling Insurer Policies, Class 4B claimants specifically reserve, and do not release, any and all claims that they may have against the Diocese, Reorganized Debtor, or any other Protected Party that implicate coverage under Non-Settling Insurer Policies, but recourse is limited to the proceeds of Non-Settling Insurer Policies and all other damages (including extra-contractual damages), awards, judgments in excess of policy limits, penalties, punitive damages and attorney's fees and costs that may be recoverable against any Non-Settling Insurers because of their conduct concerning insurance coverage for, or defense or settlement of, any Tort Claim, and any such judgments or awards will be turned over to the Trust for handling in accordance with Sections 6.14(i) and (j) of the Plan. The Class 4B Claims will not be released or enjoined as against the Diocese, the Reorganized Debtor, or any other Protected Party for any Abuse that may be covered under Non-Settling Insurer Policies until such claims are settled with the Diocese, the Reorganized Debtor, or any other Protected Party, as applicable, and such Non-Settling Insurer or are fully adjudicated, resolved and subject to Final Order, but recourse is limited as described above.

The Trust will fund the defense of the Diocese, the Reorganized Debtor, and any other Protected Party as against any Litigation Claims brought by Class 4B Claimants, but only to the extent that the Diocese, the Reorganized Debtor, or any other Protected Party, as applicable, is not defended or otherwise reimbursed for its defense expenses on an advance basis by any Insurer. The Trust will advance funding to the Diocese, the Reorganized Debtor, or any other Protected Party, as applicable, with respect to any judgments or settlements of any Litigation Claims brought by Class 4B Claimants, but only to the extent that such judgments or settlements are not funded by any Insurer. The Trust will pursue recoveries against any Non-Settling Insurers in respect of the Transferred Insurance Interests.

The Non-Settling Insurers remain fully liable for their obligations related in any way to the Class 4B Claims. Any release of Class 4B Claims, in whole or in part, will be pursuant to the principles set forth in *Pierringer v. Hoger*, 124 N.W.2d 106 (Wis. 1963) and *Frey v. Snelgrove*, 269 N.W.2d 918 (Minn. 1978). The Class 4B Claimants will expressly reserve their rights against other Persons, including joint tortfeasors, who will remain severally liable on any Class 4B Claims. Nothing in Article IV of the Plan requires any Impaired Unknown Tort Claimant to release any Claims against any joint tortfeasor who is not a Protected Party or a Settling Insurer and such Claims are reserved. But in no event may a Class 4B Claimant collect on that portion of any judgment or obtain reallocation of any judgment based on the causal fault or share of liability of any Protected Party.

Responsibility for preserving and managing Trust Assets and distributing Trust Assets to Class 4B Claimants will be assigned to, assumed, and treated by the Trust as further provided in Article V of the Plan, the Trust Agreement, and the Trust Distribution Plan. Class 4B Claims will be paid in accordance with the provisions of the Trust and Trust Distribution Plan. Class 4B

Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 34 of 222

Claimants will provide sufficient information to allow the Tort Claims Reviewer to make an evaluation of the Class 4B Claim pursuant to the factors in the Trust Distribution Plan.

#### H. GENERAL UNSECURED CLAIMS (CLASS 5)

- 1. <u>Definition</u>. A Class 5 Claim means (1) any Claim arising out of the rejection of any executory contract, or (2) any Unsecured Claim that is not included in another class under the Plan and is not listed as disputed, contingent or unliquidated on the Debtor's schedules filed in connection with this Chapter 11 case ("**Debtor's Schedules**") or as to which the holder of such Claim timely filed a Claim.
- 2. <u>Treatment.</u> Each holder of a Class 5 Claim will receive, directly from the Reorganized Debtor, payment in full of such allowed Class 5 Claim, without interest, on the Effective Date.

#### I. <u>Unimpaired Abuse Related Contingent Claims (Class 6)</u>

#### 1. Class 6 Definition.

A Class 6 Claim means (i) any Claim for contribution, indemnity or reimbursement arising out of the Diocese's liability to pay or defend any Class 4A Claim, and (ii) the Claim of any insurers or other Persons who are subrogated to the Claims identified in clause (i) above.

#### 2. <u>Class 6 Treatment</u>.

As of the date of the Plan, the Debtor is not aware of the existence of any Class 6 Claims. The Reorganized Debtor will be responsible for defending any Class 6 Claim that is asserted and for payment of any amount determined to be owed on a Class 6 Claim upon settlement or adjudication of such Class 6 Claim. The Trust will not be responsible for paying Class 6 Claims and will not have a role in the settlement or adjudication of Class 6 Claims.

#### J. MINNESOTA DEPARTMENT OF COMMERCE (CLASS 7)

- 1. <u>Class 7 Definition</u>. A Class 7 Claim means the claim of the Minnesota Department of Commerce against cash collateral posted at Merchants Bank that secures the Debtor's obligation for potential workers' compensation liability.
- 2. <u>Class 7 Treatment</u>. The cash collateral securing the Class 7 Claim will vest in the Reorganized Debtor and will continue to secure the Class 7 Claim. The security interest of the holder of the Class 7 Claim will remain in place and the holder of such claim may exercise any and all rights and remedies against the collateral posted at Merchants Bank that secures the Debtor's obligation for potential workers' compensation liability.

#### K. COMMUNITY BANK OF MANKATO (CLASS 8)

1. <u>Class 8 Definition</u>. A Class 8 Claim means the claim of the Community Bank of Mankato under that certain mortgage and security agreement encumbering the real property known as 1502 Warren Street, Mankato, Minnesota.

Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 35 of 222

2. <u>Class 8 Treatment</u>. The collateral securing the Class 8 Claim will vest in the Reorganized Debtor, provided, however, that the Reorganized Debtor may transfer legal title to such collateral (subject to the mortgage and security agreement) to the equitable owner as set forth in Section 15.2 of the Plan, but in any case the collateral will continue to secure the Class 8 Claim. The mortgage and security interest of the holder of the Class 8 Claim will remain in place and the holder of such claim may exercise any and all rights and remedies against the collateral referenced in such mortgage and security agreement, available to the holder.

# VIII. MEANS OF IMPLEMENTATION OF THE PLAN

#### A. TRUST FORMATION AND FUNDING

1. <u>Purpose, Formation, and Assets.</u> The Trust will be established for the purpose of receiving, liquidating, and distributing Trust Assets in accordance with the Plan and the Trust Distribution Plan. The proposed Trust Agreement is attached to the Plan as **Exhibit D**.

#### 2. Funding.

- 1. <u>Summary</u>. The Plan will be funded from the sources and in the manner set forth in this Section. In addition to the contributions described herein, the Catholic Entities and Other Insured Entities will waive certain Claims against the Diocese and the Settling Insurers.
- 2. <u>Contributions</u>. Cash and other assets will be paid or transferred, as applicable, to the Trust Account as provided in the Plan and as described herein.
  - (i) Debtor Cash Contribution. The Debtor will transfer (a) \$13,560,000, less (A) Professional Claims (including fees and expenses of Insurance Archaeology Group), mediation fees and expenses (including the fees and expenses of John Vukelich, Paul Van Osselaer and Janice Symchych), certain other administrative expenses as agreed between the Debtor and the Committee, the fees of the Unknown Claims Representative, and U.S. Trustee quarterly fees, in each case to the extent paid after February 29, 2020, and (B) counseling expenses for Tort Claimants (whether paid before or after the Petition Date) (collectively, the "Permitted Deductions"), to the Trust Account within within five days after the Effective Date of the Plan (which payment shall constitute substantial consummation of the Plan), plus (b) \$7,552,500 as soon as practical after the sale or other monetization of certain assets of the Diocese, as well as a contribution from certain non-Diocesan entity resources, but in no event more than 12 months after the date the Confirmation Order becomes a Non-Appealable Order (the "Debtor Cash Contribution"). In the event there are

Permitted Deductions paid after the payment of the amount required under subparagraph 2(i)(a) above, then the Debtor may deduct any remaining Permitted Deductions from the \$7,552,500 payment. The Debtor Cash Contribution will be primarily comprised of funds from the following sources:

- i. non-restricted cash accounts held by the Diocese; and
- ii. sale or other monetization of certain assets of the Diocese, as well as a contribution from certain non-Diocesan entity resources (specifically, \$112,500 of the \$7,552,500 payment will be contributed by the following Catholic Entities: Lourdes High School of Rochester, Inc. a/k/a Lourdes Catholic High School and Rochester Catholic High Schools).
  - (ii) Settling Insurer Contributions.

Each Settling Insurer will pay to the Trust the sums set forth in the LMI/Interstate Settlement Agreement within the time set forth in the LMI/Interstate Settlement Agreement. In addition, all rights to receive payment of the amounts to be paid under the LMI/Interstate Settlement Agreement will be assigned to the Trust. The total amount that will be paid to the Trust by LMI/Interstate is \$6,500,000 (the "Additional Settlement Amounts").

#### 3. <u>Additional Trust Assets</u>.

(i) All Rights and Recoveries Against Non-Settling Insurers.

In addition to the funds transferred to the Trust, the Transferred Insurance Interests of the Diocese are automatically and without further act or deed assigned and transferred to the Trust on the Effective Date. In addition, the Interests of the other Protected Parties in the Transferred Insurance Interests are automatically and without further act or deed assigned and transferred to the Trust on the Effective Date. The foregoing assignment and transfer will not be construed as an assignment and transfer of the Non-Settling Insurer Policies.

- 3. <u>Vesting</u>. On the Effective Date, all Trust Assets will vest in the Trust, and the Diocese and other Protected Parties will be deemed for all purposes to have transferred all of their respective Interests in the Trust Assets to the Trust. On the Effective Date, or as soon as practicable thereafter, the Reorganized Debtor, any other Protected Party, and the Settling Insurers, as applicable, will take all actions reasonably necessary to transfer any Trust Assets to the Trust. Upon the transfer of control of Trust Assets in accordance with this paragraph, the Diocese, the other Protected Parties and the Settling Insurers will have no further interest in or with respect to the Trust Assets except as otherwise explicitly provided in the Plan.
- **B.** PAYMENT OF PROFESSIONAL FEES. The Reorganized Debtor will pay all unpaid Allowed Professional Claims accruing through the Effective Date, (i) within seven (7) days after the later of the Effective Date or the Bankruptcy Court's order on such Claims, or (ii) upon such terms as may exist pursuant to Order of the Bankruptcy Court or an agreement between such holder of an Allowed Professional Claim and the Debtor.

C. PAYMENTS EFFECTIVE UPON TENDER. Whenever the Plan requires payment to be made to a creditor, such payment will be deemed made and effective upon tender thereof by the Trustee, the Debtor, or the Reorganized Debtor to the creditor to whom payment is due. If any creditor refuses a tender, the amount tendered and refused will be held by the Trust, the Debtor, or the Reorganized Debtor for the benefit of that creditor pending final adjudication of the dispute. However, when and if the dispute is finally adjudicated and the creditor receives the funds previously tendered and refused, the creditor will be obliged to apply the funds in accordance with the Plan as of the date of the tender; and while the dispute is pending and after adjudication thereof, the creditor will not have the right to claim interest or other charges or to exercise any other rights which would be enforceable by the creditor, if the Trust, the Debtor, or the Reorganized Debtor failed to pay the tendered payment.

### IX. TRUST

- **A.** <u>ESTABLISHMENT OF TRUST</u>. On or before the Confirmation Date, the Trust will be established in accordance with the Trust Documents. The Trust is intended to qualify as a "Designated" or "Qualified Settlement Fund" pursuant to Section 468B of the Internal Revenue Code and the Treasury Regulations promulgated thereunder. The Debtor is the "transferor" within the meaning of Treasury Regulation Section 1.468B-1(d)(1). The Trustee will be classified as the "administrator" within the meaning of Treasury Regulation Section 1.468B-2(k)(3). The Trust Documents, including the Trust Agreement, are incorporated herein by reference.
- В. PURPOSE, FORMATION AND ASSETS. The Trust will be established for the purposes described in this paragraph. The Trust will receive the transfer and assignment of assets as provided in Articles IV and V of the Plan, including the Debtor Cash Contribution, Additional Settlement Amounts, and the Transferred Insurance Interests, of which the Trust is, and will be deemed to be, the sole assignee. The Trust will make distributions to the Class 3 and Class 4B claimants, as provided by the Plan, the Trust Agreement, and the Trust Distribution Plan. The Trust will pursue recoveries against any Non-Settling Insurers in respect of the Transferred Insurance Interests. The Trust will fund the defense of the Diocese, the Reorganized Debtor, and any other Protected Party as against any Litigation Claims brought by Class 3 and Class 4B claimants, but only to the extent that the Diocese, the Reorganized Debtor, or any other Protected Party are not defended or otherwise reimbursed for their defense expenses by any Non-Settling Insurer. The Trust will advance funding to the Diocese, the Reorganized Debtor, and any other Protected Party, as applicable, with respect to any judgments or settlements of any Litigation Claims brought by Class 3 and Class 4B claimants, but only to the extent that such judgments or settlements are not funded by any Insurer. The Trust will fund the costs and expenses in executing these functions, all such functions to be executed in accordance with the Plan, the Trust Agreement, and the Trust Distribution Plan, with the aim of preserving, managing, and maximizing Trust Assets to pay Class 3 and Class 4B claimants and with no objective to continue or engage in the conduct of a trade or business. The proposed Trust Agreement is attached to the Plan as **Exhibit D**.

#### C. ALLOCATIONS WITHIN AND DISTRIBUTIONS AND PAYMENTS FROM THE TRUST

1. General Corpus. The following distributions and payments will be made from the general corpus of the trust:

Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 38 of 222

- i. *Distributions*. Distributions on Class 3 Claims and Class 4B Claims as determined by the Tort Claims Reviewer in accordance with the Plan, the Trust Agreement, and the Trust Distribution Plan.
- ii. *Tort Claims Reviewer*. The Trustee will retain the Tort Claims Reviewer. Fees payable to the Tort Claims Reviewer for review of Class 3 and Class 4B Claims will be paid from the Trust.
- iii. Trust Administrative Fees. All fees, costs, and expenses of administering the Trust as provided in the Plan and the Trust Agreement will be paid by the Trust, including: (i) as reasonably necessary to meet current liabilities and to maintain the value of the respective Assets of the Trust; (ii) to pay reasonable administrative expenses (including any taxes imposed on the Trust and any professional fees); and (iii) to satisfy other liabilities incurred by the Trust in accordance with the Plan or the Trust Agreement.
- iv. *Indemnity*. The Trust's obligations, if any, to defend, indemnify, or hold harmless any Person expressly set out in the Plan will be made from the corpus of the Trust.
- 2. <u>Impaired Unknown Tort Claim Reserve Fund</u>. The Trust will establish an Impaired Unknown Tort Claim Reserve Fund for the benefit of Class 4B Claimants, as detailed in the Trust Agreement, in an amount to be determined after consultation with the Unknown Claims Representative. The Trust will maintain the Impaired Unknown Tort Claim Reserve Fund until the later of (i) the date that the Impaired Unknown Tort Claim Reserve Fund has been exhausted or (ii) the occurrence of the sixth (6th) anniversary of the Effective Date. Neither the Diocese, the Reorganized Debtor nor any other Protected Party will have any obligation to make any contribution to the Trust to establish an Impaired Unknown Tort Claim Reserve Fund in excess of those contributions specified in Article V, Section 5.1(b) of the Plan as the Debtor Cash Contribution and Transferred Insurance Interests.
- **D.** Tax Matters. The Trust will not be deemed to be the same legal entity as the Diocese, but only the assignee of certain assets of the Diocese and a representative of the Estate for delineated purposes within the meaning of Section 1123(b)(3) of the Bankruptcy Code. The Trust is expected to be tax exempt. The Trustee will file such income tax and other returns and documents as are required to comply with the applicable provisions of the Internal Revenue Code of 1986, 26 U.S.C. §§ 1 *et seq.*, as may be amended, and the regulations promulgated thereunder, 31 C.F.R. §§ 900 *et seq.*, and Minnesota law and the regulations promulgated thereunder, and will pay from the Trust all taxes, assessments, and levies upon the Trust, if any.
- **E.** APPOINTMENT OF THE TRUSTEE. The initial Trustee has been identified in **Exhibit D** to the Plan. The Trustee will commence serving as the Trustee on the Confirmation Date; provided, however, that the Trustee will be permitted to act in accordance with the terms of the Trust Agreement from such earlier date, as authorized by the Diocese and the Committee, and will be entitled to seek compensation in accordance with the terms of the Trust Agreement and the Plan.

#### F. RIGHTS AND RESPONSIBILITIES OF TRUSTEE

- The Trustee will be deemed the Estate's representative in accordance with Section 1123 of the Bankruptcy Code and will have all the rights, powers, authority, responsibilities, and benefits specified in the Plan and the Trust Agreement, including (to the extent necessary to enforce those rights, powers, authority, responsibilities, and benefits only) the powers of a trustee under Sections 704, 108 and 1106 of the Bankruptcy Code and Bankruptcy Rule 2004 (including commencing, prosecuting or settling Causes of Action, enforcing contracts, and asserting Claims, defenses, offsets and privileges). If there is any inconsistency or ambiguity between the Plan and Confirmation Order, on the one hand, and the Trust Agreement, on the other hand, with respect to the Trustee's authority to act, the provisions of the Plan and Confirmation Order will control. Among other things, the Trustee: (1) will liquidate and convert to cash the Trust Assets, make timely distributions and not unduly prolong the duration of the Trust; (2) may request an expedited determination of taxes of the Trust under Section 505(b) of the Bankruptcy Code for all returns filed for, or on behalf of, the Trust for all taxable periods through the dissolution of the Trust; and (3) may retain professionals, including legal counsel, accountants, financial advisors, auditors, and other agents on behalf of the Trust, and at the Trust's sole expense, as reasonably necessary and to carry out the obligations of the Trustee hereunder and under the Trust Agreement.
- 2. Notwithstanding the foregoing, the Diocese, the Reorganized Debtor, and the Trust acting for itself and on behalf the Estate, will be deemed to have waived, effective upon the Effective Date:
  - i. Any and all Claims under Sections 547, 548, 549 and 550 of the Bankruptcy Code for the recovery of any sums paid to any Person who provided goods and services to the Diocese in the ordinary course of business prior to the Effective Date; and
  - ii. Any and all Claims and Causes of Action: (i) seeking the substantive consolidation of the Diocese and any other Person or an order deeming any such Person and the Diocese to be an "alter-ego" of the other or any other similar Claim or Cause of Action; (ii) to avoid, set aside or recover any payment or other transfer made to any Person under Sections 547, 548, 549, and 550 of the Bankruptcy Code; and (iii) any proceeding to avoid or set aside any interest of a Person in property under Section 544 of the Bankruptcy Code.

The Confirmation Order will state that, absent permission of the Bankruptcy Court, no judicial, administrative, arbitral, or other action or proceeding will be commenced in any forum other than the Bankruptcy Court against the Trustee in its/his/her official capacity, with respect to its/his/her status, duties, powers, acts, or omissions as Trustee.

#### G. TRANSFERRED INSURANCE INTERESTS

1. <u>Enforcement of Transferred Insurance Interests Against Non-Settling</u> Insurers.

- As set forth in Article V of the Plan, the Transferred Insurance Interests are assigned and transferred to the Trust. The Trust will be entitled to all policy proceeds due by virtue of a judgment or settlement of a Class 3 Claim or a Class 4B Claim. The Trust will also have the right to pursue judgment against Non-Settling Insurers to determine the amount of coverage available for Protected Parties' liability for Tort Claims. The foregoing transfer will not be construed to entitle any Person to insurance coverage other than those Persons entitled to such coverage from Non-Settling Insurers. The Trust will be fully authorized to act in its own name, or in the name of any Protected Party, to enforce any right, title, or interest of any Protected Party in the Transferred Insurance Interests. No limitations on recovery from Non-Settling Insurers will be imposed by virtue of the fact that the Diocese is in bankruptcy or by any distribution from the Trust to any Tort Claimant. The transfer of the Transferred Insurance Interests will not affect any Non-Settling Insurer's duty to defend, but to the extent that a failure to defend or a separate agreement between the Diocese, the Reorganized Debtor, or any other Protected Party and any Non-Settling Insurer gives rise to a monetary obligation or policy proceeds to reimburse defense costs in lieu of a duty to defend, the Trust will be entitled to the benefit of such monetary obligation or policy proceeds. Any recovery by the Trust on an action against a Non-Settling Insurer for a determination of coverage for Protected Parties' liability for Tort Claims will become a Trust Asset and will be distributed as provided in the Plan, the Trust Agreement, and the Trust Distribution Plan. The Trust's recourse to the Diocese and the other Protected Parties will be limited to the Transferred Insurance Interests and any other rights or interests expressly granted to the Trust under the Plan, including any indemnification obligations of the Reorganized Debtor for Covered Non-Tort Claims under section 16.4 of the Plan, or as otherwise provided by the Plan. The Trust will have no liability for Covered Non-Tort Claims and holders of Covered Non-Tort Claims will have no recourse to the Trust with respect to such Claims.
- 2. The Trust will have full access to coverage issued by the Non-Settling Insurers to the greatest extent permitted by applicable non-bankruptcy law, in the same manner and to the same extent as the Protected Parties prior to the confirmation of the Plan and the transfer of the Transferred Insurance Interests to the Trust, subject to the assertion of any coverage defenses except any defense (a) regarding the assignment and transfer of the Transferred Insurance Interests; (b) effected by operation of law because of confirmation of the Plan; or (c) based on res judicata or collateral estoppel related to facts determined by the Bankruptcy Court.

The assignment and transfer of the Transferred Insurance Interests to the Trust does not affect the Diocese's, the Reorganized Debtor's, other Protected Parties', or any Non-Settling Insurer's right to contest the Diocese's, or any other insured's, liability or the amount of damages in respect of any Tort Claims. Notwithstanding the assignment and transfer of the Transferred Insurance Interests, the Diocese, the Reorganized Debtor, and any other Protected Party will not be relieved of any obligations or duties under any Non-Settling Insurer Policy (including without limitation any duty to cooperate) and will continue to honor such duties and obligations as required by such applicable Non-Settling Insurer Policies and applicable law. The transfer and assignment of the Transferred Insurance Interests does not affect any insurers' rights, obligations, or duties under applicable Non-Settling Insurer Policies or applicable law. If the Trust brings an action against a Non-Settling Insurer to assert any claim or to determine the Non-Settling Insurer's obligation to provide coverage for any Tort Claim, the Non-Settling Insurer may raise any defense to coverage as if the action had been brought by the Diocese, the Reorganized Debtor, or any other Protected Party.

- 3. The Bankruptcy Court will determine at the Confirmation Hearing (i) whether the assignment of the Transferred Insurance Interests provided for in Section 6.7 of the Plan is valid, and (ii) whether such transfer or the discharge and injunctions set forth in Sections 13.2 and 13.3 of the Plan, or any other term of the Plan, void, defeat, or impair the insurance coverage issued by the Non-Settling Insurers. If a party in interest (which, for this purpose, will include the Non-Settling Insurers) fails to timely file an objection to the proposed assignment and transfer of the Transferred Insurance Interests to the Trust or other terms of the Plan related to the Non-Settling Insurer Policies by the date set to file such objections, that party in interest will be deemed to have irrevocably consented to the assignment of Transferred Insurance Interests and other Plan terms related to such Non-Settling Insurer Policies and will be forever barred from asserting that the assignment of the Transferred Insurance Interests or other Plan terms affect the ability of the Trust or Tort Claimants to pursue the Non-Settling Insurers, or any of them, for insurance coverage.
- 4. In the event that the Bankruptcy Court enters a Final Order determining that the assignment of the Transferred Insurance Interests is valid and does not defeat or impair coverage under the Non-Settling Insurer Policies, following the Effective Date, the Trust will assume responsibility for, and be bound by, only such obligations of the Diocese and other Protected Parties under the Non-Settling Insurer Policies as are necessary to enforce the Transferred Insurance Interests; provided, however, that the Protected Parties will not be relieved of any obligations the Protected Parties may have under Non-Settling Insurer Policies.
- 5. The Reorganized Debtor will cooperate and assist the Trust in enforcing any right or prosecuting any claim based on the Transferred Insurance Interests. This cooperation includes, but is not limited to, providing access to documents and electronic information and providing clergy, employees, agents, and volunteers to testify in depositions and at trial.
- 2. Appointment of Trustee as Estate Representative to Enforce Insurance Interests and Obtain Insurance Recoveries.
- 1. If the Bankruptcy Court does not enter a Final Order approving the assignment and transfer of the Transferred Insurance Interests to the Trust, then the assignment will not occur and pursuant to the provisions of Section 1123(b)(3)(B) of the Bankruptcy Code, the Trustee is hereby appointed as the representative of the Diocese's Estate for the purpose of retaining and enforcing all of the Diocese's and the Estate's Interests against the Non-Settling Insurers with respect to the Tort Claims. Any recoveries on such Interests by the Trustee will be paid to the Trust. The determination of whether the appointment of the Trustee as the Diocese's and the Estate's representative provided for in Section 6.7(b)(1) of the Plan is valid, and does not defeat or impair the insurance coverage Non-Settling Insurers are responsible for under Non-Settling Insurer Policies, will be made by the Bankruptcy Court at the confirmation hearing. If a party in interest (which, for this purpose, will include the Non-Settling Insurers) fails to timely file an objection to the proposed appointment by the deadline for filing objections to confirmation of the Plan, that party in interest will be deemed to have irrevocably consented to the appointment and will be forever barred from asserting that the appointment in any way affects the ability of the Trustee to pursue Non-Settling Insurers, or any of them, for insurance coverage. In the event that the Bankruptcy Court determines that the appointment is valid and does not defeat or impair

coverage Non-Settling Insurers are responsible for under Non-Settling Insurer Policies, then, following the Effective Date, the Trustee will assume responsibility for, and be bound by, only such obligations of the Diocese and other Protected Parties under Non-Settling Insurer Policies as are necessary to act as the representative of the Diocese and the Estate for the purpose of retaining and enforcing their Interests, if any, against the Non-Settling Insurers; provided, however, that the Trustee's appointment will not relieve the Diocese, the Reorganized Debtor or the other Protected Parties from any obligation that such entities may have under the Non-Settling Insurer Policies. Nothing contained in Section 6.7(b)(1) of the Plan will affect the rights and remedies of a Person who is not a Protected Party but is an insured or additional insured with the Diocese or is asserting rights under a Non-Settling Insurer Policy.

- 2. In the event that a Final Order is entered holding that: (a) the assignment of the Transferred Insurance Interests, or (b) the appointment of the Trustee as the Diocese's and the Estate's representative are invalid or would defeat or impair the insurance coverage issued by the Non-Settling Insurers, then the assignment and/or appointment, as the case may be, will be deemed not to have been made, and the Diocese, the Reorganized Debtor, and each of the Protected Parties will retain their Interests under each Non-Settling Insurer Policy.
  - (i) At the request of the Trust, the Reorganized Debtor and the other Protected Parties will assert their Interests against a Non-Settling Insurer, including, but not limited to, by filing a lawsuit for recovery of policy proceeds. All recoveries by the Reorganized Debtor and the other Protected Parties will be paid to the Trust. The Reorganized Debtor and the other Protected Parties will select and retain counsel to pursue their Interests against Non-Settling Insurers pursuant to Section 6.7(b) of the Plan, subject to the Trustee's approval, which approval will not be unreasonably withheld.
  - (ii) The Trust will pay the reasonable attorneys' fees, costs and expenses that are incurred by the Reorganized Debtor and the other Protected Parties in pursuing, pursuant to Section 6.7(b) of the Plan, its Interests against Non-Settling Insurers.
  - (iii) The Trust will, in addition to reasonable attorneys' fees, costs and expenses provided for in Section 6.7(b) of the Plan, reimburse the Reorganized Debtor and each of the Protected Parties for any reasonable out of pocket costs and expenses it incurs as a direct consequence of pursuing its Interests against Non-Settling Insurers, but will not compensate the Reorganized Debtor or any other Protected Party for any time any of its employees expend. Upon receipt by the Reorganized Debtor or other Protected Party, all recoveries received by the Reorganized Debtor or other Protected Party from Non-Settling Insurers will be deemed to be held in trust for the benefit of the Trust and will be remitted by the Reorganized Debtor or other Protected Party to the Trust as

soon as practicable following the Reorganized Debtor's or other Protected Party's receipt of such recoveries.

#### H. SPECIAL DISTRIBUTION CONDITIONS

- 1. With respect to Class 3, the Trust will maintain sufficient funds to pay any potential reimbursements to Medicare and will complete the following "Medicare Procedures": (i) the Trustee will determine whether each Tort Claimant with a Date of Injury after December 5, 1980 is eligible to receive, is receiving, or has received Medicare benefits ("Medicare Eligible"); (ii) upon request, the Trust will provide to a Settling Insurer or the Diocese information sufficient to allow them to perform their own SSA queries to the extent they wish to do so; (iii) in the event that one or more Tort Claimants with Dates of Injury after December 5, 1980 is/are identified as Medicare Eligible, the Trust will complete a query to the CMS for each such Tort Claimant to determine whether any payment ("Conditional Payment") made pursuant to Section 1395y(b)(2)(B) of the MSPA has been made to or on behalf of that Tort Claimant arising from or relating to treatment for Abuse; (iv) if any Conditional Payment has been made to or on behalf of that Tort Claimant, the Trustee will, within the time period called for by the MSPA, reimburse the appropriate Medicare Trust Fund for the appropriate amount, and submit the required information for that Tort Claimant to the appropriate agency of the United States government.
- 2. Compliance with the provisions of Section 6.8 of the Plan will be a material obligation of the Trust in favor of LMI/Interstate under the LMI/Interstate Settlement Agreement, which authorizes funding to the Trust.
- 3. The Trust will defend, indemnify and hold harmless the Protected Parties and the Settling Insurers from any Medicare Claims, and any Claims related to the Trust's obligations under this Section.
- **I.** <u>INVESTMENT POWERS; PERMITTED CASH EXPENDITURES</u>. All funds held by the Trust will be invested in cash or short-term highly liquid investments that are readily convertible to known amounts of cash as more particularly described in the Trust Agreement. The Trustee may expend the cash of the Trust.
- **J.** REGISTRY OF BENEFICIAL INTERESTS. To evidence the beneficial interest in the Trust of each holder of such an interest, the Trustee will maintain a registry of beneficiaries.
- **K.** <u>Non-Transferability of Interests</u>. Any transfer of an interest in the Trust will not be effective until and unless the Trustee receives written notice of such transfer.
- **L.** <u>TERMINATION</u>. The Trust will terminate after its liquidation, administration, and distribution of the Trust Assets in accordance with the Plan and its full performance of all other duties and functions set forth herein or in the Trust Agreement. The Trust will terminate no later than the sixth (6th) anniversary of the Effective Date.

#### M. <u>IMMUNITY; LIABILITY; INDEMNIFICATION</u>

1. Neither the Reorganized Debtor or its respective members, designees, or professionals, nor the Trustee or any duly designated agent or representative of the Trustee, nor

their respective employees, will be liable for the acts or omissions of any other member, designee, agent, or representative of such Trustee, except that the Trustee will be liable for his/her/its specific acts or omissions resulting from such Trustee's misconduct, gross negligence, fraud, or breach of the fiduciary duty of loyalty. The Trustee may, in connection with the performance of his/her/its functions and in his/her/its sole and absolute discretion, consult with his/her/its attorneys, accountants, financial advisors, and agents, and will not be liable for any act taken, omitted to be taken, or suffered to be done in accordance with advice or opinions rendered by such Persons. Notwithstanding such authority, the Trustee will not be under any obligation to consult with his/her/its attorneys, accountants, financial advisors, or agents, and his/her/its determination not to do so will not result in the imposition of liability on the Trustee unless such determination is based on the Trustee's recklessness, gross negligence, willful misconduct, or fraud.

- 2. No recourse will ever be had, directly or indirectly, against the Trustee personally, or against any employee, contractor, agent, attorney, accountant, or other professional retained in accordance with the terms of the Trust Agreement or the Plan by the Trustee, by legal or equitable proceedings or by virtue of any statute or otherwise, nor upon any promise, contract, instrument, undertaking, obligation, covenant or Trust Agreement whatsoever executed by the Trustee in implementation of the Trust Agreement or the Plan, it being expressly understood and agreed that all such liabilities, covenants, and Trust Agreements of the Trust whether in writing or otherwise, will be enforceable only against and be satisfied only out of the Trust Assets or such part thereof as will under the term of any such Trust Agreement be liable therefore or will be evidence only of a right of payment out of the Trust Assets. Notwithstanding the foregoing, the Trustee may be held liable for his/her/its recklessness, gross negligence, willful misconduct, knowing and material violation of law, or fraud; and if liability on such grounds is established, recourse may be had directly against the Trustee. The Trust will not be covered by a bond.
- 3. The Trust will defend, indemnify, and hold harmless the Trustee, his/her/its officers, directors, agents, representatives, and employees to the fullest extent that a corporation or trust organized under the laws of Minnesota is entitled to indemnify and defend its directors, trustees, officers, and employees against any and all liabilities, expenses, Claims, damages or losses incurred by them in the performance of their duties hereunder.
- 1. Additionally, the Reorganized Debtor, and each of its respective agents, who was or is a party, or is threatened to be made a party to any threatened or pending judicial, administrative, or arbitrative action, by reason of any act or omission of the Trust or Trustee or respective agents, with respect to: (i) the Chapter 11 case and any act or omission undertaken by them prior to the commencement thereof, (ii) the assessment or liquidation of any Class 3 and Class 4B Claims, (iii) the administration of the Trust and the implementation of the Trust Distribution Plan, or (iv) any and all activities in connection with the Trust Agreement, will be indemnified and defended by the Trust, to the same extent that a corporation or trust organized under the laws of Minnesota is from time to time entitled to indemnify and defend its own officers, directors, trustees, and employees, against reasonable expenses, costs and fees (including attorneys' fees and costs), judgments, awards, amounts paid in settlement and liabilities of all kinds incurred by the Debtor or Reorganized Debtor, and their respective professionals, officers, and directors, in connection with or resulting from such action, suit or proceeding, provided that, with respect to amounts paid in settlement, the Trust has approved such amounts in advance, such approval not to be unreasonably withheld.

2. Reasonable expenses, costs, and fees (including attorneys' fees and costs) incurred by or on behalf of a Trustee, the Debtor, the Reorganized Debtor, and their respective agents in connection with any action, suit, or proceeding, whether civil, administrative, or arbitrative, from which they are entitled to be indemnified by the Trust, will be paid by the Trust in advance of the final disposition thereof upon receipt of an undertaking, by or on behalf of such Trustee, the Debtor, the Reorganized Debtor, and their respective agents, to repay such amount in the event that it will be determined ultimately by Non-Appealable Order that such Trustee, the Debtor, the Reorganized Debtor, and their respective professionals, officers, and directors is not entitled to be indemnified by the Trust.

#### N. TREATMENT OF TORT CLAIMS

1. <u>Trust Liability</u>. On the Effective Date, the Trust will automatically and without further act or deed assume: (i) all liability, if any, of the Protected Parties and the Settling Insurers in respect of Channeled Claims, subject to section 16.4 of the Plan; (ii) the responsibility for preserving, managing and distributing Trust Assets pursuant to the Trust Distribution Plan; and (iii) the right to pursue the Transferred Insurance Interests. Except as otherwise provided in the Plan, the Trust does not assume any liabilities of the Diocese or Reorganized Debtor, in whole or in part, in regards to any Tort Claims that are not released nor the liabilities of the Settling Insurers.

#### 2. Assessment of Tort Claims.

- 1. Each Tort Claim will be assessed by the Tort Claims Reviewer in accordance with the Trust Distribution Plan to determine whether the Tort Claimant is entitled to a distribution under the Trust. The Diocese or the Reorganized Debtor will reasonably cooperate with the Tort Claims Reviewer and the Trustee as requested by the Tort Claims Reviewer or the Trustee in connection with any inquiries by either in the administration of the Trust Distribution Plan, but will not be required to act in any way that violates any duty to cooperate with a Non-Settling Insurer. Under no circumstance will the Tort Claims Reviewer's review of a Class 3 or Class 4B Claim or a determination regarding a distribution thereon have any effect on the rights of a Non-Settling Insurer.
- 2. Each Tort Claimant may elect, in lieu of assessment by the Tort Claims Reviewer, to have his Tort Claim treated pursuant to the convenience Claim process as provided by the Trust Distribution Plan ("Convenience Claim").
- 3. <u>Election</u>. No later than thirty (30) days after a Tort Claimant is notified of the amount of the award under the Trust Distribution Plan, the Tort Claimant will elect in writing one of the following treatment alternatives:
- 1. Receiving a payment from the Trust in the amount determined by the Tort Claims Reviewer pursuant to the Trust Distribution Plan ("**Distribution Plan Claim**"). A Tort Claimant who elects to receive a distribution for a Distribution Plan Claim (a "**Distribution Plan Claimant**") must execute the release of all his or her Tort Claims against the Settling Insurers and the Protected Parties as set forth in Plan **Exhibit E** or Plan **Exhibit F**, as applicable, and waives his right to pursue a direct action under Minn. Stat. § 60A.08, subd. 6 or other applicable law against any Non-Settling Insurer; or

2. Treatment of the Tort Claim as a Litigation Claim. A Tort Claimant electing treatment as a Litigation Claim ("Litigation Claimant") will have rights, to the extent set forth in the Trust Distribution Plan, to initial and future distributions from the Trust. Each Litigation Claimant also retains the right to: (i) pursue his or her Tort Claim for its full amount according to proof in order to determine the liability of any Protected Party for purposes of recovering against any Non-Settling Insurer that is or may be liable on the Tort Claim and (ii) proceed in a direct action against any Non-Settling Insurer to the extent allowed by applicable law, including Minn. Stat. § 60A.08, subd. 6 (each a "Litigation Claim"). A Litigation Claimant's recovery on a Litigation Claim is limited as provided herein. The Settling Insurers will not be obligated to defend or indemnify any Person in connection with a Litigation Claim and the Settling Insurers will not have any other duties or obligations to any Person in connection with a Litigation Claim. Under no circumstances will a Tort Claimant or any other Person be able to recover any amount from the Settling Insurers in connection with a Litigation Claim.

#### 4. Modification of Treatment Election.

- 1. If a Tort Claimant does not make one of the elections in Section 6.14(c) of the Plan, the Tort Claimant will be treated as a Litigation Claimant.
- 2. Upon written notice to the Trustee, subject to the Trustee's sole and absolute discretion, a Tort Claimant may rescind the election to be treated as a Litigation Claimant in favor of being treated as a Distribution Plan Claimant. Notwithstanding the foregoing, the Trustee will consent to a Tort Claimant's rescission if such written notice of rescission is given prior to entry of an order of dismissal or a final judgment on the Litigation Claim in favor of a Protected Party.
- 3. No later than sixty (60) days after being notified of the amount of their award under the Trust Distribution Plan, a Tort Claimant may rescind the election to be treated as a Distribution Plan Claimant in favor of being treated as a Litigation Claimant.
- 5. Legal Effect of Estimation of Claims and Distributions under Trust Distribution Plan. The Tort Claims Reviewer's determinations are for estimation purposes only and will not be a finding or fixing of the fact or liability or the amount payable for any Tort Claim with any binding legal effect, other than for distribution purposes by the Trust pursuant to the Trust Distribution Plan. The determination of qualification, estimation of claims, and payment of distributions is not an admission of liability by any Protected Party or the Trust with respect to any Tort Claims and has no res judicata or collateral estoppel effect on any Protected Party, the Trust, or any Non-Settling Insurer. Any payments by the Trust to Tort Claimants in connection with their Tort Claims is not a release of the Debtor nor an accord or novation of the Debtor's liability on account of the Class 3 and Class 4B Claims. The Trust's act of making a distribution is immaterial to, and will not be construed as, a determination or admission of the Diocese's, the Reorganized Debtor's, or any other Protected Party's liability for, or damages with respect to, any Class 3 or Class 4B Claim. The determination of qualification, estimation of claims, and payment of distributions is not a settlement, release, accord, or novation of Class 3 or Class 4B Claims and cannot be used by any third party as a defense to any alleged joint liability. The determination of qualification, estimation of claims, and payment of partial distributions does not impair a Litigation Claimant's rights to obtain a judgment, including a judgment based on joint and several liability,

against a Protected Party or any Non-Settling Insurer or other Person, for purposes of establishing the Protected Party's liability on the Tort Claim, but any such judgment awarded to a Litigation Claimant will be reduced by the amount of distributions already paid by the Trust to such Litigation Claimant on his or her Tort Claim(s). Neither the Tort Claims Reviewer's review of a Tort Claim and determination of qualification, nor the Trust's estimation of claims or payment of distributions will (1) constitute a trial, an adjudication on the merits, or evidence of liability or damages in any litigation with the Protected Parties, Non-Settling Insurers, or any other Person, or (2) constitute, or be deemed, a determination of the reasonableness of the amount of any Tort Claim, either individually or in the aggregate with other Tort Claims, in any coverage litigation with any Non-Settling Insurers. The Trust's estimation of claims and payment of distributions does not constitute a triggering event for liability under any Non-Settling Insurer Policy nor does it create an admission of the fact of liability or the extent of damages on behalf of the Protected Parties.

payment on any award unless and until such Tort Claimant has executed the Release attached as **Exhibit E** or **Exhibit F** to the Plan, as applicable. Each Tort Claimant must execute a release of all claims against the Settling Insurers and must release all claims against the Diocese, the Reorganized Debtor, and the other Protected Parties that do not implicate insurance coverage under Non-Settling Insurer Policies. To preserve coverage under Non-Settling Insurer Policies, Tort Claimants specifically reserve, and do not release, any and all claims that they may have against the Protected Parties that implicate coverage under Non-Settling Insurer Policies, but recourse is limited to the proceeds of Non-Settling Insurer Policies and all other damages (including extracontractual damages), awards, judgments in excess of policy limits, penalties, punitive damages and attorney's fees and costs that may be recoverable against any Non-Settling Insurers because of their conduct concerning insurance coverage for, or defense or settlement of, any Tort Claim, and any such judgments or awards will be handled in accordance with Sections 6.14(i) and (j) of the Plan.

The Tort Claims will not be released or enjoined as against the Protected Parties for any Abuse that may be covered under Non-Settling Insurer Policies until such claims are settled with the Protected Parties and their Non-Settling Insurers, or are fully adjudicated, resolved, and subject to Final Order, unless the Tort Claimant elects to proceed as a Distribution Plan Claimant, but recourse will be limited as described above.

With respect to all other Claims, except as otherwise provided in the Plan, the Debtor's liability on account of such Claims will be discharged pursuant to the provisions of 1141(d). The Tort Claimants' release, in whole or in part, of their Class 3 or Class 4B Claims will be pursuant to the principles set forth in *Pierringer v. Hoger*, 124 N.W.2d 106 (Wis. 1963) and *Frey v. Snelgrove*, 269 N.W.2d 918 (Minn. 1978). The Class 3 and Class 4B Claimants will expressly reserve their rights against other Persons (other than Protected Parties), including joint tortfeasors, who will remain severally liable on any Class 3 and Class 4B Claims. Any Person (other than a Protected Party) that is, or was alleged to be a joint tortfeasor with any of the Protected Parties in connection with the Abuse that forms the basis of a Tort Claim will not be liable for any Protected Party's share of causal liability or fault.

7. <u>Trust Rights Against Non-Settling Insurers</u>. The Trust retains the right to pursue Non-Settling Insurers for coverage of the Diocese's, the Reorganized Debtor's, and any

Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 48 of 222

other Protected Party's liability to Tort Claimants regardless of whether the Tort Claimants elect treatment as Distribution Plan Claimants or Litigation Claimants.

- 8. <u>Distributions to Tort Claimants</u>. A Tort Claimant electing to be treated as a Distribution Plan Claimant, and who the Tort Claims Reviewer determines to be entitled to a distribution, will receive a distribution from the Trust in the amount(s) and at the time(s) provided for in the Trust Distribution Plan. Any payment on a Tort Claim constitutes payment for damages on account of personal physical injuries or sickness arising from an occurrence, within the meaning of Section 104(a)(2) of the Internal Revenue Code of 1986, as amended.
- 9. Litigation Claims. If necessary in the Trustee's discretion, the Trustee may establish a reserve for payment of a claim held by a Litigation Claimant in the amount that would have been awarded to the Litigation Claimant if such Claimant had elected to proceed under the Trust Distribution Plan. The creation and existence of this reserve is not a settlement, release, accord or novation of the Litigation Claims and cannot be used by any third party as a defense to any alleged joint liability with any Protected Party. For avoidance of doubt, the creation and existence of this reserve does not affect, diminish or impair a Litigation Claimant's rights to collect a judgment, including a judgment based on joint and several liability, against any Non-Settling Insurer or Person that is not a Protected Party, except as expressly provided herein. The Trustee may establish one reserve for all of the Litigation Claims but no Litigation Claimant will have any interest in any portion of the reserve in excess of the amount determined for that Litigation Claimant under the Trust Distribution Plan, and then only in the event that the Litigation Claimant prevails on his Litigation Claim. Neither the Trust's payment of, or reserving monies on account of, the Tort Claims nor the Tort Claims Reviewer's review of a Tort Claim, will: (1) constitute a trial, an adjudication on the merits, or evidence of liability or damages in any litigation with the Protected Parties, Non-Settling Insurers, or any other Person, or (2) constitute, or be deemed, a determination of the reasonableness of the amount of any Tort Claim, either individually or in the aggregate with other Tort Claims, in any coverage litigation with any Non-Settling Insurers.
- 1. In the event that a Litigation Claimant obtains a judgment against any Protected Party and no Non-Settling Insurer is implicated by the Litigation Claim, the judgment will be satisfied by the Trust in the amount of such judgment against such Protected Party, up to the amount of any reserve set for that Litigation Claimant's Litigation Claim plus an additional \$1,000.
- 2. In the event that any Non-Settling Insurer is implicated by the Litigation Claim, and either a settlement is achieved with such Non-Settling Insurer(s) as to such Litigation Claim or the Litigation Claimant obtains a judgment against a Protected Party and either the Trust or the Litigation Claimant obtains a recovery from any such Non-Settling Insurer(s) as to such judgment, then such recovery will be turned over to the Trust for handling pursuant to the Plan. Such recovery will first go to reimburse the Trust or the Litigation Claimant, as the case may be, for all costs (including attorneys' fees) incurred in connection with pursuing the recovery against the Non-Settling Insurer(s) relating to the Litigation Claim, so long as such amounts are reasonable and were agreed to in advance by the Trust. Any amount remaining will be distributed in a manner consistent with the Trust Distribution Plan.

Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 49 of 222

- 10. The Trust's payment to a Litigation Claimant that has recovered a judgment or settlement does not affect, diminish or impair the Trust's right to collect the policy proceeds respecting such Class 3 or Class 4B Claim from any Non-Settling Insurer, nor does it affect, diminish or impair the Trust's right to bring any claims against the Non-Settling Insurer that have been assigned to the Trust or that belong to the Trust by operation of law.
- If a Non-Settling Insurer has refused to defend a Protected Party with 11. respect to any Litigation Claim, the Trust will advance or reimburse the Protected Party for reasonable and necessary attorneys' fees and other expenses incurred in defending the Litigation Claim. If any Non-Settling Insurer has refused to indemnify a Protected Party with respect to any settlement or judgment of a Litigation Claim, the Trust will advance or reimburse the Protected Party for any judgment or settlement incurred by the Protected Party on such Litigation Claim, provided the Trust has consented in advance to any such settlement, such consent not to be withheld unreasonably. If all insurers that could potentially have responsibility to defend and/or indemnify for a Tort Claim have denied coverage, that Litigation Claimant must sign a covenant not to execute against that Protected Party's assets (other than Non-Settling Insurer Policies or proceeds or other assigned rights or interests), under which the Litigation Claimant will agree to seek recovery only from Non-Settling Insurers for any judgment the Litigation Claimant obtains against any Protected Party, in exchange for a stipulated judgment and assignment of insurance rights, as authorized by law. If any judgment on any Tort Claim is within the retention of any Non-Settling Insurer Policy, and all insurers have denied indemnity for such judgment, then the Trust will fund the judgment. The Trust's advancement or reimbursement of the Protected Party for such defense costs and/or judgment or settlement payments, and any distributions made by the Trust to the Litigation Claimant and other Class 3 Claimants, will not affect, diminish or impair the Trust's right to bring any claims against any Non-Settling Insurers for refusing to defend and/or indemnify the Protected Party, including but not limited to claims for payment of policy proceeds, bad faith, wrongful failure to settle, and extra-contractual damages authorized by law.
- 12. As of the Effective Date, the Trustee will be deemed to have the right to join or intervene into the Insurance Coverage Adversary Proceeding and to pursue recoveries against any Non-Settling Insurers.
- 13. Nothing in the Plan, Confirmation Order or any Plan Document will impose any obligation on any Non-Settling Insurer to provide a defense for, settle, or pay any judgment with respect to, any Tort Claim, or grant to any Person any right to sue any Non-Settling Insurer directly, in connection with a Tort Claim. All such obligations with respect to Non-Settling Insurers will be determined by and in accordance with the terms of the Non-Settling Insurer Policies and with applicable non-bankruptcy law. Nothing provided for in the Plan, the Trust Agreement, or the Trust Distribution Plan constitutes an endorsement of a Class 3 or Class 4B Claimant's right to pursue their remedies under Minn. Stat. § 60A.08.
- 14. If the Litigation Claimant fails to prosecute his Litigation Claim to final judgment or settlement of the claim, or a Final Order is entered finding that no Protected Party has liability to such Tort Claimant on account of his Tort Claim, any reserve maintained by the Trust on account of such Tort Claim will revert to the non-reserved assets of the Trust and the Litigation Claimant will have no recourse against the Trustee, the Trust, any Protected Party, or any Settling Insurer.

#### O. OBJECTIONS AND LITIGATION AFTER THE EFFECTIVE DATE.

- 1. Regardless of whether a Class 3 or Class 4B Claimant elects treatment as a Distribution Plan Claimant or a Litigation Claimant, the Trustee may object to that Class 3 or Class 4B Claimant's Claim. The Trustee's right to object to a Class 3 or Class 4B Claimant's Claim after the Effective Date will not affect or impair any right the Diocese, the Reorganized Debtor, other Protected Parties, and/or Non-Settling Insurers may have under the Non-Settling Insurer Policies or applicable law to object to such Class 3 and Class 4B Claims.
- 2. The Protected Parties will comply with all obligations under the Non-Settling Insurer Policies and applicable law. The Trustee, to the extent required by the Non-Settling Insurer Policies implicated by such Tort Claims and applicable law, will also cooperate with the Non-Settling Insurer in the defense of such judicial proceeding contemplated in Section 6.13(p)(1) of the Plan. In the event of a dispute between a Non-Settling Insurer and the Trustee regarding whether the Trustee must allow such Non-Settling Insurer to control the defense of such Tort Claim, or the extent of anyone's duty to cooperate, such dispute will be resolved by the Bankruptcy Court and the Bankruptcy Court will retain jurisdiction over such disputes. In the event the Non-Settling Insurer fails to seek a determination from the Bankruptcy Court over the existence and extent of the Trustee's obligation to cooperate, or the Non-Settling Insurer's right to control the defense, the Non-Settling Insurer will be deemed to have waived such claim.
- P. <u>CLAIM WITHDRAWAL</u>. A Tort Claimant may withdraw his or her Tort Claim at any time on written notice to the Trustee. If withdrawn, (a) the Tort Claim will be withdrawn with prejudice and may not be reasserted, and such Tort Claimant will still be subject to Section 13.2 of the Plan, the Channeling Injunction, and the Supplemental Settling Insurer Injunction as provided by the Plan; and (b) any reserve maintained by the Trust on account of such Tort Claim will revert to the Trust as a Trust Asset for distribution in accordance with the Plan and Trust Distribution Plan. Each Protected Party, Non-Settling Insurer, Settling Insurer, and the Trust will retain any and all defenses that may exist in respect to such Tort Claim.

#### X. SETTLING INSURERS

- **A.** <u>SETTLING INSURER SETTLEMENT AGREEMENT</u>. Upon satisfaction of the conditions precedent to the LMI/Interstate Settlement Agreement becoming effective, including the Confirmation Order and the order approving the LMI/Interstate Settlement Agreement becoming Non-Appealable Orders, the LMI/Interstate Settlement Agreement will be fully binding on the Trust, Protected Parties, the Reorganized Debtor, the Committee, LMI/Interstate, the Tort Claimants, and parties in interest, and any of the foregoing Persons' successors.
- **B.** SALE FREE AND CLEAR OF INTERESTS OF SETTLING INSURER POLICIES. To the extent provided in the LMI/Interstate Settlement Agreement and effective on the later of (i) the Effective Date of the Plan or (ii) the payment by each Settling Insurer of the settlement payment(s) due under such agreement, each and every Settling Insurer Policy issued by LMI or Interstate will be sold to the issuing Settling Insurer pursuant to Sections 105, 363, and 1123 of the Bankruptcy Code, free and clear of all liens, Claims and Interests, including those of the Diocese, Diocesan Parishes, and Tort Claimants. As set forth in the LMI/Interstate Settlement Agreement and the

corresponding Approval Order, LMI and Interstate are good faith purchasers of such insurance policies and certificates of insurance within the meaning of Section 363(m) of the Bankruptcy Code, the consideration exchanged constitutes a fair and reasonable settlement of the Parties' disputes and of their respective rights and obligations relating to each such Settling Insurer Policy and constitutes reasonably equivalent value, the releases in the LMI/Interstate Settlement Agreement and the policy buyback comply with the Bankruptcy Code and applicable non-bankruptcy laws, and each such Settling Insurer Policy will be terminated and be of no further force and effect with the issuing Settling Insurer having fully and completely performed any and all obligations under each such Settling Insurer Policy, including any performance owed to the Diocese and Diocesan Parishes, and all limits of liability of each such Settling Insurer Policy will be exhausted.

- C. <u>RESOLUTION OF CLAIMS INVOLVING SETTLING INSURERS</u>. The Confirmation Order will provide that within 20 days after receipt of the settlement payment required by the LMI/Interstate Settlement Agreement, the Diocese and LMI/Interstate will effect dismissal with prejudice of their Claims against each other in the Insurance Coverage Adversary Proceeding, and will be prohibited from pursuing the claims therein against each other at any time, with each side to bear its own fees and costs. The Diocese will not be required to dismiss the Insurance Coverage Adversary Proceeding as against any Non-Settling Insurers.
- **D.** <u>THE SETTLING INSURERS' PAYMENTS</u>. LMI/Interstate will pay to the Trust the sums set forth in the LMI/Interstate Settlement Agreement within the time set forth in the LMI/Interstate Settlement Agreement.

#### E. <u>Judgment Reduction</u>

- 1. In any proceeding, suit, or action, including the Insurance Coverage Adversary Proceeding, to recover or obtain insurance coverage or proceeds from a Non-Settling Insurer for a Tort Claim, the following will apply:
- 1. If a Non-Settling Insurer has asserted, asserts, or could assert, any Contribution Claim against a Settling Insurer, then any judgment or award obtained by any Protected Party, the Trust, or a Tort Claimant against such Non-Settling Insurer will be automatically reduced by the amount, if any, that a court or arbitrator determine such Settling Insurer would have been liable to pay such Non-Settling Insurer as a result of its Contribution Claim, so that the Contribution Claim is thereby satisfied and extinguished entirely ("Reduction **Amount**"). In any action by a Protected Party, the Trust or a Tort Claimant against a Non-Settling Insurer to obtain insurance coverage or proceeds for a Tort Claim, where the Settling Insurers are not parties, the Non-Settling Insurers' Contribution Claim may be asserted as a defense, and to the extent the Non-Settling Insurers' Contribution Claim against a Settling Insurer is determined to be valid by the court presiding over such action, the liability of the Non-Settling Insurer will be reduced dollar for dollar by the amount so determined. In the event that a reduction is not made as described above, then any Contribution Claim by any Non-Settling Insurer against any Settling Insurer will be determined by the court or arbitration proceeding in which such Contribution Claim is filed. To the extent possible, no Settling Insurer will be required to answer or otherwise respond to a complaint alleging a Contribution Claim against such Settling Insurer until such Reduction Amount is determined by such court or arbitrator(s). If, notwithstanding the foregoing, a court

Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 52 of 222

refuses to reduce the liability of the Non-Settling Insurer, then once the order establishing the Settling Insurer's liability for the Contribution Claim is a Final Order, the Trust will promptly indemnify and hold harmless the Settling Insurer for such amount of such Contribution Claim.

- 2. As provided in the LMI/Interstate Settlement Agreement, each Settling Insurer agrees that it will not pursue any contribution claim that it might have against any other insurer.
- Effective Date, the Reorganized Debtor and the Settling Insurers will be authorized to enter into, execute, adopt, deliver, or implement all notes, contracts, security agreements, instruments, releases, and other agreements or documents necessary to effectuate or memorialize the settlements contained in Article VII of the Plan without further order of the Bankruptcy Court. The Reorganized Debtor and the Settling Insurers may make technical or immaterial alterations, amendments, modifications, waivers, or supplements to the terms of any Insurance Settlement Agreement. A class of Claims that has accepted the Plan will be deemed to have accepted the Plan, as altered, amended, modified, or supplemented under Section 7.6 of the Plan, if the proposed alteration, amendment, modification, or supplement does not materially and adversely change the treatment of the Claims within such class. An order of the Bankruptcy Court approving any amendment or modification made pursuant to Section 7.6 of the Plan will constitute an order in aid of consummation of the Plan and will not require the re-solicitation of votes on the Plan.
- G. <u>INDEMNIFICATION OBLIGATIONS</u>. With respect to the indemnification obligations of the Trust or the Reorganized Debtor, including with respect to Tort Claims made by Persons over whom the Diocese or Diocesan Parishes do not have control, including any other Person who asserts Tort Claims against or right to coverage under the Settling Insurer Policies, The Settling Insurers may undertake the defense of any Claim on receipt of such Claim without affecting such indemnification obligations. The Settling Insurers shall notify the Trust or Reorganized Debtor, as applicable, as soon as practicable of any such Claims identified in this section and of their choice of preferred counsel. The obligation of the Trust or Reorganized Debtor, as applicable, to indemnify the Settling Insurer under Section 7.7 of the Plan will not exceed the settlement amount set forth in the Insurance Settlement Agreement as actually paid by the corresponding Settling Insurer. In defense of any such Claims, the Settling Insurers may settle or otherwise resolve a Claim consistent with the terms of this Plan and with the prior consent of the indemnifying party, which consent shall not be unreasonably withheld. Any such indemnification obligations with respect to Tort Claims will be channeled to and paid by the Trust.

#### H. WAIVER/ CONSENT

1. In consideration of the releases and Channeling Injunction, the Supplemental Settling Insurer Injunction, and other covenants set forth herein, subject to the occurrence of the Effective Date and the satisfaction of the other conditions precedent to the effectiveness of the LMI/Interstate Settlement Agreement, and upon receipt by the Trust of the settlement payment required by the LMI/Interstate Settlement Agreement, each of the Protected Parties:

Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 53 of 222

- 1. Irrevocably and unconditionally, without limitation, releases, acquits, forever discharges, and waives any Claims and/or Interests it has or might have now or in the future against the other Protected Parties and the Settling Insurers with respect to any contribution, subrogation, indemnification, or other similar Claim arising from or relating to released Tort Claims covered or alleged to be covered under the Settling Insurer Policies, and any Settling Insurer Policies; and
- 2. Consents to the sale of the Protected Parties' Claims and/or Interests, if any, in the Settling Insurer Policies issued by LMI and Interstate in accordance with the LMI/Interstate Settlement Agreement and to the contribution of the proceeds from such sale and settlement to the Trust, as provided in the Plan.
- 2. Nothing in Section 7.8 of the Plan will be construed to bar either (i) a Claim based on Abuse against a Person who is not a Protected Party or a Settling Insurer or (ii) a Claim by such Person for insurance coverage in connection with a Claim described in the foregoing subsection (i) under an insurance policy other than the Settling Insurer Policies.
- be made by the Settling Insurers and other consideration provided by each Settling Insurer, upon payment by the Settling Insurers of their respective settlement amounts under the LMI/Interstate Settlement Agreement, the Diocese irrevocably and unconditionally, without limitation, releases, acquits, forever discharges, and waives any Interests it has or might have now or in the future (i) under the Settling Insurer Policies issued by LMI and Interstate to the extent those Settling Insurer Policies are bought back under the LMI/Interstate Settlement Agreement and the Plan; (ii) against LMI/Interstate with respect to any Tort Claim; and (iii) against the other Protected Parties with respect to any Channeled Claim. For the avoidance of doubt, the Diocese irrevocably and unconditionally, without limitation, releases, acquits, forever discharges, and waives any Interests it has or might have now or in the future under the Settling Insurer Policies issued by, subscribed to, or underwritten by LMI and/or Interstate, which are bought back under the terms of the LMI/Interstate Settlement Agreement and the Plan.

#### J. SUPPLEMENTAL SETTLING INSURER INJUNCTION.

1. <u>Supplemental Injunction Preventing Prosecution of Claims Against Settling Insurers</u>. Pursuant to Sections 105(a) and 363 of the Bankruptcy Code and in consideration of the undertakings of the Settling Insurers pursuant to the Insurance Settlement Agreements, including LMI's and Interstate's purchases of insurance policies or Interests in insurance policies free and clear of all interests pursuant to Section 363(f) of the Bankruptcy Code:

Any and all Persons who have held, now hold or who may in the future hold any Interests (including all debt holders, all equity holders, governmental, tax and regulatory authorities, lenders, trade and other creditors, Tort Claimants, perpetrators, and all others holding Interests of any kind or nature whatsoever, including those Claims released or to be released pursuant to the Insurance Settlement Agreements) against any of the Settling Insurers, or any other Person covered or allegedly covered under the Settling Insurer Policies, including Contribution Claims, Extra-Contractual Claims, and Indirect Claims, are hereby permanently stayed, enjoined, barred, and

restrained from taking any action, directly or indirectly, to assert, enforce or attempt to assert or enforce any such Interest against the Settling Insurers or the Settling Insurer Policies:

- 1. Commencing or continuing in any manner any action or other proceeding against the Settling Insurers or the property of the Settling Insurers;
- 2. Enforcing, attaching, collecting, or recovering, by any manner or means, any judgment, award, decree or order against the Settling Insurers or the property of the Settling Insurers;
- 3. Creating, perfecting, or enforcing any lien of any kind against the Settling Insurers or the property of the Settling Insurers;
- 4. Asserting or accomplishing any setoff, right of indemnity, subrogation, contribution, or recoupment of any kind against any obligation due the Settling Insurers or the property of the Settling Insurers; and
- 5. Taking any action, in any manner, in any place whatsoever, that does not conform to, or comply with, the provisions of the Plan.

For the avoidance of doubt, this Supplemental Settling Insurer Injunction bars only the above-referenced actions against the Settling Insurers or the Settling Insurer Policies issued by LMI or Interstate. The foregoing injunctive provisions are an integral part of the Plan and are essential to its implementation.

### XI. <u>NON-SETTLING INSURERS</u>

#### A. PRESERVATION OF RIGHTS AND OBLIGATIONS

In the event: (i) a Tort Claim is pursued in state or federal court by a Tort 1. Claimant against a Protected Party or Non-Settling Insurer, or (ii) the Trust asserts an objection to or otherwise seeks a determination of liability as to a Tort Claim, then the Protected Parties, the Trust and each Non-Settling Insurer will retain any and all legal and factual defenses that may exist in respect to such Tort Claim and, except as set forth in Section 8.1 of the Plan, all coverage defenses. The rights, duties and obligations of each Non-Settling Insurer under the Non-Settling Insurer Policies with respect to Tort Claims are not impaired, altered, reduced, or diminished by: (a) the discharge in bankruptcy of the Debtor; (b) any distribution to Tort Claimants pursuant to the Plan, the Trust Agreement, and the Trust Distribution Plan; (c) the transfer of the Protected Parties' Transferred Insurance Interests; (d) protections granted to Protected Parties and the Settling Insurers under the Plan; or (e) any other provision of the Plan, the Trust Agreement, or the Trust Distribution Plan. Non-Settling Insurers retain any defenses that they would be able to raise if the claim for coverage were brought by the Diocese, the Reorganized Debtor, or any other Protected Party, except any defense (1) related to the transfer of Transferred Insurance Interests to the Trust; (2) effected by operation of bankruptcy law as a result of confirmation; or (3) based on facts determined by the Bankruptcy Court in connection with the confirmation.

- 2. The rights and obligations of the Diocese, the Reorganized Debtor, and other insureds and every Non-Settling Insurer under the terms of the Non-Settling Insurer Policies and at law (including without limitation any duty of an insured to cooperate) will not be affected by the assessment of any Tort Claim and will be treated as if the Tort Claim had never been assessed for distribution purposes by the Trust.
- 3. Each Non-Settling Insurer will be entitled to all rights as are provided under the terms of its Non-Settling Insurer Policies as if the Tort Claim had never been assessed for distribution purposes by the Trust.
- 4. After the Effective Date, upon consent of the Trustee, a Person may become a Settling Insurer if the Bankruptcy Court, after notice and hearing, approves the agreement between the Person, the Reorganized Debtor and the Trustee. After the Effective Date, the Trustee will have the exclusive authority to seek approval of such agreement. Upon the Bankruptcy Court's entry of a Final Order approving such agreement, **Exhibit N** of the Plan will be amended by the Trustee to include such Person. Any such Person will have all of the rights, remedies and duties of a Settling Insurer notwithstanding that such Person originally may have been a Non-Settling Insurer under any provision of the Plan. Such rights, remedies and duties will include the terms and conditions of the Channeling Injunction and Supplemental Settling Insurer Injunction. The Bankruptcy Court's retained jurisdiction to approve an agreement under Article XIII of the Plan will include jurisdiction to determine the adequacy of notice of a motion to approve such an agreement.
- **B.** <u>ESTIMATIONS/ASSESSMENTS ARE NOT BINDING</u>. Estimations of Class 3 and Class 4B Claims for purposes of voting, and the determination of qualification, assignment of points, and payment of distributions of Tort Claims under the Trust Distribution Plan:
- 1. Will not (i) constitute an admission of liability by any Person with respect to such Claims; (ii) have any res judicata or collateral estoppel effect on any Person; (iii) constitute a settlement, release, accord, satisfaction, or novation of such Claims; (iv) be used by any third-party as a defense to any alleged joint liability; or (v) otherwise prejudice any rights of the Trust, Protected Parties, Settling Insurers, Non-Settling Insurers, and Claimants in all other contexts or forums;
- 2. Will be without prejudice to any and all rights of the Trust, the Diocese, the Reorganized Debtor, other Protected Parties, the Non-Settling Insurers and Tort Claimants in all other contexts and forums and will not be deemed to be a determination of liability of the Diocese or a determination of whether, or the extent to which, such claim is covered under any Non-Settling Insurer Policy. The fact that a claim has been estimated for distribution purposes has no res judicata or collateral estoppel effect and is not a binding determination on any issue or the creation of a liquidated non-bankruptcy claim. The assessment by the Tort Claims Reviewer under the Trust Distribution Plan will have no effect upon any "no action" provisions contained in any Non-Settling Insurer Policy to the extent any such provision remains enforceable by a Non-Settling Insurer under applicable law. Rather, the liability of the Diocese, the Reorganized Debtor, or any other Protected Party and the amount owed by the Diocese, the Reorganized Debtor, other Protected Parties, and any Non-Settling Insurer on any Class 3 or Class 4B Claim, will be determined by: (i) the amount of any court judgment obtained by the Class 3 or Class 4B Claimant;

- or (ii) through a settlement agreement either to which such Non-Settling Insurer has consented or, if such Non-Settling Insurer has not consented, a settlement agreement which does not breach any duty of the Trust, Trustee, Diocese, the Reorganized Debtor, or any other Protected Party to the Non-Settling Insurer under the respective Non-Settling Insurer Policy or applicable law.
- **C.** <u>RIGHTS UNDER INSURANCE SETTLEMENT AGREEMENTS</u>. The rights of the parties under any Insurance Settlement Agreement will be determined exclusively under the applicable Insurance Settlement Agreement and those provisions of the Approval Order approving such Insurance Settlement Agreements, the Plan and the Confirmation Order.
- D. THE PLAN IS NEUTRAL AS TO NON-SETTLING INSURER POLICIES. For the avoidance of doubt, solely with respect to the Non-Settling Insurers, nothing in the Plan, the Trust Agreement, the Trust Distribution Plan, the Confirmation Order, or any other order of the Bankruptcy Court to the contrary (including any other provision that purports to be preemptory or supervening or grants a release): (i) will affect, impair, or prejudice the rights and defenses of any Non-Settling Insurer, any Protected Party, the Trust, or any other insureds under Non-Settling Insurer Policies in any manner, including any defenses to any claim for insurance; (ii) will constitute a settlement or resolution of any Protected Party's liability to a Tort Claimant; (iii) will in any way operate to, or have the effect of, impairing or having any res judicata, collateral estoppel, or other preclusive effect on, any party's legal, equitable, or contractual rights or obligations under any Non-Settling Insurer Policy; or (iv) will otherwise determine the applicability or nonapplicability of any provision of any Non-Settling Insurer Policy and any such rights and obligations will be determined under the Non-Settling Insurer Policy and applicable law.
- E. THE DIOCESE'S OBLIGATIONS SURVIVE. Notwithstanding the transfer of the Transferred Insurance Interests to the Trust, the Diocese will not be relieved of its duties or obligations under any Non-Settling Insurer Policies (except as provided to the contrary in any subsequent Insurance Settlement Agreement), and will continue to perform such duties as required by such Non-Settling Insurer Policies and applicable law. If the Trust asserts any claim that the Diocese has breached such duties or obligations under the Non-Settling Insurer Policies resulting in a loss of coverage, it will give the Diocese notice and an opportunity to cure any alleged breach, and in any event, the Diocese will not be liable for any alleged breach resulting in a loss of coverage except to the extent that (i) the breach relates to post-Effective Date conduct of the Diocese, and (ii) the Diocese willfully or intentionally fails to comply with its continuing obligations under the Non-Settling Insurer Policies. In addition, any such claim will not be automatically allowed; the Diocese will have the right to defend against such claim.

#### F. TRUST POWERS WITH RESPECT TO TORT CLAIMS AND NON-SETTLING INSURERS

1. A Tort Claimant or the Trust, as applicable, may enter into a settlement of a Tort Claim allowed by applicable non-bankruptcy law including but not limited to settlements consistent with *Miller v. Shugart*, 316 N.W.2d 729 (Minn. 1982) or *Drake v. Ryan*, 514 N.W.2d 785 (Minn. 1994), and may enter into an arrangement with Tort Claimant's counsel providing such counsel will receive reasonable compensation from any recovery from a Non-Settling Insurer as provided in Section 4.3 of the Plan.

- The Trustee may use the Trust Assets to prosecute litigation against the Non-Settling Insurers.
- If the Trust successfully resolves an insurance coverage dispute with a Non-3. Settling Insurer or otherwise receives a recovery of insurance proceeds relating to Tort Claim(s) from a Non-Settling Insurer, such proceeds will become Trust Assets available to pay, and will increase the amount available to pay, Tort Claims, pursuant to the Trust Distribution Plan. In such event, and on a periodic basis accumulating all such recoveries, the Trust will make supplemental payments to Tort Claimants in accordance with the Trust Agreement and Trust Distribution Plan.

### XII. **INSURANCE POLICIES**

- **CONTINUATION OF INSURANCE POLICIES.** Except to the extent any Diocese Entity Insurance Policies are bought back as set forth in and pursuant to any Insurance Settlement Agreement or as otherwise provided by the terms of the Plan, all Diocese Entity Insurance Policies (including, without limitation, any Non-Implicated Insurance Policies) will, as applicable, either be deemed assumed by the Reorganized Debtor pursuant to Sections 365, 1123(a)(5)(A), and 1123(b)(2) of the Bankruptcy Code to the extent such Diocese Entity Insurance Policy is or was an executory contract of the Diocese, or continued in accordance with its terms pursuant to Section 1123(a)(5)(A) of the Bankruptcy Code, to the extent such Diocese Entity Insurance Policy is not an executory contract of the Diocese, such that each of the parties' contractual, legal, and equitable rights under each such Diocese Entity Insurance Policy will remain unaltered. All known Diocese Entity Insurance Policies are listed on Plan Exhibit I. To the extent that any or all such Diocese Entity Insurance Policies are considered to be executory contracts, then the Plan will constitute a motion to assume such Diocese Entity Insurance Policies in connection with the Plan. Subject to the occurrence of the Effective Date, the Confirmation Order will approve such assumption pursuant to §§ 365(a), 1123(a)(5)(A), and 1123(b)(2) of the Bankruptcy Code and include a finding by the Bankruptcy Court that each such assumption is in the best interest of the Debtor, the Estate, and all parties in interest in this Chapter 11 case. Unless otherwise determined by the Bankruptcy Court pursuant to an order which becomes a Non-Appealable Order or agreed to by the parties thereto prior to the Effective Date, no payments are required to cure any defaults of the Diocese existing as of the Effective Date with respect to any Diocese Entity Insurance Policy. The Diocese reserves the right to seek rejection of any Diocese Entity Insurance Policy or other available relief prior to the Effective Date.
- THE PLAN IS NEUTRAL AS TO NON-IMPLICATED INSURANCE POLICIES. For the В. avoidance of doubt, solely with respect to the Non-Implicated Insurer, nothing in the Plan, the Trust Agreement, the Trust Distribution Plan, the Confirmation Order, or any other order of the Bankruptcy Court to the contrary (including any other provision that purports to be preemptory or supervening or grants a release): (i) will affect, impair, or prejudice the rights and defenses of the Non-Implicated Insurer, any Protected Party, or any other insureds under Non-Implicated Insurance Policies in any manner, including any defenses to any claim for insurance; (ii) will constitute a settlement or resolution of any Protected Party's liability to a Class 4A Claimant; (iii) will in any way operate to, or have the effect of, impairing or having any res judicata, collateral estoppel, or other preclusive effect on, any party's legal, equitable, or contractual rights or obligations under any Non-Implicated Insurance Policy; or (iv) will otherwise determine the

applicability or nonapplicability of any provision of any Non-Implicated Insurance Policy and any such rights and obligations will be determined under the Non-Implicated Insurance Policy and applicable law.

### XIII. PROCEDURES FOR GENERAL CLAIMS ADMINISTRATION

- A. RESERVATION OF RIGHTS TO OBJECT TO NON-TORT CLAIMS. Unless a Claim is expressly described as an allowed Claim pursuant to or under the Plan, or otherwise becomes an allowed Claim prior to the Effective Date, upon the Effective Date, the Reorganized Debtor or the Trustee, as applicable, will be deemed to have a reservation of any and all rights, Interests, and objections of the Diocese, the Committee, or the Estate to any and all Claims and motions or requests for the payment of or on account of Claims, whether administrative expense, priority, secured, or unsecured, including any and all rights, Interests and objections to the validity or amount of any and all alleged Claims, Liens, and Interests, whether under the Bankruptcy Code, other applicable law, or contract. The failure to object to any Claim in this Chapter 11 case will be without prejudice to the Reorganized Debtor's or the Trustee's, as applicable, right to contest or otherwise defend against such Claim in the Bankruptcy Court as set forth Section 10.1 of the Plan when and if such Claim is sought to be enforced by the holder of such Claim.
- В. **OBJECTIONS TO NON-TORT CLAIMS.** Prior to the Effective Date, the Diocese will have the authority to pursue any objection to the allowance of any non-Tort Claim. From and after the Effective Date, the Reorganized Debtor will retain responsibility for administering, disputing, objecting to, compromising, or otherwise resolving and making distributions, if any, with respect to non-Tort Claims (including those Claims that are subject to objection by the Diocese as of the Effective Date); provided, however, that nothing in Section 10.2 of the Plan will affect the right of any party-in-interest (including the Reorganized Debtor and the Trustee) to object to any non-Tort Claim to the extent such objection is otherwise permitted by the Bankruptcy Code, the Bankruptcy Rules, and the Plan. Unless otherwise provided in the Plan or by order of the Bankruptcy Court, any objections to non-Tort Claims will be filed and served not later than thirty (30) days after the later of: (i) the Effective Date, or (ii) the date such Claim is filed. Such deadline or any Bankruptcy Court approved extension thereof, may be extended upon request by the Reorganized Debtor by filing a motion without any requirement to provide notice to any Person, based upon a reasonable exercise of the Reorganized Debtor's business judgment. A motion seeking to extend the deadline to object to any Claim will not be deemed an amendment to the Plan. Notwithstanding the foregoing, there will be no deadline for the Reorganized Debtor to object to Class 4A or Class 6 Claims.
- C. <u>DETERMINATION OF CLAIMS</u>. From and after the Effective Date, any Claim that is not a Tort Claim, and as to which a Proof of Claim or motion or request for payment was timely filed in this Chapter 11 case, or deemed timely filed by order of the Bankruptcy Court, may be determined and (so long as such determination has not been stayed, reversed, or amended, as to which determination (or any revision, modification, or amendment thereof) the time to appeal or seek review or rehearing has expired, (and as to which no appeal or petition for review or rehearing was filed or, if filed, remains pending)), liquidated pursuant to: (i) an order of the Bankruptcy Court; (ii) applicable bankruptcy law; (iii) agreement of the parties without the need for Bankruptcy Court approval; (iv) applicable non-bankruptcy law; or (v) the lack of (a) an objection

to such Claim, (b) an application to equitably subordinate such Claim, and (c) an application to otherwise limit recovery with respect to such Claim, filed by the Diocese, the Reorganized Debtor, or any other party in interest on or prior to any applicable deadline for filing such objection or application with respect to such Claim. Any such Claim so determined and liquidated will be deemed to be an allowed Claim for such liquidated amount and will be satisfied in accordance with the Plan. Nothing contained in Section 10.3 of the Plan will constitute or be deemed a waiver of any Claims, rights, Interests, or Causes of Action that the Debtor or the Reorganized Debtor may have against any Person in connection with or arising out of any Claim or Claims, including any rights under 28 U.S.C. § 157.

- **D.** <u>No Distributions Pending Allowance</u>. No payments or distributions will be made with respect to a Disputed Claim, or any portion thereof, unless and until all objections to such Disputed Claim have been settled or withdrawn or have been determined by an order which has become a Non-Appealable Order, and the Disputed Claim has become an allowed Claim.
- **E.** <u>CLAIM ESTIMATION</u>. To effectuate distributions pursuant to the Plan and avoid undue delay in the administration of the Chapter 11 case, with respect to Disputed Claims, the Diocese (if prior to the Effective Date) and the Reorganized Debtor (on and after the Effective Date), after notice and a hearing (which notice may be limited to the holder of such Disputed Claim), will have the right to seek an order of the Bankruptcy Court or the District Court, pursuant to Section 502(c) of the Bankruptcy Code, estimating or limiting the amount of: (i) property that must be withheld from or reserved for distribution purposes on account of such Disputed Claim(s), (ii) such Claim for allowance or disallowance purposes, or (iii) such Claim for any other purpose permitted under the Bankruptcy Code; provided, however, that the Bankruptcy Court or the District Court, as applicable, will determine: (y) whether such Claims are subject to estimation pursuant to Section 502(c) of the Bankruptcy Code, and (z) the timing and procedures for such estimation proceedings, if any, such matters being beyond the scope of the Plan.

### XIV. DISTRIBUTIONS UNDER THE PLAN

- **A.** <u>PAYMENT DATE</u>. Whenever any payment or distribution to be made under the Plan will be due on a day other than a business day, such payment or distribution will instead be made, without interest, on the immediately following business day.
- B. <u>UNDELIVERABLE DISTRIBUTIONS</u>. If payment or distribution to the holder of an allowed non-Tort Claim under the Plan is returned for lack of a current address for the holder or otherwise, the Reorganized Debtor will file with the Bankruptcy Court the name, if known, and last known address of the holder and the reason for its inability to make payment. All allowed Claims paid as provided in Section 11.2 of the Plan will be deemed addressed to the same extent as if payment or distribution had been made to the holder of the allowed Claim with no recourse to the Reorganized Debtor or property of the Reorganized Debtor. If, after the passage of six (6) months, the payment or distribution still cannot be made, the Reorganized Debtor will make the payment to the Trust. All allowed Claims paid as provided in Section 11.2 of the Plan will be deemed satisfied and released, with no recourse to the Reorganized Debtor or property of the Reorganized Debtor upon payment to the Trust, to the same extent as if payment or distribution has been made to the holder of the allowed Claim.

- C. <u>SETOFFS</u>. The Reorganized Debtor or the Trustee, as applicable, may, to the extent permitted under applicable law, set off against any allowed Claim and the distributions to be made pursuant to the Plan on account of such allowed Claim, the Claims, rights and Causes of Action of any nature that the Reorganized Debtor or the Trustee, as applicable, may hold against the holder of such allowed Claim that are not otherwise waived, released or compromised in accordance with the Plan; provided, however, that neither such a setoff nor the allowance of any Claim hereunder will constitute a waiver or release by the Reorganized Debtor or the Trustee, as applicable, of any such Claims, rights, and Causes of Action that the Reorganized Debtor or the Trustee, as applicable, possesses against such holder.
- **D.** <u>No Interest On Claims</u>. Unless otherwise specifically provided for in the Plan, the Confirmation Order, or a post-petition agreement in writing between a claimant and the Diocese, the Reorganized Debtor, or the Trust, and approved by an order of the Bankruptcy Court, postpetition interest will not accrue or be paid on any Claim, and claimants will not be entitled to interest accruing on or after the Petition Date on any Claim. In addition, and without limiting the foregoing or any other provision of the Plan, Confirmation Order, or Trust Agreement, interest will not accrue on or be paid on any Disputed Claim in respect of the period from the Effective Date to the date a final distribution is made when and if such Disputed Claim becomes an allowed Claim.
- **E.** <u>WITHHOLDING TAXES</u>. The Reorganized Debtor and the Trust will comply with all withholding and reporting requirements imposed by any federal, state, local, or foreign taxing authority, and all distributions hereunder will be subject to any such withholding and reporting requirements. As a condition to making any distribution under the Plan, the Reorganized Debtor and the Trust may require that the holder of an allowed Claim provide such holder's taxpayer identification number and such other information and certification as may be deemed necessary to comply with applicable tax reporting and withholding laws.

# XV. CONDITIONS TO EFFECTIVE DATE

- **A.** <u>CONDITIONS TO OCCURRENCE OF EFFECTIVE DATE</u>. The Plan will not become effective unless and until each of the following conditions will have been satisfied in full in accordance with the provisions specified below:
- 1. <u>Entry of Confirmation Order</u>. The Confirmation Order has become a Non-Appealable Order;
- 2. All Schedules and Exhibits to the Plan will have been duly completed by the Plan Proponents and filed with the Court and all agreements and releases referred to in the Plan will have been duly executed by all parties thereto and filed with the Court, in each case in form and substance satisfactory to the Debtor, the Committee, and LMI/Interstate; and
  - 3. <u>The Trust</u>. The Trust will have been formed.
- **B.** Notice of Effective Date. The Plan Proponents will file a Notice of Effective Date with the Bankruptcy Court within three (3) days after the occurrence of the Effective Date.

Such notice will include all relevant deadlines put into effect by the occurrence of the Effective Date.

C. <u>EFFECT OF NON-OCCURRENCE OF CONDITIONS</u>. If substantial consummation of the Plan does not occur, the Plan will be null and void in all respects and nothing contained in the Plan or the disclosure statement will: (i) constitute a waiver or release of any Claims by or against the Protected Parties or the Settling Insurers; (ii) prejudice in any manner the rights of the Protected Parties, the Trust or the Settling Insurers; or (iii) constitute an admission, acknowledgement, offer, or undertaking by the Protected Parties or the Settling Insurers in any respect, including but not limited to, in any proceeding or case against the Debtor; or (iv) be admissible in any action, proceeding or case against the Protected Parties or Settling Insurers in any court or other forum.

### XVI. EFFECT OF PLAN CONFIRMATION

THE DISCHARGE AND INJUNCTIONS CONTAINED IN THE PLAN AND THE RELEASES PROVIDED UNDER THE PLAN AND THE LMI/INTERSTATE SETTLEMENT AGREEMENT DO NOT RELEASE OR IMPAIR A TORT CLAIMANT'S RIGHT TO RECOVER ON ANY TORT CLAIM AGAINST (1) ANY PERPETRATOR OF SEXUAL ABUSE; (2) ANY PARTY THAT IS NOT A PROTECTED PARTY; (3) ANY PROTECTED PARTY WITH RESPECT TO ANY CLAIM NOT RELEASED OR ENJOINED UNDER THE PLAN (BUT RECOURSE IS LIMITED AS DESCRIBED ABOVE); AND (4) ANY SUCCESSOR OF THE DEBTOR AFTER THE EFFECTIVE DATE OF THE PLAN FOR ACTS OF ABUSE THAT ARE INDEPENDENT OF THE LIABILITY OF THE DEBTOR OR ANY PROTECTED PARTY.

- A. <u>DISSOLUTION OF COMMITTEE</u>. On the Effective Date, the Committee will dissolve automatically, whereupon their members, Professionals and agents will be released from any further duties and responsibilities in this Chapter 11 case and under the Bankruptcy Code, except that such parties will continue to be bound by any obligations arising under confidentiality agreements, joint defense/common interest agreements (whether formal or informal), and protective orders entered during this Chapter 11 case, including any orders regarding confidentiality issued by the Bankruptcy Court or mediators, which will remain in full force and effect according to their terms, provided that such parties will continue to have a right to be heard with respect to any and all applications for Professional Claims.
- **B.** <u>DISCHARGE AND INJUNCTION</u>. Except as otherwise expressly provided in the Plan or in the Confirmation Order, on the Effective Date, the Diocese will be discharged from, and its liability will be extinguished completely in respect to, any Claim and debt, whether reduced to judgment or not, liquidated or unliquidated, contingent or noncontingent, asserted or unasserted, fixed or not, matured or unmatured, disputed or undisputed, legal or equitable, known or future, based on conduct occurring before the Confirmation Date, including, without limitation, all interest, if any, on any such Claims and debts, whether such interest accrued before or after the Petition Date, and including all Claims and debts of the kind specified in Bankruptcy Code Sections 502(g), 502(h), and 502(i), whether or not a Proof of Claim is filed or is deemed filed under the Bankruptcy Code Section 501, such Claim is allowed under Bankruptcy Code Section 502, or the holder of such Claim has accepted the Plan.

Tort Claimants and the Trust will be permitted to name the Diocese in any proceeding to resolve whether the Diocese has liability for Tort Claims and the amount of any such liability, solely for the purpose of obtaining insurance coverage from Non-Settling Insurers. The discharge will not limit in any way the obligations of Non-Settling Insurers to defend and pay the Diocese's liability for Tort Claims under Non-Settling Insurer Policies. The limitations otherwise set forth in the Plan on a Tort Claimant's recovery will not in any way limit the Non-Settling Insurers' obligations under the Non-Settling Insurer Policies or the Tort Claimants' and/or Trust's recoveries against the Non-Settling Insurers for the Non-Settling Insurers' conduct in connection with the defense or settlement of a Tort Claim, including on any judgments in excess of the limits of a Non-Settling Insurer Policy.

- C. <u>Channeling Injunction</u>. Channeling Injunction Preventing Prosecution of Channeled Claims Against Protected Parties and Settling Insurers. In consideration of the undertakings of the Protected Parties and Settling Insurers under the Plan, their contributions to the Trust, and other consideration, and pursuant to their respective settlements with the Debtor and to further preserve and promote the agreements between and among the Protected Parties and any Settling Insurers, and pursuant to Section 105 of the Bankruptcy Code:
- 1. any and all Channeled Claims are channeled into the Trust and will be treated, administered, determined, and resolved under the procedures and protocols and in the amounts as established under the Plan and the Trust Agreement as the sole and exclusive remedy for all holders of Channeled Claims; and
- 2. all Persons who have held or asserted, hold or assert, or may in the future hold or assert any Channeled Claims are hereby permanently stayed, enjoined, barred and restrained from taking any action, directly or indirectly, for the purposes of asserting, enforcing, or attempting to assert or enforce any Channeled Claim against the Protected Parties or Settling Insurers, including:
- 1. commencing or continuing in any manner any action or other proceeding of any kind with respect to any Channeled Claim against any of the Protected Parties or Settling Insurers or against the property of any of the Protected Parties or Settling Insurers;
- 2. enforcing, attaching, collecting or recovering, by any manner or means, from any of the Protected Parties or Settling Insurers, or the property of any of the Protected Parties or Settling Insurers, any judgment, award, decree, or order with respect to any Channeled Claim against any of the Protected Parties, Settling Insurers, or any other Person;
- 3. creating, perfecting or enforcing any lien of any kind relating to any Channeled Claim against any of the Protected Parties or the Settling Insurers, or the property of the Protected Parties or the Settling Insurers;
- 4. asserting, implementing or effectuating any Channeled Claim of any kind against:

- (i) any obligation due any of the Protected Parties or Settling Insurers;
- (ii) any of the Protected Parties or Settling Insurers; or
- (iii) the property of any of the Protected Parties or Settling Insurers.
- 5. taking any act, in any manner, in any place whatsoever, that does not conform to, or comply with, the provisions of the Plan; and
- 6. asserting or accomplishing any setoff, right of indemnity, subrogation, contribution, or recoupment of any kind against an obligation due to any of the Protected Parties, the Settling Insurers, or the property of any of the Protected Parties or the Settling Insurers.

Notwithstanding anything to the contrary specifically in Article XIII of the Plan, or anything generally in the Plan, Tort Claimants and the Trust will be permitted to name the Diocese and any other Protected Party in any proceeding to resolve whether the Diocese or such other Protected Party has liability for a Tort Claim, and the amount of any such liability, for the purpose of obtaining insurance coverage from Non-Settling Insurers under the Non-Settling Insurer Policies, but recourse is limited to the proceeds of Non-Settling Insurer Policies and all other damages (including extra-contractual damages), awards, judgments in excess of policy limits, penalties, punitive damages and attorney's fees and costs that may be recoverable against any Non-Settling Insurers because of their conduct concerning insurance coverage for, or defense or settlement of, any Tort Claim, and any such judgments or awards will be turned over to the Trust for handling in accordance with Sections 6.14(i) and (j) of the Plan.

The Channeling Injunction is an integral part of the Plan and is essential to the Plan's consummation and implementation. It is intended that the channeling of the Channeled Claims as provided in Section 13.3 of the Plan will inure to the benefit of the Protected Parties and Settling Insurers. In a successful action to enforce the injunctive provisions of this Section in response to a willful violation thereof, the moving party may seek an award of costs (including reasonable attorneys' fees) against the non-moving party, and such other legal or equitable remedies as are just and proper, after notice and a hearing.

D. <u>Supplemental Settling Insurer Injunction</u>. Supplemental Injunction Preventing Prosecution of Claims Against Settling Insurers. Pursuant to Sections 105(a) and 363 of the Bankruptcy Code and in consideration of the undertakings of the Settling Insurers pursuant to any Insurance Settlement Agreement, including LMI's and Interstate's purchases of insurance policies or Interests in insurance policies free and clear of all interests pursuant to Section 363(f) of the Bankruptcy Code:

Any and all Persons who have held, now hold or who may in the future hold any Interests (including all debt holders, all equity holders, governmental, tax and regulatory authorities, lenders, trade and other creditors, Tort Claimants, perpetrators, and all others holding Interests of any kind or nature whatsoever, including those Claims released or to be released

pursuant to any Insurance Settlement Agreement) against any of the Settling Insurers, or any other Person covered or allegedly covered under the Settling Insurer Policies, including Contribution Claims, Extra-Contractual Claims, and Indirect Claims, are hereby permanently stayed, enjoined, barred, and restrained from taking any action, directly or indirectly, to assert, enforce or attempt to assert or enforce any such Interest against the Settling Insurers or the Settling Insurer Policies issued by LMI or Interstate:

- 1. Commencing or continuing in any manner any action or other proceeding against the Settling Insurers or the property of the Settling Insurers;
- 2. Enforcing, attaching, collecting, or recovering, by any manner or means, any judgment, award, decree or order against the Settling Insurers or the property of the Settling Insurers;
- 3. Creating, perfecting, or enforcing any lien of any kind against the Settling Insurers or the property of the Settling Insurers;
- 4. Asserting or accomplishing any setoff, right of indemnity, subrogation, contribution, or recoupment of any kind against any obligation due the Settling Insurers or the property of the Settling Insurers; and
- 5. Taking any action, in any manner, in any place whatsoever, that does not conform to, or comply with, the provisions of the Plan.

For the avoidance of doubt, this Supplemental Settling Insurer Injunction bars only the above-referenced actions against the Settling Insurers or the Settling Insurer Policies issued by LMI or Interstate. The foregoing injunctive provisions are an integral part of the Plan and are essential to its implementation.

E. EXCULPATION AND LIMITATION OF LIABILITY OF EXCULPATED PARTIES. From and after the Effective Date, none of the Exculpated Parties will have or incur any liability for, and each Exculpated Party will be released from, any Claim, Cause of Action or liability to any other Exculpated Party, to any holder of a Claim, or to any other party in interest, for any act or omission that occurred during and in connection with this Chapter 11 case or in connection with the preparation and filing of this Chapter 11 case, the formulation, negotiation, or pursuit of confirmation of the Plan, the consummation of the Plan, and the administration of the Plan or the property to be distributed under the Plan, except for Claims, Causes of Action or liabilities arising from the gross negligence, willful misconduct, fraud, or breach of the fiduciary duty of lovalty of any Exculpated Party, in each case subject to determination of such by Non-Appealable Order of a court of competent jurisdiction and provided that any Exculpated Party will be entitled to reasonably rely upon the advice of counsel with respect to its duties and responsibilities (if any) under the Plan. Without limiting the generality of the foregoing, the Committee and the Diocese and their respective officers, board and committee members, employees, attorneys, financial advisors, and other Professionals will be entitled to and granted the benefits of Section 1125(e) of the Bankruptcy **Code and the Channeling Injunction.** 

Under the Plan, "Exculpated Parties" means collectively, (i) the Diocese, the Estate, and the Committee; (ii) the respective officers, directors, employees, members, attorneys, financial advisors, members of subcommittees of the board of directors, volunteers, and members of consultative bodies and councils formed under Canon Law of the persons identified in the preceding clause including with respect to their service or participation in an outside board on which they serve at the request of the Diocese or the Bishop, in their capacity as such; (iii) the Settling Insurers with respect to their Settling Insurer Policies; and (iv) professionals of a Person identified in the preceding clause (i) through (iii).

- F. <u>TIMING</u>. The injunctions, releases, and discharges to which any Settling Insurer is entitled pursuant to such Insurance Settlement Agreement, the Plan, the Confirmation Order, the Approval Orders, and the Bankruptcy Code will only become effective when the Trust receives payment in full from the corresponding Settling Insurer pursuant to the terms of such Settling Insurer's Insurance Settlement Agreement, and the other conditions to effectiveness of the Insurance Settlement Agreement are fully met.
- **G.** No BAR ON CERTAIN CLAIMS. Notwithstanding the foregoing, nothing in the Plan will be construed to bar (a) a Claim based on Abuse against a Person who is not a Protected Party or a Settling Insurer (b) a Claim by such Person for insurance coverage in connection with a Claim described in the foregoing clause under an insurance policy other than the Settling Insurer Policies; or (c) a Tort Claim against a Protected Party that is not released or enjoined under the Plan, to the extent set forth in the Plan.

# XVII. CHILD PROTECTION PROTOCOLS

The Child Protection Protocols are incorporated into the Plan.

# XVIII. THE REORGANIZED DEBTOR

- A. <u>Continued Corporate Existence</u>. The Diocese will, as the Reorganized Debtor, continue to exist after the Effective Date as a separate entity in accordance with Minn. Stat. Section 315.16 having tax-exempt status under 26 U.S.C. § 501(c)(3) under applicable law and without prejudice to any right to alter or terminate such existence under applicable state law, except as such rights may be limited and conditioned by the Plan and the documents and instruments executed and delivered in connection therewith.
- B. <u>Vesting of Assets</u>. In accordance with Sections 1141 and 1123(a)(5) of the Bankruptcy Code, and except as otherwise provided in the Plan or the Confirmation Order, all Assets of the Diocese and the Estate will vest in the Reorganized Debtor (or such other entity or entities specified by the Debtor in a Supplemental Plan Document, and subject to approval by the Bankruptcy Court at the confirmation hearing) on the Effective Date free and clear of all liens, claims, and interests of creditors, including successor liability claims. On and after the Effective Date, the Reorganized Debtor may operate and manage its affairs and may use, acquire, and dispose of property without notice to any Person, and without supervision or approval by the Bankruptcy Court and free of any restrictions imposed by the Bankruptcy Code, Bankruptcy Rules,

or the Bankruptcy Court, other than those restrictions expressly imposed by the Plan or the Confirmation Order. The Reorganized Debtor is expressly authorized, as of the Effective Date, to transfer legal title to the Inadvertently Titled Real Property to the respective equitable title owners of such properties without further order of the Bankruptcy Court.

- **C.** <u>IDENTITY OF OFFICERS OF REORGANIZED DEBTOR</u>. In accordance with § 1129(a)(5) of the Bankruptcy Code, the identities and affiliations of the Persons proposed to serve as the corporate Members of the Reorganized Debtor and the persons proposed to serve as directors and officers of the Reorganized Debtor on and after the Effective Date are set forth on **Exhibit J** of the Plan.
- **D.** <u>FURTHER AUTHORIZATION</u>. The Reorganized Debtor will be entitled to seek such orders, judgments, injunctions, rulings, and other assistance as it deems necessary to carry out the intentions and purposes, and to give full effect to the provisions, of the Plan.

### XIX. CERTAIN OTHER GENERAL PROVISIONS OF THE PLAN

The Plan contains other provisions consistent with the requirements of Chapter 11 of the Bankruptcy Code as set forth below:

A. <u>Causes of Action/Avoidance Actions</u>. Except as otherwise provided in the Plan, the Reorganized Debtor will retain and exclusively enforce the Debtor's Causes of Action, whether arising before or after the Petition Date, in any Court or other tribunal, including without limitation, an adversary proceeding filed in this case.

#### **B.** RETENTION OF JURISDICTION.

- 1. <u>By the Bankruptcy Court</u>. Pursuant to Sections 105, 1123(a)(5), and 1142(b) of the Bankruptcy Code, and 28 U.S.C. Sections 1334 and 157, on and after the Effective Date, the Bankruptcy Court will retain: (i) original and exclusive jurisdiction over this Chapter 11 case, (ii) original, but not exclusive, jurisdiction to hear and determine all core proceedings arising under the Bankruptcy Code or arising in this Chapter 11 case, and (iii) original, but not exclusive, jurisdiction to hear and make proposed findings of fact and conclusions of law in any non-core proceedings related to this Chapter 11 case and the Plan, including matters concerning the interpretation, implementation, consummation, execution, or administration of the Plan. Subject to, but without limiting the generality of the foregoing, the Bankruptcy Court's post-Effective Date jurisdiction will include jurisdiction:
  - 1. over disputes concerning the ownership of Claims;
- 2. over disputes concerning the distribution or retention of assets under the Plan;
- 3. over objections to Claims, motions to allow late-filed Claims, and motions to estimate Claims;

### Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 67 of 222

- 4. over proceedings to determine the extent, validity, or priority of any Lien asserted against property of the Diocese, the Estate, or Trust, or property abandoned or transferred by the Diocese, the Estate, or the Trust;
- 5. over motions to approve any insurance settlement agreements entered into after the Effective Date by the Trustee;
- 6. over matters related to the assets of the Estate or of the Trust, including the terms of the Trust, or the recovery, liquidation, or abandonment of Trust Assets;
- 7. the removal of the Trustee and the appointment of a successor Trustee;
  - 8. over matters relating to the subordination of Claims;
- 9. to enter and implement such orders as may be necessary or appropriate in the event the Confirmation Order is for any reason stayed, revoked, modified or vacated;
- 10. to consider and approve modifications of or amendments to the Plan, to cure any defects or omissions or to reconcile any inconsistencies in any order of the Bankruptcy Court, including the Confirmation Order;
- 11. to issue orders in aid of execution, implementation, or consummation of the Plan, including the issuance of orders enforcing any and all releases and injunctions issued under or pursuant to the Plan and any Insurance Settlement Agreement;
- 12. over disputes arising from or relating to the Plan, the Confirmation Order, or any agreements, documents, or instruments executed in connection therewith;
- 13. over requests for allowance of payment of Claims entitled to priority under Sections 507(a)(2) and 503(b)(9) of the Bankruptcy Code and any objections thereto;
  - 14. over all Fee Applications;
- 15. over matters concerning state, local, or federal taxes in accordance with Sections 346, 505, and 1146 of the Bankruptcy Code;
- 16. over conflicts and disputes among the Trust, the Reorganized Debtor, and holders of Claims, including holders of Class 3, Class 4A or Class 4B Claims;
- 17. over disputes concerning the existence, nature, or scope of the Diocese's discharge, including any dispute relating to any liability arising out of the termination of employment or the termination of any employee or retiree benefit program, regardless of whether such termination occurred prior to or after the Effective Date;
- 18. to issue injunctions, provide declaratory relief, or grant such other legal or equitable relief as may be necessary or appropriate to restrain interference with the Plan,

Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 68 of 222

the Diocese or its property, the Reorganized Debtor or its property, the Estate or its property, the Trust or its property, Trustee, the Professionals, or the Confirmation Order;

- 19. to enter a Final Decree closing the Chapter 11 case;
- 20. to enforce all orders previously entered by the Bankruptcy Court; and
- 21. over any and all other suits, adversary proceedings, motions, applications, and contested matters that may be commenced or maintained pursuant to this Chapter 11 case or the Plan, including, but not limited to, disputes arising out of or related to the Transferred Insurance Interests or the rights and obligations of the Non-Settling Insurers on and after the Effective Date.
- 2. <u>By the District Court</u>. Pursuant to Sections 105, 1123(a)(5), and 1142(b) of the Bankruptcy Code, and 28 U.S.C. Section 1334, on and after the Effective Date, the District Court will retain original, but not exclusive, jurisdiction to hear and determine all matters arising under the Bankruptcy Code or arising in or related to this Chapter 11 case.
- 3. Actions to Collect Amounts Owed Pursuant to the Plan. Notwithstanding anything to the contrary in Section 16.1 of the Plan, the Diocese, the Reorganized Debtor and the Trustee may, but are not required to, commence an adversary proceeding to collect amounts owed pursuant to the Plan for any settlements embodied in the Plan or later approved by the Bankruptcy Court, which are not paid in accordance with the Plan. Any such action may be commenced by filing a motion in aid of confirmation with the Bankruptcy Court.
- 4. <u>Case Closure</u>. The existence and continued operation of the Trust will not prevent the Bankruptcy Court from closing this Chapter 11 case. In an action involving the Trust, any costs incurred in reopening the Chapter 11 case, including any statutory fees will be paid by the Trustee from the Trust Assets in accordance with an order of the Bankruptcy Court.
- **C. REVOCATION OF THE PLAN**. The Plan Proponents reserve the right to revoke and withdraw the Plan prior to the Confirmation Date.
- **D.** <u>Non-Severability</u>. Except as specifically provided in the Plan, the terms of the Plan constitute interrelated compromises and are not severable, and no provision of the Plan may be stricken, altered, or invalidated, except by amendment of the Plan by the Plan Proponents.
- E. SECTION 1146 EXEMPTION. Pursuant to Section 1146 of the Bankruptcy Code, the delivery or recording of an instrument of transfer on or after the Confirmation Date shall be deemed to be made pursuant to and under the Plan, including any such acts by the Diocese (if prior to the Effective Date), and the Reorganized Debtor (if on or after the Effective Date), including any subsequent transfers of property by the Reorganized Debtor, including without limitation the transfer by the Reorganized Debtor of title to the Inadvertently Titled Real Property as referenced in Section 15.2 of the Plan, and shall not be taxed under any law imposing a stamp tax, transfer tax, state deed tax, or similar tax or fee. Consistent with the foregoing, each recorder of deeds or similar official for any county, city, or governmental unit in which any instrument hereunder is to be recorded will, pursuant to the Confirmation Order and the Plan, be ordered and directed to

Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 69 of 222

accept such instrument, without requiring the payment of any documentary stamp, tax, deed stamps, stamp tax, transfer tax, intangible tax, or similar tax.

**F.** MANAGEMENT OF THE REORGANIZED DEBTOR. The Reorganized Debtor will continue to be managed by current management. Those managers, their titles and salaries consist of:

Most Rev. John M. Quinn, Bishop and President of Diocese Civil Corporation - Salary \$33,372 (full time; amount excludes housing and other allowances);

Very Rev. William D. Thompson, Vicar General and Vice President of Diocese Civil Corporation - Salary \$14,784 (part-time (50%); amount excludes housing and other allowances);

Very Rev. Glenn K. Frerichs, Chancellor, Judicial Vicar and Secretary of Diocese Civil Corporation - Salary \$17,442 (part-time (50%); amount excludes housing and other allowances);

Mr. Andrew D. Brannon, Chief Operating Officer, Chief Financial Officer and Treasurer of Diocese Civil Corporation - Salary \$123,624 (full-time); and

Mr. Timothy McManimon, Director At-Large of Diocese Civil Corporation (Volunteer, No Salary).

# XX. CONFIRMATION PROCEDURES

In order for the Plan to be confirmed by the Bankruptcy Court, all of the applicable requirements of Bankruptcy Code § 1129 must be met. This includes, among others, requirements that the Plan: (i) is accepted by all impaired Classes or, if rejected by an impaired Class, "does not discriminate unfairly" and is "fair and equitable" as to each rejecting Class; (ii) is feasible; and (iii) is in the best interests of holders of Claims in each impaired Class.

**A.** <u>SOLICITATION OF VOTES; ACCEPTANCE</u>. The Debtor and the Committee are soliciting the acceptance of the Plan from all holders of Claims in Classes that are impaired under the Plan. Using this criteria, holders of only Claims in Classes 3 and 4B are entitled to vote on the Plan. In addition, a vote may be disregarded if the Bankruptcy Court determines, after notice and a hearing, that such vote was not solicited or procured or made in good faith or in accordance with the applicable provisions of the Bankruptcy Code.

Classes 3 and 4B will be deemed to have accepted the Plan if the Plan is accepted by holders of at least two-thirds in dollar amount and more than one-half in number of the Claims in that Class that has cast Ballots on the Plan and that are eligible to vote for or against the Plan. The Debtor will file a motion to approve solicitation and voting procedures, which motion will propose to estimate Class 3 Claims at \$1.00 each and Class 4B Claims at \$1.00 for voting purposes only. The

Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 70 of 222

actual amount payable to holders of Allowed Claims in such classes may exceed \$1.00. This is consistent with the procedure followed in other mass tort bankruptcy cases.

Confirmation Hearing. Bankruptcy Code § 1128(a) requires the Bankruptcy Court, after notice, to hold the Confirmation Hearing after the period for submission of Ballots has expired. The Confirmation Hearing may be postponed from time to time by the Bankruptcy Court without further notice except for an announcement of the postponement made at the Confirmation Hearing. Bankruptcy Code § 1128(b) provides that any party in interest may object to confirmation of the Plan. Objections must be made in writing, specifying in detail the name and address of the person or entity objecting, the grounds for the objection and the nature and amount of the Claim held by the objector, and must otherwise comply with the requirements of the Bankruptcy Rules and the Local Bankruptcy Rules. Objections must be filed with the Clerk of the Bankruptcy Court and served upon the parties so designated in the Order and Notice accompanying this Disclosure Statement on or before the time and date designated in such Order and Notice. FAILURE TO TIMELY FILE AND SERVE AN OBJECTION TO CONFIRMATION MAY BE DEEMED BY THE BANKRUPTCY COURT TO BE CONSENT TO CONFIRMATION OF THE PLAN.

At the Confirmation Hearing, the Bankruptcy Court will determine, among other things, whether the following confirmation requirements specified in Bankruptcy Code Section 1129 have been satisfied:

- 1. The Plan complies with the applicable provisions of the Bankruptcy Code.
- 2. The Plan Proponents have complied with the applicable provisions of the Bankruptcy Code.
- 3. The Plan has been proposed in good faith and not by any means proscribed by law.
- 4. Any payment made or promised by the Debtor for services or for costs and expenses in, or in connection with, this Chapter 11 case, or in connection with the Plan and incident to the Chapter 11 case, has been approved by, or is subject to approval of, the Bankruptcy Court as reasonable.
- 5. The Plan Proponents have disclosed the identity and affiliations of all individuals proposed to serve, after confirmation, as directors or officers of the Debtor and the appointment to or continuance in such positions by those individuals is consistent with the interests of creditors and with public policy; and (b) the Plan Proponents have disclosed the identities of any insider(s) that will be employed or retained by the Reorganized Debtor and the nature of any proposed compensation for such insider(s).
- 6. Each holder of a Claim in an impaired Class either has accepted the Plan or will receive or retain under the Plan property of a value, as of the Effective Date, that is not less than the amount that such entity would receive or retain if the Debtor were liquidated on such date under Chapter 7 of the Bankruptcy Code. *See* "Best Interests of Creditors Test" below.

- 7. Unless the Plan Proponents are required to seek nonconsensual confirmation of the Plan, each Class of Claims has either accepted the Plan or is not impaired under the Plan.
- 8. Except to the extent that the holder of a Claim has agreed to different treatment, the Plan provides that: (a) allowed Administrative Claims will be paid in full on the later of the Effective Date, or the date the Claim is allowed; (b) other Priority Claims will be paid in full on the Effective Date; and (c) Priority Tax Claims will receive payment in full plus interest over five (5) years.
- 9. At least one impaired Class has accepted the Plan, determined without including any acceptance of the Plan by any insider holding a Claim in such Class.
- 10. Confirmation of the Plan is not likely to be followed by the liquidation of or the need for further financial reorganization by the Debtor or the Reorganized Debtor.
- 11. The Plan Proponents believe that the Plan has been submitted in good faith and that, upon acceptance of the Plan by the voting Classes, the Plan will satisfy all of the foregoing statutory requirements.
- C. <u>BEST INTERESTS OF CREDITORS TEST</u>. As mentioned above, confirmation of the Plan requires that each holder of a Claim in an impaired Class must either: (i) accept the Plan; or (ii) receive or retain under the Plan property of a value, as of the Effective Date, that is not less than the value such holder would receive or retain if the Debtor were liquidated under Chapter 7 of the Bankruptcy Code.

The Debtor believes that in a hypothetical liquidation all creditors will receive less than they will likely receive under the Plan. Chapter 7 liquidation carries potential costs and risks that are resolved through the Plan, as follows:

- 1. The Plan incorporates the Trust Distribution Plan, which was negotiated by counsel representing in excess of 85% of Tort Claimants. There is likelihood that a Chapter 7 Trustee will be unable to implement the Trust Distribution Plan or a similar plan in the absence of a confirmed Chapter 11 Plan. As such, substantial estate resources would likely be expended adjudicating or analyzing Tort Claims in a Chapter 7 case.
- 2. A Chapter 7 Trustee would be entitled to compensation of a percentage of all funds distributed to parties in interest, excluding the Debtor, under 11 U.S.C. § 326. Any such payment would dilute the amount of funds available to pay creditors.
- 3. LMI/Interstate would not get the Channeling Injunction and/or releases provided in the Plan, nor would they make the substantial contributions they are making under the Plan without such injunctions and/or releases.

In addition, it is unlikely that LMI/Interstate would voluntarily contribute without the corresponding benefit of final resolution of Tort Claims. Annexed hereto as  $\underline{Exhibit}\ C$  is a Liquidation Analysis. The Liquidation Analysis indicates that in a liquidation, holders of Unsecured Claims would receive a lesser distribution than provided under the Plan.

- **D.** <u>FEASIBILITY</u>. The Bankruptcy Code requires that confirmation of a Plan is not likely to be followed by liquidation or the need for further financial reorganization. The Plan provides that the Debtor will pay the sum of \$13,560,000 (minus (i) amounts paid by the Debtor for certain administrative expenses paid after February 29, 2020 and (ii) counseling expenses for Tort Claimants paid by the Debtor) to the Trust within five days after the Effective Date of the Plan (which payment shall constitute substantial consummation of the Plan). It further provides for the Debtor to pay an additional \$7,940,000 to the Trust as soon as practical but in no event more than 12 months after the order confirming the Plan becomes final and non-appealable. The Debtor currently has sufficient funds on hand to make the initial cash payment required by the Plan, and the Plan provides the Debtor up to 12 months to make the second cash payment, which the Debtor believes is sufficient time to raise such funds via the sale or financing of certain assets and contributions from certain non-Diocesan entity resources. Accordingly, the Plan Proponents believe the Plan is feasible.
- **E.** <u>CRAM DOWN</u>. The Bankruptcy Code provides a mechanism by which a Plan may be confirmed even if it has been rejected by an impaired Class of Claims. Under the "cram down" provisions of the Bankruptcy Code (§ 1129(b)), the proponent of the Plan (in this case the Debtor and the Committee) may request that it be confirmed despite its rejection by an impaired Class, and the Bankruptcy Court will confirm the Plan if it (i) does not discriminate unfairly against a dissenting impaired Class and (ii) is fair and equitable with respect to such Class.

The Bankruptcy Code sets forth specific guidelines for determining whether a Plan is fair and equitable with respect to a particular Class of Claims. For unsecured Claims, as are those in all impaired Classes, a Plan must provide that equity interest holders do not receive or retain any property on account of their interest. The Debtor submits that this test which is applied to traditional corporations is inapplicable because the Debtor, as a not-for-profit entity, does not have any equity holders.

In the event the Bankruptcy Court refuses to impose a "cram down" on the rights of a non-consenting Class unless certain modifications are made to the terms and conditions of such non-consenting Class's treatment under the Plan, the Debtor reserves the right, without re-solicitation, to propose such modification to such non-consenting Class's treatment and to confirm the Plan.

# XXI. <u>ALTERNATIVES TO CONFIRMATION AND</u> CONSUMMATION OF THE PLAN

The Plan Proponents believe the Plan is in the best interests of creditors and should accordingly be accepted and confirmed. If the Plan as proposed, however, is not confirmed, the following two alternatives may be available: (A) an alternative plan of reorganization may be proposed and confirmed or (B) the Debtor's Chapter 11 case may be dismissed. As discussed below, a third option, liquidation under Chapter 7, is not a viable alternative in this Chapter 11 case.

**A.** <u>ALTERNATIVE CHAPTER 11 PLANS</u>. If the Plan is not confirmed, the Plan Proponents, either jointly or separately, may propose a different plan, which might involve an alternative means for reorganization of the Diocese. However, the Plan Proponents believe that the

terms of the Plan provide for the best result for the Diocese's creditors. The negotiation and drafting required for a different plan would likely add substantially greater administrative expenses with no guarantee of a better result for the Diocese's creditors. For these reasons, the Plan Proponents do not believe that alternative plans of reorganization are preferable alternatives to the Plan.

**B.** <u>DISMISSAL OF THE CHAPTER 11 CASE</u>. If the Plan is not confirmed, the Debtor or another party in interest may seek to dismiss the Chapter 11 case. After appropriate notice and a hearing, the Bankruptcy Court may grant the request and dismiss the Chapter 11 case. Dismissal of the Debtor's Chapter 11 case would have the effect of restoring, or attempting to restore, all parties to the position they were in immediately prior to the Petition Date.

Upon the dismissal of the Debtor's Chapter 11 case, the protection of the Bankruptcy Code would be lost, resulting in the expensive and time-consuming process of negotiation and protracted litigation between the Diocese and individual Tort Claimants and the Diocese and its insurers. In addition to the expense and delay, the Plan Proponents believe that these actions would lead to an unfair distribution to Tort Claimants, with the first Tort Claimants to obtain and enforce a judgment against the Diocese depleting the Diocese's assets and resulting in insufficient assets to satisfy any further judgments. Therefore, the Plan Proponents believe that dismissal of the Diocese's Chapter 11 case is not a preferable alternative to the Plan.

C. CONVERSION TO CHAPTER 7 NOT A VIABLE ALTERNATIVE. Under the Bankruptcy Code, it is also possible for a Chapter 11 case to be converted to a case under Chapter 7, in which event a Trustee would be elected or appointed to liquidate the Debtor's assets for distribution to its creditors in accordance with the priorities established by the Bankruptcy Code. However, under § 1112(c) of the Bankruptcy Code, if a debtor is "not a moneyed, business, or commercial corporation," the debtor's Chapter 11 case cannot be converted to a Chapter 7 case without the debtor's consent. The Diocese, as a non-profit entity, may not be forced to convert its Chapter 11 case to a Chapter 7 case. The Diocese will not consent to a conversion because the Diocese must continue to carry out the mission of the Catholic Church in the Region that it serves. Moreover, the Diocese believes that liquidation under Chapter 7 would result in smaller distributions being made to creditors than those provided for under the Plan because the Debtor's cash contribution under the Plan is well in excess of the liquidation value of the Debtor's unencumbered and unrestricted assets, particularly when taking into account (1) the increased costs and expenses of a liquidation under Chapter 7 arising from fees payable to a Chapter 7 Trustee and professional advisors to such Chapter 7 Trustee and (2) the erosion in value of assets in the context of the expeditious liquidation required under Chapter 7 and the "forced sale" environment in which such a liquidation would likely occur. Thus, conversion to a Chapter 7 case is not a viable alternative to the Plan.

# XXII. CERTAIN FEDERAL INCOME TAX CONSIDERATIONS

THE FEDERAL, STATE, LOCAL, AND FOREIGN TAX CONSEQUENCES OF THE PLAN ARE COMPLEX AND, IN MANY AREAS, UNCERTAIN. ACCORDINGLY, ALL HOLDERS OF CLAIMS ARE STRONGLY URGED TO CONSULT THEIR TAX ADVISORS WITH SPECIFIC REFERENCE TO THE FEDERAL, STATE, AND LOCAL TAX CONSEQUENCES OF THE PLAN WITH RESPECT TO SUCH HOLDER. NEITHER THE

Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 74 of 222

DEBTOR NOR THE DEBTOR'S COUNSEL MAKE ANY REPRESENTATIONS REGARDING THE PARTICULAR TAX CONSEQUENCES OF CONFIRMATION AND CONSUMMATION OF THE PLAN AS TO THE DEBTOR OR ANY CREDITOR.

Under the Internal Revenue Code of 1986, as amended (the "Code"), there may be significant federal income tax issues arising under the Plan described in this Disclosure Statement that affect holders of Claims in this Case.

- **A.** <u>THE TRUST</u>. The Trust is a "qualified settlement fund" ("**QSF**") within the meaning of Treasury Regulations enacted under the Internal Revenue Code at 26 U.S.C. § 468B(g). The Trust is characterized as a QSF because:
- 1. The Trust is established pursuant to an Order of, or is approved by, the United States, any state or political subdivision thereof, or any agency or instrumentality (including a court of law) of any of the foregoing and is subject to the continuing jurisdiction of that governmental authority;
- 2. The Trust is established to resolve or satisfy one or more contested or uncontested claims that has resulted or may result from an event that has occurred and that has given rise to at least one claim asserting liability arising out of, among other things, a tort, breach of contract, or violation of law related to sexual abuse (but excluding non-tort obligations of the Debtor to make payments to its general trade creditors or debt holders that relate to: a case under Title 11 of the United States Code, a receivership, foreclosure of similar proceeding in a Federal or State court, or a workout); and,
  - 3. The Trust is a trust under state law.

The primary tax consequences of the Trust being characterized as a QSF are the following:

- i. The Trust must use a calendar taxable year and the accrual method of accounting.
- ii. If the Debtor funds the Trust with appreciated property, the Debtor is deemed to sell the property to the Trust. Accordingly, any gain or loss from the deemed sale must be reported by the Debtor.
- iii. The Trust takes a fair market value basis in property contributed to it by the Debtor.
- iv. The Trust's gross income less certain modifications is taxable at the highest federal tax rate applicable to trusts and Estate (currently 35%). The Debtor's funding of the Trust with cash and other property is not reported by the Trust as taxable income. However, earnings recognized from, for example, the short-term investment of the Trust's funds will be subject to tax.
- v. The Trust may deduct from its gross income a limited number of administrative expenses; the Trust in not entitled to deduct distributions paid to its beneficiaries.

Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 75 of 222

- vi. The Trust will have a separate taxpayer identification number and will be required to file annual tax returns (which are due on March 15). The Trust will also be required to comply with a number of other administrative tax rules including filing information returns (generally IRS Form 1099) when approved payments are made to claimants.
- **B.** FEDERAL INCOME TAX CONSEQUENCES TO HOLDERS OF CLAIMS. The federal income tax consequences to a holder of a Claim receiving, or entitled to receive, a Distribution in partial or total satisfaction of a Claim may depend on a number of factors, including the nature of the Claim, the claimants' method of accounting, and their own particular tax situation. Because each claimant's tax situation differs, claimants should consult their own tax advisors to determine how the Plan affects them for federal, state and local tax purposes, based on its particular tax situations.

Among other things, the federal income tax consequences of a distribution to a claimant may depend initially on the nature of the original transaction pursuant to which the Claim arose. For example, a distribution in repayment of the principal amount of a loan is generally not included in the claimant's gross income. Distributions to Tort Claimants may not be taxable as it may be considered compensation for personal injuries.

The federal income tax consequences of a distribution to a claimant may also depend on whether the item to which the distribution relates has previously been included in the claimant's gross income or has previously been subject to a loss or bad debt deduction. For example, if a distribution is made in satisfaction of a receivable acquired in the ordinary course of the claimant's trade or business, and the claimant had previously included the amount of such receivable distribution in his or her gross income under his or her method of accounting, and had not previously claimed a loss or bad debt deduction for that amount, the receipt of the distribution should not result in additional income to the claimant but may, as discussed below, result in a loss. Conversely, if the claimant had previously claimed a loss or bad debt deduction with respect to the item previously included in income, the claimant generally would be required to include the amount of the distribution in income when received.

A claimant receiving a distribution in satisfaction of his or her claim generally may recognize taxable income or loss measured by the difference between (i) the cash and the fair market value (if any) of the property received and (ii) its adjusted tax basis in the Claim. For this purpose, the adjusted tax basis may include amounts previously included in income (less any bad debt or loss deduction) with respect to that item. This income or loss may be ordinary income or loss if the distribution is in satisfaction of accounts or notes receivable acquired in the ordinary course of the claimant's trade or business for the performance of services or for the sale of goods or merchandise. In addition, if a claimant had claimed an ordinary bad debt deduction for the worthlessness of his or her Claim in whole or in part in a prior taxable year, any income realized by the claimant as a result of receiving a distribution may be taxed as ordinary income to the extent of the ordinary deduction previously claimed. Generally, the income or loss will be capital gain or loss if the Claim is a capital asset in the claimant's hands.

## XXIII. VOTING INSTRUCTIONS

Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 76 of 222

Solicitation Packages which will include copies of (i) the Disclosure Statement Approval Order, (ii) the Notice of Disclosure Statement Approval and Confirmation Hearing, (iii) the approved Form of Disclosure Statement (together with the Plan annexed thereto), and (iv) the form of Ballot will be sent to creditors. Procedures and deadlines for submitting the Ballot will be included in such Solicitation Package.

### XXIV. CONCLUSION

The Debtor believes that confirmation and implementation of the Plan is preferable to any other alternative. Accordingly, the Debtor urges holders of Claims to vote to accept the Plan by so indicating on its Ballots and returning them as specified in the instructions set forth in the Solicitation Packages.

Dated: May 19, 2021

[Signature page for Disclosure Statement]

THE DIOCESE OF WINONA-ROCHESTER

Very Rev. William D. Phompson

#### **RESTOVICH BRAUN & ASSOCIATES**

/e/Thomas R. Braun

Thomas R. Braun (#350631) Christopher W. Coon (#390083) 117 East Center Street Rochester, MN 55904 (507) 216-8652 thomas@restovichlaw.com christopher@restovichlaw.com

and

#### **BODMAN PLC**

/e/Robert J. Diehl, Jr.

Robert J. Diehl, Jr. (P31264)
Brian R. Trumbauer (P57747)
Jaimee L. Witten (P70068)
6th Floor at Ford Field
1901 St. Antoine Street
Detroit, Michigan 48226
(313) 259-7777
rdiehl@bodmanlaw.com
btrumbauer@bodmanlaw.com
jwitten@bodmanlaw.com

ATTORNEYS FOR THE DIOCESE OF WINONA-ROCHESTER

[Signature page for Disclosure Statement]

Respectfully submitted,

### THE OFFICIAL COMMITTEE OF UNSECURED CREDITORS

/e/James W. Keenan

By: James W. Keenan Its: Chairperson

and

#### /e/Robert T. Kugler

#### STINSON, LLP

Robert T. Kugler (#194116) Edwin H. Caldie (#388930) Andrew J. Glasnovich (#398366) 50 South Sixth Street, Suite 2600 Minneapolis, MN 55402 robert.kugler@stinson.com ed.caldie@stinson.com drew.glasnovich@stinson.com

Telephone: 612-335-1500 Facsimile: 612-335-1657

ATTORNEYS FOR THE OFFICIAL COMMITTEE OF UNSECURED CREDITORS FOR THE DIOCESE OF WINONAROCHESTER

Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 79 of 222

#### **EXHIBIT A**

First Amended Joint Chapter 11 Plan of Reorganization of the Diocese of Winona-Rochester (Filed as a Separate Document)

### IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF MINNESOTA

In re:

Case No. 18-33707

Diocese of Winona-Rochester,

Chapter 11

Debtor.

### FIRST AMENDED JOINT CHAPTER 11 PLAN OF REORGANIZATION OF THE DIOCESE OF WINONA-ROCHESTER

#### RESTOVICH BRAUN & ASSOCIATES

Thomas R. Braun (#350631) Christopher W. Coon (#390083) 117 East Center Street Rochester, MN 55904 (507) 216-8652 thomas@restovichlaw.com christopher@restovichlaw.com

#### **BODMAN PLC**

Robert J. Diehl, Jr. (P31264) Brian R. Trumbauer (P57747) Jaimee L. Witten (P70068) 6th Floor at Ford Field 1901 St. Antoine Street Detroit, Michigan 48226 (313) 259-7777 rdiehl@bodmanlaw.com btrumbauer@bodmanlaw.com jwitten@bodmanlaw.com

### ATTORNEYS FOR THE DIOCESE OF WINONA-ROCHESTER

Dated: May 19, 2021

#### STINSON, LLP

Robert T. Kugler (#194116) Edwin H. Caldie (#388930) Andrew J. Glasnovich (#398366) 50 South Sixth Street, Suite 2600 Minneapolis, MN 55402 robert.kugler@stinson.com ed.caldie@stinson.com drew.glasnovich@stinson.com

Telephone: 612-335-1500 Facsimile: 612-335-1657

ATTORNEYS FOR THE OFFICIAL COMMITTEE OF UNSECURED CREDITORS FOR THE DIOCESE OF WINONA-ROCHESTER

#### TABLE OF CONTENTS

INTRODUC	CTION	V
ARTICI E I	DEFINITIONS AND INTERPRETATION	1
1.1	DEFINED TERMS	
1.2	INTERPRETATION	
1.3	TIME PERIODS	
1.4	EXHIBITS AND SCHEDULES	
ADTICLE		1.5
	I TREATMENT OF UNCLASSIFIED CLAIMS	
2.1	ADMINISTRATIVE CLAIMS	
2.2	STATUTORY FEES	
2.3	PRIORITY TAX CLAIMS	17
ARTICLE I	II CLASSIFICATION OF CLAIMS	17
3.1	SUMMARY	17
3.2	CLASSIFICATION AND VOTING	18
ARTICI E I	V TREATMENT OF CLASSIFIED CLAIMS	1.9
4.1	PRIORITY CLAIMS (CLASS 1)	
4.2	GOVERNMENTAL UNIT CLAIMS (CLASS 2)	
4.3	TORT CLAIMS OTHER THAN IMPAIRED UNKNOWN TORT	10
4.3	CLAIMS (CLASS 3)	10
4.4	UNIMPAIRED UNKNOWN TORT CLAIMS (CLASS 4A)	
4.5	IMPAIRED UNKNOWN TORT CLAIMS (CLASS 4B)	
<b>4.</b> 6	GENERAL UNSECURED CLAIMS (CLASS 5)	
<b>4.</b> 7	UNIMPAIRED ABUSE RELATED CONTINGENT CLAIMS	
4.7	(CLASS 6)	23
4.8	MINNESOTA DEPARTMENT OF COMMERCE (CLASS 7)	
4.9	COMMUNITY BANK OF MANKATO (CLASS 8)	
4.7	COMMUNITY DANK OF MANKATO (CLASS 6)	23
ARTICLE V	MEANS OF IMPLEMENTATION OF THE PLAN	24
5.1	TRUST FORMATION AND FUNDING	24
5.2	PAYMENT OF PROFESSIONAL FEES	
5.3	PAYMENTS EFFECTIVE UPON TENDER	26
ARTICLE V	/I TRUST	26
6.1	ESTABLISHMENT OF TRUST	
6.2	PURPOSE, FORMATION AND ASSETS	
6.3	ALLOCATIONS WITHIN AND DISTRIBUTIONS AND	20
0.0	PAYMENTS FROM THE TRUST	27
6.4	TAX MATTERS	
6.5	APPOINTMENT OF THE TRUSTEE	
6.6	RIGHTS AND RESPONSIBILITIES OF TRUSTEE	
6.7	TRANSFERRED INSURANCE INTERESTS	
0.7		·····

6.8	SPECIAL DISTRIBUTION CONDITIONS	32
6.9	INVESTMENT POWERS; PERMITTED CASH EXPENDITURES	33
6.1	0 REGISTRY OF BENEFICIAL INTERESTS	33
6.1	1 NON-TRANSFERABILITY OF INTERESTS	33
6.1	2 TERMINATION	33
6.1	3 IMMUNITY; LIABILITY; INDEMNIFICATION	33
6.1	4 TREATMENT OF TORT CLAIMS	35
ARTICLE	VII SETTLING INSURERS	
7.1		41
7.2	SALE FREE AND CLEAR OF INTERESTS OF SETTLING INSURER POLICIES	41
7.3	RESOLUTION OF CLAIMS INVOLVING SETTLING INSURERS	42
7.4	THE SETTLING INSURERS' PAYMENTS	42
7.5	JUDGMENT REDUCTION	42
7.6	FURTHER ASSURANCES; NON-MATERIAL MODIFICATIONS	42
7.7	INDEMNIFICATION OBLIGATIONS	43
7.8		
7.9	DEBTOR WAIVER AND RELEASE OF CLAIMS	44
7.1	0 SUPPLEMENTAL SETTLING INSURER INJUNCTION	44
ARTICLE	VIII	45
NON-SET	TLING INSURERS	45
8.1	PRESERVATION OF RIGHTS AND OBLIGATIONS	
8.2	ESTIMATIONS/ASSESSMENTS ARE NOT BINDING	46
8.3		
8.4	THE PLAN IS NEUTRAL AS TO NON-SETTLING INSURER	
	POLICIES	46
8.5	THE DIOCESE'S OBLIGATIONS SURVIVE	47
8.6		
	NON-SETTLING INSURERS	47
ARTICLE	IX INSURANCE POLICIES	
9.1	CONTINUATION OF INSURANCE POLICIES	48
9.2	THE PLAN IS NEUTRAL AS TO NON-IMPLICATED	
	INSURANCE POLICIES	48
ARTICLE	X PROCEDURES FOR GENERAL CLAIMS ADMINISTRATION	48
10.	1 RESERVATION OF RIGHTS TO OBJECT TO NON-TORT	
	CLAIMS	
10.		
10.		
10.	4 NO DISTRIBUTIONS PENDING ALLOWANCE	50
10.	5 CLAIM ESTIMATION	50
ARTICLE	XI DISTRIBUTIONS UNDER THE PLAN	50

11.1	PAYMENT DATE	50		
11.2	UNDELIVERABLE DISTRIBUTIONS	50		
11.3	SETOFFS	50		
11.4	NO INTEREST ON CLAIMS	51		
11.5	WITHHOLDING TAXES	51		
ARTICLE X	II EFFECTIVENESS OF THE PLAN	51		
12.1	12.1 CONDITIONS TO OCCURRENCE OF EFFECTIVE DATE			
12.2	NOTICE OF EFFECTIVE DATE	51		
12.3	EFFECT OF NON-OCCURRENCE OF CONDITIONS	51		
ARTICLE X	III EFFECTS OF CONFIRMATION			
13.1	DISSOLUTION OF UCC			
13.2	DISCHARGE AND INJUNCTION			
13.3	CHANNELING INJUNCTION			
13.4	EXCULPATION; LIMITATION OF LIABILITY			
13.5	TIMING			
13.6	NO BAR ON CERTAIN CLAIMS	55		
ARTICLE X	IV INCORPORATION OF CHILD PROTECTION PROTOCOLS			
14.1	CHILD PROTECTION PROTOCOLS	55		
ARTICLE X	V THE REORGANIZED DEBTOR	55		
15.1	CONTINUED CORPORATE EXISTENCE	55		
15.2	VESTING OF ASSETS			
15.3	IDENTITY OF OFFICERS OF REORGANIZED DEBTOR	56		
15.4	FURTHER AUTHORIZATION	56		
ARTICLE X	VI MISCELLANEOUS PROVISIONS			
16.1	RETENTION OF JURISDICTION	56		
16.2	ASSUMPTION OF EXECUTORY CONTRACTS	58		
16.3	INDEMNIFICATION OF MEMBERS, MANAGERS, OFFICERS,			
16.4	AND EMPLOYEES DEFENSE AND INDEMNITY FOR COVERED NON-TORT	38		
16.4		50		
16.5	CLAIMSRESERVATION OF RIGHTS			
16.5 16.6	NON-APPEALABLE ORDER			
16.0 16.7	AMENDMENTS AND MODIFICATIONS			
16.7	U.S. TRUSTEE REPORTS			
10.8 16.9	NO WAIVER			
16.9 16.10				
	NON-SEVERABILITY			
	REVOCATION			
	CONTROLLING DOCUMENTS			
	GOVERNING LAW			
	NOTICES			
	FILING OF ADDITIONAL DOCUMENTS			
10.10		1		

# Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 84 of 222

16.18 16.19 16.20 16.21 16.22	POWERS OF OFFICERS DIRECTION TO A PARTY SUCCESSORS AND ASSIGNS CERTAIN ACTIONS FINAL DECREE PLAN AS SETTLEMENT COMMUNICATION OTHER RIGHTS	61 62 62 62
ARTICLE XV	II BANKRUPTCY RULE 9019 REQUEST	62
ARTICLE XV	III CONFIRMATION REQUEST	63
	EXHIBITS AND SCHEDULES	
EXHIBIT A:	UNKNOWN CLAIMS REPRESENTATIVE'S REPORT AND	
	RECOMMENDATION	
EXHIBIT B:	NON-SETTLING INSURERS	
EXHIBIT C:	NON-SETTLING INSURER POLICIES	
EXHIBIT D:	TRUST AGREEMENT AND TRUST DISTRIBUTION PROTOCOL	
EXHIBIT E:	TORT CLAIM RELEASE	
EXHIBIT F:	IMPAIRED UNKNOWN TORT CLAIM RELEASE	
EXHIBIT G:	INADVERTENTLY TITLED REAL PROPERTY	
EXHIBIT H:	LIST OF CURRENT PARISHES	
EXHIBIT I:	KNOWN DIOCESE ENTITY INSURANCE POLICIES	
EXHIBIT J:	OFFICERS AND DIRECTORS OF REORGANIZED DEBTOR	
EXHIBIT K:	CHILD PROTECTION PROTOCOLS	
EXHIBIT K(1)	,	
EXHIBIT L:	OTHER INSURED ENTITIES	
EXHIBIT M:	LIST OF CATHOLIC ENTITIES	
EXHIBIT N:	SETTLING INSURER POLICIES	

#### INTRODUCTION

The Official Committee of Unsecured Creditors and the Diocese of Winona-Rochester, the debtor and debtor in possession in the above-captioned Chapter 11 case, propose this First Amended Joint Chapter 11 Plan of Reorganization (the "Plan") pursuant to the provisions of the Bankruptcy Code.

All creditors are encouraged to consult the disclosure statement for the Chapter 11 Plan of Reorganization (the "Disclosure Statement"), before voting to accept or reject this Plan. Among other information, the Disclosure Statement contains discussions of the Diocese of Winona-Rochester, events prior to and during this Chapter 11 case, and a summary and analysis of the Plan. No solicitation materials, other than the Disclosure Statement, have been authorized by the Bankruptcy Court for use in soliciting acceptances or rejections of the Plan.

### ARTICLE I DEFINITIONS AND INTERPRETATION

- **1.1 DEFINED TERMS.** For the purposes of the Plan, except as expressly provided, all capitalized terms not otherwise defined herein have the meanings ascribed to them below:
  - 1. "Abuse" means (i) any actual or alleged act of sexual conduct, misconduct, abuse, or molestation, including actual or alleged "sexual abuse" as that phrase is defined in Minnesota Statutes Section 541.073(1); (ii) indecent assault or battery, rape, lascivious behavior, undue familiarity, pedophilia, ephebophilia, or sexually-related physical, psychological, or emotional harm; (iii) contacts or interactions of a sexual nature; or (iv) assault, battery, corporal punishment, or other act of physical, psychological, or emotional abuse, humiliation, or intimidation.
  - 2. "Administrative Claim" means a Claim for costs and expenses of administration that is allowable and entitled to priority under Sections 503, 507(a)(2), or 507(b) of the Bankruptcy Code, including any post-petition tax Claims, any actual and necessary expenses of preserving the Estate, any actual and necessary expenses of operating the business of the Debtor, all Professional Claims, and any fees or charges assessed against the Estate under 28 U.S.C. § 1930.
  - 3. "Allowed Professional Claim" means a Professional Claim for which the Bankruptcy Court has entered an Order, which has become a Non-Appealable Order allowing the relevant Fee Application.
  - 4. "Approval Order" means an order of the Bankruptcy Court approving one or more Insurance Settlement Agreements.
  - 5. "Assets" of the Diocese or the Estate means, collectively, any and all property of the Diocese or the Estate, respectively, of every kind and character, wherever located, whether real or personal, tangible or intangible, and specifically including cash (including the residual balance of any reserves established under the Plan, but not the Trust) and Causes of Action.
  - 6. "Ballot" means the ballot, the form of which has been approved by the Bankruptcy Court, accompanying the disclosure statement provided to each holder of a Claim entitled to vote to accept or reject the Plan.
  - 7. "Bankruptcy Code" means the Bankruptcy Reform Act of 1978, as amended and codified in Title 11 of the United States Code.
  - 8. "Bankruptcy Court" means the United States Bankruptcy Court for the District of Minnesota.
  - 9. "Bankruptcy Rules" means the Federal Rules of Bankruptcy Procedure as currently promulgated.

- 10. "Canon Law" means the Code of Canon Law of the Roman Catholic Church, as codified in 1983 and as may hereafter be amended, and all binding universal and particular laws of the Roman Catholic Church.
- 11. "Catholic Entities" means the Diocesan Parishes, the Catholic schools and cemeteries located within the Diocese, including those listed on Exhibit M, and other non-Diocesan Catholic entities located within the Diocese that are listed on Exhibit M.
- 12. "Cause of Action" or "Causes of Action" means, except as provided otherwise in the Plan, the Confirmation Order, or any document, instrument, release, or other agreement entered into in connection with the Plan, all Claims, actions, causes of action, suits, debts, dues, sums of money, accounts, reckonings, bonds, bills, specialties, controversies, variances, trespasses, damages, judgments, third-party Claims, counterclaims, and cross claims of the Diocese or its Estate, the Committees, or the Trust (as successor to the Diocese or its Estate), including an action that is or may be pending on the Effective Date or instituted by the Reorganized Debtor after the Effective Date against any Person based on law or equity (in each case, in respect of a Cause of Action that arose before the Effective Date), including under the Bankruptcy Code, whether direct, indirect, derivative, or otherwise and whether asserted or unasserted, known or unknown, any action brought pursuant to Sections 522, 541-45, 547-51, and 553 of the Bankruptcy Code; provided, however, that any affirmative defense or crossclaim asserted with respect to a Claim shall not be deemed a Cause of Action to the extent that it seeks to disallow or reduce, or is offset against, such Claim.
- "Channeled Claim(s)" means any released Tort Claim, Contribution Claim, Medicare Claim, Extra-Contractual Claim or other released Claim against any of the Protected Parties or the Settling Insurers to the extent such released Claim arises from the same injury or damages asserted as a Tort Claim against the Protected Parties or Settling Insurers, that directly or indirectly arises out of, relates to, or is in connection with such Tort Claim or other Claim covered by the Channeling Injunction and Supplemental Settling Insurer Injunction in Articles VII and XIII; provided, however, that Channeled Claims shall not include any (i) Claim against an individual who perpetrated an act of Abuse that forms the basis of a Tort Claim with respect to that Tort Claim; (ii) Claim against any religious order, diocese or archdiocese (other than the Diocese itself); (iii) Class 4A Claim, or (iv) Class 6 Claim.
- 14. "Channeling Injunction" means the injunction imposed pursuant to Section 13.3 of the Plan.
- 15. "Child Protection Protocols" means the document entitled "Child Protection Protocols" and the related "Appendix A" included as Exhibits K and K(1).

- 16. "Claim" means any past, present or (to the extent it arises prior to the Effective Date) future Claim, demand, action, request, cause of action, suit, proceeding or liability of any kind or nature whatsoever, whether at law or equity, known or unknown, actual or alleged, asserted or not asserted, suspected or not suspected, anticipated or unanticipated, accrued or not accrued, fixed or contingent, which has been or may be asserted by or on behalf of any Person, whether seeking damages (including compensatory, punitive or exemplary damages) or equitable, mandatory, injunctive, or any other type of relief, including cross-claims, counterclaims, third-party Claims, suits, lawsuits, administrative proceedings, notices of liability or potential liability, arbitrations, actions, rights, causes of action or orders, and any other Claim within the definition of "Claim" in Section 101(5) of the Bankruptcy Code.
- 17. "Claims Filing Date" means April 8, 2019.
- 18. "Confirmation Date" means the date on which the Bankruptcy Court enters the Confirmation Order.
- 19. "Confirmation Order" means the order entered by the Bankruptcy Court confirming the Plan pursuant to Section 1129 of the Bankruptcy Code which becomes a Non-Appealable Order.
- 20. "Contribution Claim" means any Claim, most commonly expressed in terms of contribution, indemnity, equitable indemnity, subrogation, or equitable subrogation, or reimbursement, or any other indirect or derivative recovery, by a Non-Settling Insurer against a Settling Insurer for the payment of money where the holder of such Claim contends that it has paid more than its equitable or proportionate share of a Tort Claim asserted against any Protected Party.
- 21. "Covered Non-Tort Claim" means any Claim, other than Tort Claims, Unimpaired Unknown Tort Claims, Contribution Claims, Indirect Claims, or Medicare Claims, for which the Diocese, a Catholic Entity or Other Insured Entity would otherwise have coverage under a Settling Insurer Policy but for the sale, transfer, or release by the Debtor, Catholic Entity, or Other Insured Entity of such Settling Insurer Policy in connection with an Insurance Settlement Agreement.
- 22. "Debtor" means the Diocese.
- 23. "Diocesan Parishes" means all past and present parishes of or in the Diocese and includes all the parishes identified on Exhibit H.
- 24. "Diocese" and "Diocesan" refers to the Diocese of Winona-Rochester, which is the diocesan corporation formed pursuant to Minnesota Statutes Section 315.16 that is the public juridic person of the Roman Catholic Diocese of Winona-Rochester, as now constituted or as it may have been

- constituted, and the Estate (pursuant to Section 541 of the Bankruptcy Code).
- 25. "Diocese Entity Insurance Policies" mean the insurance policies that are listed on Exhibit I.
- 26. "Disputed" when used with respect to a Claim against the Diocese or property of the Diocese, means a Claim: (i) designated as disputed, contingent, or unliquidated in the Debtor's Schedules; (ii) which is the subject of an objection, appeal, or motion to estimate that has been or will be timely filed by a party in interest and which objection, appeal, or motion has not been determined by a Non-Appealable Order; or (iii) which during the period prior to the deadline fixed by the Plan or the Bankruptcy Court for objecting to such Claim, is in excess of the amount scheduled as other than disputed, unliquidated, or contingent. The processes for handling "Disputed Claims" do not apply to Class 3 or Class 4B Claims. The process for Class 3 and 4B Claims will be addressed in the Trust Agreement. In the event that any part of a Claim is Disputed, such Claim in its entirety shall be deemed to constitute a Disputed Claim for purposes of distribution under the Plan unless the Debtor or the Reorganized Debtor, as applicable, and the holder thereof agree otherwise. To the extent the term "Disputed" is used in the Plan with respect to a specified class of Claims or an unclassified category of Claims (i.e., "Disputed [class designation/unclassified Claim category] Claim"), the resulting phrase shall mean a Disputed Claim of the specified class or unclassified category of Claims.
- 27. "Distribution Plan Claimants" are Tort Claimants who elect to receive a distribution in the amount determined by the Tort Claims Reviewer pursuant to the Trust Distribution Plan and who have released all their Tort Claims against the Settling Insurers and the Protected Parties as set forth in Exhibit E or Exhibit F, as applicable.
- 28. "Distribution Plan Claims" are Tort Claims asserted by Distribution Plan Claimants.
- 29. "District Court" means the United States District Court for the District of Minnesota.
- 30. "Effective Date" means the date upon which the conditions in Article XII of the Plan have been satisfied.
- 31. "Estate" means the estate created in this Chapter 11 case pursuant to Section 541 of the Bankruptcy Code.
- 32. "Exculpated Parties" means collectively, (i) the Diocese, the Estate, and the UCC; (ii) the respective officers, directors, employees, members, attorneys, financial advisors, members of subcommittees of the board of directors, volunteers, and members of consultative bodies and councils formed under

(iii).

(iv) professionals of a Person identified in the preceding clause (i) through

- "Extra-Contractual Claims" means any Claim against the Settling Insurers 33. seeking any type of relief in connection with any alleged obligations of the Settling Insurers to the Protected Parties before the Effective Date (including compensatory, exemplary, or punitive damages, or attorneys' fees, interest, costs or any other type of relief) alleging any of the following: bad faith; failure to provide insurance coverage under any Settling Insurer Policy; failure or refusal to compromise and settle any Claim insured under any Settling Insurer Policy; failure to act in good faith; violation of any covenant or duty of good faith and fair dealing; violation of any state insurance codes, state surplus lines statutes or similar codes or statutes; violation of any unfair claims practices act or similar statute, regulation or code; any type of misconduct or any other act or omission of any type for which the claimant seeks relief other than coverage or benefits under an insurance policy. The term "Extra-Contractual Claims" includes all Claims relating to the Settling Insurers' (i) handling of any request for insurance coverage for any Claim under the Settling Insurer Policies; (ii) conduct relating to the negotiation of the Insurance Settlement Agreements; and (iii) conduct relating to the settlement of any coverage Claim concerning the Settling Insurer Policies.
- 34. "Fee Application" means an application filed with the Bankruptcy Court in accordance with the Bankruptcy Code and Bankruptcy Rules for payment of a Professional Claim.
- 35. "Final Decree" means the decree contemplated under Bankruptcy Rule 3022.
- 36. "Impaired Unknown Tort Claim" means any Tort Claim that arose prior to July 1, 2002 that was neither filed, nor deemed filed by the Claims Filing Date, and is held by (i) an individual who has a Tort Claim that was barred by the statute of limitations as of the Claims Filing Date but is no longer barred by the applicable statute of limitations for any reason as of the Effective Date, including the enactment of legislation that revises previously time-barred Tort Claims; (ii) an individual who was incapable of knowing of the existence of his or her Tort Claim as of the Effective Date for any reason, including memory repression or memory suppression; or (iii) any other individual or class of individuals the Unknown Tort Claims Representative can identify that would have a Tort Claim on or prior to the Effective Date.

- 37. "Impaired Unknown Tort Claimant" means the holder of an Impaired Unknown Tort Claim, the estate of a deceased individual who held an Impaired Unknown Tort Claim, or the personal executor or personal representative of the estate of a deceased individual who held an Impaired Unknown Tort Claim, as the case may be.
- 38. "Impaired Unknown Tort Claim Reserve Fund" means the reserve established for the benefit of Impaired Unknown Tort Claimants pursuant to Section 6.3(b) of the Plan.
- 39. "Inadvertently Titled Real Property" means that certain real property in which Debtor inadvertently holds bare legal title with no equitable or possessory interest, which is described on the attached Exhibit G.
- 40. "Indirect Claim" means a Claim asserted by any Person that is not a Non-Settling Insurer for contribution, indemnity, equitable indemnity, subrogation, or equitable subrogation, or reimbursement, or any other indirect or derivative recovery, on account of or with respect to any Tort Claim against any Protected Party.
- 41. "Insolvent Insurers" means Northwestern National Insurance Company of Milwaukee Wisconsin (f/k/a Bellefonte Insurance Co.), Stronghold Insurance Co. Ltd. and CX Reinsurance Company Ltd.
- 42. "Insurance Coverage Adversary Proceeding" means the Adversary Proceeding against United States Fire Insurance Company and others, removed by the Diocese from the State of Minnesota, County of Winona, Third Judicial District Court, to Minnesota Federal District Court and referred to the Bankruptcy Court as Adv. Proceeding No. 18-03094.
- 43. "Insurance Litigation" means any actual or potential litigation as to any recoveries from any Non-Settling Insurer or any rights under any Non-Settling Insurer Policies.
- 44. "Insurance Settlement Agreements" means the settlement agreements among the Diocese and the other Protected Parties and the Settling Insurers, which are included as Exhibit D to the Disclosure Statement.
- 45. "Interest" means all liens, Claims, encumbrances, interests, and other rights of any nature, whether at law or in equity, including any rights of contribution, indemnity, defense, subrogation, or similar relief.
- 46. "Interstate" means Interstate Fire & Casualty Company.
- 47. "Lien" means any mortgage, lien, pledge, security interest or other charge or encumbrance or security device of any kind in, upon, or affecting any Asset of the Debtor as contemplated by Section 101(37) of the Bankruptcy Code.

- 48. "Litigation Claimants" are Tort Claimants who elect to litigate their Tort Claims after the Effective Date for purposes of determining the Non-Settling Insurers' liability, if any, for such Tort Claims, including those Tort Claimants alleging Abuse during any period from July 1, 1967 to July 1, 1978.
- 49. "Litigation Claims" are Tort Claims asserted by Litigation Claimants.
- 50. "LMI" means certain Underwriters at Lloyd's, London, and certain London Market Companies (as defined in the LMI/Interstate Settlement Agreement). "LMI" does not include the Insolvent Insurers.
- 51. "LMI/Interstate Settlement Agreement" means the Insurance Settlement Agreement among the Debtor, certain Catholic Entities, LMI and Interstate.
- 52. "MAO" means Medicare Advantage Organizations under parts C & D of the MMSEA.
- 53. "Medicare Claims" means any and all Claims relating to Tort Claims by the Centers for Medicare & Medicaid Services of the United States Department of Health and Human Services and/or any other agent or successor Person charged with responsibility for monitoring, assessing, or receiving reports made under MMSEA and pursuing Claims under MSPA, including Claims for reimbursement of payments made to Tort Claimants who recover or receive any distribution from the Trust and Claims relating to reporting obligations.
- 54. "Medicare Trust Fund" means a U.S. Treasury-held trust fund account from which Medicare is funded or from which Medicare disbursements are paid, including the Hospital Insurance Trust Fund and the Supplementary Medical Insurance (SMI) Trust Fund.
- 55. "MMSEA" means § 111 of the Medicare, Medicaid, and SCHIP Extension Act of 2007 (P.L.110-173), which imposes reporting obligations on those Persons with payment obligations under the MSPA.
- 56. "MSPA" means 42 U.S.C. § 1395y et seq., or any other similar statute or regulation, and any related rules, regulations, or guidance issued in connection therewith or amendments thereto.
- 57. "Non-Appealable Order" means an order, judgment, or other decree (including any modification or amendment thereof) that remains in effect and is final and has not been reversed, withdrawn, vacated, or stayed, and as to which the time to appeal or seek review, rehearing, or writ of certiorari has expired and as to which no appeal, petition for certiorari, or other proceedings for reargument or rehearing shall then be pending or as to which, if such an appeal, writ of certiorari, review, reargument, or rehearing has been timely sought, then no order, judgment, or other decree is a Non-

- 58. "Non-Implicated Insurer" means Catholic Mutual Relief Society of America.
- 59. "Non-Implicated Insurance Policy(ies") means any known or unknown contracts, binders, certificates, or policies of insurance that the Non-Implicated Insurer issued, subscribed in any interest in, or has underwritten any risk in, in effect on or before the Effective Date, and that were issued to, allegedly issued to, or for the benefit of, or that otherwise actually, allegedly, or potentially insure, the Diocese, any Diocesan Parish or any other Catholic Entity or any of their predecessors in interest, successors, or assigns, and that actually, allegedly or could potentially afford coverage with respect to any Unimpaired Unknown Tort Claim, but as to which there are no Unimpaired Unknown Tort Claims for the applicable coverage periods that have been asserted as of the date of this Plan.
- 60. "Non-Settling Insurer Policy(ies)" means any known or unknown contracts, binders, certificates, or policies of insurance that any Non-Settling Insurer issued, subscribed in any interest in, or has underwritten any risk in, in effect on or before the Effective Date, and that were issued to, allegedly issued to, or for the benefit of, or that otherwise actually, allegedly, or potentially insure, the Diocese or any Diocesan Parish or any of their predecessors in interest, successors, or assigns, and that actually, allegedly or could potentially afford coverage with respect to any Tort Claim.
- 61. "Non-Settling Insurer(s)" means any and all Persons (including all their respective affiliates, predecessors, successors and assigns) that are not "Settling Insurers" and that have, or are alleged to have, extended insurance coverage for Tort Claims to the Diocese or any Diocesan Parish, or have issued, subscribed in any interest in, assumed any liability for, or underwritten any risk in, any Non-Settling Insurer Policy. The "Non-Settling Insurers" include, but are not limited to, those Persons listed on Exhibit B.
- 62. "Other Insured Entity(ies)" means those Persons listed on Exhibit L that are insured or covered or allegedly insured or covered under a Settling Insurer

Policy that was issued or allegedly issued to the Diocese, but only with respect to any Tort Claim based on alleged Abuse that occurred during the effective periods of that Settling Insurer Policy and that would be covered or alleged to be covered under that Settling Insurer Policy but for an Insurance Settlement Agreement. Notwithstanding the foregoing, "Other Insured Entities" does not include the Diocese or the Diocesan Parishes. An individual who perpetrated an act of Abuse that forms the basis of a Tort Claim is not an Other Insured Entity. No religious order, archdiocese or diocese is an Other Insured Entity.

- 63. "Person" means any individual or entity, including any corporation, limited liability company, partnership, general partnership, limited partnership, limited liability partnership, limited liability limited partnership, proprietorship, association, joint stock company, joint venture, estate, trust, trustee, personal executor or personal representative, unincorporated association, or other entity, including any federal, international, foreign, state, or local governmental or quasi-governmental entity, body, or political subdivision or any agency or instrumentality thereof and any other individual or entity within the definition of (i) "person" in Section 101(41) of the Bankruptcy Code; or (ii) "entity" in Section 101(15) of the Bankruptcy Code.
- 64. "Petition Date" means November 30, 2018, the date on which the Diocese commenced the Chapter 11 case.
- 65. "Plan" means this joint Chapter 11 plan of reorganization, either in its present form or as it may be altered, amended, or modified from time to time in accordance with the provisions of the Bankruptcy Code and the Bankruptcy Rules.
- 66. "Plan Proponents" means the Diocese and the UCC.
- 67. "Priority Tax Claim" means a Claim of a governmental unit of the kind specified in Section 507(a)(8) of the Bankruptcy Code.
- 68. "Pro Rata" means, with respect to any distribution on account of any allowed Claim in any class, the ratio of the amount of such allowed Claim to the sum of (i) all allowed Claims in such class and (ii) the aggregate maximum of all Claims in such class that are not yet allowed Claims.
- 69. "Professional" means any professional employed or to be compensated pursuant to §§ 327, 328, 330, 331, 503(b), or 1103 of the Bankruptcy Code.
- 70. "Professional Claim" means a Claim for compensation for services and/or reimbursement of expenses pursuant to §§ 327, 328, 330, 331, or 503(b) of the Bankruptcy Code in connection with an application made to the Bankruptcy Court in the Chapter 11 case.

- 71. "Proof of Claim" means a proof of Claim filed in the Chapter 11 case pursuant to § 501 of the Bankruptcy Code and/or pursuant to any order of the Bankruptcy Court, together with supporting documents.
- 72. "Protected Party" means any of (i) the Diocese, (ii) the Reorganized Debtor, (iii) the Diocesan Parishes; (iv) the Catholic Entities, (v) the Other Insured Entities (but, as set forth in the definition of "Other Insured Entities," such "Other Insured Entities" are Protected Parties only as to certain Tort Claims); (vi) each of the foregoing Persons' respective past, present and future parents, subsidiaries, affiliates, holding companies, merged companies, related companies, divisions and acquired companies; (vii) each of the foregoing Persons' respective predecessors, successors and assigns; (viii) solely to the extent of and in their capacity as such, each of the foregoing Persons' respective past and present employees, officers, directors, shareholders, principals, teachers, staff, members, boards, administrators, priests, deacons, brothers, sisters, nuns, other clergy or persons bound by monastic vows, volunteers; and (ix) agents, attorneys and representatives, in their capacity as such, of the Persons identified in the foregoing subsections (i)-(viii). Nothing in the foregoing is intended to suggest that such Persons are "employees" or agents of the Diocese, or subject to its control. An individual who perpetrated an act of abuse that forms the basis of a Tort Claim or an Unimpaired Unknown Tort Claim is not a Protected Party. No religious order, archdiocese or diocese, other than the Diocese itself, is a Protected Party.
- 73. "Reorganization Assets" means, collectively, all Assets of the Debtor and the Estate. For the avoidance of doubt, the Reorganization Assets do not include the Trust Assets.
- 74. "Reorganized Debtor" means the Diocese, on and after the Effective Date. Unless otherwise expressly stated or the context otherwise requires, references to the "Diocese" and the "Reorganized Debtor" throughout various provisions of the Plan are an effort to anticipate whether an event may occur before or after the Effective Date. In this regard, and generally for purposes of the Plan, any written agreement signed after the Petition Date made by the Diocese as part of the Plan before the Effective Date (unless otherwise provided) will survive the Confirmation Date and the Effective Date and will bind the Reorganized Debtor and every other party to such agreement (including, but not limited to, the provisions of the Plan if confirmed).
- 75. "Settling Insurer Policies" means a) all policies or certificates of insurance listed on Exhibit N; and b) with respect to LMI (but not with respect to Interstate), any and all other known and unknown contracts, binders, certificates, or policies of insurance, including all of the insurance policies mentioned or referred to in any Insurance Settlement Agreement, and which (i) were issued by a Settling Insurer; (ii) were in effect on or before the

Effective Date; (iii) were issued to, allegedly issued to, or for the benefit of, or otherwise actually, allegedly, or potentially insure, the Diocese or any of its predecessors in interest, successors, or assigns; and (iv) actually, allegedly, or could potentially afford coverage with respect to any Tort Claim. If, however, a contract, binder, certificate, or policy of insurance that is not identified on Exhibit N and was not issued to or allegedly issued to the Diocese insures or covers both the Diocese and and any other Person, such contract, binder, certificate, or policy of insurance, as applicable, is a "Settling Insurer Policy" to the extent it insures or covers the Diocese and the Diocesan Parishes but not to the extent it insures or covers any other Person. For clarity, the following policies do not constitute Settling Insurer Policies: (1) a Cyber Liability policy issued to Catholic Mutual and effected through NAS Insurance Services with Certain Underwriters at Lloyds, London under Master Policy No. 510641, which expires July 1, 2021; (2) a Malicious Attack Liability Coverage and Crisis Response policy issued to Catholic Mutual and effected through Aon with Certain Underwriters at Lloyds, London under Master Policy No. CMCTR2003035, which expires July 1, 2021; or (3) any policy issued by LMI or Interstate to the Boy Scouts of America, Delaware BSA, LLC, any Boy Scout local council or other organization or affiliate of the Boy Scouts of America, under which the Diocese or any Diocesan Parish is insured.

76. "Settling Insurers" means the Persons listed on Exhibit N whose Insurance Settlement Agreements are approved by the Approval Orders and such orders become Non-Appealable Orders. Solely in connection with insurance under any Settling Insurer Policies, Settling Insurers also includes each of their past, present and future parents, subsidiaries, affiliates, divisions, reinsurers, and retrocessionaires, including Persons released pursuant to the Insurance Settlement Agreements; each of the foregoing Persons' respective past, present and future parents, subsidiaries, affiliates, holding companies, merged companies, related companies, divisions and acquired companies, including the Persons released pursuant to the respective Insurance Settlement Agreements; each of the foregoing Persons' respective past, present and future directors, officers, shareholders, employees, subrogees, partners, principals, agents, attorneys, joint ventures, joint venturers, representatives, and Claims handling administrators; and each of the foregoing Persons' respective predecessors, successors, assignors, and assigns, whether known or unknown, and all Persons acting on behalf of, by, through or in concert with them, except to the extent, if any, such Person's actual or alleged rights, duties, obligations, or liabilities arise out of or relate to their status as, or conduct, acts or omissions on behalf of, a Non-Settling Insurer. Any Non-Settling Insurer who enters into a final and binding Insurance Settlement Agreement with the Trust after the Effective Date is also a Settling Insurer, except to the extent, if any, such Person's actual or alleged rights, duties, obligations or liabilities arise out of or relate to their status as, or conduct, acts or omissions on behalf of, a Non-Settling Insurer.

- 77. "Supplemental Plan Documents" means, collectively, the documents included (or to be included) in the supplemental appendix to the Plan and filed with the Bankruptcy Court at least 14 days prior to the confirmation hearing.
- 78. "Supplemental Settling Insurer Injunction" means the injunction imposed pursuant to Section 7.9 of the Plan.
- 79. "Tort Claim" means any Claim (other than an Unimpaired Unknown Tort Claim) against any of the Protected Parties or any of the Settling Insurers that arises out of, relates to, results from, or is in connection with, in whole or in part, directly or indirectly, Abuse that took place in whole or in part prior to the Effective Date, including any such Claim that seeks monetary damages or any other relief, under any theory of liability, including vicarious liability; respondeat superior; any fraud-based theory, including fraud in the inducement; any negligence-based or employment-based theory, including negligent hiring, supervision, retention misrepresentation; any other theory based on misrepresentation, concealment, or unfair practice; public or private nuisance; or any other theory, including any theory based on public policy or any acts or failures to act by any of the Protected Parties, any of the Settling Insurers or any other Person for whom any of the Protected Parties are allegedly responsible, including any such Claim asserted against any of the Protected Parties in connection with this Chapter 11 case. Tort Claim includes any Impaired Unknown Tort Claim. For clarity, Tort Claim does not include any Claim for contribution, indemnity or reimbursement held by a joint tortfeasor, or the Claim of any insurers or other Persons who are subrogated to such Claims.
- 80. "Tort Claimant" means the holder of a Tort Claim.
- 81. "Tort Claims Reviewer" means the Person, including the designee of such Person, who will assess Class 3 and Class 4B Claims under the Trust Distribution Plan.
- 82. "Transferred Insurance Interests" means the following rights and interests of the Protected Parties in Non-Settling Insurer Policies in respect of actual or potential coverage for any Class 3 Claim or Class 4B Claim: (1) the proceeds of such Non-Settling Insurer Policies and all claims for such proceeds, including the claims filed by the Diocese in the insolvency proceedings of Northwestern National Insurance Company of Milwaukee (f/k/a Bellefonte Insurance Co.), Wisconsin, Stronghold Insurance Co. Ltd. and CX Reinsurance Company Ltd.; and (2) all claims and causes of action that currently exist or may arise in the future against Non-Settling Insurers based on their conduct concerning insurance coverage for, or defense or settlement of, any Class 3 Claim or Class 4B Claim, including but not limited to all claims and causes of action for breach of the Non-Settling

Insurer Policies, vexatious refusal, bad faith, wrongful failure to settle, and for any other similar claim or cause of action, including any and all such claims or causes of action providing for penalties, extra-contractual damages, punitive damages and attorneys' fees and costs.

- 83. "Trust" means the trust created for the benefit of certain Tort Claimants in accordance with the Plan and Confirmation Order and the Trust Agreement.
- 84. "Trust Agreement" or "Trust Documents" shall mean the trust agreement establishing the Trust, as it may be amended, together with such additional documents as may be executed in connection with the Trust Agreement.
- 85. "Trust Assets" means the cash, Transferred Insurance Interests and other assets and rights to be transferred to the Trust under Article V of the Plan.
- 86. "Trust Distribution Plan" means the Trust Distribution Plan established under the Trust Agreement.
- 87. "Trustee" means the Person appointed as Trustee of the Trust in accordance with the terms of the Plan, the Confirmation Order, and the Trust Agreement, or any successor appointed in accordance with the terms of the Plan, Confirmation Order, and the Trust Agreement.
- 88. "UCC" means the Official Committee of Unsecured Creditors appointed in this Chapter 11 case, as such committee may be constituted from time to time.
- 89. "Unimpaired" means, with respect to a class of Claims, that such class is not impaired.
- 90. "Unimpaired Unknown Tort Claim" means any Claim against any of the Protected Parties or the Non-Implicated Insurer that arises out of, relates to, results from, or is in connection with, in whole or in part, directly or indirectly, Abuse that took place in whole or in part prior to the Effective Date but after June 30, 2002, including any such Claim that seeks monetary damages or any other relief, under any theory of liability, including vicarious liability; respondeat superior; any fraud-based theory, including fraud in the inducement; any negligence-based or employment-based theory, including negligent hiring, supervision, retention misrepresentation; any other theory based on misrepresentation, concealment, or unfair practice; contribution; indemnity; public or private nuisance; or any other theory, including any theory based on public policy or any acts or failures to act by any of the Protected Parties, the Non-Implicated Insurer or any other Person for whom any of the Protected Parties are allegedly responsible, including any such Claim asserted against any of the Protected Parties in connection with this Chapter 11 case, which Claim was neither filed, nor deemed filed by the Claims Filing Date.

### Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 99 of 222

- 91. "Unimpaired Unknown Tort Claimant" means the holder of an Unimpaired Unknown Tort Claim.
- 92. "Unknown Claims Representative" means Judge Michael R. Hogan in accordance with the Bankruptcy Court's order dated \_\_\_\_\_\_\_, and any successor or such other person appointed by the Bankruptcy Court or otherwise.
- 93. "Unsecured Claims" means Claims which are not secured by any property of the Debtor's Estate and which are not part of any other class defined in this Plan.
- 94. "U.S. Trustee" means the Office of the United States Trustee for Region 12, which includes the District of Minnesota.

#### **1.2 INTERPRETATION**. For purposes of the Plan:

- (a) any term that is not defined herein, but that is used in the Bankruptcy Code or the Bankruptcy Rules, shall have the meaning assigned to that term in the Bankruptcy Code or the Bankruptcy Rules, as applicable;
- **(b)** the terms "including" or "include(s)" are intended to be illustrative and not exhaustive, and shall be construed as "including, but not limited to" or "include(s), but is not limited to":
- (c) whenever the context requires, terms shall include the plural as well as the singular number, and the masculine gender shall include the feminine and the feminine gender shall include the masculine;
- (d) the rules of construction set forth in Section 102 of the Bankruptcy Code and in the Bankruptcy Rules shall apply;
- (e) unless the context should otherwise require, all references to documents to be filed shall refer to filing with the Bankruptcy Court in accordance with the Bankruptcy Code and Bankruptcy Rules;
- (f) any reference in the Plan to a contract, instrument, release, indenture, or other agreement or document being in a particular form or on particular terms and conditions means that such document shall be substantially in such form or substantially on such terms and conditions;
- (g) any reference in the Plan to an existing document or exhibit filed or to be filed means such document or exhibit, as it may have been or may be amended, modified or supplemented;
- **(h)** unless otherwise specified, all references in the Plan to "Articles," "Sections," "Schedules" and "Exhibits" are references to Articles, Sections, Schedules and Exhibits of or to the Plan;

- (i) the words "herein," "hereof," and "hereto" refer to the Plan in its entirety rather than to a particular portion of the Plan;
- (j) captions and headings to Articles and Sections are inserted for ease of reference only and shall not be considered a part of the Plan or otherwise affect the interpretation of the Plan; and
- (k) the Plan supersedes all prior drafts of the Plan, and all prior negotiations, agreements, and understandings with respect to the Plan, evidence of which shall not affect the interpretation of any provision of the Plan.
- 1.3 TIME PERIODS. In computing any period of time prescribed or allowed by the Plan, unless otherwise expressly provided, the provisions of Bankruptcy Rule 9006(a) shall apply. Enlargement of any period of time prescribed or allowed by the Plan shall be governed by the provisions of Bankruptcy Rule 9006(b).

#### 1.4 EXHIBITS AND SCHEDULES.

- (a) All Exhibits and Schedules to the Plan (including any Supplemental Plan Documents) (with the Plan, the "Plan Documents") are hereby incorporated by reference and made part of the Plan as if set forth fully herein.
  - **(b)** The Exhibits to the Plan include the following:

Exhibit A: Unknown Claims Representative's Report and

Recommendation

Exhibit B: Non-Settling Insurers

Exhibit C: Non-Settling Insurer Policies

Exhibit D: Trust Agreement and Trust Distribution Protocol

Exhibit E: Tort Claim Release

Exhibit F: Impaired Unknown Tort Claim Release

Exhibit G: Inadvertently Titled Real Property

Exhibit H: List of Current Parishes

Exhibit I: Known Diocese Entity Insurance Policies
Exhibit J: Officers and Directors of Reorganized Debtor

Exhibit K: Child Protection Protocols

Exhibit K(1): Appendix A

Exhibit L: Other Insured Entities
Exhibit M: List of Catholic Entities
Exhibit N: Settling Insurer Policies

## ARTICLE II TREATMENT OF UNCLASSIFIED CLAIMS

**2.1 ADMINISTRATIVE CLAIMS**. As provided in Section 1123(a)(1) of the Bankruptcy Code, Administrative Claims including Professional Claims, and Priority Tax Claims shall not be classified for the purposes of voting or receiving distributions under the Plan. Rather,

all such Claims shall be treated separately as unclassified Claims on the terms set forth in this Article.

(a) Treatment. Each holder of an allowed Administrative Claim, excluding Professional Claims, against the Diocese shall receive, in full satisfaction, settlement, release, and extinguishment of such Claim, an amount from the Reorganized Debtor equal to the allowed amount of such Administrative Claim, unless the holder agrees in writing to other treatment of such Claim. Each holder of an Allowed Professional Claim shall receive, in full satisfaction, settlement, release, and extinguishment of such Claim, an amount from the Reorganized Debtor equal to the allowed amount of such Professional Claim, unless the holder agrees in writing to other treatment of such Claim and subject to the provisions of Section 5.2 of this Plan.

#### (b) Administrative Filing Deadline.

- 1. Except as otherwise set forth in this Plan, requests for allowance and payment of Administrative Claims, excluding Professional Claims, must be filed and served no later than thirty (30) days after a notice of the Effective Date is filed with the Bankruptcy Court (the "Administrative Claims Filing Deadline"). Administrative Claims holders, excluding Professional Claims, that do not file a request for payment by the Administrative Claims Filing Deadline shall be forever barred from asserting such Claims against the Diocese, the Reorganized Debtor, any Settling Insurer (to the extent applicable), the Trust, or any of their property. Administrative Claims representing obligations incurred by the Diocese after the Effective Date (including, without limitation, Claims for professionals' fees and expenses) will not be subject to application to the Bankruptcy Court and may be paid by the Reorganized Debtor in the ordinary course of business and without Bankruptcy Court approval. In addition, holders of Administrative Claims representing trade debt incurred after the Petition Date in the ordinary course of Debtor's operations are not required to file requests for allowance of an Administrative Claim and will be paid by the Debtor in the ordinary course.
- 2. All objections to the allowance of Administrative Claims (excluding Professional Claims) must be served and filed by any parties-in-interest no later than fourteen (14) days after the Administrative Claim Filing Deadline (the "Administrative Claim Objection Deadline"). If no objection to the applicable Administrative Claim is filed on or before the Administrative Claim Objection Deadline, such Administrative Claim will be deemed allowed. For the avoidance of doubt, the Administrative Claim Objection Deadline established by this subparagraph shall control over any contrary deadline set forth in any requests for payment of Administrative Claims.

#### (c) Professional Claim Filing Deadline.

All Professionals or other Persons holding a Professional Claim for services rendered on or before the Effective Date (including, among other things, any

compensation requested by any Professional or any other Person for making a substantial contribution in the Chapter 11 case) shall file and serve an application for final allowance of compensation and reimbursement of expenses accruing from the Petition Date to the Effective Date, no later than thirty (30) days after a notice of the Effective Date is filed (the "Professional Claim Filing Deadline").

- 2.2 STATUTORY FEES. All fees due and payable pursuant to 28 U.S.C. § 1930 and not paid prior to the Effective Date shall be paid by the Reorganized Debtor as soon as practicable after the Effective Date. After the Effective Date, the Trust shall pay quarterly fees to the U.S. Trustee until the Chapter 11 case is closed, but in no event shall the payments made to the Trust made pursuant to Article IV, V, or VI by any Person other than the Debtor be considered "disbursements" under 28 U.S.C. § 1930, nor shall any payment made by the Trust to any Person be considered a disbursement under 28 U.S.C. § 1930. The Reorganized Debtor shall file post-Confirmation Date quarterly reports in conformance with the U.S. Trustee guidelines. The U.S. Trustee shall not be required to file a request for payment of its quarterly fees, which will be deemed Administrative Claims against the Debtor and its Estate. The Reorganized Debtor shall remain responsible for any pre-Effective Date reporting and pre-Effective Date unpaid fees.
- **2.3 PRIORITY TAX CLAIMS**. With respect to each allowed Priority Tax Claim not paid prior to the Effective Date, the Reorganized Debtor shall (i) pay such Claim in cash as soon as practicable after the Effective Date from its ongoing operations, or (ii) provide such other treatment agreed to by the holder of such allowed Priority Tax Claim and the Diocese or Reorganized Debtor, as applicable, in writing, provided such treatment is no less favorable to the Diocese than the treatment set forth in clause (i) of this sentence.

# ARTICLE III CLASSIFICATION OF CLAIMS

**3.1 SUMMARY**. The categories of Claims listed below classify Claims (except for Administrative Claims and Priority Tax Claims) for all purposes, including voting, confirmation of the Plan, and distribution pursuant to the Plan.

CLASS	DESCRIPTION	IMPAIRMENT	VOTING
1	Priority Claim	Unimpaired	No
2	Governmental Unit Claims	Unimpaired	No
3	Tort Claims Other Than Impaired Unknown Tort Claims	Impaired	Yes
4A	Unimpaired Unknown Tort Claims	Unimpaired	No
4B	Impaired Unknown Tort Claims	Impaired	Yes
5	General Unsecured Claims	Unimpaired	No
6	Unimpaired Abuse Related Contingent Claims	Unimpaired	No
7	Secured Claim of Minnesota Department of Commerce	Unimpaired	No
8	Secured Claim of Community Bank of Mankato	Unimpaired	No

#### 3.2 CLASSIFICATION AND VOTING.

The Claims against the Debtor shall be classified as specified above (other than Administrative Claims and Priority Tax Claims, which shall be unclassified and treated in accordance with Article II). Consistent with Section 1122 of the Bankruptcy Code, a Claim is classified by the Plan in a particular class only to the extent the Claim is within the description of the class, and a Claim is classified in a different class to the extent it is within the description of that different class.

### ARTICLE IV TREATMENT OF CLASSIFIED CLAIMS

#### **4.1 PRIORITY CLAIMS (CLASS 1).**

- (a) **Definition**. A Class 1 Claim means an allowed Claim described in, and entitled to priority under Sections 507(a) and 503(b)(9) of the Bankruptcy Code other than an Administrative Claim or a Priority Tax Claim.
- **(b) Treatment**. Unless the holder of an allowed Class 1 Claim and the Diocese or the Reorganized Debtor (as applicable) agree to a different treatment, the Reorganized Debtor shall pay each such allowed Class 1 Claim in full, in cash, without interest, from ongoing operations on the later of the Effective Date (or as soon thereafter as is practicable) and the date a Class 1 Claim becomes an allowed Claim (or as soon thereafter as is practicable).

#### **4.2** GOVERNMENTAL UNIT CLAIMS (CLASS 2).

(a) **Definition**. A "Class 2 Claim" means an allowed Claim of Governmental Units not otherwise included in Article II or Section 4.1 above.

**(b) Treatment**. Unless the holder of an allowed Class 2 Claim and the Diocese or the Reorganized Debtor (as applicable) agree to a different treatment, the Reorganized Debtor shall pay each such allowed Class 2 Claim in full, in cash, without interest, from ongoing operations on the later of the Effective Date (or as soon thereafter as is practicable) and the date a Class 2 Claim becomes an allowed Claim (or as soon thereafter as is practicable).

### 4.3 TORT CLAIMS OTHER THAN IMPAIRED UNKNOWN TORT CLAIMS (CLASS 3).

(a) **Definition**. A Class 3 Claim means a Tort Claim other than an Impaired Unknown Tort Claim ("Class 3 Claim"). A "Class 3 Claimant" shall mean a holder of a Class 3 Claim.

**Summary**. The Plan creates a Trust to fund payments to Class 3 Claimants entitled to such payments under the Plan, Trust Agreement, and Trust Distribution Plan. The Trust shall be funded as provided in Articles IV ,V, and VI, including by contributions from the Diocese and others and the assignment of the Transferred Insurance Interests. The Trust shall make distributions to the Class 3 Claimants, as provided by this Plan, the Trust Agreement, and the Trust Distribution Plan, which shall represent the sole recovery available to Class 3 Claimants in respect to any obligation owed by Settling Insurers. Distribution from the Trust, however, does not preclude or affect claims or recoveries by Class 3 Claimants against the Non-Settling Insurers.

No Class 3 Claimant shall receive any payment on any award unless and until such Class 3 Claimant has executed the Release attached as Exhibit E to this Plan. Each Class 3 Claimant must execute a release of all claims against the Settling Insurers and must release all claims against the Diocese, the Reorganized Debtor, and any other Protected Party that do not implicate insurance coverage under Non-Settling Insurer Policies. To preserve coverage under Non-Settling Insurer Policies, Class 3 Claimants specifically reserve, and do not release, any and all claims that they may have against the Diocese, the Reorganized Debtor, or any other Protected Party that implicate coverage under Non-Settling Insurer Policies, but recourse is limited to the proceeds of Non-Settling Insurer Policies and all other damages (including extra-contractual damages), awards, judgments in excess of policy limits, penalties, punitive damages and attorney's fees and costs that may be recoverable against any Non-Settling Insurers because of their conduct concerning insurance coverage for, or defense or settlement of, any Tort Claim, and any such judgments or awards will be handled in accordance with Sections 6.14(i) and (j). The Class 3 Claims will not be released or enjoined as against the Diocese, the Reorganized Debtor, or any other Protected Party for any Abuse that may be covered under Non-Settling Insurer Policies until such claims are settled with the Diocese, the Reorganized Debtor, any other Protected Party and such Non-Settling Insurer or are fully adjudicated, resolved, and subject to Final Order, but recourse is limited as described above.

Any release of Class 3 Claims, in whole or in part, will be pursuant to the principles set forth in *Pierringer v. Hoger*, 124 N.W.2d 106 (Wis. 1963) and *Frey v. Snelgrove*, 269 N.W.2d 918 (Minn. 1978). The Claimants will expressly reserve their rights against other

Persons, including joint tortfeasors, who will remain severally liable on any Class 3 Claims. Any Person that is or was alleged to be a joint tortfeasor with any of the Protected Parties in connection with the Abuse that forms the basis of a Class 3 Claim shall not be liable for any Protected Party's share of causal liability or fault. In no event may a Class 3 Claimant collect on that portion of any judgment or obtain any reallocation of any judgment based on the causal fault or share of liability of any Protected Parties. Any Person that is or was alleged to be a joint tortfeasor with any of the Protected Parties in connection with the Abuse that forms the basis of a Class 3 Claim shall be provided by the Trustee with a copy of the executed Release upon reasonable request and provision of an appropriate, executed confidentiality agreement and shall not be liable for any Protected Parties' share of liability or fault. The Trust shall be obligated to provide copies of the Class 3 Claimants' releases and certifications to any of the Protected Parties or Settling Insurers upon request provided that such Protected Parties or Settling Insurers have signed a confidentiality agreement encompassing such information.

The Trust shall fund the defense of the Diocese, the Reorganized Debtor, and any other Protected Party as against any Litigation Claims brought by Class 3 Claimants, but only to the extent that the Diocese, the Reorganized Debtor, or any other Protected Party, as applicable, is not defended or otherwise reimbursed for its defense expenses on an advance basis by any Insurer. The Trust shall advance funding to the Diocese, the Reorganized Debtor, or any other Protected Party, as applicable, with respect to any judgments or settlements of any Litigation Claims brought by Class 3 Claimants, but only to the extent that such judgments or settlements are not funded by any Insurer. The Trust shall pursue recoveries against any Non-Settling Insurers in respect of the Transferred Insurance Interests.

The Non-Settling Insurers remain fully liable for their obligations related in any way to the Class 3 Claims, and their obligations are not reduced by the fact that the Diocese is in bankruptcy or by the amount of distributions Class 3 Claimants receive, or are entitled to receive, based on the Trust Distribution Plan. For the avoidance of doubt, determinations by the Tort Claims Reviewer and/or any distributions entitled to be received from the Trust shall not constitute a determination of any Protected Party's liability or damages for Class 3 Claims. The Trust may continue efforts to obtain recoveries from Non-Settling Insurers related to the Class 3 Claims. Any such recoveries by the Trust from Non-Settling Insurers will likewise become Trust Assets to be distributed pursuant to the Trust Distribution Plan. To bar any argument by the Non-Settling Insurers that any provision of this Plan, including the assignment and transfer of the Transferred Insurance Interests to the Trust, results in a forfeiture of coverage, this Plan preserves the Non-Settling Insurers' rights to the extent required under their respective Non-Settling Insurer Policies and applicable law.

**(b) Treatment**. Responsibility for preserving and managing Trust Assets and distributing Trust Assets to Class 3 Claimants shall be assigned to, assumed and treated by the Trust as further provided in Article VI, the Trust Agreement, and the Trust Distribution Plan. Class 3 Claims shall be paid in accordance with the provisions of the Trust and Trust Distribution Plan.

(c) Stipulated Judgments. Certain Class 3 Claimants have entered into, or may enter into, agreements with the Diocese, the Reorganized Debtor, or any other Protected Party for settlement of a Tort Claim allocated by applicable non-bankruptcy law, including but not limited to settlements consistent with *Miller v. Shugart*, 316 N.W.2d 729 (Minn. 1982) or *Drake v. Ryan*, 514 N.W.2d 785 (Minn. 1994). If a Class 3 Claimant enters into such an agreement with the Diocese, the Reorganized Debtor, or any other Protected Party, the Trust will pursue any judgment against the Non-Settling Insurer on behalf of the Class 3 Claimant. Any recoveries by the Trust from Non-Settling Insurers will become Trust Assets to be distributed pursuant to the Trust Distribution Plan.

#### 4.4 UNIMPAIRED UNKNOWN TORT CLAIMS (CLASS 4A).

- (a) **Definition**. A Class 4A Claim means an Unimpaired Unknown Tort Claim ("Class 4A Claim"). A "Class 4A Claimant" shall mean a holder of a Class 4A Claim.
- **(b) Treatment**. As of the date of this Plan, the Debtor is not aware of the existence of any Class 4A Claims. The Reorganized Debtor will be responsible for defending any Class 4A Claim that is asserted and for payment of any amount determined to be owed on a Class 4A Claim upon settlement or adjudication of such Class 4A Claim. The Trust will not be responsible for paying Class 4A Claims and will not have a role in the settlement or adjudication of Class 4A Claims.

#### 4.5 IMPAIRED UNKNOWN TORT CLAIMS (CLASS 4B).

- (a) **Definition**. A Class 4B Claim means an Impaired Unknown Tort Claim ("Class 4B Claim"). A "Class 4B Claimant" shall mean a holder of a Class 4B Claim.
- (b) Treatment. The Plan creates a Trust to administer payments to Class 4B Claimants entitled to such payments under the Plan, Trust Agreement, and Trust Distribution Plan. The Trust shall be funded as provided in Articles IV, V, and VI, including from post-confirmation payments from the Reorganized Debtor and third parties. The Trust shall make distributions to the Class 4B Claimants, as provided by this Plan, the Trust Agreement, and the Trust Distribution Plan, which shall represent the sole recovery available to Class 4B Claimants in respect to any obligation owed by Settling Insurers. Distribution from the Trust, however, does not preclude or affect claims or recoveries by Class 4B Claimants against the Non-Settling Insurers.

No Class 4B Claimant shall receive any payment on any award unless and until such Class 4B Claimant has executed the Release attached as <a href="Exhibit F">Exhibit F</a> to this Plan. Each Class 4B Claimant must execute a release of all claims against the Settling Insurers and must release all claims against the Diocese, the Reorganized Debtor, and any other Protected Party that do not implicate insurance coverage under Non-Settling Insurer Policies. To preserve coverage under Non-Settling Insurer Policies, Class 4B claimants specifically reserve, and do not release, any and all claims that they may have against the Diocese, Reorganized Debtor, or any other Protected Party that implicate coverage under Non-Settling Insurer Policies, but recourse is limited to the proceeds of Non-Settling Insurer Policies and all other damages (including extra-contractual damages), awards,

judgments in excess of policy limits, penalties, punitive damages and attorney's fees and costs that may be recoverable against any Non-Settling Insurers because of their conduct concerning insurance coverage for, or defense or settlement of, any Tort Claim, and any such judgments or awards will be handled in accordance with Sections 6.14(i) and (j). The Class 4B Claims will not be released or enjoined as against the Diocese, the Reorganized Debtor, or any other Protected Party for any Abuse that may be covered under Non-Settling Insurer Policies until such claims are settled with the Diocese, the Reorganized Debtor, or any other Protected Party, as applicable, and such Non-Settling Insurer or are fully adjudicated, resolved and subject to Final Order, but recourse is limited as described above.

The Trust shall fund the defense of the Diocese, the Reorganized Debtor, and any other Protected Party as against any Litigation Claims brought by Class 4B Claimants, but only to the extent that the Diocese, the Reorganized Debtor, or any other Protected Party, as applicable, is not defended or otherwise reimbursed for its defense expenses on an advance basis by any Insurer. The Trust shall advance funding to the Diocese, the Reorganized Debtor, or any other Protected Party, as applicable, with respect to any judgments or settlements of any Litigation Claims brought by Class 4B Claimants, but only to the extent that such judgments or settlements are not funded by any Insurer. The Trust shall pursue recoveries against any Non-Settling Insurers in respect of the Transferred Insurance Interests.

The Non-Settling Insurers remain fully liable for their obligations related in any way to the Class 4B Claims. Any release of Class 4B Claims, in whole or in part, will be pursuant to the principles set forth in *Pierringer v. Hoger*, 124 N.W.2d 106 (Wis. 1963) and *Frey v. Snelgrove*, 269 N.W.2d 918 (Minn. 1978). The Class 4B Claimants will expressly reserve their rights against other Persons, including joint tortfeasors, who will remain severally liable on any Class 4B Claims. Nothing in this Article requires any Impaired Unknown Tort Claimant to release any Claims against any joint tortfeasor who is not a Protected Party or a Settling Insurer and such Claims are reserved. But in no event may a Class 4B Claimant collect on that portion of any judgment or obtain reallocation of any judgment based on the causal fault or share of liability of any Protected Party.

Responsibility for preserving and managing Trust Assets and distributing Trust Assets to Class 4B Claimants shall be assigned to, assumed, and treated by the Trust as further provided in Article V, the Trust Agreement, and the Trust Distribution Plan. Class 4B Claims shall be paid in accordance with the provisions of the Trust and Trust Distribution Plan. Class 4B Claimants shall provide sufficient information to allow the Tort Claims Reviewer to make an evaluation of the Class 4B Claim pursuant to the factors in the Trust Distribution Plan.

#### 4.6 GENERAL UNSECURED CLAIMS (CLASS 5).

(a) **Definition**. A Class 5 Claim means (1) any Claim arising out of the rejection of any executory contract, or (2) any Unsecured Claim that is not included in another class under the Plan and is not listed as disputed, contingent or unliquidated on the Debtor's schedules filed in connection with this Chapter 11 case ("Debtor's Schedules") or as to which the holder of such Claim timely filed a Claim.

**(b) Treatment**. Each holder of a Class 5 Claim shall receive, directly from the Reorganized Debtor, payment in full of such allowed Class 5 Claim, without interest, on the Effective Date.

#### 4.7 UNIMPAIRED ABUSE RELATED CONTINGENT CLAIMS (CLASS 6).

- (a) Class 6 Definition. A Class 6 Claim means (i) any Claim for contribution, indemnity or reimbursement arising out of the Diocese's liability to pay or defend any Class 4A Claim, and (ii) the Claim of any insurers or other Persons who are subrogated to the Claims identified in Section 4.7(a) clause (i).
- **(b)** Class 6 Treatment. As of the date of this Plan, the Debtor is not aware of the existence of any Class 6 Claims. The Reorganized Debtor will be responsible for defending any Class 6 Claim that is asserted and for payment of any amount determined to be owed on a Class 6 Claim upon settlement or adjudication of such Class 6 Claim. The Trust will not be responsible for paying Class 6 Claims and will not have a role in the settlement or adjudication of Class 6 Claims.

#### 4.8 MINNESOTA DEPARTMENT OF COMMERCE (CLASS 7).

- (a) Class 7 Definition. A Class 7 Claim means the claim of the Minnesota Department of Commerce against cash collateral posted at Merchants Bank that secures the Debtor's obligation for potential workers' compensation liability.
- **(b)** Class 7 Treatment. The cash collateral securing the Class 7 Claim shall vest in the Reorganized Debtor and shall continue to secure the Class 7 Claim. The security interest of the holder of the Class 7 Claim shall remain in place and the holder of such claim may exercise any and all rights and remedies against the collateral posted at Merchants Bank that secures the Debtor's obligation for potential workers' compensation liability.

#### 4.9 COMMUNITY BANK OF MANKATO (CLASS 8).

- (a) Class 8 Definition. A Class 8 Claim means the claim of the Community Bank of Mankato under that certain mortgage and security agreement encumbering the real property known as 1502 Warren Street, Mankato, Minnesota.
- (b) Class 8 Treatment. The collateral securing the Class 8 Claim shall vest in the Reorganized Debtor, provided, however, that the Reorganized Debtor may transfer legal title to such collateral (subject to the mortgage and security agreement) to the equitable owner as set forth in Section 15.2 below, but in any case the collateral shall continue to secure the Class 8 Claim. The mortgage and security interest of the holder of the Class 8 Claim shall remain in place and the holder of such claim may exercise any and all rights and remedies against the collateral referenced in such mortgage and security agreement, available to the holder.

# ARTICLE V MEANS OF IMPLEMENTATION OF THE PLAN

#### 5.1 TRUST FORMATION AND FUNDING.

(a) **Purpose, Formation, and Assets**. The Trust shall be established for the purpose of receiving, liquidating, and distributing Trust Assets in accordance with this Plan and the Trust Distribution Plan. The proposed Trust Agreement is attached hereto as <u>Exhibit D</u>.

### (b) Funding.

- 1. **Summary**. This Plan will be funded from the sources and in the manner set forth in this Section. In addition to the contributions described herein, the Catholic Entities and Other Insured Entities will waive certain Claims against the Diocese and Settling Insurers, as set forth below or in the LMI/Interstate Settlement Agreement.
- 2. **Contributions**. Cash and other assets will be paid or transferred, as applicable, to the Trust Account as provided in the Plan and as described herein.
  - **Debtor Cash Contribution**. The Debtor will transfer (a) (i) \$13,560,000, less (A) Professional Claims (including fees and expenses of Insurance Archaeology Group), mediation fees and expenses (including the fees and expenses of John Vukelich, Paul Van Osselaer and Janice Symchych), certain other administrative expenses as agreed between the Debtor and the UCC, the fees of the Unknown Claims Representative, and U.S. Trustee quarterly fees, in each case to the extent paid after February 29, 2020, and (B) counseling expenses for Tort Claimants (whether paid before or after the Petition Date) (collectively, the "Permitted Deductions"), to the Trust Account within five days after the Effective Date (which payment shall constitute substantial consummation of the Plan), plus (b) \$7,552,500 as soon as practical after the sale or other monetization of certain assets of the Diocese, as well as a contribution from certain non-Diocesan entity resources, but in no event more than 12 months after the date the Confirmation Order becomes a Non-Appealable Order (the "Debtor Cash Contribution"). In the event there are Permitted Deductions paid after the payment of the amount required under subparagraph 2(i)(a) above, then the Debtor may deduct any remaining Permitted Deductions from the \$7,552,500 payment. The Debtor Cash Contribution will be primarily comprised of funds from the following sources:
    - 1. non-restricted cash accounts held by the Diocese; and
    - 2. sale or other monetization of certain assets of the Diocese, as well as a contribution from certain non-Diocesan entity resources (specifically, \$112,500 of the \$7,552,500 payment will be

For clarity, the Debtor Cash Contribution (including the portion contributed by the Contributing Catholic Entities) is being made in respect of uninsured exposure of the Diocese and the Contributing Catholic Entities for Tort Claims, and not in respect of any Tort Claims that may be covered under Non-Settling Insurer Policies. The Debtor Cash Contribution (including the portion contributed by the Contributing Catholic Entities) shall not be construed as a release of any Tort Claims that may be covered under Non-Settling Insurer Policies. The Debtor and the Contributing Catholic Entities will not be released for any Abuse that may be covered under Non-Settling Insurer Policies until such claims are settled with the Diocese, the Reorganized Debtor, the Contributing Catholic Entities or any other Protected Party, as applicable, and such Non-Settling Insurer or are fully adjudicated, resolved and subject to Final Order, but recourse will be limited as set forth in Sections 4.3 and 4.5.

- (ii) **Settling Insurer Contributions**. LMI and Interstate shall pay to the Trust the sums set forth in the LMI/Interstate Settlement Agreement within the time set forth in the LMI/Interstate Settlement Agreement. In addition, all rights to receive payment of the amounts to be paid under the LMI/Interstate Settlement Agreement shall be assigned to the Trust. The total amount that will be paid to the Trust by LMI and Interstate is \$6,500,000 (the "Additional Settlement Amounts").
- 3. Additional Trust Assets: All Rights and Recoveries Against Non-Settling Insurers. In addition to the funds transferred to the Trust, the Transferred Insurance Interests of the Diocese are automatically and without further act or deed assigned and transferred to the Trust on the Effective Date. In addition, the Interests of the other Protected Parties in the Transferred Insurance Interests are automatically and without further act or deed assigned and transferred to the Trust on the Effective Date. The foregoing assignment and transfer shall not be construed as an assignment and transfer of the Non-Settling Insurer Policies.
- (c) Vesting. On the Effective Date, all Trust Assets shall vest in the Trust, and the Diocese and other Protected Parties shall be deemed for all purposes to have transferred all of their respective Interests in the Trust Assets to the Trust. On the Effective Date, or as soon as practicable thereafter, the Reorganized Debtor, any other Protected Party, and Settling Insurers, as applicable, shall take all actions reasonably necessary to transfer any Trust Assets to the Trust. Upon the transfer of control of Trust Assets in accordance with this paragraph, the Diocese, the other Protected Parties and the Settling Insurers shall have no further interest in or with respect to the Trust Assets except as otherwise explicitly provided in this Plan.

- **5.2 PAYMENT OF PROFESSIONAL FEES**. The Reorganized Debtor shall pay all unpaid Allowed Professional Claims accruing through the Effective Date, (i) within seven (7) days after the later of the Effective Date or the Bankruptcy Court's order on such Claims, or (ii) upon such terms as may exist pursuant to Order of the Bankruptcy Court or an agreement between such holder of an Allowed Professional Claim and the Debtor.
- 5.3 PAYMENTS EFFECTIVE UPON TENDER. Whenever the Plan requires payment to be made to a creditor, such payment will be deemed made and effective upon tender thereof by the Trustee, the Debtor, or the Reorganized Debtor to the creditor to whom payment is due. If any creditor refuses a tender, the amount tendered and refused will be held by the Trust, the Debtor, or the Reorganized Debtor for the benefit of that creditor pending final adjudication of the dispute. However, when and if the dispute is finally adjudicated and the creditor receives the funds previously tendered and refused, the creditor will be obliged to apply the funds in accordance with the Plan as of the date of the tender; and while the dispute is pending and after adjudication thereof, the creditor will not have the right to claim interest or other charges or to exercise any other rights which would be enforceable by the creditor, if the Trust, the Debtor, or the Reorganized Debtor failed to pay the tendered payment.

### ARTICLE VI TRUST

- **6.1 ESTABLISHMENT OF TRUST**. On or before the Confirmation Date, the Trust shall be established in accordance with the Trust Documents. The Trust is intended to qualify as a "Designated" or "Qualified Settlement Fund" pursuant to Section 468B of the Internal Revenue Code and the Treasury Regulations promulgated thereunder. The Debtor is the "transferor" within the meaning of Treasury Regulation Section 1.468B-1(d)(1). The Trustee shall be classified as the "administrator" within the meaning of Treasury Regulation Section 1.468B-2(k)(3). The Trust Documents, including the Trust Agreement, are incorporated herein by reference.
- **6.2** PURPOSE, FORMATION AND ASSETS. The Trust shall be established for the purposes described in this paragraph. The Trust shall receive the transfer and assignment of assets as provided in Articles IV and V, including the Debtor Cash Contribution, the Additional Settlement Amounts, and the Transferred Insurance Interests, of which the Trust is, and shall be deemed to be, the sole assignee. The Trust shall make distributions to the Class 3 and Class 4B claimants, as provided by this Plan, the Trust Agreement, and the Trust Distribution Plan. The Trust shall pursue recoveries against any Non-Settling Insurers in respect of the Transferred Insurance Interests. The Trust shall fund the defense of the Diocese, the Reorganized Debtor, and any other Protected Party as against any Litigation Claims brought by Class 3 and Class 4B claimants, but only to the extent that the Diocese, the Reorganized Debtor, or any other Protected Party are not defended or otherwise reimbursed for their defense expenses by any Non-Settling Insurer. The Trust shall advance funding to the Diocese, the Reorganized Debtor, and any other Protected Party, as applicable, with respect to any judgments or settlements of any Litigation Claims brought by Class 3 and Class 4B claimants, but only to the extent that such judgments or settlements are not funded by any Insurer. The Trust shall fund the costs and expenses in executing these functions, all such functions to be executed in accordance with this Plan, the Trust Agreement, and the Trust Distribution Plan, with the aim of preserving, managing, and maximizing Trust Assets to pay Class 3 and Class 4B claimants and with no objective to continue or engage in

the conduct of a trade or business. The proposed Trust Agreement is attached to this Plan as <u>Exhibit</u> <u>D</u>.

### 6.3 ALLOCATIONS WITHIN AND DISTRIBUTIONS AND PAYMENTS FROM THE TRUST.

- (a) General Corpus. The following distributions and payments will be made from the general corpus of the trust:
  - 1. **Distributions**. Distributions on Class 3 Claims and Class 4B Claims as determined by the Tort Claims Reviewer in accordance with this Plan, the Trust Agreement, and the Trust Distribution Plan.
  - 2. **Tort Claims Reviewer**. The Trustee shall retain the Tort Claims Reviewer. Fees payable to the Tort Claims Reviewer for review of Class 3 and Class 4B Claims shall be paid from the Trust.
  - 3. **Trust Administrative Fees**. All fees, costs, and expenses of administering the Trust as provided in the Plan and the Trust Agreement shall be paid by the Trust, including: (i) as reasonably necessary to meet current liabilities and to maintain the value of the respective Assets of the Trust; (ii) to pay reasonable administrative expenses (including any taxes imposed on the Trust and any professional fees); and (iii) to satisfy other liabilities incurred by the Trust in accordance with the Plan or the Trust Agreement.
  - 4. **Indemnity**. The Trust's obligations, if any, to defend, indemnify, or hold harmless any Person expressly set out in the Plan shall be made from the corpus of the Trust.
- (b) IMPAIRED UNKNOWN TORT CLAIM RESERVE FUND. The Trust shall establish an Impaired Unknown Tort Claim Reserve Fund for the benefit of Class 4B Claimants, as detailed in the Trust Agreement, in an amount to be determined after consultation with the Unknown Claims Representative. The Trust shall maintain the Impaired Unknown Tort Claim Reserve Fund until the later of (i) the date that the Impaired Unknown Tort Claim Reserve Fund has been exhausted or (ii) the occurrence of the sixth (6th) anniversary of the Effective Date. Neither the Diocese, the Reorganized Debtor nor any other Protected Party shall have any obligation to make any contribution to the Trust to establish an Impaired Unknown Tort Claim Reserve Fund in excess of those contributions specified in Article V, Section 5.1(b) as the Debtor Cash Contribution and Transferred Insurance Interests.
- **6.4 TAX MATTERS**. The Trust shall not be deemed to be the same legal entity as the Diocese, but only the assignee of certain assets of the Diocese and a representative of the Estate for delineated purposes within the meaning of Section 1123(b)(3) of the Bankruptcy Code. The Trust is expected to be tax exempt. The Trustee shall file such income tax and other returns and documents as are required to comply with the applicable provisions of the Internal Revenue Code of 1986, 26 U.S.C. §§ 1 et seq., as may be amended, and the regulations promulgated thereunder,

- 31 C.F.R. §§ 900 et seq., and Minnesota law and the regulations promulgated thereunder, and shall pay from the Trust all taxes, assessments, and levies upon the Trust, if any.
- 6.5 APPOINTMENT OF THE TRUSTEE. The initial Trustee has been identified in Exhibit D to this Plan. The Trustee shall commence serving as the Trustee on the Confirmation Date; provided, however, that the Trustee shall be permitted to act in accordance with the terms of the Trust Agreement from such earlier date, as authorized by the Diocese and the UCC, and shall be entitled to seek compensation in accordance with the terms of the Trust Agreement and the Plan.

#### 6.6 RIGHTS AND RESPONSIBILITIES OF TRUSTEE.

- The Trustee shall be deemed the Estate's representative in accordance with (a) Section 1123 of the Bankruptcy Code and shall have all the rights, powers, authority, responsibilities, and benefits specified in the Plan and the Trust Agreement, including (to the extent necessary to enforce those rights, powers, authority, responsibilities, and benefits only) the powers of a trustee under Sections 704, 108 and 1106 of the Bankruptcy Code and Bankruptcy Rule 2004 (including commencing, prosecuting or settling Causes of Action, enforcing contracts, and asserting Claims, defenses, offsets and privileges). If there is any inconsistency or ambiguity between the Plan and Confirmation Order, on the one hand, and the Trust Agreement, on the other hand, with respect to the Trustee's authority to act, the provisions of the Plan and Confirmation Order shall control. Among other things, the Trustee: (1) shall liquidate and convert to cash the Trust Assets, make timely distributions and not unduly prolong the duration of the Trust; (2) may request an expedited determination of taxes of the Trust under Section 505(b) of the Bankruptcy Code for all returns filed for, or on behalf of, the Trust for all taxable periods through the dissolution of the Trust; and (3) may retain professionals, including legal counsel, accountants, financial advisors, auditors, and other agents on behalf of the Trust, and at the Trust's sole expense, as reasonably necessary and to carry out the obligations of the Trustee hereunder and under the Trust Agreement.
- **(b)** Notwithstanding the foregoing, the Diocese, the Reorganized Debtor, and the Trust acting for itself and on behalf the Estate, shall be deemed to have waived, effective upon the Effective Date:
  - 1. Any and all Claims under Sections 547, 548, 549 and 550 of the Bankruptcy Code for the recovery of any sums paid to any Person who provided goods and services to the Diocese in the ordinary course of business prior to the Effective Date; and
  - 2. Any and all Claims and Causes of Action: (i) seeking the substantive consolidation of the Diocese and any other Person or an order deeming any such Person and the Diocese to be an "alter-ego" of the other or any other similar Claim or Cause of Action; (ii) to avoid, set aside or recover any payment or other transfer made to any Person under Sections 547, 548, 549, and 550 of the Bankruptcy Code; and (iii) any proceeding to avoid or set aside any interest of a Person in property under Section 544 of the Bankruptcy Code.

The Confirmation Order shall state that, absent permission of the Bankruptcy Court, no judicial, administrative, arbitral, or other action or proceeding shall be commenced in any forum other than the Bankruptcy Court against the Trustee in its/his/her official capacity, with respect to its/his/her status, duties, powers, acts, or omissions as Trustee.

#### **6.7 TRANSFERRED INSURANCE INTERESTS.**

### (a) Enforcement of Transferred Insurance Interests Against Non-Settling Insurers.

1. As set forth in Article V, the Transferred Insurance Interests are assigned and transferred to the Trust. The Trust shall be entitled to all policy proceeds due by virtue of a judgment or settlement of a Class 3 Claim or a Class 4B Claim and will be entitled to assert and/or assign to any Tort Claimant all claims and causes of action that currently exist or may arise in the future against Non-Settling Insurers based on their conduct concerning insurance coverage for, or defense or settlement of, any Class 3 Claim or Class 4B Claim, including but not limited to all claims and causes of action for breach of the Non-Settling Insurer Policies, vexatious refusal, bad faith, wrongful failure to settle, and for any other similar claim or cause of action, including any and all such claims or causes of action providing for penalties, extra-contractual damages, punitive damages and attorneys' fees and costs. The Trust shall also have the right to pursue judgment against Non-Settling Insurers to determine the amount of coverage available for Protected Parties' liability for Tort Claims. The foregoing transfer shall not be construed to entitle any Person to insurance coverage other than those Persons entitled to such coverage from Non-Settling Insurers. The Trust shall be fully authorized to act in its own name, or in the name of any Protected Party, to enforce any right, title, or interest of any Protected Party in the Transferred Insurance Interests. No limitations on recovery from Non-Settling Insurers shall be imposed by virtue of the fact that the Diocese is in bankruptcy or by any distribution from the Trust to any Tort Claimant. The transfer of the Transferred Insurance Interests shall not affect any Non-Settling Insurer's duty to defend, but to the extent that a failure to defend or a separate agreement between the Diocese, the Reorganized Debtor, or any other Protected Party and any Non-Settling Insurer gives rise to a monetary obligation or policy proceeds to reimburse defense costs in lieu of a duty to defend, the Trust shall be entitled to the benefit of such monetary obligation or policy proceeds. Any recovery by the Trust on an action against a Non-Settling Insurer for a determination of coverage for Protected Parties' liability for Tort Claims shall become a Trust Asset and shall be distributed as provided in this Plan, the Trust Agreement, and the Trust Distribution Plan. The Trust's recourse to the Diocese and the other Protected Parties shall be limited to the Transferred Insurance Interests and any other rights or interests expressly granted to the Trust under this Plan, including any indemnification obligations of the Reorganized Debtor for Covered Non-Tort Claims under section 16.4, or as otherwise provided by the Plan. The Trust shall have no liability for Covered Non-Tort Claims and holders of Covered

Non-Tort Claims shall have no recourse to the Trust with respect to such Claims.

2. The Trust shall have full access to coverage issued by the Non-Settling Insurers to the greatest extent permitted by applicable non-bankruptcy law, in the same manner and to the same extent as the Protected Parties prior to the confirmation of the Plan and the transfer of the Transferred Insurance Interests to the Trust, subject to the assertion of any coverage defenses except any defense (a) regarding the assignment and transfer of the Transferred Insurance Interests; (b) effected by operation of law because of confirmation of this Plan; or (c) based on res judicata or collateral estoppel related to facts determined by the Bankruptcy Court.

The assignment and transfer of the Transferred Insurance Interests to the Trust does not affect the Diocese's, the Reorganized Debtor's, other Protected Parties', or any Non-Settling Insurer's right to contest the Diocese's, or any other insured's, liability or the amount of damages in respect of any Tort Claims. Notwithstanding the assignment and transfer of the Transferred Insurance Interests, the Diocese, the Reorganized Debtor, and any other Protected Party shall not be relieved of any obligations or duties under any Non-Settling Insurer Policy (including without limitation any duty to cooperate) and shall continue to honor such duties and obligations as required by such applicable Non-Settling Insurer Policies and applicable law. The transfer and assignment of the Transferred Insurance Interests does not affect any insurers' rights, obligations, or duties under applicable Non-Settling Insurer Policies or applicable law. If the Trust brings an action against a Non-Settling Insurer to assert any claim or to determine the Non-Settling Insurer's obligation to provide coverage for any Tort Claim, the Non-Settling Insurer may raise any defense to coverage as if the action had been brought by the Diocese, the Reorganized Debtor, or any other Protected Party.

3. The Bankruptcy Court shall determine at the Confirmation Hearing (i) whether the assignment of the Transferred Insurance Interests provided for in this Section is valid, and (ii) whether such transfer or the discharge and injunctions set forth in Sections 13.2 and 13.3, or any other term of the Plan, void, defeat, or impair the insurance coverage issued by the Non-Settling Insurers. If a party in interest (which, for this purpose, shall include the Non-Settling Insurers) fails to timely file an objection to the proposed assignment and transfer of the Transferred Insurance Interests to the Trust or other terms of the Plan related to the Non-Settling Insurer Policies by the date set to file such objections, that party in interest shall be deemed to have irrevocably consented to the assignment of Transferred Insurance Interests and other Plan terms related to such Non-Settling Insurer Policies and will be forever barred from asserting that the assignment of the Transferred Insurance Interests or other Plan terms affect the ability of the Trust or Tort Claimants to pursue the Non-Settling Insurers, or any of them, for insurance coverage.

Policies.

- 5. The Reorganized Debtor will cooperate and assist the Trust in enforcing any right or prosecuting any claim based on the Transferred Insurance Interests. This cooperation includes, but is not limited to, providing access to documents and electronic information and providing clergy, employees, agents, and volunteers to testify in depositions and at trial.
- (b) Appointment of Trustee as Estate Representative to Enforce Insurance Interests and Obtain Insurance Recoveries.
  - 1. If the Bankruptcy Court does not enter a Final Order approving the assignment and transfer of the Transferred Insurance Interests to the Trust, then the assignment shall not occur and pursuant to the provisions of Section 1123(b)(3)(B) of the Bankruptcy Code, the Trustee is hereby appointed as the representative of the Diocese's Estate for the purpose of retaining and enforcing all of the Diocese's and the Estate's Interests against the Non-Settling Insurers with respect to the Tort Claims. Any recoveries on such Interests by the Trustee will be paid to the Trust. The determination of whether the appointment of the Trustee as the Diocese's and the Estate's representative provided for in this Section 6.7(b)(1) is valid, and does not defeat or impair the insurance coverage Non-Settling Insurers are responsible for under Non-Settling Insurer Policies, shall be made by the Bankruptcy Court at the confirmation hearing. If a party in interest (which, for this purpose, shall include the Non-Settling Insurers) fails to timely file an objection to the proposed appointment by the deadline for filing objections to confirmation of this Plan, that party in interest shall be deemed to have irrevocably consented to the appointment and will be forever barred from asserting that the appointment in any way affects the ability of the Trustee to pursue Non-Settling Insurers, or any of them, for insurance coverage. In the event that the Bankruptcy Court determines that the appointment is valid and does not defeat or impair coverage Non-Settling Insurers are responsible for under Non-Settling Insurer Policies, then, following the Effective Date, the Trustee shall assume responsibility for, and be bound by, only such obligations of the Diocese and other Protected Parties under Non-Settling Insurer Policies as are necessary to act as the representative of the Diocese and the Estate for the purpose of retaining and enforcing their Interests, if any, against the Non-Settling Insurers; provided, however, that the Trustee's appointment shall not relieve the Diocese, the Reorganized Debtor or the other Protected Parties from any obligation that such entities may have under the Non-Settling Insurer

Policies. Nothing contained in this Section 6.7(b)(1) shall affect the rights and remedies of a Person who is not a Protected Party but is an insured or additional insured with the Diocese or is asserting rights under a Non-Settling Insurer Policy.

- 2. In the event that a Final Order is entered holding that: (a) the assignment of the Transferred Insurance Interests, or (b) the appointment of the Trustee as the Diocese's and the Estate's representative are invalid or would defeat or impair the insurance coverage issued by the Non-Settling Insurers, then the assignment and/or appointment, as the case may be, will be deemed not to have been made, and the Diocese, the Reorganized Debtor, and each of the Protected Parties will retain their Interests under each Settling Insurer and Non-Settling Insurer Policy.
  - (i) At the request of the Trust, the Reorganized Debtor and the other Protected Parties will assert their Interests against a Non-Settling Insurer, including, but not limited to, by filing a lawsuit for recovery of policy proceeds. All recoveries by the Reorganized Debtor and the other Protected Parties will be paid to the Trust. The Reorganized Debtor and the other Protected Parties will select and retain counsel to pursue their Interests against Non-Settling Insurers pursuant to this Section 6.7(b), subject to the Trustee's approval, which approval shall not be unreasonably withheld.
  - (ii) The Trust shall pay the reasonable attorneys' fees, costs and expenses that are incurred by the Reorganized Debtor and the other Protected Parties in pursuing, pursuant to this Section 6.7(b), its Interests against Non-Settling Insurers.
  - (iii) The Trust shall, in addition to reasonable attorneys' fees, costs and expenses provided for in this Section 6.7(b), reimburse the Reorganized Debtor and each of the Protected Parties for any reasonable out of pocket costs and expenses it incurs as a direct consequence of pursuing its Interests against Non-Settling Insurers, but will not compensate the Reorganized Debtor or any other Protected Party for any time any of its employees expend. Upon receipt by the Reorganized Debtor or other Protected Party, all recoveries received by the Reorganized Debtor or other Protected Party from Non-Settling Insurers shall be deemed to be held in trust for the benefit of the Trust and shall be remitted by the Reorganized Debtor or other Protected Party to the Trust as soon as practicable following the Reorganized Debtor's or other Protected Party's receipt of such recoveries.

#### **6.8** SPECIAL DISTRIBUTION CONDITIONS.

(a) With respect to Class 3, the Trust shall maintain sufficient funds to pay any potential reimbursements to Medicare and shall complete the following "Medicare Procedures": (i) the Trustee shall determine whether each Tort Claimant with a Date of

Injury after December 5, 1980 is eligible to receive, is receiving, or has received Medicare benefits ("Medicare Eligible"); (ii) upon request, the Trust shall provide to a Settling Insurer or the Diocese information sufficient to allow them to perform their own SSA queries to the extent they wish to do so; (iii) in the event that one or more Tort Claimants with Dates of Injury after December 5, 1980 is/are identified as Medicare Eligible, the Trust shall complete a query to the CMS for each such Tort Claimant to determine whether any payment ("Conditional Payment") made pursuant to Section 1395y(b)(2)(B) of the MSPA has been made to or on behalf of that Tort Claimant arising from or relating to treatment for Abuse; (iv) if any Conditional Payment has been made to or on behalf of that Tort Claimant, the Trustee shall, within the time period called for by the MSPA, reimburse the appropriate Medicare Trust Fund for the appropriate amount, and submit the required information for that Tort Claimant to the appropriate agency of the United States government.

- **(b)** Compliance with the provisions of this Section 6.8 shall be a material obligation of the Trust in favor of the Settling Insurers under any settlement agreements between any of those Settling Insurers and the Diocese, which authorizes funding to the Trust.
- (c) The Trust shall defend, indemnify and hold harmless the Protected Parties and the Settling Insurers from any Medicare Claims, and any Claims related to the Trust's obligations under this Section.
- **6.9 INVESTMENT POWERS; PERMITTED CASH EXPENDITURES**. All funds held by the Trust shall be invested in cash or short-term highly liquid investments that are readily convertible to known amounts of cash as more particularly described in the Trust Agreement. The Trustee may expend the cash of the Trust.
- **6.10 REGISTRY OF BENEFICIAL INTERESTS**. To evidence the beneficial interest in the Trust of each holder of such an interest, the Trustee shall maintain a registry of beneficiaries.
- **6.11 NON-TRANSFERABILITY OF INTERESTS**. Any transfer of an interest in the Trust shall not be effective until and unless the Trustee receives written notice of such transfer.
- **6.12 TERMINATION**. The Trust shall terminate after its liquidation, administration, and distribution of the Trust Assets in accordance with the Plan and its full performance of all other duties and functions set forth herein or in the Trust Agreement. The Trust shall terminate no later than the sixth (6th) anniversary of the Effective Date.

### 6.13 IMMUNITY; LIABILITY; INDEMNIFICATION.

(a) Neither the Reorganized Debtor or its respective members, designees, or professionals, nor the Trustee or any duly designated agent or representative of the Trustee, nor their respective employees, shall be liable for the acts or omissions of any other member, designee, agent, or representative of such Trustee, except that the Trustee shall be liable for his/her/its specific acts or omissions resulting from such Trustee's misconduct, gross negligence, fraud, or breach of the fiduciary duty of loyalty. The Trustee may, in connection with the performance of his/her/its functions and in his/her/its sole and absolute

discretion, consult with his/her/its attorneys, accountants, financial advisors, and agents, and shall not be liable for any act taken, omitted to be taken, or suffered to be done in accordance with advice or opinions rendered by such Persons. Notwithstanding such authority, the Trustee shall not be under any obligation to consult with his/her/its attorneys, accountants, financial advisors, or agents, and his/her/its determination not to do so shall not result in the imposition of liability on the Trustee unless such determination is based on the Trustee's recklessness, gross negligence, willful misconduct, or fraud.

- (b) No recourse shall ever be had, directly or indirectly, against the Trustee personally, or against any employee, contractor, agent, attorney, accountant, or other professional retained in accordance with the terms of the Trust Agreement or the Plan by the Trustee, by legal or equitable proceedings or by virtue of any statute or otherwise, nor upon any promise, contract, instrument, undertaking, obligation, covenant or Trust Agreement whatsoever executed by the Trustee in implementation of the Trust Agreement or this Plan, it being expressly understood and agreed that all such liabilities, covenants, and Trust Agreements of the Trust whether in writing or otherwise, shall be enforceable only against and be satisfied only out of the Trust Assets or such part thereof as shall under the term of any such Trust Agreement be liable therefore or shall be evidence only of a right of payment out of the Trust Assets. Notwithstanding the foregoing, the Trustee may be held liable for his/her/its recklessness, gross negligence, willful misconduct, knowing and material violation of law, or fraud; and if liability on such grounds is established, recourse may be had directly against the Trustee. The Trust shall not be covered by a bond.
- (c) The Trust shall defend, indemnify, and hold harmless the Trustee, his/her/its officers, directors, agents, representatives, and employees to the fullest extent that a corporation or trust organized under the laws of Minnesota is entitled to indemnify and defend its directors, trustees, officers, and employees against any and all liabilities, expenses, Claims, damages or losses incurred by them in the performance of their duties hereunder.
  - 1. Additionally, the Reorganized Debtor, and each of its respective agents, who was or is a party, or is threatened to be made a party to any threatened or pending judicial, administrative, or arbitrative action, by reason of any act or omission of the Trust or Trustee or respective agents, with respect to: (i) the Chapter 11 case and any act or omission undertaken by them prior to the commencement thereof, (ii) the assessment or liquidation of any Class 3 and Class 4B Claims, (iii) the administration of the Trust and the implementation of the Trust Distribution Plan, or (iv) any and all activities in connection with the Trust Agreement, shall be indemnified and defended by the Trust, to the same extent that a corporation or trust organized under the laws of Minnesota is from time to time entitled to indemnify and defend its own officers, directors, trustees, and employees, against reasonable expenses, costs and fees (including attorneys' fees and costs), judgments, awards, amounts paid in settlement and liabilities of all kinds incurred by the Debtor or Reorganized Debtor, and their respective professionals, officers, and directors, in connection with or resulting from such action, suit or proceeding, provided that, with respect to amounts paid in

- settlement, the Trust has approved such amounts in advance, such approval not to be unreasonably withheld.
- 2. Reasonable expenses, costs, and fees (including attorneys' fees and costs) incurred by or on behalf of a Trustee, the Debtor, the Reorganized Debtor, and their respective agents in connection with any action, suit, or proceeding, whether civil, administrative, or arbitrative, from which they are entitled to be indemnified by the Trust, shall be paid by the Trust in advance of the final disposition thereof upon receipt of an undertaking, by or on behalf of such Trustee, the Debtor, the Reorganized Debtor, and their respective agents, to repay such amount in the event that it shall be determined ultimately by Non-Appealable Order that such Trustee, the Debtor, the Reorganized Debtor, and their respective professionals, officers, and directors is not entitled to be indemnified by the Trust.

#### 6.14 TREATMENT OF TORT CLAIMS.

(a) TRUST LIABILITY. On the Effective Date, the Trust shall automatically and without further act or deed assume: (i) all liability, if any, of the Protected Parties and Settling Insurers in respect of Channeled Claims, subject to section 16.4; (ii) the responsibility for preserving, managing and distributing Trust Assets pursuant to the Trust Distribution Plan; and (iii) the right to pursue the Transferred Insurance Interests. Except as otherwise provided herein, the Trust does not assume any liabilities of the Diocese or Reorganized Debtor, in whole or in part, in regards to any Tort Claims that are not released, nor the liabilities of the Settling Insurers.

#### (b) ASSESSMENT OF TORT CLAIMS.

- 1. Each Tort Claim will be assessed by the Tort Claims Reviewer in accordance with the Trust Distribution Plan to determine whether the Tort Claimant is entitled to a distribution under the Trust. The Diocese or the Reorganized Debtor shall reasonably cooperate with the Tort Claims Reviewer and the Trustee as requested by the Tort Claims Reviewer or the Trustee in connection with any inquiries by either in the administration of the Trust Distribution Plan, but shall not be required to act in any way that violates any duty to cooperate with a Non-Settling Insurer. Under no circumstance shall the Tort Claims Reviewer's review of a Class 3 or Class 4B Claim or a determination regarding a distribution thereon have any effect on the rights of a Non-Settling Insurer.
- 2. Each Tort Claimant may elect, in lieu of assessment by the Tort Claims Reviewer, to have his Tort Claim treated pursuant to the convenience Claim process as provided by the Trust Distribution Plan ("Convenience Claim").
- (c) **ELECTION.** No later than thirty (30) days after a Tort Claimant is notified of the amount of the award under the Trust Distribution Plan, the Tort Claimant shall elect in writing one of the following treatment alternatives:

- 1. Receiving a payment from the Trust in the amount determined by the Tort Claims Reviewer pursuant to the Trust Distribution Plan ("Distribution Plan Claim"). A Tort Claimant who elects to receive a distribution for a Distribution Plan Claim (a "Distribution Plan Claimant") must execute the release of all his or her Tort Claims against the Settling Insurers and the Protected Parties as set forth in Exhibit E or Exhibit F, as applicable, and waives his right to pursue a direct action under Minn. Stat. § 60A.08, subd. 6 or other applicable law against any Non-Settling Insurer; or
- 2. Treatment of the Tort Claim as a Litigation Claim. A Tort Claimant electing treatment as a Litigation Claim ("Litigation Clamant") will have rights, to the extent set forth in the Trust Distribution Plan, to initial and future distributions from the Trust. Each Litigation Claimant also retains the right to: (i) pursue his or her Tort Claim for its full amount according to proof in order to determine the liability of any Protected Party for purposes of recovering against any Non-Settling Insurer that is or may be liable on the Tort Claim and (ii) proceed in a direct action against any Non-Settling Insurer to the extent allowed by applicable law, including Minn. Stat. § 60A.08, subd. 6 (each a "Litigation Claim"). A Litigation Claimant's recovery on a Litigation Claim is limited as provided herein. The Settling Insurers shall not be obligated to defend or indemnify any Person in connection with a Litigation Claim and the Settling Insurers shall not have any other duties or obligations to any Person in connection with a Litigation Claim. Under no circumstances will a Tort Claimant or any other Person be able to recover any amount from a Settling Insurer in connection with a Litigation Claim.

### (d) MODIFICATION OF TREATMENT ELECTION.

- 1. If a Tort Claimant does not make one of the elections in Section 6.14(c), the Tort Claimant will be treated as a Litigation Claimant.
- 2. Upon written notice to the Trustee, subject to the Trustee's sole and absolute discretion, a Tort Claimant may rescind the election to be treated as a Litigation Claimant in favor of being treated as a Distribution Plan Claimant. Notwithstanding the foregoing, the Trustee shall consent to a Tort Claimant's rescission if such written notice of rescission is given prior to entry of an order of dismissal or a final judgment on the Litigation Claim in favor of a Protected Party.
- 3. No later than ten (60) days after a Tort Claimant is notified of the amount of the award under the Trust Distribution Plan, a Tort Claimant may rescind the election to be treated as a Distribution Plan Claimant in favor of being treated as a Litigation Claimant.
- (e) Legal Effect of Estimation of Claims and Distributions under Trust Distribution Plan. The Tort Claims Reviewer's determinations are for estimation purposes only and shall not be a finding or fixing of the fact or liability or the amount payable for

any Tort Claim with any binding legal effect, other than for distribution purposes by the Trust pursuant to the Trust Distribution Plan. The determination of qualification, estimation of claims, and payment of distributions is not an admission of liability by any Protected Party or the Trust with respect to any Tort Claims and has no res judicata or collateral estoppel effect on any Protected Party, the Trust, or any Non-Settling Insurer. Any payments by the Trust to Tort Claimants in connection with their Tort Claims is not a release of the Debtor nor an accord or novation of the Debtor's liability on account of the Class 3 and Class 4B Claims. The Trust's act of making a distribution is immaterial to, and shall not be construed as, a determination or admission of the Diocese's, the Reorganized Debtor's, or any other Protected Party's liability for, or damages with respect to, any Class 3 or Class 4B Claim. The determination of qualification, estimation of claims, and payment of distributions is not a settlement, release, accord, or novation of Class 3 or Class 4B Claims and cannot be used by any third party as a defense to any alleged joint liability. The determination of qualification, estimation of claims, and payment of partial distributions does not impair a Litigation Claimant's rights to obtain a judgment, including a judgment based on joint and several liability, against a Protected Party or any Non-Settling Insurer or other Person, for purposes of establishing the Protected Party's liability on the Tort Claim, but any such judgment awarded to a Litigation Claimant will be reduced by the amount of distributions already paid by the Trust to such Litigation Claimant on his or her Tort Claim(s). Neither the Tort Claims Reviewer's review of a Tort Claim and determination of qualification, nor the Trust's estimation of claims or payment of distributions shall (1) constitute a trial, an adjudication on the merits, or evidence of liability or damages in any litigation with the Protected Parties, Non-Settling Insurers, or any other Person, or (2) constitute, or be deemed, a determination of the reasonableness of the amount of any Tort Claim, either individually or in the aggregate with other Tort Claims, in any coverage litigation with any Non-Settling Insurers. The Trust's estimation of claims and payment of distributions does not constitute a triggering event for liability under any Non-Settling Insurer Policy nor does it create an admission of the fact of liability or the extent of damages on behalf of the Protected Parties.

**(f) RELEASE AND DISCHARGE OF TORT CLAIMS.** No Tort Claimant shall receive any payment on any award unless and until such Tort Claimant has executed the Release attached as Exhibit E or Exhibit F to this Plan, as applicable. Each Tort Claimant must execute a release of all claims against the Settling Insurers and must release all claims against the Diocese, the Reorganized Debtor, and the other Protected Parties that do not implicate insurance coverage under Non-Settling Insurer Policies. To preserve coverage under Non-Settling Insurer Policies, Tort Claimants specifically reserve, and do not release, any and all claims that they may have against the Protected Parties that implicate coverage under Non-Settling Insurer Policies, but recourse is limited to the proceeds of Non-Settling Insurer Policies and all other damages (including extracontractual damages), awards, judgments in excess of policy limits, penalties, punitive damages and attorney's fees and costs that may be recoverable against any Non-Settling Insurers because of their conduct concerning insurance coverage for, or defense or settlement of, any Tort Claim, and any such judgments or awards will be handled in accordance with Sections 6.14(i) and (j).

(g) The Tort Claims will not be released or enjoined as against the Protected Parties for any Abuse that may be covered under Non-Settling Insurer Policies until such claims are settled with the Protected Parties and their Non-Settling Insurers, or are fully adjudicated, resolved, and subject to Final Order, unless the Tort Claimant elects to proceed as a Distribution Plan Claimant, but recourse will be limited as described in Sections 4.3 and 4.5 above.

With respect to all other Claims, except as otherwise provided in the Plan, the Debtor's liability on account of such Claims shall be discharged pursuant to the provisions of 1141(d). The Tort Claimants' release, in whole or in part, of their Class 3 or Class 4B Claims will be pursuant to the principles set forth in *Pierringer v. Hoger*, 124 N.W.2d 106 (Wis. 1963) and *Frey v. Snelgrove*, 269 N.W.2d 918 (Minn. 1978). The Class 3 and Class 4B Claimants will expressly reserve their rights against other Persons (other than Protected Parties), including joint tortfeasors, who will remain severally liable on any Class 3 and Class 4B Claims. Any Person (other than a Protected Party) that is, or was alleged to be a joint tortfeasor with any of the Protected Parties in connection with the Abuse that forms the basis of a Tort Claim shall not be liable for any Protected Party's share of causal liability or fault.

- (h) TRUST RIGHTS AGAINST NON-SETTLING INSURERS. The Trust retains the right to pursue Non-Settling Insurers for the Diocese's, the Reorganized Debtor's, and any other Protected Party's liability to Tort Claimants regardless of whether the Tort Claimants elect treatment as Distribution Plan Claimants or Litigation Claimants.
- (i) **DISTRIBUTIONS TO TORT CLAIMANTS**. A Tort Claimant electing to be treated as a Distribution Plan Claimant, and who the Tort Claims Reviewer determines to be entitled to a distribution, will receive a distribution from the Trust in the amount(s) and at the time(s) provided for in the Trust Distribution Plan. Any payment on a Tort Claim constitutes payment for damages on account of personal physical injuries or sickness arising from an occurrence, within the meaning of Section 104(a)(2) of the Internal Revenue Code of 1986, as amended.
- LITIGATION CLAIMS. If necessary in the Trustee's discretion, the **(i)** Trustee may establish a reserve for payment of a claim held by a Litigation Claimant in the amount that would have been awarded to the Litigation Claimant if such Claimant had elected to proceed under the Trust Distribution Plan. The creation and existence of this reserve is not a settlement, release, accord or novation of the Litigation Claims and cannot be used by any third party as a defense to any alleged joint liability with any Protected Party. For avoidance of doubt, the creation and existence of this reserve does not affect, diminish or impair a Litigation Claimant's rights to collect a judgment, including a judgment based on joint and several liability, against any Non-Settling Insurer or Person that is not a Protected Party, except as expressly provided herein. The Trustee may establish one reserve for all of the Litigation Claims but no Litigation Claimant shall have any interest in any portion of the reserve in excess of the amount determined for that Litigation Claimant under the Trust Distribution Plan, and then only in the event that the Litigation Claimant prevails on his Litigation Claim. Neither the Trust's payment of, or reserving monies on account of, the Tort Claims nor the Tort Claims Reviewer's review of a Tort

Claim, shall: (1) constitute a trial, an adjudication on the merits, or evidence of liability or damages in any litigation with the Protected Parties, Non-Settling Insurers, or any other Person, or (2) constitute, or be deemed, a determination of the reasonableness of the amount of any Tort Claim, either individually or in the aggregate with other Tort Claims, in any coverage litigation with any Non-Settling Insurers.

- 1. In the event that a Litigation Claimant obtains a judgment against any Protected Party and no Non-Settling Insurer is implicated by the Litigation Claim, the judgment will be satisfied by the Trust in the amount of such judgment against such Protected Party, up to the amount of any reserve set for that Litigation Claimant's Litigation Claim plus an additional \$1,000.
- 2. In the event that any Non-Settling Insurer is implicated by the Litigation Claim, and either a settlement is achieved with such Non-Settling Insurer(s) as to such Litigation Claim or the Litigation Claimant obtains a judgment against a Protected Party and either the Trust or the Litigation Claimant obtains a recovery from any such Non-Settling Insurer(s) as to such judgment, then such recovery shall be turned over to the Trust for handling pursuant to this Plan. Such recovery shall first go to reimburse the Trust or the Litigation Claimant, as the case may be, for all costs (including attorneys' fees) incurred in connection with pursuing the recovery against the Non-Settling Insurer(s) relating to the Litigation Claim, so long as such amounts are reasonable and were agreed to in advance by the Trust. Any amount remaining shall be distributed in a manner consistent with the Trust Distribution Plan.
- (k) The Trust's payment to a Litigation Claimant that has recovered a judgment or settlement does not affect, diminish or impair the Trust's right to collect the policy proceeds respecting such Class 3 or Class 4B Claim from any Non-Settling Insurer, nor does it affect, diminish or impair the Trust's right to bring any claims against the Non-Settling Insurer that have been assigned to the Trust or that belong to the Trust by operation of law.
- **(I)** If a Non-Settling Insurer has refused to defend a Protected Party with respect to any Litigation Claim, the Trust will advance or reimburse the Protected Party for reasonable and necessary attorneys' fees and other expenses incurred in defending the Litigation Claim. If any Non-Settling Insurer has refused to indemnify a Protected Party with respect to any settlement or judgment of a Litigation Claim, the Trust will advance or reimburse the Protected Party for any judgment or settlement incurred by the Protected Party on such Litigation Claim, provided the Trust has consented in advance to any such settlement, such consent not to be withheld unreasonably. If all insurers that could potentially have responsibility to defend and/or indemnify for a Tort Claim have denied coverage, that Litigation Claimant must sign a covenant not to execute against that Protected Party's assets (other than Non-Settling Insurer Policies or proceeds or other assigned rights or interests), under which the Litigation Claimant will agree to seek recovery only from Non-Settling Insurers for any judgment the Litigation Claimant obtains against any Protected Party, in exchange for a stipulated judgment and assignment of insurance rights, as authorized by law. If any judgment on any Tort Claim is within the

retention of any Non-Settling Insurer Policy, and all insurers have denied indemnity for such judgment, then the Trust will fund the judgment. The Trust's advancement or reimbursement of the Protected Party for such defense costs and/or judgment or settlement payments, and any distributions made by the Trust to the Litigation Claimant and other Class 3 Claimants, will not affect, diminish or impair the Trust's right to bring any claims against any Non-Settling Insurers for refusing to defend and/or indemnify the Protected Party, including but not limited to claims for payment of policy proceeds, bad faith, wrongful failure to settle, and extra-contractual damages authorized by law.

- (m) As of the Effective Date, the Trustee will be deemed to have the right to join or intervene into the Insurance Coverage Adversary Proceeding and to pursue recoveries against any Non-Settling Insurers.
- (n) Nothing in the Plan, Confirmation Order or any Plan Document shall impose any obligation on any Non-Settling Insurer to provide a defense for, settle, or pay any judgment with respect to, any Tort Claim, or grant to any Person any right to sue any Non-Settling Insurer directly, in connection with a Tort Claim. All such obligations with respect to Non-Settling Insurers shall be determined by and in accordance with the terms of the Non-Settling Insurer Policies and with applicable non-bankruptcy law. Nothing provided for in this Plan, the Trust Agreement, or the Trust Distribution Plan constitutes an endorsement of a Class 3 or Class 4B Claimant's right to pursue their remedies under Minn. Stat. § 60A.08.
- (o) If the Litigation Claimant fails to prosecute his Litigation Claim to final judgment or settlement of the claim, or a Final Order is entered finding that no Protected Party has liability to such Tort Claimant on account of his Tort Claim, any reserve maintained by the Trust on account of such Tort Claim shall revert to the non-reserved assets of the Trust and the Litigation Claimant shall have no recourse against the Trustee, the Trust, any Protected Party, or any Settling Insurer.

### (p) OBJECTIONS AND LITIGATION AFTER THE EFFECTIVE DATE.

- 1. Regardless of whether a Class 3 or Class 4B Claimant elects treatment as a Distribution Plan Claimant or a Litigation Claimant, the Trustee may object to that Class 3 or Class 4B Claimant's Claim. The Trustee's right to object to a Class 3 or Class 4B Claimant's Claim after the Effective Date will not affect or impair any right the Diocese, the Reorganized Debtor, other Protected Parties, and/or Non-Settling Insurers may have under the Non-Settling Insurer Policies or applicable law to object to such Class 3 and Class 4B Claims.
- 2. The Protected Parties will comply with all obligations under the Non-Settling Insurer Policies and applicable law. The Trustee, to the extent required by the Non-Settling Insurer Policies implicated by such Tort Claims and applicable law, shall also cooperate with the Non-Settling Insurer in the defense of such judicial proceeding contemplated in Section 6.13(p)(1). In the event of a dispute between a Non-Settling Insurer and the Trustee regarding whether the Trustee must allow such Non-Settling Insurer to control the defense of such Tort Claim,

or the extent of anyone's duty to cooperate, such dispute shall be resolved by the Bankruptcy Court and the Bankruptcy Court shall retain jurisdiction over such disputes. In the event the Non-Settling Insurer fails to seek a determination from the Bankruptcy Court over the existence and extent of the Trustee's obligation to cooperate, or the Non-Settling Insurer's right to control the defense, the Non-Settling Insurer shall be deemed to have waived such claim.

**CLAIM WITHDRAWAL**. A Tort Claimant may withdraw his or her Tort **(q)** Claim at any time on written notice to the Trustee. If withdrawn, (a) the Tort Claim will be withdrawn with prejudice and may not be reasserted, and such Tort Claimant shall still be subject to Section 13.2 of the Plan, the Channeling Injunction, and the Supplemental Settling Insurer Injunction as provided by this Plan; and (b) any reserve maintained by the Trust on account of such Tort Claim shall revert to the Trust as a Trust Asset for distribution in accordance with the Plan and Trust Distribution Plan. Each Protected Party, Non-Settling Insurer, Settling Insurer, and the Trust shall retain any and all defenses that may exist in respect to such Tort Claim.

### **ARTICLE VII SETTLING INSURERS**

7.1 SETTLING INSURER SETTLEMENT AGREEMENT. Upon satisfaction of the conditions precedent to the LMI/Interstate Settlement Agreement becoming effective, including the Confirmation Order and the order approving the LMI/Interstate Settlement Agreement becoming Non-Appealable Orders, the LMI/Interstate Settlement Agreement will be fully binding on the Trust, Protected Parties, the Reorganized Debtor, the UCC, LMI and Interstate, the Tort Claimants, and parties in interest, and any of the foregoing Persons' successors.

#### 7.2 FREE AND CLEAR OF INTERESTS OF SETTLING INSURER POLICIES.

To the extent provided in the LMI/Interstate Settlement Agreement and effective on the later of (i) the Effective Date of the Plan or (ii) the payment by each Settling Insurer of the settlement payment(s) due under such agreement, each and every Settling Insurer Policy issued by LMI or Interstate shall be sold to the issuing Settling Insurer pursuant to Sections 105, 363, and 1123 of the Bankruptcy Code, free and clear of all liens, Claims and Interests, including those of the Diocese, Diocesan Parishes, and Tort Claimants. As set forth in the LMI/Interstate Settlement Agreement and the corresponding Approval Order, LMI and Interstate are good faith purchasers of such insurance policies and certificates of insurance within the meaning of Section 363(m) of the Bankruptcy Code, the consideration exchanged constitutes a fair and reasonable settlement of the Parties' disputes and of their respective rights and obligations relating to each such Settling Insurer Policy and constitutes reasonably equivalent value, the releases in the LMI/Interstate Settlement Agreement and the policy buyback comply with the Bankruptcy Code and applicable non-bankruptcy laws, and each such Settling Insurer Policy shall be terminated and be of no further force and effect with the issuing Settling Insurer having fully and completely performed any and all obligations under each such Settling Insurer Policy, including any performance owed to the Diocese and Diocesan Parishes, and all limits of liability of each such Settling Insurer Policy shall be exhausted.

- 7.3 RESOLUTION OF CLAIMS INVOLVING SETTLING INSURERS. The Confirmation Order shall provide that within 20 days after receipt of the settlement payment required by the LMI/Interstate Settlement Agreement, the Diocese, LMI and Interstate shall effect dismissal with prejudice of their Claims against each other in the Insurance Coverage Adversary Proceeding, and shall be prohibited from pursuing the claims therein against each other at any time, with each side to bear its own fees and costs. The Diocese shall not be required to dismiss the Insurance Coverage Adversary Proceeding as against any Non-Settling Insurers.
- **7.4 THE SETTLING INSURERS' PAYMENTS**. LMI/Interstate will pay to the Trust the sums set forth in the LMI/Interstate Settlement Agreement within the time set forth in the LMI/Interstate Settlement Agreement.

#### 7.5 JUDGMENT REDUCTION.

- (a) In any proceeding, suit, or action, including the Insurance Coverage Adversary Proceeding, to recover or obtain insurance coverage or proceeds from a Non-Settling Insurer for a Tort Claim, the following shall apply:
- (1) If a Non-Settling Insurer has asserted, asserts, or could assert, any Contribution Claim against a Settling Insurer, then any judgment or award obtained by any Protected Party, the Trust, or a Tort Claimant against such Non-Settling Insurer shall be automatically reduced by the amount, if any, that a court or arbitrator determine such Settling Insurer would have been liable to pay such Non-Settling Insurer as a result of its Contribution Claim, so that the Contribution Claim is thereby satisfied and extinguished entirely ("Reduction Amount"). In any action by a Protected Party, the Trust or a Tort Claimant against a Non-Settling Insurer to obtain insurance coverage or proceeds for a Tort Claim, where the Settling Insurers are not parties, the Non-Settling Insurers' Contribution Claim may be asserted as a defense, and to the extent the Non-Settling Insurers' Contribution Claim against a Settling Insurer is determined to be valid by the court presiding over such action, the liability of the Non-Settling Insurer will be reduced dollar for dollar by the amount so determined. In the event that a reduction is not made as described above, then any Contribution Claim by any Non-Settling Insurer against any Settling Insurer will be determined by the court or arbitration proceeding in which such Contribution Claim is filed. To the extent possible, no Settling Insurer will be required to answer or otherwise respond to a complaint alleging a Contribution Claim against such Settling Insurer until such Reduction Amount is determined by such court or arbitrator(s). If, notwithstanding the foregoing, a court refuses to reduce the liability of the Non-Settling Insurer, then once the order establishing the Settling Insurer's liability for the Contribution Claim is a Final Order, the Trust shall promptly indemnify and hold harmless the Settling Insurer for such amount of such Contribution Claim.
- **(b)** As provided in the LMI/Interstate Settlement Agreement, each Settling Insurer agrees that it will not pursue any contribution claim that it might have against any other insurer.
- **7.6 FURTHER ASSURANCES; NON-MATERIAL MODIFICATIONS**. From and after the Effective Date, the Reorganized Debtor and the Settling Insurers shall be authorized to enter into, execute, adopt, deliver, or implement all notes, contracts, security agreements, instruments, releases, and other agreements or documents necessary to effectuate or memorialize the settlements contained in this Article without further order of the Bankruptcy Court. The

Reorganized Debtor and the Settling Insurers may make technical or immaterial alterations, amendments, modifications, waivers, or supplements to the terms of any Insurance Settlement Agreement. A class of Claims that has accepted the Plan shall be deemed to have accepted the Plan, as altered, amended, modified, or supplemented under this Section, if the proposed alteration, amendment, modification, or supplement does not materially and adversely change the treatment of the Claims within such class. An order of the Bankruptcy Court approving any amendment or modification made pursuant to this Section shall constitute an order in aid of consummation of the Plan and shall not require the re-solicitation of votes on the Plan.

#### 7.7 INDEMNIFICATION OBLIGATIONS.

(a) With respect to the indemnification obligations of the Trust or the Reorganized Debtor, including with respect to Tort Claims made by Persons over whom the Diocese or Diocesan Parishes do not have control, including any other Person who asserts Tort Claims against or right to coverage under the Settling Insurer Policies, the Settling Insurers may undertake the defense of any Claim on receipt of such Claim without affecting such indemnification obligations. The Settling Insurers shall notify the Trust or Reorganized Debtor, as applicable, as soon as practicable of any such Claims identified in this section and of their choice of preferred counsel. The obligation of the Trust or Reorganized Debtor, as applicable, to indemnify the Settling Insurer under this Section 7.7 shall not exceed the settlement amount set forth in the Insurance Settlement Agreement as actually paid by the corresponding Settling Insurer. In defense of any such Claims, the Settling Insurers may settle or otherwise resolve a Claim consistent with the terms of this Plan and with the prior consent of the indemnifying party, which consent shall not be unreasonably withheld. Any such indemnification obligations with respect to Tort Claims will be channeled to and paid by the Trust.

#### 7.8 WAIVER/CONSENT.

In consideration of the releases and Channeling Injunction, the Supplemental Settling Insurer Injunction, and other covenants set forth herein, subject to the occurrence of the Effective Date and the satisfaction of the other conditions precedent to the effectiveness of the LMI/Interstate Settlement Agreement, and upon receipt by the Trust of the settlement payment required by the LMI/Interstate Settlement Agreement, each of the Protected Parties: (1) irrevocably and unconditionally, without limitation, releases, acquits, forever discharges, and waives any Claims and/or Interests it has or might have now or in the future against the other Protected Parties with respect to any contribution, subrogation, indemnification, or other similar Claim arising from or relating to released Tort Claims covered or alleged to be covered under the Settling Insurer Policies, and any Settling Insurer Policies; and (2) consents to the sale of the Protected Parties' Claims and/or Interests, if any, in the Settling Insurer Policies in accordance with the Insurance Settlement Agreements and to the contribution of the proceeds from such sale and settlement to the Trust, as provided in the Plan. (b) Nothing in this Section 7.8 shall be construed to bar either (i) a Claim based on Abuse against a Person who is not a Protected Party or a Settling Insurer or (ii) a Claim by such Person for insurance coverage in connection with a Claim described in the foregoing subsection (i) under an insurance policy other than the Settling Insurer Policies.

#### 7.9 DEBTOR WAIVER AND RELEASE OF CLAIMS

In consideration of any payments to be made by the Settling Insurers and other consideration provided by each Settling Insurer, upon payment by the Settling Insurers of their respective settlement amounts under the corresponding Insurance Settlement Agreements, the Diocese irrevocably and unconditionally, without limitation, releases, acquits, forever discharges, and waives any Interests it has or might have now or in the future (i) under the Settling Insurer Policies issued by LMI and Interstate to the extent those Settling Insurer Policies are bought back under the LMI/Interstate Settlement Agreement and this Plan; (ii) against the Settling Insurers with respect to any Tort Claim; and (iii) against the other Protected Parties with respect to any Channeled Claim. For the avoidance of doubt, the Diocese irrevocably and unconditionally, without limitation, releases, acquits, forever discharges, and waives any Interests it has or might have now or in the future under the Settling Insurer Policies issued by, subscribed to, or underwritten by LMI and/or Interstate, which are bought back under the terms of LMI/Interstate Settlement Agreement and this Plan.

#### 7.10 SUPPLEMENTAL SETTLING INSURER INJUNCTION

(a) <u>Supplemental Injunction Preventing Prosecution of Claims Against Settling Insurers</u>. Pursuant to Sections 105(a) and 363 of the Bankruptcy Code and in consideration of the undertakings of the Settling Insurers pursuant to any Insurance Settlement Agreement, including LMI's and Interstate's purchases of insurance policies or Interests in insurance policies free and clear of all interests pursuant to Section 363(f) of the Bankruptcy Code:

Any and all Persons who have held, now hold or who may in the future hold any Interests (including all debt holders, all equity holders, governmental, tax and regulatory authorities, lenders, trade and other creditors, Tort Claimants, perpetrators, and all others holding Interests of any kind or nature whatsoever, including those Claims released or to be released pursuant to any Insurance Settlement Agreement) against any of the Settling Insurers, or any other Person covered or allegedly covered under the Settling Insurer Policies, including Contribution Claims, Extra-Contractual Claims, and Indirect Claims, are hereby permanently stayed, enjoined, barred, and restrained from taking any action, directly or indirectly, to assert, enforce or attempt to assert or enforce any such Interest against the Settling Insurers or the Settling Insurer Policies:

- 1. Commencing or continuing in any manner any action or other proceeding against the Settling Insurers or the property of the Settling Insurers;
- 2. Enforcing, attaching, collecting, or recovering, by any manner or means, any judgment, award, decree or order against the Settling Insurers or the property of the Settling Insurers;
- 3. Creating, perfecting, or enforcing any lien of any kind against the Settling Insurers or the property of the Settling Insurers;

- 4. Asserting or accomplishing any setoff, right of indemnity, subrogation, contribution, or recoupment of any kind against any obligation due the Settling Insurers or the property of the Settling Insurers; and
- 5. Taking any action, in any manner, in any place whatsoever, that does not conform to, or comply with, the provisions of the Plan.

For the avoidance of doubt, this Supplemental Settling Insurer Injunction bars only the above-referenced actions against the Settling Insurers or the Settling Insurer Policies issued by LMI or Interstate. The foregoing injunctive provisions are an integral part of this Plan and are essential to its implementation.

# ARTICLE VIII NON-SETTLING INSURERS

#### 8.1 PRESERVATION OF RIGHTS AND OBLIGATIONS.

- In the event: (i) a Tort Claim is pursued in state or federal court by a Tort Claimant (a) against a Protected Party or Non-Settling Insurer, or (ii) the Trust asserts an objection to or otherwise seeks a determination of liability as to a Tort Claim, then the Protected Parties, the Trust and each Non-Settling Insurer shall retain any and all legal and factual defenses that may exist in respect to such Tort Claim and, except as set forth in this Section 8.1, all coverage defenses. The rights, duties and obligations of each Non-Settling Insurer under the Non-Settling Insurer Policies with respect to Tort Claims are not impaired, altered, reduced, or diminished by: (a) the discharge in bankruptcy of the Debtor; (b) any distribution to Tort Claimants pursuant to this Plan, the Trust Agreement, and the Trust Distribution Plan; (c) the transfer of the Protected Parties' Transferred Insurance Interests; (d) protections granted to Protected Parties and Settling Insurers under the Plan; or (e) any other provision of this Plan, the Trust Agreement, or the Trust Distribution Plan. Non-Settling Insurers retain any defenses that they would be able to raise if the claim for coverage were brought by the Diocese, the Reorganized Debtor, or any other Protected Party, except any defense (1) related to the transfer of Transferred Insurance Interests to the Trust; (2) effected by operation of bankruptcy law as a result of confirmation; or (3) based on facts determined by the Bankruptcy Court in connection with the confirmation.
- **(b)** The rights and obligations of the Diocese, the Reorganized Debtor, and other insureds and every Non-Settling Insurer under the terms of the Non-Settling Insurer Policies and at law (including without limitation any duty of an insured to cooperate) shall not be affected by the assessment of any Tort Claim, and shall be treated as if the Tort Claim had never been assessed for distribution purposes by the Trust.
- (c) Each Non-Settling Insurer shall be entitled to all rights as are provided under the terms of its Non-Settling Insurer Policies as if the Tort Claim had never been assessed for distribution purposes by the Trust.
- (d) After the Effective Date, upon consent of the Trustee, a Person may become a Settling Insurer if the Bankruptcy Court, after notice and hearing, approves the agreement between the Person, the Reorganized Debtor and the Trustee. After the Effective Date, the Trustee shall have the exclusive authority to seek approval of such agreement. Upon the Bankruptcy Court's

entry of a Final Order approving such agreement, Exhibit N shall be amended by the Trustee to include such Person. Any such Person shall have all of the rights, remedies and duties of a Settling Insurer notwithstanding that such Person originally may have been a Non-Settling Insurer under any provision of the Plan. Such rights, remedies and duties shall include the terms and conditions of the Channeling Injunction and Supplemental Settling Insurer Injunction. The Bankruptcy Court's retained jurisdiction to approve an agreement under this Article shall include jurisdiction to determine the adequacy of notice of a motion to approve such an agreement.

- **8.2 ESTIMATIONS/ASSESSMENTS ARE NOT BINDING**. Estimations of Class 3 and Class 4B Claims for purposes of voting, and the determination of qualification, assignment of points, and payment of distributions of Tort Claims under the Trust Distribution Plan:
  - (a) Shall not (i) constitute an admission of liability by any Person with respect to such Claims; (ii) have any res judicata or collateral estoppel effect on any Person; (iii) constitute a settlement, release, accord, satisfaction, or novation of such Claims; (iv) be used by any third-party as a defense to any alleged joint liability; or (v) otherwise prejudice any rights of the Trust, Protected Parties, Settling Insurers, Non-Settling Insurers, and Claimants in all other contexts or forums;
  - Shall be without prejudice to any and all rights of the Trust, the Diocese, the Reorganized Debtor, other Protected Parties, the Non-Settling Insurers and Tort Claimants in all other contexts and forums and shall not be deemed to be a determination of liability of the Diocese or a determination of whether, or the extent to which, such claim is covered under any Non-Settling Insurer Policy. The fact that a claim has been estimated for distribution purposes has no res judicata or collateral estoppel effect and is not a binding determination on any issue or the creation of a liquidated non-bankruptcy claim. The assessment by the Tort Claims Reviewer under the Trust Distribution Plan shall have no effect upon any "no action" provisions contained in any Non-Settling Insurer Policy to the extent any such provision remains enforceable by a Non-Settling Insurer under applicable law. Rather, the liability of the Diocese, the Reorganized Debtor, or any other Protected Party and the amount owed by the Diocese, the Reorganized Debtor, other Protected Parties, and any Non-Settling Insurer on any Class 3 or Class 4B Claim, shall be determined by: (i) the amount of any court judgment obtained by the Class 3 or Class 4B Claimant; or (ii) through a settlement agreement either to which such Non-Settling Insurer has consented or, if such Non-Settling Insurer has not consented, a settlement agreement which does not breach any duty of the Trust, Trustee, Diocese, the Reorganized Debtor, or any other Protected Party to the Non-Settling Insurer under the respective Non-Settling Insurer Policy or applicable law.
- **8.3 RIGHTS UNDER INSURANCE SETTLEMENT AGREEMENTS.** The rights of the parties under any Insurance Settlement Agreement shall be determined exclusively under the applicable Insurance Settlement Agreement and those provisions of the Approval Order approving such Insurance Settlement Agreement, the Plan and the Confirmation Order.
- **8.4** THE PLAN IS NEUTRAL AS TO NON-SETTLING INSURER POLICIES. For the avoidance of doubt, solely with respect to the Non-Settling Insurers, nothing in the Plan, the Trust Agreement, the Trust Distribution Plan, the Confirmation Order, or any other order of

the Bankruptcy Court to the contrary (including any other provision that purports to be preemptory or supervening or grants a release): (i) shall affect, impair, or prejudice the rights and defenses of any Non-Settling Insurer, any Protected Party, the Trust, or any other insureds under Non-Settling Insurer Policies in any manner, including any defenses to any claim for insurance; (ii) shall constitute a settlement or resolution of any Protected Party's liability to a Tort Claimant; (iii) shall in any way operate to, or have the effect of, impairing or having any res judicata, collateral estoppel, or other preclusive effect on, any party's legal, equitable, or contractual rights or obligations under any Non-Settling Insurer Policy; or (iv) shall otherwise determine the applicability or nonapplicability of any provision of any Non-Settling Insurer Policy and applicable law.

8.5 THE DIOCESE'S OBLIGATIONS SURVIVE. Notwithstanding the transfer of the Transferred Insurance Interests to the Trust, the Diocese shall not be relieved of its duties or obligations under any Non-Settling Insurer Policies (except as provided to the contrary in any subsequent Insurance Settlement Agreement), and shall continue to perform such duties as required by such Non-Settling Insurer Policies and applicable law. If the Trust asserts any claim that the Diocese has breached such duties or obligations under the Non-Settling Insurer Policies resulting in a loss of coverage, it shall give the Diocese notice and an opportunity to cure any alleged breach, and in any event, the Diocese shall not be liable for any alleged breach resulting in a loss of coverage except to the extent that (i) the breach relates to post-Effective Date conduct of the Diocese, and (ii) the Diocese willfully or intentionally fails to comply with its continuing obligations under the Non-Settling Insurer Policies. In addition, any such claim will not be automatically allowed; the Diocese will have the right to defend against such claim.

# 8.6 TRUST POWERS WITH RESPECT TO TORT CLAIMS AND NON-SETTLING INSURERS.

- (a) A Tort Claimant or the Trust, as applicable, may enter into a settlement of a Tort Claim allowed by applicable non-bankruptcy law including but not limited to settlements consistent with *Miller v. Shugart*, 316 N.W.2d 729 (Minn. 1982) or *Drake v. Ryan*, 514 N.W.2d 785 (Minn. 1994), and may enter into an arrangement with Tort Claimant's counsel providing such counsel will receive reasonable compensation from any recovery from a Non-Settling Insurer as provided in Section 4.3
- **(b)** The Trustee may use the Trust Assets to prosecute litigation against the Non-Settling Insurers.
- (c) If the Trust successfully resolves an insurance coverage dispute with a Non-Settling Insurer or otherwise receives a recovery of insurance proceeds relating to Tort Claim(s) from a Non-Settling Insurer, such proceeds shall become Trust Assets available to pay, and shall increase the amount available to pay, Tort Claims, pursuant to the Trust Distribution Plan. In such event, and on a periodic basis accumulating all such recoveries, the Trust shall make supplemental payments to Tort Claimants in accordance with the Trust Agreement and Trust Distribution Plan.

# ARTICLE IX INSURANCE POLICIES

- 9.1 CONTINUATION OF INSURANCE POLICIES. Except to the extent any Diocese Entity Insurance Policies are bought back as set forth in and pursuant to any Insurance Settlement Agreement or as otherwise provided by the terms of the Plan, all Diocese Entity Insurance Policies (including, without limitation, any Non-Implicated Insurance Policies) shall, as applicable, either be deemed assumed by the Reorganized Debtor pursuant to Sections 365, 1123(a)(5)(A), and 1123(b)(2) of the Bankruptcy Code to the extent such Diocese Entity Insurance Policy is or was an executory contract of the Diocese, or continued in accordance with its terms pursuant to Section 1123(a)(5)(A) of the Bankruptcy Code, to the extent such Diocese Entity Insurance Policy is not an executory contract of the Diocese, such that each of the parties' contractual, legal, and equitable rights under each such Diocese Entity Insurance Policy shall remain unaltered. All known Diocese Entity Insurance Policies are listed on Exhibit I. To the extent that any or all such Diocese Entity Insurance Policies are considered to be executory contracts, then the Plan shall constitute a motion to assume such Diocese Entity Insurance Policies in connection with the Plan. Subject to the occurrence of the Effective Date, the Confirmation Order shall approve such assumption pursuant to §§ 365(a), 1123(a)(5)(A), and 1123(b)(2) of the Bankruptcy Code and include a finding by the Bankruptcy Court that each such assumption is in the best interest of the Debtor, the Estate, and all parties in interest in this Chapter 11 case. Unless otherwise determined by the Bankruptcy Court pursuant to an order which becomes a Non-Appealable Order or agreed to by the parties thereto prior to the Effective Date, no payments are required to cure any defaults of the Diocese existing as of the Effective Date with respect to any Diocese Entity Insurance Policy. The Diocese reserves the right to seek rejection of any Diocese Entity Insurance Policy or other available relief prior to the Effective Date.
- **9.2 THE PLAN IS NEUTRAL AS TO NON-IMPLICATED INSURANCE POLICIES.** For the avoidance of doubt, solely with respect to the Non-Implicated Insurer, nothing in the Plan, the Trust Agreement, the Trust Distribution Plan, the Confirmation Order, or any other order of the Bankruptcy Court to the contrary (including any other provision that purports to be preemptory or supervening or grants a release): (i) shall affect, impair, or prejudice the rights and defenses of the Non-Implicated Insurer, any Protected Party, or any other insureds under Non-Implicated Insurance Policies in any manner, including any defenses to any claim for insurance; (ii) shall constitute a settlement or resolution of any Protected Party's liability to a Class 4A Claimant; (iii) shall in any way operate to, or have the effect of, impairing or having any res judicata, collateral estoppel, or other preclusive effect on, any party's legal, equitable, or contractual rights or obligations under any Non-Implicated Insurance Policy; or (iv) shall otherwise determine the applicability or nonapplicability of any provision of any Non-Implicated Insurance Policy and any such rights and obligations shall be determined under the Non-Implicated Insurance Policy and applicable law.

# ARTICLE X PROCEDURES FOR GENERAL CLAIMS ADMINISTRATION

**10.1 RESERVATION OF RIGHTS TO OBJECT TO NON-TORT CLAIMS**. Unless a Claim is expressly described as an allowed Claim pursuant to or under the Plan, or otherwise becomes an allowed Claim prior to the Effective Date, upon the Effective Date, the Reorganized

Debtor or the Trustee, as applicable, shall be deemed to have a reservation of any and all rights, Interests, and objections of the Diocese, the UCC, or the Estate to any and all Claims and motions or requests for the payment of or on account of Claims, whether administrative expense, priority, secured, or unsecured, including any and all rights, Interests and objections to the validity or amount of any and all alleged Claims, Liens, and Interests, whether under the Bankruptcy Code, other applicable law, or contract. The failure to object to any Claim in this Chapter 11 case shall be without prejudice to the Reorganized Debtor's or the Trustee's, as applicable, right to contest or otherwise defend against such Claim in the Bankruptcy Court as set forth in this Section when and if such Claim is sought to be enforced by the holder of such Claim.

OBJECTIONS TO NON-TORT CLAIMS. Prior to the Effective Date, the 10.2 Diocese shall have the authority to pursue any objection to the allowance of any non-Tort Claim. From and after the Effective Date, the Reorganized Debtor will retain responsibility for administering, disputing, objecting to, compromising, or otherwise resolving and making distributions, if any, with respect to non-Tort Claims (including those Claims that are subject to objection by the Diocese as of the Effective Date); provided, however, that nothing in this Section shall affect the right of any party-in-interest (including the Reorganized Debtor and the Trustee) to object to any non-Tort Claim to the extent such objection is otherwise permitted by the Bankruptcy Code, the Bankruptcy Rules, and this Plan. Unless otherwise provided in the Plan or by order of the Bankruptcy Court, any objections to non-Tort Claims will be filed and served not later than thirty (30) days after the later of: (i) the Effective Date, or (ii) the date such Claim is filed. Such deadline or any Bankruptcy Court approved extension thereof, may be extended upon request by the Reorganized Debtor by filing a motion without any requirement to provide notice to any Person, based upon a reasonable exercise of the Reorganized Debtor's business judgment. A motion seeking to extend the deadline to object to any Claim shall not be deemed an amendment to the Plan. Notwithstanding the foregoing, there shall be no deadline for the Reorganized Debtor to object to Class 4A or Class 6 Claims.

**DETERMINATION OF CLAIMS**. From and after the Effective Date, any Claim that is not a Tort Claim, and as to which a Proof of Claim or motion or request for payment was timely filed in this Chapter 11 case, or deemed timely filed by order of the Bankruptcy Court, may be determined and (so long as such determination has not been stayed, reversed, or amended, as to which determination (or any revision, modification, or amendment thereof) the time to appeal or seek review or rehearing has expired, (and as to which no appeal or petition for review or rehearing was filed or, if filed, remains pending)), liquidated pursuant to: (i) an order of the Bankruptcy Court; (ii) applicable bankruptcy law; (iii) agreement of the parties without the need for Bankruptcy Court approval; (iv) applicable non-bankruptcy law; or (v) the lack of (a) an objection to such Claim, (b) an application to equitably subordinate such Claim, and (c) an application to otherwise limit recovery with respect to such Claim, filed by the Diocese, the Reorganized Debtor, or any other party in interest on or prior to any applicable deadline for filing such objection or application with respect to such Claim. Any such Claim so determined and liquidated shall be deemed to be an allowed Claim for such liquidated amount and shall be satisfied in accordance with the Plan. Nothing contained in this Section shall constitute or be deemed a waiver of any Claims, rights, Interests, or Causes of Action that the Debtor or the Reorganized Debtor may have against any Person in connection with or arising out of any Claim or Claims, including any rights under 28 U.S.C. § 157.

- **10.4 NO DISTRIBUTIONS PENDING ALLOWANCE**. No payments or distributions will be made with respect to a Disputed Claim, or any portion thereof, unless and until all objections to such Disputed Claim have been settled or withdrawn or have been determined by an order which has become a Non-Appealable Order, and the Disputed Claim has become an allowed Claim.
- undue delay in the administration of the Chapter 11 case, with respect to Disputed Claims, the Diocese (if prior to the Effective Date) and the Reorganized Debtor (on and after the Effective Date), after notice and a hearing (which notice may be limited to the holder of such Disputed Claim), shall have the right to seek an order of the Bankruptcy Court or the District Court, pursuant to Section 502(c) of the Bankruptcy Code, estimating or limiting the amount of: (i) property that must be withheld from or reserved for distribution purposes on account of such Disputed Claim(s), (ii) such Claim for allowance or disallowance purposes, or (iii) such Claim for any other purpose permitted under the Bankruptcy Code; provided, however, that the Bankruptcy Court or the District Court, as applicable, shall determine: (y) whether such Claims are subject to estimation pursuant to Section 502(c) of the Bankruptcy Code, and (z) the timing and procedures for such estimation proceedings, if any, such matters being beyond the scope of the Plan.

# ARTICLE XI DISTRIBUTIONS UNDER THE PLAN

- 11.1 PAYMENT DATE. Whenever any payment or distribution to be made under the Plan shall be due on a day other than a business day, such payment or distribution shall instead be made, without interest, on the immediately following business day.
- of an allowed non-Tort Claim under the Plan is returned for lack of a current address for the holder or otherwise, the Reorganized Debtor shall file with the Bankruptcy Court the name, if known, and last known address of the holder and the reason for its inability to make payment. All allowed Claims paid as provided in this Section shall be deemed addressed to the same extent as if payment or distribution had been made to the holder of the allowed Claim with no recourse to the Reorganized Debtor or property of the Reorganized Debtor. If, after the passage of six (6) months, the payment or distribution still cannot be made, the Reorganized Debtor shall make the payment to the Trust. All allowed Claims paid as provided in this Section shall be deemed satisfied and released, with no recourse to the Reorganized Debtor or property of the Reorganized Debtor upon payment to the Trust, to the same extent as if payment or distribution has been made to the holder of the allowed Claim.
- 11.3 SETOFFS. The Reorganized Debtor or the Trustee, as applicable, may, to the extent permitted under applicable law, set off against any allowed Claim and the distributions to be made pursuant to the Plan on account of such allowed Claim, the Claims, rights and Causes of Action of any nature that the Reorganized Debtor or the Trustee, as applicable, may hold against the holder of such allowed Claim that are not otherwise waived, released or compromised in accordance with the Plan; provided, however, that neither such a setoff nor the allowance of any Claim hereunder shall constitute a waiver or release by the Reorganized Debtor or the Trustee, as

applicable, of any such Claims, rights, and Causes of Action that the Reorganized Debtor or the Trustee, as applicable, possesses against such holder.

- 11.4 NO INTEREST ON CLAIMS. Unless otherwise specifically provided for in the Plan, the Confirmation Order, or a post-petition agreement in writing between a claimant and the Diocese, the Reorganized Debtor, or the Trust, and approved by an order of the Bankruptcy Court, postpetition interest shall not accrue or be paid on any Claim, and claimants shall not be entitled to interest accruing on or after the Petition Date on any Claim. In addition, and without limiting the foregoing or any other provision of the Plan, Confirmation Order, or Trust Agreement, interest shall not accrue on or be paid on any Disputed Claim in respect of the period from the Effective Date to the date a final distribution is made when and if such Disputed Claim becomes an allowed Claim.
- 11.5 WITHHOLDING TAXES. The Reorganized Debtor and the Trust shall comply with all withholding and reporting requirements imposed by any federal, state, local, or foreign taxing authority, and all distributions hereunder shall be subject to any such withholding and reporting requirements. As a condition to making any distribution under the Plan, the Reorganized Debtor and the Trust may require that the holder of an allowed Claim provide such holder's taxpayer identification number and such other information and certification as may be deemed necessary to comply with applicable tax reporting and withholding laws.

# ARTICLE XII EFFECTIVENESS OF THE PLAN

- **12.1 CONDITIONS TO OCCURRENCE OF EFFECTIVE DATE**. The Plan shall not become effective unless and until each of the following conditions shall have been satisfied in full in accordance with the provisions specified below:
  - **(a) Entry of Confirmation Order**. The Confirmation Order has become a Non-Appealable Order;
  - **(b) Plan Documents.** All Schedules and Exhibits to the Plan shall have been duly completed by the Plan Proponents and filed with the Court and all agreements and releases referred to in the Plan shall have been duly executed by all parties thereto and filed with the Court, in each case in form and substance satisfactory to the Debtor, the Committee, and LMI and Interstate; and
    - (c) The Trust. The Trust shall have been formed.
- **12.2 NOTICE OF EFFECTIVE DATE**. The Plan Proponents shall file a Notice of Effective Date with the Bankruptcy Court within three (3) days after the occurrence of the Effective Date. Such notice will include all relevant deadlines put into effect by the occurrence of the Effective Date.
- **12.3 EFFECT OF NON-OCCURRENCE OF CONDITIONS.** If substantial consummation of the Plan does not occur, the Plan will be null and void in all respects and nothing contained in the Plan or the disclosure statement will: (i) constitute a waiver or release of any Claims by or against the Protected Parties or the Settling Insurers; (ii) prejudice in any manner the

rights of the Protected Parties, the Trust or the Settling Insurers; or (iii) constitute an admission, acknowledgement, offer, or undertaking by the Protected Parties or the Settling Insurers in any respect, including but not limited to, in any proceeding or case against the Debtor; or (iv) be admissible in any action, proceeding or case against the Protected Parties or Settling Insurers in any court or other forum.

# ARTICLE XIII EFFECTS OF CONFIRMATION

13.1 DISSOLUTION OF UCC. On the Effective Date, the UCC shall dissolve automatically, whereupon their members, Professionals and agents shall be released from any further duties and responsibilities in this Chapter 11 case and under the Bankruptcy Code, except that such parties shall continue to be bound by any obligations arising under confidentiality agreements, joint defense/common interest agreements (whether formal or informal), and protective orders entered during this Chapter 11 case, including any orders regarding confidentiality issued by the Bankruptcy Court or mediators, which shall remain in full force and effect according to their terms, provided that such parties shall continue to have a right to be heard with respect to any and all applications for Professional Claims.

### 13.2 DISCHARGE AND INJUNCTION.

Except as otherwise expressly provided in the Plan or in the Confirmation Order, on the Effective Date, the Diocese will be discharged from, and its liability will be extinguished completely in respect to, any Claim and debt, whether reduced to judgment or not, liquidated or unliquidated, contingent or noncontingent, asserted or unasserted, fixed or not, matured or unmatured, disputed or undisputed, legal or equitable, known or future, based on conduct occurring before the Confirmation Date, including, without limitation, all interest, if any, on any such Claims and debts, whether such interest accrued before or after the Petition Date, and including all Claims and debts of the kind specified in Bankruptcy Code Sections 502(g), 502(h), and 502(i), whether or not a Proof of Claim is filed or is deemed filed under the Bankruptcy Code Section 501, such Claim is allowed under Bankruptcy Code Section 502, or the holder of such Claim has accepted the Plan.

Tort Claimants and the Trust shall be permitted to name the Diocese in any proceeding to resolve whether the Diocese has liability for Tort Claims and the amount of any such liability, solely for the purpose of obtaining insurance coverage from Non-Settling Insurers. The discharge hereunder does not apply to, and shall not limit in any way the obligations of Non-Settling Insurers to defend and pay, the Diocese's liability for Tort Claims under Non-Settling Insurer Policies. The limitations otherwise set forth in the Plan on a Tort Claimant's recovery will not in any way limit the Non-Settling Insurers' obligations under the Non-Settling Insurer Policies or the Tort Claimants' and/or Trust's recoveries against the Non-Settling Insurers for the Non-Settling Insurers' conduct in connection with the defense or settlement of a Tort Claim, including on any judgments in excess of the limits of a Non-Settling Insurer Policy.

13.3 CHANNELING INJUNCTION. Channeling Injunction Preventing Prosecution of Channeled Claims Against Protected Parties and Settling Insurers.

- (a) In consideration of the undertakings of the Protected Parties and Settling Insurers under the Plan, their contributions to the Trust, and other consideration, and pursuant to their respective settlements with the Debtor and to further preserve and promote the agreements between and among the Protected Parties and any Settling Insurers, and pursuant to Section 105 of the Bankruptcy Code:
  - 1. any and all Channeled Claims are channeled into the Trust and shall be treated, administered, determined, and resolved under the procedures and protocols and in the amounts as established under the Plan and the Trust Agreement as the sole and exclusive remedy for all holders of Channeled Claims; and
  - 2. all Persons who have held or asserted, hold or assert, or may in the future hold or assert any Channeled Claims are hereby permanently stayed, enjoined, barred and restrained from taking any action, directly or indirectly, for the purposes of asserting, enforcing, or attempting to assert or enforce any Channeled Claim against the Protected Parties or Settling Insurers, including:
    - (i) commencing or continuing in any manner any action or other proceeding of any kind with respect to any Channeled Claim against any of the Protected Parties or Settling Insurers or against the property of any of the Protected Parties or Settling Insurers;
    - (ii) enforcing, attaching, collecting or recovering, by any manner or means, from any of the Protected Parties or Settling Insurers, or the property of any of the Protected Parties or Settling Insurers, any judgment, award, decree, or order with respect to any Channeled Claim against any of the Protected Parties, Settling Insurers, or any other Person;
    - (iii) creating, perfecting or enforcing any lien of any kind relating to any Channeled Claim against any of the Protected Parties or the Settling Insurers, or the property of the Protected Parties or the Settling Insurers;
    - (iv) asserting, implementing or effectuating any Channeled Claim of any kind against:
      - 1. any obligation due any of the Protected Parties or Settling Insurers;
      - 2. any of the Protected Parties or Settling Insurers; or
      - 3. the property of any of the Protected Parties or Settling Insurers.

- (v) taking any act, in any manner, in any place whatsoever, that does not conform to, or comply with, the provisions of the Plan; and
- (vi) asserting or accomplishing any setoff, right of indemnity, subrogation, contribution, or recoupment of any kind against an obligation due to any of the Protected Parties, the Settling Insurers, or the property of any of the Protected Parties or the Settling Insurers.

Notwithstanding anything to the contrary in this Article or the Plan, Tort Claimants and the Trust shall be permitted to name the Diocese and any other Protected Party in any proceeding to resolve whether the Diocese or such other Protected Party has liability for a Tort Claim, and the amount of any such liability, for the purpose of obtaining insurance coverage from Non-Settling Insurers under the Non-Settling Insurer Policies, but recourse is limited to the proceeds of Non-Settling Insurer Policies and all other damages (including extra-contractual damages), awards, judgments in excess of policy limits, penalties, punitive damages and attorney's fees and costs that may be recoverable against any Non-Settling Insurers because of their conduct concerning insurance coverage for, or defense or settlement of, any Tort Claim, and any such judgments or awards will be turned over to the Trust for handling in accordance with Sections 6.14(i) and (j).

The Channeling Injunction is an integral part of the Plan and is essential to the Plan's consummation and implementation. It is intended that the channeling of the Channeled Claims as provided in this Section 13.3 shall inure to the benefit of the Protected Parties and Settling Insurers. In a successful action to enforce the injunctive provisions of this Section in response to a willful violation thereof, the moving party may seek an award of costs (including reasonable attorneys' fees) against the non-moving party, and such other legal or equitable remedies as are just and proper, after notice and a hearing.

The Channeling Injunction and Supplemental Insurer Injunction will not be for the benefit of Non-Settling Insurers, including the Insolvent Insurers. Nothing herein will be construed to authorize a violation of any stay that is applicable in light of the pending insolvency proceedings with respect to such Insolvent Insurers.

13.4 EXCULPATION; LIMITATION OF LIABILITY. From and after the Effective Date, none of the Exculpated Parties shall have or incur any liability for, and each Exculpated Party shall be released from, any Claim, Cause of Action or liability to any other Exculpated Party, to any holder of a Claim, or to any other party in interest, for any act or omission that occurred during and in connection with this Chapter 11 case or in connection with the preparation and filing of this Chapter 11 case, the formulation, negotiation, or pursuit of confirmation of the Plan, the consummation of the Plan, and the administration of the Plan or the property to be distributed under the Plan, except for Claims, Causes of Action or liabilities arising from the gross negligence, willful misconduct, fraud, or breach of the fiduciary duty of loyalty of any Exculpated Party, in each case subject to determination of such by Non-Appealable Order of a court of competent jurisdiction and provided that any Exculpated Party shall be entitled to reasonably rely upon the advice of counsel with respect to its duties and responsibilities (if any) under the Plan. Without limiting the generality of

the foregoing, the UCC and the Diocese and their respective officers, board and committee members, employees, attorneys, financial advisors, and other Professionals shall be entitled to and granted the benefits of Section 1125(e) of the Bankruptcy Code and the Channeling Injunction.

- 13.5 TIMING. The injunctions, releases, and discharges to which any Settling Insurer is entitled pursuant to such Insurance Settlement Agreement, the Plan, the Confirmation Order, the Approval Orders, and the Bankruptcy Code shall only become effective when the Trust receives payment in full from the corresponding Settling Insurer pursuant to the terms of such Settling Insurer's Insurance Settlement Agreement, and the other conditions to effectiveness of the Insurance Settlement Agreement are fully met.
- 13.6 NO BAR ON CERTAIN CLAIMS. Notwithstanding the foregoing, nothing in this Plan shall be construed to bar (a) a Claim based on Abuse against a Person who is not a Protected Party or a Settling Insurer (b) a Claim by such Person for insurance coverage in connection with a Claim described in the foregoing clause under an insurance policy other than the Settling Insurer Policies; or (c) a Tort Claim against a Protected Party that is not released or enjoined under the Plan, to the extent set forth herein.

### ARTICLE XIV INCORPORATION OF CHILD PROTECTION PROTOCOLS

**14.1 CHILD PROTECTION PROTOCOLS**. The Child Protection Protocols are incorporated into the Plan.

# ARTICLE XV THE REORGANIZED DEBTOR

- 15.1 CONTINUED CORPORATE EXISTENCE. The Diocese will, as the Reorganized Debtor, continue to exist after the Effective Date as a separate entity in accordance with Minn. Stat. Section 315.16 having tax-exempt status under 26 U.S.C. § 501(c)(3) under applicable law and without prejudice to any right to alter or terminate such existence under applicable state law, except as such rights may be limited and conditioned by the Plan and the documents and instruments executed and delivered in connection therewith.
- 15.2 VESTING OF ASSETS. In accordance with Sections 1141 and 1123(a)(5) of the Bankruptcy Code, and except as otherwise provided in the Plan or the Confirmation Order, the Reorganization Assets shall vest in the Reorganized Debtor (or such other entity or entities specified by the Debtor in a Supplemental Plan Document, and subject to approval by the Bankruptcy Court at the confirmation hearing) on the Effective Date free and clear of all liens, Claims, and Interests of creditors, including successor liability Claims. On and after the Effective Date, the Reorganized Debtor may operate and manage its affairs and may use, acquire, and dispose of property without notice to any Person, and without supervision or approval by the Bankruptcy Court and free of any restrictions imposed by the Bankruptcy Code, Bankruptcy Rules, or the Bankruptcy Court, other than those restrictions expressly imposed by the Plan or the Confirmation Order. The Reorganized Debtor is expressly authorized, as of the Effective Date, to

transfer legal title to the Inadvertently Titled Real Property to the respective equitable title owners of such properties without further order of the Bankruptcy Court.

- 15.3 IDENTITY OF OFFICERS OF REORGANIZED DEBTOR. In accordance with § 1129(a)(5) of the Bankruptcy Code, the identities and affiliations of the Persons proposed to serve as the corporate Members of the Reorganized Debtor and the persons proposed to serve as directors and officers of the Reorganized Debtor on and after the Effective Date are set forth on Exhibit J.
- **15.4 FURTHER AUTHORIZATION**. The Reorganized Debtor shall be entitled to seek such orders, judgments, injunctions, rulings, and other assistance as it deems necessary to carry out the intentions and purposes, and to give full effect to the provisions, of the Plan.

### ARTICLE XVI MISCELLANEOUS PROVISIONS

#### 16.1 RETENTION OF JURISDICTION.

- (a) By the Bankruptcy Court. Pursuant to Sections 105, 1123(a)(5), and 1142(b) of the Bankruptcy Code, and 28 U.S.C. Sections 1334 and 157, on and after the Effective Date, the Bankruptcy Court shall retain: (i) original and exclusive jurisdiction over this Chapter 11 case, (ii) original, but not exclusive, jurisdiction to hear and determine all core proceedings arising under the Bankruptcy Code or arising in this Chapter 11 case, and (iii) original, but not exclusive, jurisdiction to hear and make proposed findings of fact and conclusions of law in any non-core proceedings related to this Chapter 11 case and the Plan, including matters concerning the interpretation, implementation, consummation, execution, or administration of the Plan. Subject to, but without limiting the generality of the foregoing, the Bankruptcy Court's post-Effective Date jurisdiction shall include jurisdiction:
  - 1. over disputes concerning the ownership of Claims;
  - 2. over disputes concerning the distribution or retention of assets under the Plan;
  - 3. over objections to Claims, motions to allow late-filed Claims, and motions to estimate Claims;
  - 4. over proceedings to determine the extent, validity, or priority of any Lien asserted against property of the Diocese, the Estate, or Trust, or property abandoned or transferred by the Diocese, the Estate, or the Trust;
  - 5. over motions to approve any Insurance Settlement Agreements entered into after the Effective Date by the Trustee;
  - 6. over matters related to the assets of the Estate or of the Trust, including the terms of the Trust, or the recovery, liquidation, or abandonment of Trust Assets;
  - 7. the removal of the Trustee and the appointment of a successor Trustee;

- 8. over matters relating to the subordination of Claims;
- 9. to enter and implement such orders as may be necessary or appropriate in the event the Confirmation Order is for any reason stayed, revoked, modified or vacated;
- 10. to consider and approve modifications of or amendments to the Plan, to cure any defects or omissions or to reconcile any inconsistencies in any order of the Bankruptcy Court, including the Confirmation Order;
- 11. to issue orders in aid of execution, implementation, or consummation of the Plan, including the issuance of orders enforcing any and all releases and injunctions issued under or pursuant to this Plan and any Insurance Settlement Agreement;
- 12. over disputes arising from or relating to the Plan, the Confirmation Order, or any agreements, documents, or instruments executed in connection therewith;
- 13. over requests for allowance of payment of Claims entitled to priority under Sections 507(a)(2) and 503(b)(9) of the Bankruptcy Code and any objections thereto;
- 14. over all Fee Applications;
- 15. over matters concerning state, local, or federal taxes in accordance with Sections 346, 505, and 1146 of the Bankruptcy Code;
- 16. over conflicts and disputes among the Trust, the Reorganized Debtor, and holders of Claims, including holders of Class 3, Class 4A or Class 4B Claims;
- 17. over disputes concerning the existence, nature, or scope of the Diocese's discharge, including any dispute relating to any liability arising out of the termination of employment or the termination of any employee or retiree benefit program, regardless of whether such termination occurred prior to or after the Effective Date;
- 18. to issue injunctions, provide declaratory relief, or grant such other legal or equitable relief as may be necessary or appropriate to restrain interference with the Plan, the Diocese or its property, the Reorganized Debtor or its property, the Estate or its property, the Trust or its property, Trustee, the Professionals, or the Confirmation Order;
- 19. to enter a Final Decree closing the Chapter 11 case;
- 20. to enforce all orders previously entered by the Bankruptcy Court; and
- 21. over any and all other suits, adversary proceedings, motions, applications, and contested matters that may be commenced or maintained pursuant to this

Chapter 11 case or the Plan, including, but not limited to, disputes arising out of or related to the Transferred Insurance Interests or the rights and obligations of the Non-Settling Insurers on and after the Effective Date.

- **(b)** By the District Court. Pursuant to Sections 105, 1123(a)(5), and 1142(b) of the Bankruptcy Code, and 28 U.S.C. Section 1334, on and after the Effective Date, the District Court shall retain original, but not exclusive, jurisdiction to hear and determine all matters arising under the Bankruptcy Code or arising in or related to this Chapter 11 case.
- (c) Actions to Collect Amounts Owed Pursuant to the Plan. Notwithstanding anything to the contrary in this Section, the Diocese, the Reorganized Debtor and the Trustee may, but are not required to, commence an adversary proceeding to collect amounts owed pursuant to the Plan for any settlements embodied in the Plan or later approved by the Bankruptcy Court, which are not paid in accordance with this Plan. Any such action may be commenced by filing a motion in aid of confirmation with the Bankruptcy Court.
- (d) Case Closure. The existence and continued operation of the Trust shall not prevent the Bankruptcy Court from closing this Chapter 11 case. In an action involving the Trust, any costs incurred in reopening the Chapter 11 case, including any statutory fees will be paid by the Trustee from the Trust Assets in accordance with an order of the Bankruptcy Court.
- 16.2 ASSUMPTION OF EXECUTORY CONTRACTS. On the Effective Date, except for any executory contract: (i) that was previously rejected by an order of the Bankruptcy Court or otherwise pursuant to Section 365 of the Bankruptcy Code; or (ii) that is subject to a pending motion to reject before the Bankruptcy Court, and except as otherwise provided in the Plan, Confirmation Order, or Insurance Settlement Agreements, each executory contract entered into by the Debtor prior to the Petition Date that has not previously expired or terminated pursuant to its own terms, shall be assumed pursuant to Sections 365 and 1123 of the Bankruptcy Code, effective as of the Confirmation Date with no cure amount due. The Confirmation Order shall constitute an order of the Bankruptcy Court approving such assumption pursuant to Sections 365 and 1123 of the Bankruptcy Code as of the Effective Date.
- **16.3 INDEMNIFICATION OF MEMBERS, MANAGERS, OFFICERS, AND EMPLOYEES**. The obligation of the Diocese to indemnify any individual serving at any time on or prior to the Effective Date, as one of its officers, employees, council members, or volunteers by reason of such individual's service in such capacity, to the extent provided in any of the Diocese's constituent documents or by a written agreement with the Debtor or under the laws of the State of Minnesota pertaining to the Diocese, will be deemed and treated as executory contracts that are assumed by the Reorganized Debtor, pursuant to the Plan and Bankruptcy Code Section 365 as of the Effective Date, with no cure amount due. Notwithstanding the foregoing, under no circumstances will the Diocese, the Trust, or the Reorganized Debtor assume or be responsible for any alleged indemnification of any Person against whom the Diocese has determined or may, in the future, determine, that there are credible allegations of Abuse asserted against such Person or such Person has or may have engaged in some other conduct that would excuse the Reorganized Debtor from providing any indemnification to such Person.

- 16.4 DEFENSE AND INDEMNITY FOR COVERED NON-TORT CLAIMS. After the Effective Date, the Reorganized Debtor will defend and indemnify any Protected Party with respect to any Covered Non-Tort Claim and, if so required by any Insurance Settlement Agreements, will defend and indemnify the Settling Insurers with respect to any Covered Non-Tort Claims. As to any Claim against the Trust that qualifies as a Covered Non-Tort Claim, the Reorganized Debtor will also undertake on behalf of the Trust the enforcement of the injunctions set forth in Articles VII and XIII, will defend the Covered Non-Tort Claim, and, if judgment is entered on such Claim, will indemnify the Trust for any liability for such Claim. The Reorganized Debtor may not seek insurance coverage for the Claims defended or indemnified under this Section from the Settling Insurers under any Settling Insurer Policy. Nothing in this provision or any other Plan provision is intended to suggest that any Person is entitled to obtain a judgment on a Covered Non-Tort Claim or Channeled Claim, that such judgment would be covered under any Settling Insurer Policy, or that any Person is entitled to seek coverage for such judgment against any Protected Party or Settling Insurer in violation of the discharge, Channeling Injunction, or Supplemental Settling Insurer Injunction, as applicable. For the avoidance of doubt, nothing contained in this Section or the Plan is intended to provide, expand, modify or add coverage for the Diocese or any other Protected Party under any Settling Insurer Policy to cover the Diocese's indemnification of any Covered Non-Tort Claims.
- 16.5 RESERVATION OF RIGHTS. In accordance with the provisions of this Plan, the Diocese reserves the right to sell property of the Estate or compromise Causes of Action on behalf of the Estate at any time prior to the Effective Date, subject to Bankruptcy Court approval. Notice of any such sale or compromise sought as part of the Plan shall be filed as a Supplemental Plan Document, and approval of such sale or settlement shall be considered at the confirmation hearing or as soon thereafter as is practicable.
- **16.6 NON-APPEALABLE ORDER**. Except as otherwise expressly provided in the Plan, any requirement in the Plan for a Non-Appealable Order may be jointly waived by the Plan Proponents upon written notice to the Bankruptcy Court provided that Plan Proponents first obtain the consent of all Diocesan Settling Insurers.
- 16.7 AMENDMENTS AND MODIFICATIONS. The Plan Proponents may jointly modify the Plan at any time prior to the confirmation hearing in accordance with Section 1127(a) of the Bankruptcy Code. After the Confirmation Date and prior to substantial consummation, the Plan Proponents may jointly modify the Plan in accordance with Section 1127(b) of the Bankruptcy Code by filing a motion on notice as required under the applicable Bankruptcy Rules, and the solicitation of all creditors and other parties in interest shall not be required unless directed by the Bankruptcy Court. Notwithstanding any provision of this Plan to the contrary, the provisions of Article VII and Sections 13.2 through 13.4 are intended to be an integrated set of provisions that implement and supplement the Insurance Settlement Agreements that may not be severed, waived, amended, deleted, or otherwise modified without the prior written approval of all of the Settling Insurers affected by such severance, waiver, amendment, deletion, or modification.
- **16.8 U.S. TRUSTEE REPORTS**. From the Effective Date until the case is closed, the Reorganized Debtor shall, within thirty (30) days of the end of each fiscal quarter, file with the Bankruptcy Court and submit to the U.S. Trustee, quarterly reports setting forth all receipts and

disbursements as required by the U.S. Trustee guidelines. The Reorganized Debtor will not be required to file monthly operating reports or provide copies of bank account statements.

- **16.9 NO WAIVER**. The failure of the Diocese to object to any Claim for purposes of voting shall not be deemed a waiver of the Diocese's, the Reorganized Debtor's, or the Trustee's right to object to such Claim, in whole or in part.
- 16.10 TAX EXEMPTION. Pursuant to Section 1146 of the Bankruptcy Code, the delivery or recording of an instrument of transfer on or after the Confirmation Date shall be deemed to be made pursuant to and under the Plan, including any such acts by the Diocese (if prior to the Effective Date), and the Reorganized Debtor (if on or after the Effective Date), including any subsequent transfers of property by the Reorganized Debtor, including without limitation the transfer by the Reorganized Debtor of title to the Inadvertently Titled Real Property as referenced in Section 15.2 above, and shall not be taxed under any law imposing a stamp tax, transfer tax, state deed tax, or similar tax or fee. Consistent with the foregoing, each recorder of deeds or similar official for any county, city, or governmental unit in which any instrument hereunder is to be recorded shall, pursuant to the Confirmation Order and the Plan, be ordered and directed to accept such instrument, without requiring the payment of any documentary stamp, tax, deed stamps, stamp tax, transfer tax, intangible tax, or similar tax.
- **16.11 NON-SEVERABILITY**. Except as specifically provided herein, the terms of the Plan constitute interrelated compromises and are not severable, and no provision of the Plan may be stricken, altered, or invalidated, except by amendment of the Plan by the Plan Proponents.
- **16.12 REVOCATION**. The Plan Proponents reserve the right to revoke and withdraw the Plan prior to the Confirmation Date.
- 16.13 CONTROLLING DOCUMENTS. In the event and to the extent that any provision of the Plan or Trust Agreement is inconsistent with any provision of the disclosure statement, the provisions of the Plan or Trust Agreement, as applicable, shall control and take precedence. In the event and to the extent that any provision of the Trust Agreement (other than provisions relating to the Trustee's authority to act) is inconsistent with any provision of this Plan, this Plan shall control and take precedence. In the event and to the extent that any provision of the Confirmation Order is inconsistent with any provision of the Plan or the Trust Agreement, the provisions of the Confirmation Order shall control and take precedence.
- 16.14 GOVERNING LAW. Except to the extent a rule of law or procedure is supplied by federal law (including the Bankruptcy Code and the Bankruptcy Rules), and unless specifically stated, the rights, duties, and obligations arising under the Plan, any agreements, documents, and instruments executed in connection with the Plan (except as otherwise set forth in those agreements, in which case the governing law of such agreements shall control) shall be governed by, and construed and enforced in accordance with, the laws of the State of Minnesota, without giving effect to conflicts of law principles.
- **16.15 NOTICES**. Any notices or requests by parties in interest under or in connection with the Plan shall be in writing and served either by: (i) certified mail, return receipt requested,

postage prepaid, (ii) hand delivery, or (iii) reputable overnight delivery service, all charges prepaid, and shall be deemed to have been given when received by the following parties:

If to the Diocese or the Reorganized Debtor:

Diocese of Winona-Rochester c/o Andrew D. Brannon, COO/CFO 55 West Sanborn Street Winona, MN 55987

with a copy to:

Restovich Braun & Associates c/o Thomas R. Braun 117 East Center Street Rochester, MN 55904

and

Bodman PLC c/o Robert J. Diehl, Jr. 6<sup>th</sup> Floor at Ford Field 1901 St. Antoine Street Detroit, MI 48226

If to the Trust or the Trustee:

[CONTACT INFORMATION]

- 16.16 FILING OF ADDITIONAL DOCUMENTS. At any time before substantial consummation, the Diocese, the Trust, or the Reorganized Debtor, as appropriate, may file with the Bankruptcy Court or execute, as appropriate, such agreements and other documents as may be necessary or appropriate to effectuate and further evidence the terms and conditions of the Plan, or otherwise to comply with applicable law.
- **16.17 POWERS OF OFFICERS**. The officers of the Diocese or the Reorganized Debtor, as the case may be, shall have the power to enter into or execute any documents or agreements that they deem reasonable and appropriate to effectuate the terms of the Plan.
- **16.18 DIRECTION TO A PARTY**. On and after the Effective Date, the Trust or the Reorganized Debtor, as applicable, may apply to the Bankruptcy Court for entry of an order directing any Person to execute or deliver or to join in the execution or delivery of any instrument or document reasonably necessary or reasonably appropriate to effect a transfer of properties dealt with by the Plan, and to perform any other act (including satisfaction of any lien or security interest) that is reasonably necessary or reasonably appropriate for the consummation of the Plan.

- 16.19 SUCCESSORS AND ASSIGNS. The Plan shall be binding upon and inure to the benefit of the Diocese and its successors and assigns, including the Reorganized Debtor. The rights, benefits, and obligations of any entity named or referred to in the Plan shall be binding on, and shall inure to the benefit of, any heir, executor, administrator successor, or assign of such entity.
- 16.20 CERTAIN ACTIONS. By reason of entry of the Confirmation Order, prior to, on or after the Effective Date (as appropriate), all matters provided for under the Plan that would otherwise require approval of the officers of the Diocese under the Plan, including: (a) the adoption, execution, delivery, and implementation of all contracts, leases, instruments, releases, and other agreements or documents related to the Plan, and (b) the adoption, execution, and implementation of other matters provided for under the Plan involving the Diocese or organizational structure of the Diocese shall be deemed to have occurred and shall be in effect prior to, on or after the Effective Date (as appropriate), pursuant to applicable non-bankruptcy law, without any requirement of further action by the officers of the Diocese.
- **16.21 FINAL DECREE**. Once the Estate has been fully administered, as referred to in Bankruptcy Rule 3022, the Reorganized Debtor, Trustee or such other party as the Bankruptcy Court may designate in the Confirmation Order, shall file a motion with the Bankruptcy Court to obtain a Final Decree to close the Chapter 11 case.
- 16.22 PLAN AS SETTLEMENT COMMUNICATION. The Plan furnishes or offers or promises to furnish (or accepts or offers or promises to accept) valuable consideration in compromising or attempting to compromise Claims and Causes of Action that are Disputed as to validity or amount (including Tort Claims and the Insurance Litigation), except as otherwise provided above. Accordingly, the Plan, the disclosure statement, and any communications regarding the Plan or the disclosure statement are subject in all respects to Federal Rule of Evidence 408 and any comparable provision(s) of applicable state law precluding their use as evidence of liability for, or the validity or invalidity of, any Disputed Claim or Cause of Action. Nothing herein or in any confirmed Plan is intended to constitute a compromise of Tort Claims.
- 16.23 OTHER RIGHTS. Except as expressly set forth in this Plan, nothing in the Plan shall preclude any Person from asserting in any proceeding, or against any award or judgment entered in such proceeding, any and all rights that may be accorded under Minnesota law, or any other applicable statutory or common law, of contribution, indemnity, reduction, credit, or setoff, arising from the settlement and resolution of the Tort Claims.

#### ARTICLE XVII BANKRUPTCY RULE 9019 REQUEST

Pursuant to Bankruptcy Rule 9019 and through the Plan, the Plan Proponents request approval of all compromises and settlements included in the Plan.

# ARTICLE XVIII CONFIRMATION REQUEST

The Plan Proponents request confirmation of the Plan under Section 1129 of the Bankruptcy Code with respect to any impaired class that does not accept the Plan or is deemed to reject the Plan.

[Signature page for Plan of Reorganization]

THE DIOCESE OF WINONA-ROCHESTER

Very Rev. William D. Thompson

**RESTOVICH BRAUN & ASSOCIATES** 

/e/Thomas R. Braun

Thomas R. Braun (#350631) Christopher W. Coon (#390083) 117 East Center Street Rochester, MN 55904 (507) 216-8652 thomas@restovichlaw.com christopher@restovichlaw.com

and

#### **BODMAN PLC**

/e/Robert J. Diehl, Jr.

Robert J. Diehl, Jr. (P31264) Brian R. Trumbauer (P57747) Jaimee L. Witten (P70068) 6th Floor at Ford Field 1901 St. Antoine Street Detroit, Michigan 48226 (313) 259-7777 rdiehl@bodmanlaw.com btrumbauer@bodmanlaw.com jwitten@bodmanlaw.com

ATTORNEYS FOR THE DIOCESE OF WINONA-ROCHESTER

[Signature page for Plan of Reorganization]

Respectfully submitted,

# THE OFFICIAL COMMITTEE OF UNSECURED CREDITORS

/e/James W. Keenan

By: James W. Keenan Its: Chairperson

and

#### /e/Robert T. Kugler

#### STINSON, LLP

Robert T. Kugler (#194116) Edwin H. Caldie (#388930) Andrew J. Glasnovich (#398366) 50 South Sixth Street, Suite 2600 Minneapolis, MN 55402 robert.kugler@stinson.com ed.caldie@stinson.com drew.glasnovich@stinson.com

Telephone: 612-335-1500 Facsimile: 612-335-1657

ATTORNEYS FOR THE OFFICIAL COMMITTEE OF UNSECURED CREDITORS FOR THE DIOCESE OF WINONA-ROCHESTER

## **EXHIBIT A**

## **Unknown Claims Representative's Report and Recommendation**

(see attached)

Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 152 of 222

#### **EXHIBIT B**

#### **Non-Settling Insurers**

21st Century Centennial Insurance Company (f/k/a Colonial Penn Insurance Company)

CX Reinsurance Company Ltd.<sup>1</sup>

Northwestern National Insurance Company of Milwaukee, Wisconsin (f/k/a Bellefonte Insurance Co.)

Stronghold Insurance Co. Ltd.

United States Fire Insurance Company

State Farm Fire and Casualty Company

Insurance Company of North America

Liberty Mutual Insurance Company

Federated Mutual of Owatonna

The Hartford

Travelers Casualty and Surety Company (f/k/a Aetna Casualty & Surety Company)

Maryland Casualty Company

1

CX Reinsurance Company Ltd. acquired responsibility for certain policies issued to the Diocese by CNA Reinsurance of London, Ltd. and CNA International Reinsurance Co. Ltd.

## **EXHIBIT C**

## **Non-Settling Insurer Policies**

(see attached)

#### **Exhibit C**

#### Non-Settling Insurer Policies<sup>1</sup>

#### 21st Century Centennial Insurance Company (f/k/a Colonial Penn Insurance Company)

Upon information and belief, Colonial Penn Insurance Company issued at least the following policies to the Diocese:

- 1. A liability insurance policy with a policy period from July 1, 1985 to July 1, 1986, which the Diocese believes bears Policy No. XL 150023; and
- 2. A liability insurance policy with a policy period from July 1, 1986 to July 1, 1987, which the Diocese believes bears Policy No. XL 150049.

### CX Reinsurance Company Ltd.<sup>2</sup>

Upon information and belief, CNA Reinsurance of London, Ltd. and/or CNA International Reinsurance Co. Ltd. underwrote a portion of the risk covered by at least the following policies issued to the Diocese:<sup>3</sup>

- 1. A liability insurance policy with a policy period from July 1, 1982 to July 1, 1983, which the Diocese believes bears Policy No. SLC 5975;
- 2. A liability insurance policy with a policy period from July 1, 1983 to July 1, 1986, which the Diocese believes bears Policy No. ICO 4049;
- 3. A liability insurance policy with a policy period from July 1, 1986 to July 1, 1987, which the Diocese believes bears Policy No. ICO 5357;
- 4. A liability insurance policy with a policy period from July 1, 1987 to July 1, 1988, which the Diocese believes bears Policy No. ICO 5522;
- 5. A liability insurance policy with a policy period from July 1, 1988 to July 1, 1989, which the Diocese believes bears Policy No. ICO 5633;
- 6. A liability insurance policy with a policy period from July 1, 1989 to July 1, 1990, which the Diocese believes bears Policy No. ICO 5862;
- 7. A liability insurance policy with a policy period from July 1, 1990 to July 1, 1991, with an unknown policy number, but which appears to form part of Cover Note No. ISL 4569;
- 8. A liability insurance policy with a policy period from July 1, 1992 to July 1, 1993, which the Diocese believes bears Policy No. ICO 6964;
- 9. A liability insurance policy with a policy period from July 1, 1993 to July 1, 1994, which the Diocese believes bears Policy No. ICO 7315;

CORE/3515029.0002/166654194.1 Bodman\_17646317\_5

-

This Exhibit C is based on the Diocese's present knowledge, information, and belief, as informed by information and documents currently available to the Diocese. Exhibit C is meant to be illustrative only and does not in any way limit the definition of Non-Settling Insurer Policy(ies) contained in the Plan. Moreover, certain Non-Settling Insurers dispute the existence and/or terms of certain of the policies referenced herein.

<sup>2</sup> CX Reinsurance Company Ltd. acquired responsibility for certain policies issued to the Diocese by CNA Reinsurance of London, Ltd. and CNA International Reinsurance Co. Ltd.

The CX Reinsurance Company Ltd. policies are Non-Settling Insurer Policies to the extent of the risk insured by the Non-Settling Insurers.

- 10. A liability insurance policy with a policy period from July 1, 1994 to July 1, 1995, which the Diocese believes bears Policy No. ICO 7541; and
- 11. A liability insurance policy with a policy period from July 1, 1999 to July 1, 2000, which the Diocese believes bears Policy No. ICO 7893.

# Northwestern National Insurance Company of Milwaukee, Wisconsin (f/k/a Bellefonte Insurance Co.)

Upon information and belief, Bellefonte Insurance Co. underwrote a portion of the risk covered by at least the following policies issued to the Diocese:<sup>4</sup>

1. A liability insurance policy with a policy period from July 1, 1978 to July 1, 1981, which the Diocese believes bears Policy No. SLC 5421/SL 3402.

#### Stronghold Insurance Co. Ltd.

Upon information and belief, Stronghold Insurance Company, Ltd. underwrote a portion of the risk covered by at least the following policies issued to the Diocese:<sup>5</sup>

- 1. A liability insurance policy with a policy period from July 1, 1982 to July 1, 1983, which the Diocese believes bears Policy No. SLC 5975;
- 2. A liability insurance policy with a policy period from July 1, 1983 to July 1, 1986, which the Diocese believes bears Policy No. ICO 4049.

#### **United States Fire Insurance Company**

Upon information and belief, United States Fire Insurance Company issued at least the following policies to the Diocese:

- 1. A primary liability insurance policy with a policy period from July 1, 1967 to July 1, 1970, which the Diocese believes bears Policy No. ML 42768;
- 2. An excess liability insurance policy with a policy period from July 1, 1967 to July 1, 1970, which the Diocese believes bears Policy No. DCL 176017;<sup>6</sup>
- 3. A primary liability insurance policy with a policy period from July 1, 1970 to July 1, 1973, which the Diocese believes bears Policy No. ML 76766;
- 4. An excess liability insurance policy with a policy period from July 1, 1970 to July 1, 1973, which the Diocese believes bears Policy No. DCL 536095;
- 5. A primary liability insurance policy with a policy period from July 1, 1973 to July 1, 1976, which the Diocese believes bears Policy No. ML 145245;
- 6. An excess liability insurance policy with a policy period from July 1, 1973 to July 1, 1976, which the Diocese believes bears Policy No. DCL 0861307;

The Bellefonte Insurance Co. policy is a Non-Settling Insurer Policy to the extent of the risk insured by the Non-Settling Insurers.

The Stronghold Insurance Co. Ltd. policies are Non-Settling Insurer Policies to the extent of the risk insured by the Non-Settling Insurers.

<sup>&</sup>lt;sup>6</sup> US Fire disputes the existence of any excess liability insurance policies before July 1, 1976.

- 7. A primary liability insurance policy with a policy period from July 1, 1976 to July 1, 1977, which the Diocese believes bears Policy No. ML 236050;
- 8. An excess liability insurance policy with a policy period from July 1, 1976 to July 1, 1977, which the Diocese believes bears Policy No. 520-004686 5;
- 9. A primary liability insurance policy with a policy period from July 1, 1977 to July 1, 1978, which the Diocese believes bears Policy No. ML 237409; and
- 10. An excess liability insurance policy with a policy period from July 1, 1977 to July 1, 1978, which the Diocese believes bears Policy No. 520-277405 4.

#### **Insurance Company of North America**

1. Policy(ies) issued to certain parishes and/or schools located within the Diocese of Winona- Rochester.

#### **State Farm Fire & Casualty Company**

1. Liability policies issued for the policy years July 1, 1973 through June 30, 1974, July 1, 1974 through June 30, 1975, July 1, 1975 through June 30, 1976, and July 1, 1976 through June 30, 1977, to the parish of St. Charles Borromeo Catholic Church, St. Charles.

#### **Liberty Mutual Insurance Company**

- 1. A primary liability insurance policy issued by Western Casualty and Surety Co., to which Liberty Mutual is the successor, to the parish of St. Peter, Hokah.
- 2. A liability insurance policy with a policy period from March 15, 1971 to March 15, 1972, issued to the parish of St. Pius X, Rochester.

#### **Federated Mutual of Owatonna**

1. A primary liability policy issued by Federated Mutual of Owatonna to the parish of St. Peter, Hokah.

#### The Hartford

1. A primary liability policy issued by The Hartford to the parish of St. Ignatius, Spring Valley with a policy period expiring November 1952.

#### Travelers Casualty & Surety Company (f/k/a Aetna Casualty & Surety Company)

1. A primary liability policy issued by Aetna Casualty & Surety Company to the parish of Sacred Heart, Waseca.

#### **Maryland Casualty Company**

- 1. A primary liability policy issued by Maryland Casualty Company to the parish of St. Mary's Church, Winona.
- 2. A liability policy with a policy period March 1, 1968 to March 1, 1971, issued to the parish of St. Pius X, Rochester.

## **EXHIBIT D**

## **Trust Agreement and Trust Distribution Protocol**

(see attached)

## **EXHIBIT E**

## **Tort Claim Release**

(see attached)

## TO BE ENTITLED TO RECEIVE ANY COMPENSATION UNDER THE PLAN, YOU MUST EXECUTE AND DELIVER THIS RELEASE.

- 1. All capitalized terms in this Release are defined in the Plan and have the meanings stated in the Plan.
- 2. In consideration of the treatment under the Plan and the Trust Distribution Plan, and other valuable consideration, I, for myself and my heirs, successors, assigns, agents, and representatives:
  - a. Hereby fully, finally, and completely release, remise, acquit, and forever discharge the Settling Insurers with respect to the Settling Insurer Policies from any and all past, present and future Claims that, directly or indirectly, arise out of, relate to, or are connected with the i) Tort Claims; ii) Contribution Claims; iii) Extra-Contractual Claims; iv) Settling Insurer Policies; and v) all Claims that, directly or indirectly, arise from, relate to, or are connected with the Diocese's Chapter 11 case.
  - b. Hereby fully, finally, and completely release, remise, acquit, and forever discharge the Protected Parties with respect to their portion or share of liability for my Claims that do not implicate coverage under Non-Settling Insurer Policies.
  - c. Expressly reserve, and do not release, any Claims (including Tort Claims) against Protected Parties (including the Reorganized Debtor) for any Abuse that may implicate coverage under Non-Settling Insurer Policies.
  - d. Agree that, with respect to any Claims that I do not release hereunder that may implicate coverage under Non-Settling Insurer Policies, my recourse will be limited to the proceeds of the Non-Settling Insurer Policies and all other damages (including extracontractual damages), awards, judgments in excess of policy limits, penalties, punitive damages and attorney's fees and costs that may be recoverable against any Non-Settling Insurers because of their conduct concerning insurance coverage for, or defense or settlement of, any Tort Claim.
  - e. Agree that, if I recover on any judgment, award or settlement of my Tort Claim against a Protected Party for any allegations of Abuse that implicate coverage under a Non-Settling Insurer Policy, including any recovery for insurer conduct described in paragraph 2(d), the recovery will be handled in accordance with Sections 6.14(i) and (j) of the Plan and will be turned over to the Trust for handling pursuant to the terms of the Plan.
  - f. With respect to any Claims that are released under paragraph 2(a) or paragraph 2(b) of this release, covenant (i) not to sue or seek recovery or relief of any kind from the Protected Parties or Settling Insurers; (ii) to forever and irrevocably discharge that fraction, portion or percentage of damages I claim to have suffered in connection with the Abuse which is by trial or other disposition determined to be the causal fault or

responsibility, if any, of any Protected Party with respect to released Tort Claims; (iii) to voluntarily reduce any judgment that I may ever obtain against any Person relating to the same Abuse at issue in the released Tort Claims in an amount reflecting that fraction, portion or percentage of damage or injury that I suffered due to the causal fault or responsibility, if any, of any Protected Party; (iv) that filing of this Release with any court by any Protected Party will satisfy that fraction, portion or percentage of any judgment that may be rendered in my favor attributable to any Protected Party's causal fault or responsibility relating to the Abuse at issue in the released Tort Claims; (v) that I will not seek a reallocation of the causal fault or causal responsibility of any Protected Party to any other Person, whether assessed by reason of judgment or settlement, relating to the Abuse at issue in the released Tort Claims; and (vi) this Release extinguishes any potential liability of any Protected Party for contribution or indemnity to any Person who has been or may be held liable to me for any released Tort Claim.

- 3. I have been provided with copies of the Disclosure Statement, the Plan, and the exhibits thereto and have been given an opportunity to review such documents and to consult with counsel of my choice regarding those documents and this Release. The undersigned has read and understands and the undersigned's lawyer has read and explained to the undersigned, the Disclosure Statement, the Plan, and the exhibits thereto, and this Release.
- 4. I expressly reserve and retain my rights to recover from any Person for liability for any Abuse except as released under this Release, and do not intend that payment by the Trust constitutes full compensation for the damage alleged in my Tort Claim(s).
- 5. I intend the foregoing undertakings to comply with the principles set forth in *Pierringer v. Hoger*, 124 N.W.2d 106 (Wis. 1963) and *Frey v. Snelgrove*, 269 N.W.2d 918 (Minn. 1978).
- 6. I understand and agree that any payment by the Trust to me does not constitute an admission of liability of any kind or nature by the Trust or any Protected Party, Settling Insurer or Non-Settling Insurer.
- 7. I understand and agree that any payment by the Trust to me does not constitute a determination of the amount of my Tort Claim in any litigation with any Protected Parties, Non-Settling Insurers, or other Person, and that any such payment cannot be used as evidence of a Protected Party's liability for my Tort Claim, or of the amount of my damages, in any litigation on my Tort Claim. I understand that any Protected Party's liability for, and the amount of my damages on, any Tort Claim that I am reserving and not releasing under paragraph 2(b) (c) of this release, will be determined only by: (i) the amount of any court judgment obtained on my Tort Claim; or (ii) through a settlement agreement that does not breach the duty of any Protected Party to a Non-Settling Insurer under any applicable insurance policy or law.
- 8. I consent to, and agree to be bound by, the injunctions set forth in the Plan, including those injunctions contained in Articles 7.10 and 13.3 for the benefit of the Settling Insurers and those injunctions contained in Article 13.3 for the benefit of the Protected Parties.

- 9. I understand that payment from the Trust constitutes damages on account of personal physical injuries or sickness arising from an occurrence within the meaning of Section 104(a)(2) of the Internal Revenue Code of 1986, as amended.
- 10. I represent and warrant that I have not assigned or otherwise transferred any interest in my Tort Claim(s).
- 11. I hereby authorize the Center for Medicare & Medicaid Services ("CMS"), its agents and/or contractors to release, upon request, information related to my injury/illness and/or settlement since my date of birth to the Trust and/or its agents. I understand that I may revoke this "consent to release information" at any time, in writing. I consent to the release of information relating to my lifetime Medicare entitlement from the Social Security Administration and CMS to the Trustee and all other professionals retained by the Trust, and further authorize the Trustee and other Trust professionals to execute on my behalf any requests, including consents for release of information, for information relating to my Medicare entitlement and any obligations owing or potentially owing under the Medicare Secondary Payer Statute relating to my released Tort Claim(s), from the Social Security Administration and CMS. I affirm that I am the individual to whom the requested information or record applies or the authorized representative of the individual's estate. I declare under penalty of perjury (28 CFR § 16.41(d)(2004)) that I have examined all the information on this form and it is true and correct to the best of my knowledge. I understand that knowingly or willfully seeking or obtaining access to records about another person under false pretenses is punishable by a fine of up to \$5,000.
  - 12. This Release will bind my successors, heirs, assigns, agents, and representatives.

Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 163 of 222

TO BE COMPLETED BY TORT CLAIMANT OR AUTHORIZED REPRESENTATIVE OF TORT CLAIMANT'S ESTATE:

DATED:	
Name of Holder:	
Proof of Claim No. (if known):	
John/Jane Doe No. (if known):	
Signature:	
Address:	
Telephone No. (optional):	
E-mail (optional):	

## **EXHIBIT F**

## **Impaired Unknown Tort Claim Release**

(see attached)

## TO BE ENTITLED TO RECEIVE ANY COMPENSATION UNDER THE PLAN, YOU MUST EXECUTE AND DELIVER THIS RELEASE.

- 13. All capitalized terms in this Release are defined in the Plan and have the meanings stated in the Plan.
- 14. In consideration of the treatment under the Plan and the Trust Distribution Plan, and other valuable consideration, I, for myself and my heirs, successors, assigns, agents, and representatives:
  - g. Hereby fully, finally, and completely release, remise, acquit, and forever discharge the Settling Insurers with respect to the Settling Insurer Policies from any and all past, present and future Claims that, directly or indirectly, arise out of, relate to, or are connected with the i) Tort Claims; ii) Contribution Claims; iii) Extra-Contractual Claims; iv) Settling Insurer Policies; and v) all Claims that, directly or indirectly, arise from, relate to, or are connected with the Diocese's Chapter 11 case.
  - h. Hereby fully, finally, and completely release, remise, acquit, and forever discharge the Protected Parties with respect to their portion or share of liability for my Claims that do not implicate coverage under Non-Settling Insurer Policies.
  - i. Expressly reserve, and do not release, any Claims (including Tort Claims) against Protected Parties (including the Reorganized Debtor) for any Abuse that may implicate coverage under Non-Settling Insurer Policies.
  - j. Agree that, with respect to any Claims that I do not release hereunder that may implicate coverage under Non-Settling Insurer Policies, my recourse will be limited to the proceeds of the Non-Settling Insurer Policies and all other damages (including extracontractual damages), awards, judgments in excess of policy limits, penalties, punitive damages and attorney's fees and costs that may be recoverable against any Non-Settling Insurers because of their conduct concerning insurance coverage for, or defense or settlement of, any Tort Claim.
  - k. Agree that, if I recover on any judgment, award or settlement of my Tort Claim against a Protected Party for any allegations of Abuse that implicate coverage under a Non-Settling Insurer Policy, including any recovery for insurer conduct described in paragraph 2(d), the recovery will be handled in accordance with Sections 6.14(i) and (j) of the Plan and will be turned over to the Trust for handling pursuant to the terms of the Plan.
  - l. With respect to any Claims that are released under paragraph 2(a) or paragraph 2(b) of this release, covenant (i) not to sue or seek recovery or relief of any kind from the Protected Parties or Settling Insurers; (ii) to forever and irrevocably discharge that fraction, portion or percentage of damages I claim to have suffered in connection with the Abuse which is by trial or other disposition determined to be the causal fault or

responsibility, if any, of any Protected Party with respect to released Tort Claims; (iii) to voluntarily reduce any judgment that I may ever obtain against any Person relating to the same Abuse at issue in the released Tort Claims in an amount reflecting that fraction, portion or percentage of damage or injury that I suffered due to the causal fault or responsibility, if any, of any Protected Party; (iv) that filing of this Release with any court by any Protected Party will satisfy that fraction, portion or percentage of any judgment that may be rendered in my favor attributable to any Protected Party's causal fault or responsibility relating to the Abuse at issue in the released Tort Claims; (v) that I will not seek a reallocation of the causal fault or causal responsibility of any Protected Party to any other Person, whether assessed by reason of judgment or settlement, relating to the Abuse

at issue in the released Tort Claims; and (vi) this Release extinguishes any potential liability of any Protected Party for contribution or indemnity to any Person who has been or may

15. I have been provided with copies of the Disclosure Statement, the Plan, and the exhibits thereto and have been given an opportunity to review such documents and to consult with counsel of my choice regarding those documents and this Release. The undersigned has read and understands and the undersigned's lawyer has read and explained to the undersigned, the Disclosure Statement, the Plan, and the exhibits thereto, and this Release.

be held liable to me for any released Tort Claim.

- 16. I expressly reserve and retain my rights to recover from any Person for liability for any Abuse except as released under this Release, and do not intend that payment by the Trust constitutes full compensation for the damage alleged in my Tort Claim(s).
- 17. I intend the foregoing undertakings to comply with the principles set forth in *Pierringer v. Hoger*, 124 N.W.2d 106 (Wis. 1963) and *Frey v. Snelgrove*, 269 N.W.2d 918 (Minn. 1978).
- 18. I understand and agree that any payment by the Trust to me does not constitute an admission of liability of any kind or nature by the Trust or any Protected Party, Settling Insurer or Non-Settling Insurer.
- 19. I understand and agree that any payment by the Trust to me does not constitute a determination of the amount of my Tort Claim in any litigation with any Protected Parties, Non-Settling Insurers, or other Person, and that any such payment cannot be used as evidence of a Protected Party's liability for my Tort Claim, or of the amount of my damages, in any litigation on my Tort Claim. I understand that any Protected Party's liability for, and the amount of my damages on, any Tort Claim that I am reserving and not releasing under paragraph 2(b) (c) of this release, will be determined only by: (i) the amount of any court judgment obtained on my Tort Claim; or (ii) through a settlement agreement that does not breach the duty of any Protected Party to a Non-Settling Insurer under any applicable insurance policy or law.
- 20. I consent to, and agree to be bound by, the injunctions set forth in the Plan, including those injunctions contained in Articles 7.10 and 13.3 for the benefit of the Settling Insurers and those injunctions contained in Article 13.3 for the benefit of the Protected Parties.

- 21. I understand that payment from the Trust constitutes damages on account of personal physical injuries or sickness arising from an occurrence within the meaning of Section 104(a)(2) of the Internal Revenue Code of 1986, as amended.
- 22. I represent and warrant that I have not assigned or otherwise transferred any interest in my Tort Claim(s).
- 23. I hereby authorize the Center for Medicare & Medicaid Services ("CMS"), its agents and/or contractors to release, upon request, information related to my injury/illness and/or settlement since my date of birth to the Trust and/or its agents. I understand that I may revoke this "consent to release information" at any time, in writing. I consent to the release of information relating to my lifetime Medicare entitlement from the Social Security Administration and CMS to the Trustee and all other professionals retained by the Trust, and further authorize the Trustee and other Trust professionals to execute on my behalf any requests, including consents for release of information, for information relating to my Medicare entitlement and any obligations owing or potentially owing under the Medicare Secondary Payer Statute relating to my released Tort Claim(s), from the Social Security Administration and CMS. I affirm that I am the individual to whom the requested information or record applies or the authorized representative of the individual's estate. I declare under penalty of perjury (28 CFR § 16.41(d)(2004)) that I have examined all the information on this form and it is true and correct to the best of my knowledge. I understand that knowingly or willfully seeking or obtaining access to records about another person under false pretenses is punishable by a fine of up to \$5,000.
  - 24. This Release will bind my successors, heirs, assigns, agents, and representatives.

Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 168 of 222

TO BE COMPLETED BY TORT CLAIMANT OR AUTHORIZED REPRESENTATIVE OF TORT CLAIMANT'S ESTATE:

DATED:	
Name of Holder:	
Proof of Claim No. (if known):	
John/Jane Doe No. (if known):	
Signature:	
Address:	
Telephone No. (optional):	
E-mail (optional):	

#### **EXHIBIT G**

### **Inadvertently Titled Real Property**

St. Thomas Moore Student Parish 1502 Warren Street Mankato, Minnesota 56001 Blue Earth County Parcel #: R01.09.19.277.003

St. Casimir's Cemetery - Wells SW corner of 1st St. NW and Half Moon Rd. Wells, MN 56097 Faribault County Parcel #: 300081130, 300080140, 300080150

St. Mary's Cemetery - Geneva W. end of Twp. Rd. No. T-269A Ellendale, MN 56026 Freeborn County Parcel #: 160190030

Agricultural parcel on east side of street St. Mary's Geneva Ellendale, MN 56026 Freeborn County Parcel #: 160180060, 169180010

St. Patrick Cemetery east of Kubitz Rd. on County Rd. 5 northwest of New Albin, IA **Houston County** Parcel #: 070174000

Cavalry Cemetery-Caledonia West end of County Rd. 249 near S. Winnebago St. Caledonia, MN 55921 **Houston County** Parcel #: 211122000, 211127000

Sts. Peter & Paul Cemetery 22032 County Rd 18 Kellogg, MN 55945 Wabasha County

Parcel #: R06.00095.00

St. Joseph Cemetery 26185 627th St. Theilman, MN 55945 Wabasha County Parcel #: R16.00339.00, R16.00390.00, R16.00385.00

St. Joseph's Church (closed) 62683 261St. Ave. Theilman, MN 55945 Wabasha County Parcel #: R16.00375.03

Condominium 600 4th St. SW, #535 Rochester, MN 55902 Olmsted County Parcel #: 64.02.21.064175

Holy Redeemer Cemetery 2451 110th Ave SE Eyota, MN 55934 Olmsted County Parcel #: 62.16.14.032722

Holy Spirit Parish (parsonage) 4916 Wild Grass Ln NW Rochester, MN 55901 Olmsted County Parcel #: 74.17.22.055483

St. Bernard Cemetery - Stewartville 460 10th St. NW Stewartville, MN 55976 Olmsted County Parcel #: 54.34.22.030559

Corpus Christi Church 9052 66th St. NW Deerfield MN 55049 Steele County Parcel #: 130083301 St. Joachim Cemetery 10th St. NW and 7th Ave. NW Plainview, MN 55964 Wabasha County Parcel #: R26.00003.00

St. Agnes Catholic Church 135 W. Belvidere Kellogg, MN 55945 Wabasha County Parcel #: R21.00017.00, R21.00018.03

All Saint's Church 307 1st St SW New Richland, MN 56072 Waseca County Parcel #: 157770180

Calvary Cemetery - Waseca NE Corner of 336th Ave. & 128th St. Waseca, MN 56093 Waseca County Parcel #: 120301000

St. Mary's Cemetery - Preble 18441 455th Ave. Preble Township, MN 55345 Fillmore County Parcel #: 030265010

St. Kilian Cemetery - Wykoff 19675 County Rd. 117 Wykoff, MN 55990 Fillmore County Parcel #: 290333010 (1/2 portion of parcel)

St. Nicholas Cemetery - Freeburg N of residence located at 5588 County Rd. 249 approx. 6 miles E of Caledonia, MN Houston County Parcel #: 040179000 (1/5 portion of parcel) Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 172 of 222

St. Patrick Cemetery - Ridgeway S corner of County Rd. 8 and County Rd. 11 1.5 miles S of Ridgeway, MN Winona County Parcel #:100002120

Ss. Peter and Paul Cemetery - Hart 24898 Hartwood Dr. appox. 5 miles N of Rushford, MN Winona County Parcel #:040000530

#### **EXHIBIT H**

#### **List of Current Parishes**

The Church of the Sacred Heart of Adams, Minnesota 412 W Main St. Adams, MN 55909-0352

The Church of St. Adrian of Adrian, Minnesota 512 Maine Ave Adrian, MN 56110-0475

The Church of St. Theodore of Albert Lea, Minnesota 315 E. Clark St. Albert Lea, MN 56007-0604

St. Anthony of Altura, Minnesota 119 1st St. NE Altura, MN 55910

Church of Queen of Angels of Austin, Minnesota 1001 E Oakland Ave Austin, MN 55912-3896

The Church of St. Augustine of Austin, Minnesota 405 4th St NW Austin, MN 55912-3091

The Church of St. Edward of Austin, Minnesota 2000 West Oakland Ave Austin, MN 55912-1599

The Church of St. Columbanus of Blooming Prairie, Minnesota 114 E. Main St. Blooming Prairie, MN 55917-1427

The Church of Saints Peter and Paul of Blue Earth, Minnesota 214 S. Holland St. Blue Earth, MN 56013-1331

Church of Sacred Heart of Brewster, Minnesota 516 10th St.
Brewster, MN 56119-0187

Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 174 of 222

Saint Patrick's Church, of Brownsville 604 Adams St. Brownsville, MN 55919-0155

Church of Christ the King of Byron, Minnesota 202 4th St. NW Byron, MN 55920-1000

Church of St. Mary 513 S. Pine St. Caledonia, MN 55921-0406

Assumption, a/k/a Assumption Catholic Church 207 N. May St. Canton, MN 55922

The Church of St. Mary of Chatfield, Minnesota 405 Bench St SW Chatfield, MN 55923

Immaculate Conception Church 22032 County Rd. 18 Plainview, MN 55945

The Church of the Immaculate Heart of Mary, of Currie, Minnesota 510 Mill St Currie, MN 56123

The Church of Holy Cross of Dakota, Minnesota 180 Washington St Dakota, MN 55925

St. John Baptiste de la Salle a/k/a Church of St. John Baptists[sic] De La Salle 20 2nd St NE Dodge Center, MN 55927-0310

The Holy Family Catholic Church a/k/a Holy Family 2481 50th St.
East Chain, MN 56031-3098

Church of Our Lady of Mount Carmel of Easton, Minnesota 27 Main St.
Easton, MN 56025-0008

The Church of St. Aloysius of Elba, Minnesota 150 N. Main St. Elba, MN 55910

The Church of Holy Redeemer a/k/a Holy Redeemer 22 E. 2nd St. Eyota, MN 55934

Saint John Vianney Catholic Church of Fairmont, Minnesota 901 S. Prairie Ave. Fairmont, MN 56031-3098

The Church of St. Gabriel of Fulda, Minnesota 309 W Lake Ave Fulda, MN 56131-9402

The Church of St. Joseph of Good Thunder, Minnesota 130 N Ewing St Good Thunder, MN 56037

Church of St. Finbarr, of the County of Mower, Minnesota 504 1st St. SW Grand Meadow, MN 55936

Church of the Nativity of the Blessed Virgin Mary of Harmony, Minnesota 640 1st Ave. SW Harmony, MN 55939

Sacred Heart Church of Hayfield, Minnesota 150 2nd St. NE Hayfield, MN 55940-0027

The Church of the Sacred Heart of Heron Lake, Minnesota 321 9th St. Heron Lake, MN 56137

St. Peter a/k/a St. Peter's 34 Main St. Hokah, MN 55941-0355

St. Mary a/k/a St. Mary's 202 S. Sheridan Houston, MN 55943 The Church of St. Columba of Iona a/k/a The Church of ST. [sic] Columba of Iona, Minnesota 451 McDonnell Ave. Iona, MN 56141

The Church of the Good Shepherd of Jackson, Minnesota 311 Sverdrup Ave S Jackson, MN 56143

The Church of St. Ann of Janesville, Minnesota 307 W. 2nd St. Janesville, MN 56048-0218

St. Joseph a/k/a The Church of St. Joseph of Jasper, Minnesota 415 2nd St. E. Jasper, MN 56144-1212

St. John the Baptist 10343 640th Ave. Johnsburg, MN 55909

Holy Family Catholic Church 1904 N. Mantorville Ave. Kasson, MN 55944-0171

The Church of St. Agnes of Kellogg, Minnesota 125 Belvidere Ave. Kellogg, MN 55945

The Church of the Crucifixion of La Crescent a/k/a Crucifixion 423 S 2nd St La Crescent, MN 55947-1326

Saint Mary's Church of Lake City a/k/a St. Mary's Church of Lake City, Minnesota a/k/a Saint Mary's Church of Lake CitySpec Laws 1879 Chap 232 419 W. Lyon Ave.

Lake City, MN 55041-1649

Church of the Holy Family 201 N. Hunt Lake Crystal, MN 56055

Church of St. Mary of the Village of Lake Wilson 320 Paul Ave Lake Wilson, MN 56151

## Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 177 of 222

St. Joseph's 410 Broadway Lakefield, MN 56150

The Church of St. Patrick of Lanesboro, Minnesota 100 Ridgeview Lane South Lanesboro, MN 55949

St. Patrick a/k/a St. Patrick Catholic Church 436 W. Main St. LeRoy, MN 55951

St. Rose of Lima Catholic Church of Lewiston, Minnesota 180 S. Fremont St. Lewiston, MN 55952-0727

The Church of St. Anthony of Lismore, Minnesota 310 S 3rd Ave Lismore, MN 56155-0158

Holy Trinity Church of Litomysl 9946 SE 24th Ave. Litomysl, MN 55060

The Church of St. Catherine of Luverne a/k/a The Church of St. Catharine [sic] of Luverne, Minnesota a/k/a St. Catherine 203 E. Brown Luverne, MN 56156-1599

Church of Queen of Peace of the Village of Lyle, Minnesota 303 3rd Street Lyle, MN 55953

St. Olaf Catholic Church a/k/a St. Olaf 114 N. Locust Mabel, MN 55954

The Church of Saint Mary of Madelia, Minnesota 19 Crosby Ave NE Madelia, MN 56062-1702

All Saints Parish of Madison Lake, Inc. a/k/a All Saints Parish of Madison Lake.Inc. [sic] 605 4th St.

Madison Lake, MN 56063-0217

The Church of St. Peter and St. Paul of Mankato a/k/a The Church of St. Peter and St.Paul [sic] of Mankato, Minnesota 105 N. 5th St.

Mankato, MN 56001-4442

The Church of St. John the Baptist of Mankato 632 S Broad St Mankato, MN 56001

Church of St. Joseph the Worker 423 W. 7th St. Mankato, MN 56001-2197

St. Thomas More Catholic Newman Center Parish of Mankato, Minnesota a/k/a Newman Center, Diocese of Winona a/k/a St. Thomas More Catholic Newman Center of Mankato 1502 Warren Street
Mankato, MN 56001-4948

The Church of St. Theresa of Mapleton, Minnesota a/k/a St. Teresa 104 Silver St W Mapleton, MN 56065-0305

Ss. Peter and Paul 222 1st Ave. S. Mazeppa, MN 55956-0224

The Church of Christ the King of Medford, Minnesota 205 NW 2nd Ave Medford, MN 55049

St. Mary 424 Bennett Ave. Minneiska, MN 55910

Church of St. Paul, of the Village of Minnesota City, Minnesota 132 Anderson St.
Minnesota City, MN 55959

The Church of St. John the Baptist, of Minnesota Lake, Minnesota 100 Park St. N. Minnesota Lake, MN 56068-1399

The Church of All Saints, New Richland 307 SW First St. New Richland, MN 56072-0185 The Church of the Sacred Heart of Owatonna, Minnesota 810 S Cedar Ave.
Owatonna, MN 55060-3297

St. Joseph Church a/k/a St. Joseph 512 S Elm St Owatonna, MN 55060-3399

The Church of St. Leo of Pipestone, Minnesota 415 S. Hiawatha Ave. Pipestone, MN 56164-0036

The Church of St. Joachim of Plainview, Minnesota 900 W Broadway Plainview, MN 55964-1039

St. Columban a/k/a St. Columban's Preston a/k/a The Church of St. Columbanus of Preston, Minnesota 408 NW Preston St. Preston, MN 55965

Co-Cathedral of St. John the Evangelist 11 4th Ave SW Rochester, MN 55902-3098

Holy Spirit Catholic Church a/k/a Holy Spirit 5455 50th Ave. NW Rochester, MN 55901

Pax Christi Church of Rochester, Minnesota 4135 18th Ave NW Rochester, MN 55901-0460

Church of the Resurrection of Rochester, Minnesota 1600 11th Ave SE Rochester, MN 55904-5499

The Church of St. Francis of Assisi of Rochester, Minnesota 1114 3rd St. SE Rochester, MN 55904-7209

St. Pius X Catholic Church of Rochester, Minnesota 1315 12th Ave NW Rochester, MN 55901-1744 Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 180 of 222

The Church of the Holy Trinity, Rollingstone, Minnesota 83 Main St.
Rollingstone, MN 55969

The Church of St. Peter of Rose Creek 302 Maple St. SW Rose Creek, MN 55970

The Church of St. Joseph of Rushford, Minnesota 103 N. Mill St. Rushford, MN 55971-0577

The Church of St. Luke of Sherburne 303 S Lake St Sherburn, MN 56171

St. Bridget 2123 Cty Rd 16 SE Simpson, MN 55976

Church of St. Ann a/k/a The Church of St. Anne, Slayton, Minnesota 2747 29th St. Slayton, MN 56172-1485

The Church of St. Ignatius of Spring Valley, Minnesota 213 W. Franklin St. Spring Valley, MN 55975-1312

St. Charles Borromeo 1900 E 6th St. St. Charles, MN 55972-1426

Church of the Immaculate Conception of St. Clair, Minnesota 101 Church St. St. Clair, MN 56080-0100

The Church of St. James of St. James a/k/a THe [sic] Church of St. James of St. James, Minnesota 707 4th St. S St. James, MN 56081-1808

St. Bernard's Parish 116 4th Ave SE Stewartville, MN 55976

# Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 181 of 222

St. James 106 W Main St. Twin Lakes, MN 56007

The Church of St. Matthew of Vernon Center, Minnesota 200 Kendall St Vernon Center, MN 56090-0156

St. Felix Church of Wabasha, Minnesota 117 3rd St. W Wabasha, MN 55981-1201

St. Joseph a/k/a St. Joseph Catholic Church 225 3rd Ave N. Waldorf, MN 56091

The Church of the Sacred Heart of Waseca, Minnesota 111 4th St NW Waseca, MN 56093-2413

The Church of St. Casimir of Wells, Minnesota 320 2nd Ave. SW Wells, MN 56097-1399

The Church of St. Anthony 1153 1st Ave Westbrook, MN 56183

The Church of Our Lady of Good Counsel of Wilmont Minnesota 605 4th Ave Wilmont, MN 56185

Immaculate Conception Hwy 43 Wilson Township, MN 55987

The Church of St. Francis Xavier of Windom 548 17th St. Windom, MN 56101-1217

The Church of St. Mary of Winnebago City, Minnesota 32 1st St. NE Winnebago, MN 56098 St. Stanislaus Church of Winona a/k/a The Church of St. Stanislaus of Winona, Minnesota a/k/a Basilica of St. Stanislaus Kostka 625 E. 4th St. Winona, MN 55987

The Cathedral of the Sacred Heart 360 Main St. Winona, MN 55987

St. Casimir 624 W. Broadway Winona, MN 55987

St. John Nepomucene 560 E. Broadway Winona, MN 55987

St. Mary's Church of Winona, Minnesota 1303 W Broadway Winona, MN 55987

The Church of St. Martin of Woodstock, Minnesota 101 Smith St. N. Woodstock, MN 56186-1075

The Church of St. Mary of Worthington 1215 7th Ave. Worthington, MN 56187-2297

#### **EXHIBIT I**

### Known Diocese Entity Insurance Policies 1

#### **Workers' Compensation**

Upon information and belief, the Diocese has the following workers' compensation policies:

- 3. The Diocese is self-insured as authorized by the Minnesota Department of Commerce to self-insure for liabilities under Certificate of Authority W-007.
- 4. The Diocese has an excess workers' compensation policy issued by Berkley RE America under Policy No. WCB300021, which expires January 1, 2022.
- 5. The Diocese has an additional excess workers' compensation policy issued by the Workers Compensation Reinsurance Association under Policy No. 30238-2021, which expires January 1, 2022.

#### **General Liability**

Upon information and belief, the Diocese has the following general liability policies:

- 1. A general liability policy issued by Catholic Mutual Relief Society of America under Policy No.8933, which expires July 1, 2021; and
- 2. An excess general liability policy issued by Catholic Mutual Relief Society of America under Policy No. 8933, which expires July 1, 2021.

#### **Property**

Upon information and belief, the Diocese has the following property policies:

1. A property policy issued by Catholic Mutual Relief Society of America under Policy No.8933, which expires July 1, 2021.

#### Vehicle

Upon information and belief, the Diocese has the following vehicle policies:

- 1. A business auto policy issued by Church Mutual Insurance Company under Policy No. 0307149 09-118032, which expires July 1, 2021.
- 2. An excess automobile liability policy issued by the National Catholic Risk Retention Group under Policy No. RRG 10199-23, which expires July 1, 2021.

CORE/3515029.0002/166654194.1 Bodman\_17646317\_5

\_

This Exhibit I is based on the Diocese's present knowledge, information, and belief, as informed by information and documents currently available to the Diocese.

#### **Directors and Officers**

Upon information and belief, the Diocese has the following directors' and officers' policies:

- 1. A directors & officers policy issued by Catholic Mutual Relief Society of America under Policy No.8933, which expires July 1, 2021; and
- 2. An excess directors & officers policy issued by Catholic Mutual Relief Society of America, under Policy No.8933, which expires July 1, 2021.
- 3. An excess errors and omissions liability policy issued by The National Catholic Risk Retention Group under Policy No. RRG 10199-23, which expires July 1, 2021.

#### Crime

Upon information and belief, the Diocese has the following crime policies:

- 1. A crime policy issued by Catholic Mutual Relief Society of America under Policy No.8933, which expires July 1, 2021; and
- 2. An excess crime policy issued by Federal Insurance Company under Policy No.8235-1927, which expires July 1, 2021.

#### Other

Upon information and belief, the Diocese has the following additional insurance policies:

- 1. A Builder's Risk policy issued by Catholic Mutual Relief Society of America under Policy No.8933, which expires July 1, 2021;
- 2. An Equipment Breakdown policy issued by Travelers Indemnity Company under Policy No.8933, which expires July 1, 2021; and
- 3. A Priest/Religious policy covering both property and liability issued by Catholic Mutual Relief Society of America under Policy No. 8933, which expires July 1, 2021.
- 4. A Sexual Misconduct policy issued by Catholic Mutual Relief Society of America under Policy No. 8933, which expires July 1, 2021.
- 5. An excess sexual misconduct policy issued by The National Catholic Risk Retention Group under Policy No. RRG 10199-23, which expires July 1, 2021.
- 6. A Cyber Liability policy issued to Catholic Mutual and effected through NAS Insurance Services with Certain Underwriters at Lloyds, London under Master Policy No. 510641, which expires July 1, 2021.
- 7. An excess cyber liability policy issued by The National Catholic Risk Retention Group under Policy No. RRG 10199-23, which expires July 1, 2021.
- 8. A Foreign Casualty policy issued by AIG under Policy No. WS11009022, which expires July 1, 2021.
- A Malicious Attack Liability Coverage and Crisis Response policy issued to Catholic Mutual and effected through Aon with Certain Underwriters at Lloyds, London under Master Policy No. CMCTR2003035, which expires July 1, 2021.

## **EXHIBIT J**

## Officers and Directors of Reorganized Debtor

Name	Title	
Most Rev. John M. Quinn	Bishop and President of Diocese Civil Corporation	
Very Rev. William D. Thompson	Vicar General and Vice President of Diocese Civil	
	Corporation	
Very Rev. Glenn K. Frerichs	Chancellor, Judicial Vicar and Secretary of Diocese	
	Civil Corporation	
Mr. Andrew D. Brannon	Chief Operating Officer, Chief Financial Officer and	
	Treasurer of Diocese Civil Corporation	
Mr. Timothy McManimon	Director At-Large of Diocese Civil Corporation	

# **EXHIBIT K**

# **Child Protection Protocols**

(see attached)

# EXHIBIT K(1)

## **Appendix A to Child Protection Protocols**

(see attached)

# **EXHIBIT** L

## **Other Insured Entities**

None

#### **EXHIBIT M**

### **List of Catholic Entities**

## Parishes in the Diocese of Winona-Rochester

The Church of the Sacred Heart of Adams, Minnesota 412 W Main St. Adams, MN 55909-0352

The Church of St. Adrian of Adrian, Minnesota 512 Maine Ave Adrian, MN 56110-0475

The Church of St. Theodore of Albert Lea, Minnesota 315 E. Clark St. Albert Lea, MN 56007-0604

St. Anthony of Altura, Minnesota 119 1st St. NE Altura, MN 55910

Church of Queen of Angels of Austin, Minnesota 1001 E Oakland Ave Austin, MN 55912-3896

The Church of St. Augustine of Austin, Minnesota 405 4th St NW Austin, MN 55912-3091

The Church of St. Edward of Austin, Minnesota 2000 West Oakland Ave Austin, MN 55912-1599

The Church of St. Columbanus of Blooming Prairie, Minnesota 114 E. Main St. Blooming Prairie, MN 55917-1427

The Church of Saints Peter and Paul of Blue Earth, Minnesota 214 S. Holland St. Blue Earth, MN 56013-1331

Church of Sacred Heart of Brewster, Minnesota 516 10th St.
Brewster, MN 56119-0187

Saint Patrick's Church, of Brownsville 604 Adams St. Brownsville, MN 55919-0155

Church of Christ the King of Byron, Minnesota 202 4th St. NW Byron, MN 55920-1000

Church of St. Mary 513 S. Pine St. Caledonia, MN 55921-0406

Assumption, a/k/a Assumption Catholic Church 207 N. May St. Canton, MN 55922

The Church of St. Mary of Chatfield, Minnesota 405 Bench St SW Chatfield, MN 55923

Immaculate Conception Church 22032 County Rd. 18 Plainview, MN 55945

The Church of the Immaculate Heart of Mary, of Currie, Minnesota 510 Mill St Currie, MN 56123

The Church of Holy Cross of Dakota, Minnesota 180 Washington St Dakota, MN 55925

St. John Baptiste de la Salle a/k/a Church of St. John Baptists[sic] De La Salle 20 2nd St NE Dodge Center, MN 55927-0310

The Holy Family Catholic Church a/k/a Holy Family 2481 50th St.
East Chain, MN 56031-3098

Church of Our Lady of Mount Carmel of Easton, Minnesota 27 Main St.
Easton, MN 56025-0008

The Church of St. Aloysius of Elba, Minnesota 150 N. Main St. Elba, MN 55910

The Church of Holy Redeemer a/k/a Holy Redeemer 22 E. 2nd St. Eyota, MN 55934

Saint John Vianney Catholic Church of Fairmont, Minnesota 901 S. Prairie Ave. Fairmont, MN 56031-3098

The Church of St. Gabriel of Fulda, Minnesota 309 W Lake Ave Fulda, MN 56131-9402

The Church of St. Joseph of Good Thunder, Minnesota 130 N Ewing St Good Thunder, MN 56037

Church of St. Finbarr, of the County of Mower, Minnesota 504 1st St. SW Grand Meadow, MN 55936

Church of the Nativity of the Blessed Virgin Mary of Harmony, Minnesota 640 1st Ave. SW Harmony, MN 55939

Sacred Heart Church of Hayfield, Minnesota 150 2nd St. NE Hayfield, MN 55940-0027

The Church of the Sacred Heart of Heron Lake, Minnesota 321 9th St. Heron Lake, MN 56137

St. Peter a/k/a St. Peter's 34 Main St. Hokah, MN 55941-0355

St. Mary a/k/a St. Mary's 202 S. Sheridan Houston, MN 55943 The Church of St. Columba of Iona a/k/a The Church of ST. [sic] Columba of Iona, Minnesota 451 McDonnell Ave. Iona, MN 56141

The Church of the Good Shepherd of Jackson, Minnesota 311 Sverdrup Ave S Jackson, MN 56143

The Church of St. Ann of Janesville, Minnesota 307 W. 2nd St. Janesville, MN 56048-0218

St. Joseph a/k/a The Church of St. Joseph of Jasper, Minnesota 415 2nd St. E. Jasper, MN 56144-1212

St. John the Baptist 10343 640th Ave. Johnsburg, MN 55909

Holy Family Catholic Church 1904 N. Mantorville Ave. Kasson, MN 55944-0171

The Church of St. Agnes of Kellogg, Minnesota 125 Belvidere Ave. Kellogg, MN 55945

The Church of the Crucifixion of La Crescent a/k/a Crucifixion 423 S 2nd St La Crescent, MN 55947-1326

Saint Mary's Church of Lake City a/k/a St. Mary's Church of Lake City, Minnesota a/k/a Saint Mary's Church of Lake CitySpec Laws 1879 Chap 232 419 W. Lyon Ave.
Lake City, MN 55041-1649

Church of the Holy Family 201 N. Hunt Lake Crystal, MN 56055

Church of St. Mary of the Village of Lake Wilson 320 Paul Ave Lake Wilson, MN 56151 St. Joseph's 410 Broadway Lakefield, MN 56150

The Church of St. Patrick of Lanesboro, Minnesota 100 Ridgeview Lane South Lanesboro, MN 55949

St. Patrick a/k/a St. Patrick Catholic Church 436 W. Main St. LeRoy, MN 55951

St. Rose of Lima Catholic Church of Lewiston, Minnesota 180 S. Fremont St. Lewiston, MN 55952-0727

The Church of St. Anthony of Lismore, Minnesota 310 S 3rd Ave Lismore, MN 56155-0158

Holy Trinity Church of Litomysl 9946 SE 24th Ave. Litomysl, MN 55060

The Church of St. Catherine of Luverne a/k/a The Church of St. Catharine [sic] of Luverne, Minnesota a/k/a St. Catherine 203 E. Brown Luverne, MN 56156-1599

Church of Queen of Peace of the Village of Lyle, Minnesota 303 3rd Street Lyle, MN 55953

St. Olaf Catholic Church a/k/a St. Olaf 114 N. Locust Mabel, MN 55954

The Church of Saint Mary of Madelia, Minnesota 19 Crosby Ave NE Madelia, MN 56062-1702

All Saints Parish of Madison Lake, Inc. a/k/a All Saints Parish of Madison Lake.Inc. [sic] 605 4th St.

Madison Lake, MN 56063-0217

# Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 194 of 222

The Church of St. Peter and St. Paul of Mankato a/k/a The Church of St. Peter and St.Paul [sic] of Mankato, Minnesota 105 N. 5th St.

Mankato, MN 56001-4442

The Church of St. John the Baptist of Mankato 632 S Broad St Mankato, MN 56001

Church of St. Joseph the Worker 423 W. 7th St. Mankato, MN 56001-2197

St. Thomas More Catholic Newman Center Parish of Mankato, Minnesota a/k/a Newman Center, Diocese of Winona a/k/a St. Thomas More Catholic Newman Center of Mankato 1502 Warren Street
Mankato, MN 56001-4948

The Church of St. Theresa of Mapleton, Minnesota a/k/a St. Teresa 104 Silver St W Mapleton, MN 56065-0305

Ss. Peter and Paul 222 1st Ave. S. Mazeppa, MN 55956-0224

The Church of Christ the King of Medford, Minnesota 205 NW 2nd Ave Medford, MN 55049

St. Mary 424 Bennett Ave. Minneiska, MN 55910

Church of St. Paul, of the Village of Minnesota City, Minnesota 132 Anderson St.
Minnesota City, MN 55959

The Church of St. John the Baptist, of Minnesota Lake, Minnesota 100 Park St. N. Minnesota Lake, MN 56068-1399

The Church of All Saints, New Richland 307 SW First St. New Richland, MN 56072-0185 The Church of the Sacred Heart of Owatonna, Minnesota 810 S Cedar Ave.
Owatonna, MN 55060-3297

St. Joseph Church a/k/a St. Joseph 512 S Elm St Owatonna, MN 55060-3399

The Church of St. Leo of Pipestone, Minnesota 415 S. Hiawatha Ave. Pipestone, MN 56164-0036

The Church of St. Joachim of Plainview, Minnesota 900 W Broadway Plainview, MN 55964-1039

St. Columban a/k/a St. Columban's Preston a/k/a The Church of St. Columbanus of Preston, Minnesota 408 NW Preston St. Preston, MN 55965

Co-Cathedral of St. John the Evangelist 11 4th Ave SW Rochester, MN 55902-3098

Holy Spirit Catholic Church a/k/a Holy Spirit 5455 50th Ave. NW Rochester, MN 55901

Pax Christi Church of Rochester, Minnesota 4135 18th Ave NW Rochester, MN 55901-0460

Church of the Resurrection of Rochester, Minnesota 1600 11th Ave SE Rochester, MN 55904-5499

The Church of St. Francis of Assisi of Rochester, Minnesota 1114 3rd St. SE Rochester, MN 55904-7209

St. Pius X Catholic Church of Rochester, Minnesota 1315 12th Ave NW Rochester, MN 55901-1744 Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 196 of 222

The Church of the Holy Trinity, Rollingstone, Minnesota 83 Main St.
Rollingstone, MN 55969

The Church of St. Peter of Rose Creek 302 Maple St. SW Rose Creek, MN 55970

The Church of St. Joseph of Rushford, Minnesota 103 N. Mill St. Rushford, MN 55971-0577

The Church of St. Luke of Sherburne 303 S Lake St Sherburn, MN 56171

St. Bridget 2123 Cty Rd 16 SE Simpson, MN 55976

Church of St. Ann a/k/a The Church of St. Anne, Slayton, Minnesota 2747 29th St. Slayton, MN 56172-1485

The Church of St. Ignatius of Spring Valley, Minnesota 213 W. Franklin St. Spring Valley, MN 55975-1312

St. Charles Borromeo 1900 E 6th St. St. Charles, MN 55972-1426

Church of the Immaculate Conception of St. Clair, Minnesota 101 Church St. St. Clair, MN 56080-0100

The Church of St. James of St. James a/k/a THe [sic] Church of St. James of St. James, Minnesota 707 4th St. S St. James, MN 56081-1808

St. Bernard's Parish 116 4th Ave SE Stewartville, MN 55976

# Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 197 of 222

St. James 106 W Main St. Twin Lakes, MN 56007

The Church of St. Matthew of Vernon Center, Minnesota 200 Kendall St Vernon Center, MN 56090-0156

St. Felix Church of Wabasha, Minnesota 117 3rd St. W Wabasha, MN 55981-1201

St. Joseph a/k/a St. Joseph Catholic Church 225 3rd Ave N. Waldorf, MN 56091

The Church of the Sacred Heart of Waseca, Minnesota 111 4th St NW Waseca, MN 56093-2413

The Church of St. Casimir of Wells, Minnesota 320 2nd Ave. SW Wells, MN 56097-1399

The Church of St. Anthony 1153 1st Ave Westbrook, MN 56183

The Church of Our Lady of Good Counsel of Wilmont Minnesota 605 4th Ave Wilmont, MN 56185

Immaculate Conception Hwy 43 Wilson Township, MN 55987

The Church of St. Francis Xavier of Windom 548 17th St. Windom, MN 56101-1217

The Church of St. Mary of Winnebago City, Minnesota 32 1st St. NE Winnebago, MN 56098 Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 198 of 222

St. Stanislaus Church of Winona a/k/a The Church of St. Stanislaus of Winona, Minnesota a/k/a Basilica of St. Stanislaus Kostka 625 E. 4th St. Winona, MN 55987

The Cathedral of the Sacred Heart 360 Main St. Winona, MN 55987

St. Casimir 624 W. Broadway Winona, MN 55987

St. John Nepomucene 560 E. Broadway Winona, MN 55987

St. Mary's Church of Winona, Minnesota 1303 W Broadway Winona, MN 55987

The Church of St. Martin of Woodstock, Minnesota 101 Smith St. N. Woodstock, MN 56186-1075

The Church of St. Mary of Worthington 1215 7th Ave.
Worthington, MN 56187-2297

#### **Closed Parishes in the Diocese of Winona-Rochester**

St. Rose of Lima – Closed Avoca, MN 56114 Murray County

Our Lady of Loretto – Merged into Queen of Angels Catholic Church, Austin 404 W. Main Street
Brownsdale, MN 55918
Mower County

St. John the Baptist – Closed Caledonia, MN 55921 Houston County St. Peter – Closed Caledonia, MN 55921 Houston County

St. Ligouri – Closed Carimona, MN 55965 Fillmore, County

St. Francis de Sales – Closed 326 2nd St Claremont, MN 55924 Dodge County

St. John's Catholic Church – Closed Danville, MN 94506 Blue Earth County

Mater Delorosa– Closed Delavan, MN 56023 Faribault County

Corpus Christi – Closed 8914 NW 66th St Deerfield, MN 56093 Steele County

St. Mary Catholic Church – Closed Dundee, MN 56131 Nobles County

St. Mary – Merged into Church of St. Catherine of Luverne, Minnesota 204 N. Broadway St. SW Ellsworth, MN 56129
Nobles County

The Church of St. Aidan – Closed 218 5th Ave W Ellendale, MN 56026 Steele County

St. Lawrence O'Toole Catholic Church – Closed Fountain, MN 55935 Fillmore, County Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 200 of 222

St. Nicholas Catholic Church – Closed Freeburg, MN 55921 Houston County

The Church of St. Mary – Closed Geneva, MN Freeborn County

Our Lady of Mercy – Closed Guckeen, MN 56013 Faribault County

St. Clement Catholic Church – Closed Hammond, MN 55991 Wabasha County

Ss. Peter and Paul Catholic Church – Closed Hart, MN 55952 Winona County

St. Jarlath – Closed Iosco, MN 56048 Waseca County

St. Augustine – Closed Jeffers, MN 56145 Cottonwood County

St. Patrick – Closed Jefferson, MN 55921 Houston County

St. Olaf – Closed Kasson, MN 55944 Dodge County

St. Mary – Closed Kenneth, MN 56147 Rock County

Precious Blood – Closed La Moille, MN 55987 Winona County Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 201 of 222

St. Margaret – Closed Mantorville, MN 55955 Dodge County

St. John the Baptist – Closed Minneota, MN 56264 Jackson County

St. Wenceslaus – Closed Moravia, MN 55060 Steele County

Immaculate Conception— Closed Oak Ridge, MN 55910 Winona County

St. James – Closed Oakwood, MN 55957 Wabasha County

St. Hyacinth – Closed Owatonna, MN 55060 Steele County

St. Mary – Closed Preble, MN 55949 Houston County

St. Patrick – Closed Ridgeway, MN 55925 Winona County

St. Kilian – Closed 14034 Emerald Ave St. Kilian, MN 56185 Nobles County

St. Joseph – Closed Theilman, MN 55945 Wabasha County

St. Joseph – Closed Trimont, MN 56176 Martin County St. Katherine – Closed 518 E 2nd St. S. Truman, MN 56088 Martin County

St. Mary – Closed Waseca, MN 56093 Waseca County

St. Mary – Closed Wisner's Grove, MN 56025 Faribault, MN Faribault County

St. Joseph – Closed Winona, MN 55987 Winona County

St. Vincent de Paul – Closed 720 1st St West Concord, MN 55985 Dodge County

St. Thomas Pro-Cathedral – Closed Winona County

St. Kilian – Closed Wykoff, MN 55990 Fillmore County

## **Catholic Schools in the Diocese of Winona-Rochester**

Sacred Heart School 11 SW Fifth Street Adams, MN 55909

Saint Theodore School 323 E Clark Street Albert Lea, MN 56007

Pacelli Jr/Sr High School 311 NW Fourth Street Austin, MN 55912 Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 203 of 222

Pacelli Elementary School 511 NW 4th Avenue Austin, MN 55912

Saint Mary School 308 E South Street Caledonia, MN 55921

Saint John Vianney School 911 S Prairie Avenue Fairmont, MN 56031

Saint Peter School 34 Main Street Hokah, MN 55941

Crucifixion School 420 S Second Street La Crescent, MN 55947

Saint Mary School 223 NE First Street Madelia, MN 56062

Loyola High School / Loyola Primary School 145 Good Counsel Drive Mankato, MN 56001

Loyola Intermediate School 110 North Fifth St Mankato, MN 56001

Saint Mary School 730 S Cedar Avenue Owatonna, MN 55060

Noah's Ark Preschool 415 S Hiawatha Avenue Pipestone, MN 56164-0036

Holy Spirit School 5455 NW 50th Avenue Rochester, MN 55901 Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 204 of 222

Lourdes High School of Rochester, Inc. a/k/a Lourdes Catholic High School 2800 NW 19th Street Rochester, MN 55901

Saint Francis of Assisi School 318 SE 11th Avenue Rochester, MN 55904

Saint John Evangelist School 424 W Center Street Rochester, MN 55902

Saint Pius X School 1205 NW 12th Avenue Rochester, MN 55901

Saint Felix School 130 E Third Street Wabasha, MN 55981

Sacred Heart Children's House 400 NW Second Avenue Waseca, MN 56093

Sacred Heart School 308 W Elm Avenue Waseca, MN 56093

Saint Casimir School 330 Second Avenue Wells, MN 56097

Cotter Jr/Sr High School 1115 W. Broadway Street Winona, MN 55987

Winona Area Catholic Schools Inc. 1315 West Broadway Winona, MN 55987

WACS Primary School St. Mary's Educare 1315 West Broadway Winona, MN 55987 Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 205 of 222

Saint Mary's School 1206 Eighth Avenue Worthington, MN 56187

Mankato Area Catholic Schools, Inc. 145 Good Counsel Dr. Mankato, MN 56001

Rochester Catholic Schools 2800 19th St NW Rochester, MN 55901

Immaculate Heart of Mary Seminary, Inc. 750 Terrace Hts. Winona, MN 55987

Blessed Jose´ Sanchez Del Rio High School Seminary Institute of the Incarnate Word 512 E. Mulberry St. Mankato, MN 56001

### **Cemeteries in the Diocese of Winona-Rochester**

Sacred Heart Cemetery Southern terminus of Lions Street Adams, MN 55909

St Adrian Cemetery SW corner of Dillman Ave. and 250th St. Adrian, MN 56110

St Theodore Cemetery Lakewood Ave Albert Lea, MN 56007

Calvary Cemetery 4th Dr. SW, between 17th and 19th Aves. Austin, MN 55912

St Rose of Lima Cemetery 925 SW 2nd St Avoca, MN 56114

St Columbanus Cemetery (Main) 707 Hwy. Ave. S. Blooming Prairie, MN 55917

St Columbanus Cemetery (Rural) Corner of 755th St. & 115th Ave. Blooming Prairie, MN 55917

Calvary Cemetery N. side of Riverside Rd. Blue Earth, MN 56013

Sacred Heart Cemetery On Frontier Rd., Brewster, 1/2way between 11th St. and 210th St. Brewster, MN 56119

St Patrick Cemetery Cork Hollow Dr., North of County Rd. 3, W. of Brownsville Brownsville, MN 55919

Calvary Cemetery W. end of County Rd. 249, near S. Winnebago St Caledonia, MN 55921

Calvary Cemetery 12747 County Rd. 21 Canton, MN 55922

St Liguori Cemetery 22620 County. Rd. 14 Preston, MN 55965

Calvary Cemetery 33030 County Rd. 5 Chatfield, MN 55923

St. Mary Cemetery 32769 US-52 Chatfield, MN 55923

St Francis de Sales Cemetery 12872 610th Street Claremont, Minnesota 55924

Ss Peter & Paul Cemetery 22032 County Rd 18 Kellogg, MN 55945 Calvary Cemetery 510 Mill. St. Currie, MN

Holy Cross Cemetery SW of intersection of Center St. and W. Frontage Rd. Dakota, MN 55925

Corpus Christi Cemetery Corner of 755th St. & 115th Ave. Blooming Prairie, MN 55917

Calvary Cemetery 40985 180th St. Delavan, Minnesota 56023

St Mary Cemetery W. side of Whigam Ave., S. of 4th St. Dundee, MN 56131

Holy Family Cemetery 400-001 196th Street Lake Crystal, MN 56055

Our Lady of Mount Carmel Cemetery NE of Cty Rd 17 and 160th St. Barber Twp., MN, 55328

St Aloysius Cemetery Co Hwy 26 and Kieffer Dr. Elba, MN 55910

St Aidan Cemetery SE corner of 740th Ave. and 325th St. Bath, MN 56042

St Mary Cemetery Intersection of 325th St. and Ahlers Ave. Ellsworth, MN 56129

Holy Redeemer Cemetery 2451 110th Ave SE Eyota, MN 55934 Calvary Cemetery 1405 Albion Ave. Fairmont, MN 56031

Carrolton Cemetery 29057 County Rd. 8 Fountain, MN 55935

St Nicholas Cemetery N of old church, between Pleasant Valley Rd. and Crazy Corners Rd. N of Cty Rd. 249 Freeburg, MN 55921

St Gabriel Cemetery 191 240th Ave. Fulda, MN 56131

St Mary Cemetery W. end of Twp. Rd. No. T-269A Ellendale, MN 56026

St Joseph (Calvary) Cemetery 15524 555th Ave. Good Thunder, Minnesota 56037

St Finbarr Cemetery NE Corner of 730th Ave. and 255th St. Grand Meadow MN 55936

St Clement Cemetery 557 Main St E Hammond, MN 55991

Ss Peter & Paul Cemetery 24898 Hartwood Dr. Rushford, MN 55971

Sacred Heart Cemetery At intersection of 375th Ave (or County Rd 9) and 920th St. Heron Lake, MN 56137

Mount Calvary Cemetery N. of Hwy 16, between 10th St. and 8th St. Hokah, MN 55941 St Joseph Convent Cemetery NE of Hokah, N of Hwy 16 and Cty Rd 21, at N end of Pfeffer Valley Rd. Hokah, MN 55941

St Mary Cemetery N. of County Rd. 9, 1 mile E of Hwy 76 Houston, MN. 55943

St Columba Cemetery 1456 46th St. Iona, MN 56141

St Jarlath Cemetery At intersection of CR 22 and CR 35 Waterville, MN 56096

St Wenceslaus Cemetery 300 U.S. 71 Jackson, MN 56143

St Ann Cemetery 37459 35th St. Janesville, MN 56048

St Joseph Cemetery 146th St., Hwy 269 Jasper, MN 56144

St Patrick Cemetery Intersection of County Rd. 8, County Rd. 11, and County Rd. 13 Ridgeway, MN 55987

St John Cemetery County road 7 Adams, MN 55909

St Agnes Cemetery 63558 170th Ave. Kellogg, MN 55945

Crucifixion Cemetery Intersection of Cty 6 & 25 La Crescent, Minnesota 55947 Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 210 of 222

Pine Creek Cemetery Intersection of Beckman Rd. and County Rd. 6 La Crescent, MN 55947

Precious Blood Cemetery 23337 Lamoille Rd. Winona, MN 55987

St Mary Cemetery W. of Hwy 63 and Cross St. Lake City, MN 55041

Holy Family Cemetery 400-001 196th Street, Lake Crystal Lake Crystal, MN 56055

St Joseph Cemetery 44605 740th St. Lakefield, MN 56150

Holt Cemetery (St Patrick Cemetery) W. side of S. terminus of Auburn Ave Lanesboro, MN 55949

St Patrick Cemetery SW of intersection of Cemetery Rd. N. and 115th St. Leroy, MN 55951

St Rose of Lima Cemetery 925 SW 2nd St Avoca, MN 56114

St Anthony Cemetery NE corner of Dayton Ave., and 170th St. Lismore, MN

Holy Trinity Cemetery 9687 24th Ave. SE Owatonna MN 55060

St Catherine Cemetery NE portion of intersection of James St. and Hwy 75 Luverne, MN 56156 Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 211 of 222

Calvary Cemetery 73383 345th St. St. James, MN 56081

Calvary Cemetery 62286 State Highway 60 Madison Lake, MN 56063

Calvary Cemetery 200 Goodyear Avenue Mankato, MN 56001

St Margaret Cemetery 598 Cemetery Rd. Mantorville, MN 55955

Calvary Cemetery 301-020 Troendele Street Mapleton, MN 56065

Ss Peter & Paul Cemetery 1/10th of a mile N. of Oak St. NE on County Rd. 1 Mazeppa, MN 55956

Resurrection Cemetery 2913 66th St. NW Medford, MN 55049

St Patrick Cemetery (West Albany) 30932 Hwy. 60 Millville, MN 55957

St Mary Cemetery 56677 116th Ave. Altura, MN 55910

St John the Baptist Cemetery 44605 740th St. Minneota Township Minneota Township, MN

St Paul Cemetery 133 Homer Rd. Winona, MN 55987 St John the Baptist Cemetery 320 Cemetery Road Minnesota Lake (Danville Township), MN 56068

Calvary Cemetery SE corner of the intersection at 240th Ave. and 136th St. New Richland, MN, 56072

Immaculate Conception of St Mary Cemetery 15163 County Rd 31 Altura, MN 55910

Sacred Heart Cemetery 9687 24th Ave. SE Owatonna MN 55060

Seco Cemetery (St. Wenceslaus of Moravia) 5778 SW 32nd Ave. Owatonna, MN 55060

St Leo Cemetery 657 115th St. Pipestone, MN 56164

St Joachim Cemetery 10th St. NW Plainview, MN 55964

St Mary Cemetery 18441 455the Ave. Preble, MN 55954

Calvary Cemetery 22620 County. Rd. 14 Preston, MN 55965

St Patrick Cemetery Intersection of County Rd. 8, County Rd. 11, and County Rd. 13 Ridgeway, MN 55987

Calvary Cemetery 500 11th Ave. NE Rochester, MN 55906 Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 213 of 222

Holy Trinity Cemetery 83 Main St. Rollingstone, MN 55969

St Peter Cemetery 410 Maple St. SW Rose Creek, MN 55970

St Joseph Cemetery 44332 County Rd. 27 Rushford, MN 55971

St Luke Cemetery 803 120th St. Sherburn, MN 56171

St Bridget Cemetery 2003 Co. Hwy 20 Rochester, MN 55904

St Ann Cemetery 1597 101St St. Slayton, MN 56172

St Ignatius Cemetery 13665 County Rd. 12 Spring Valley, MN 55975

Calvary Cemetery 20050 25th St. SE St. Charles, MN 55972

Immaculate Conception Cemetery 15163 County Rd 31 Altura, MN 55910

St Kilian Cemetery 313 4th Ave. Wilmont, MN 56185

St Bernard Cemetery 460 10th St. NW Stewartville, MN 55976 Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 214 of 222

St Joseph Cemetery 26185 627th St. Theilman, MN

St James Cemetery E. Corner of 125th St. & 725th Ave. Twin Lakes, MN 56029

St Matthew Cemetery 52900 Firefly Rd. Vernon Center, MN 56090

St Felix Cemetery 1210 Rustic Lane E. Wabasha, MN 55981

St Joseph Cemetery Intersection of Main St. and Hwy. 83 Waldorf, MN 56091

Calvary Cemetery N. of intersection of 336th Ave. & 128th St. Waseca, MN 56093

St Mary Cemetery St Mary Cemetery Road Waseca, MN 56093

St Casimir Cemetery SW corner of 1st St. NW and Half Moon Rd. Wells, MN 56097

St Vincent de Paul Cemetery 720 1st St West Concord, MN 55985

Calvary Cemetery 22412-22000 State Hwy. 266 Wilmont, MN 56185

Immaculate Conception Cemetery 25124 MN-43 Winona, MN 55987

Case 18-33707 Doc 270 Filed 05/19/21 Entered 05/19/21 10:06:43 Desc Main Document Page 215 of 222

St Francis Xavier Cemetery Co. Rd. 26 at S. 1st St. Windom, MN 56101

St Mary Cemetery At the W. terminus of 6th Ave SW Winnebago, MN, 56098

St Mary Cemetery 1333 Homer Road Winona, MN 55987

St Martin Cemetery 1091 160th Ave. Woodstock, MN 56186

St Mary Cemetery W. Side of Read Ave, between 260th St. and E. Clary St. Worthington, MN 56187

St Kilian Cemetery 19675 County Rd. 117 Wykoff, MN 55990

### **Miscellaneous Organizations**

Catholic Charities of the Diocese of Winona-Rochester a/k/a Catholic Charities of Southern Minnesota f/k/a The Lamberton Home, Winona 111 Market Street, Suite 2 P.O. Box 379 Winona, MN 55987

Vision 2020 Education Foundation 1165 6th Street West Winona, MN 55987

St. Thomas Aquinas Catholic Newman Center Winona State University Campus Ministry 475 Huff St. Winona, MN 55987

#### Exhibit N

## **Settling Insurer Policies**<sup>1</sup>

#### **Interstate Fire & Casualty Company**

Upon information and belief, Interstate Fire & Casualty Company issued the following policies to the Diocese:

- 1. A liability insurance policy with a policy period from July 1, 1978 to July 1, 1979, which the Diocese believes bears Policy No. 183-152609, and which was renewed under the same Policy No. with a policy period July 1, 1979 to July 1, 1980;
- 2. A liability insurance policy with a policy period from July 1, 1980 to July 1, 1981, which the Diocese believes bears Policy No. 183-152609/1;
- 3. A liability insurance policy with a policy period from July 1, 1981 to July 1, 1982, which the Diocese believes bears Policy No. 183-152609/2;
- 4. A liability insurance policy with a policy period from July 1, 1982 to July 1, 1983, which the Diocese believes bears Policy No. 183-152609/3;
- 5. A liability insurance policy with a policy period from July 1, 1983 to July 1, 1984, which the Diocese believes bears Policy No. 183-0169196;
- 6. A liability insurance policy with a policy period from July 1, 1984 to July 1, 1985, which the Diocese believes bears Policy No. 183-0169196/1; and
- 7. A liability insurance policy with a policy period from July 1, 1985 to July 1, 1986, which the Diocese believes bears Policy No. 83-0172436.

#### **National Surety Corporation<sup>2</sup>**

Upon information and belief, National Surety Corporation issued the following policies to the Diocese:

1. A liability insurance policy with a policy period from July 1, 1984 to July 1, 1985, which the Diocese believes bears Policy No. 3-46 XLX-1395238.

#### **London Market Insurers<sup>3</sup>**

Upon information and belief, certain London Market Insurers issued and/or underwrote risk covered by the following policies issued to the Diocese:<sup>4</sup>

This Exhibit N is based on the Diocese's present knowledge, information, and belief, as informed by information and documents currently available to the Diocese.

National Surety Corporation is an affiliate of Interstate Fire & Casualty Company.

The term "London Market Insurers" means certain Underwriters at Lloyd's, London and certain London Market Companies (as defined in the LMI/Interstate Settlement Agreement).

The policies listed in this Exhibit N are "Settling Insurer Policies" only to the extent of the risk insured by the Settling Insurers. For the avoidance of doubt, none of the policies listed in this Exhibit N are "Settling Insurer Policies" to the extent of the risk insured by any Non-Settling Insurers, including, without limitation, (1) CNA Reinsurance of London, Ltd.; (2) CNA International Reinsurance Co. Ltd.; (3) CX Reinsurance Company Ltd.;

- 1. A liability insurance policy with a policy period from July 1, 1978 to July 1, 1980, which the Diocese believes bears Policy No. SLC 5421;
- 2. A liability insurance policy with a policy period from July 1, 1978 to July 1, 1980, which the Diocese believes bears Policy No. SL 3402;
- 3. A liability insurance policy with a policy period from July 1, 1978 to July 1, 1981, which the Diocese believes bears Policy No. SLC 5422;
- 4. A liability insurance policy with a policy period from July 1, 1978 to July 1, 1981, which the Diocese believes bears Policy No. SL 3403;
- 5. A liability insurance policy with a policy period from July 1, 1980 to July 1, 1983, which the Diocese believes bears Policy No. SLC 5527;
- 6. A liability insurance policy with a policy period from July 1, 1980 to July 1, 1983, which the Diocese believes bears Policy No. SL 3703;
- 7. A liability insurance policy with a policy period from July 1, 1982 to July 1, 1983, which the Diocese believes bears Policy No. SLC 5975;
- 8. A liability insurance policy with a policy period from July 1, 1982 to July 1, 1983, which the Diocese believes bears Policy No. SL 3986;
- 9. A liability insurance policy with a policy period from July 1, 1983 to July 1, 1986, which the Diocese believes bears Policy No. ICO 4049;
- 10. A liability insurance policy with a policy period from July 1, 1983 to July 1, 1986, which the Diocese believes bears Policy No. ISL 3089;
- 11. A liability insurance policy with a policy period from July 1, 1983 to July 1, 1986, which the Diocese believes bears Policy No. ICO 4047;
- 12. A liability insurance policy with a policy period from July 1, 1983 to July 1, 1986, which the Diocese believes bears Policy No. ISL 3087;
- 13. A liability insurance policy with a policy period from July 1, 1986 to July 1, 1987, which the Diocese believes bears Policy No. ICO 5357;
- 14. A liability insurance policy with a policy period from July 1, 1986 to July 1, 1987, which the Diocese believes bears Policy No. ISL 3520;
- 15. A liability insurance policy with a policy period from July 1, 1986 to July 1, 1987, which the Diocese believes bears Policy No. ICO 5359;
- 16. A liability insurance policy with a policy period from July 1, 1986 to July 1, 1987, which the Diocese believes bears Policy No. ISL 3522;
- 17. A liability insurance policy with a policy period from July 1, 1987 to July 1, 1988, which the Diocese believes bears Policy No. ICO 5522;
- 18. A liability insurance policy with a policy period from July 1, 1987 to July 1, 1988, which the Diocese believes bears Policy No. ISL 3822;
- 19. A liability insurance policy with a policy period from July 1, 1988 to July 1, 1989, which the Diocese believes bears Policy No. ICO 5633;

<sup>(4)</sup> Northwestern National Insurance Company of Milwaukee, Wisconsin; (5) Bellefonte Insurance Co.; (6) Stronghold Insurance Co. Ltd.; (7) United States Fire Insurance Company; (8) 21st Century Centennial Insurance Company (f/k/a Colonial Penn Insurance Company); (9) State Farm Fire and Casualty Company; (10) Insurance Company of North America; (11) Liberty Mutual Insurance Company; (12) Federated Mutual of Owatonna; (13) The Hartford; (14) Travelers Casualty and Surety Company (f/k/a Aetna Casualty & Surety Company); and/or (15) Maryland Casualty Company.

- 20. A liability insurance policy with a policy period from July 1, 1988 to July 1, 1989, which the Diocese believes bears Policy No. ISL 4060;
- 21. A liability insurance policy with a policy period from July 1, 1989 to July 1, 1990, which the Diocese believes bears Policy No. ICO 5862;
- 22. A liability insurance policy with a policy period from July 1, 1989 to July 1, 1990, which the Diocese believes bears Policy No. ISL 4282;
- 23. A liability insurance policy with a policy period from July 1, 1990 to July 1, 1991, with an unknown policy number, but which appears to form part of Cover Note No. ISL 4569;
- 24. A liability insurance policy with a policy period from July 1, 1990 to July 1, 1991, which the Diocese believes bears Policy No. ISL 4569
- 25. A liability insurance policy with a policy period from July 1, 1991 to July 1, 1992, which the Diocese believes bears Policy No. ICO 6520;
- 26. A liability insurance policy with a policy period from July 1, 1991 to July 1, 1992, which the Diocese believes bears Policy No. ISL 4907;
- 27. A liability insurance policy with a policy period from July 1, 1992 to July 1, 1993, which the Diocese believes bears Policy No. ICO 6964;
- 28. A liability insurance policy with a policy period from July 1, 1992 to July 1, 1993, which the Diocese believes bears Policy No. ISL 5286;
- 29. A liability insurance policy with a policy period from July 1, 1993 to July 1, 1994, which the Diocese believes bears Policy No. ICO 7315;
- 30. A liability insurance policy with a policy period from July 1, 1993 to July 1, 1994, which the Diocese believes bears Policy No. ISL 5605;
- 31. An additional liability insurance policy with a policy period from July 1, 1994 to July 1, 1995, with an unknown policy number, but which appears to form part of Cover Note No. No. 5833;
- 32. A liability insurance policy with a policy period from July 1, 1994 to July 1, 1995, which the Diocese believes bears Policy No. 5833;
- 33. A liability insurance policy with a policy period from July 1, 1999 to July 1, 2000, which the Diocese believes bears Policy No. ICO 7893; and
- 34. A liability insurance policy with a policy period from July 1, 1999 to July 1, 2000, which the Diocese believes bears Policy No. ISL 6304.

# **EXHIBIT B**

Order Approving the Disclosure Statement (Proposed)

(see attached)

# **EXHIBIT C**

Liquidation Analysis

(see attached)

# DIOCESE OF WINONA-ROCHESTER <u>Liquidation Analysis</u>

	Plan of Reorganization	Hypothetical Chapter 7 Liquidation <sup>1</sup>
Initial Cash Contribution from Debtor	\$13,560,000	\$13,560,000
(Less Certain Administrative Claims paid after 2/29/20) <sup>2</sup>	(\$1,961,973)	(\$1,803,428)
Additional Cash Contributions	\$7,552,500	-
Real Property	-	\$1,799,900 <sup>3</sup>
Other Non-Cash Assets	-	\$2,990,516 <sup>4</sup>
LMI/Interstate Settlement	\$6,500,000	-
Recoveries from Non-Settling Insurers <sup>5</sup>	TBD	TBD
TOTAL	\$25,650,527	\$16,546,988
Waterfall		
Total Distributable Assets	\$25,650,527	\$16,546,988
Less Administrative Claims		
Chapter 7 Administrative Expenses	N/A	\$496,409 <sup>6</sup>
Chapter 7 Trustee Fees	N/A	\$519,659 <sup>7</sup>
Net Distributable Assets	\$25,650,527	\$15,530,919
Secured Claim Distributions		
Class 7 (Minnesota Department of Commerce)	N/A – Claimant retains its lien	N/A – Claimant retains its lien
Class 8 (Community Bank of Mankato)	N/A – Claimant retains its lien	N/A – Claimant retains its lien
Unsecured Claim Distributions <sup>8</sup>		
Class 3 Tort Claims	Share of \$25,650,527 determined by Tort Claims Reviewer	Pro Rata Share of \$15,530,919
Class 4A Unimpaired Unknown Tort Claims	100% of Claims	Pro Rata Share of \$15,530,919
Class 4B Impaired Unknown Tort Claims	Share of \$25,650,527 determined by Tort Claims Reviewer	Pro Rata Share of \$15,530,919
Class 5 General Unsecured Claims	100% of Claims	Pro Rata Share of \$15,530,919
Class 6 Unimpaired Abuse Related Contingent Claims	100% of Claims	Pro Rata Share of \$15,530,919

<sup>&</sup>lt;sup>1</sup> Debtor would have to consent to conversion to Chapter 7 and would not do so because of its obligation to carry out the mission of the Catholic Church in the region that it serves.

<sup>&</sup>lt;sup>2</sup> Figure is as of February 28, 2021. The figure in the "Plan of Reorganization" column is higher because it also includes \$158,545 in counseling expenses for Tort Claimants paid prior to February 28, 2020.

<sup>&</sup>lt;sup>3</sup> Figure from Debtor's amended schedules. Does not include Holy Spirit Parish real estate; instead, note receivable from Holy Spirit Parish is included in Other Non-Cash Assets.

<sup>&</sup>lt;sup>4</sup> Unrestricted assets only; includes accounts receivable and note receivable from Holy Spirit Parish as of 2/28/21. Other amounts are from Debtor's amended schedules.

<sup>&</sup>lt;sup>5</sup> Amount of recoveries from Non-Settling Insurers is unknown. It is assumed that the amount will be the same or similar whether under the Plan or in a hypothetical Chapter 7.

<sup>&</sup>lt;sup>6</sup> Estimated at 3% of Total Distributable Assets.

<sup>&</sup>lt;sup>7</sup> Calculated under 11 U.S.C. § 326(a) based on Total Distributable Assets.

<sup>&</sup>lt;sup>8</sup> Distributions to unsecured creditors in the context of a hypothetical Chapter 7 liquidation do not reflect the expenses of adjudicating the Tort Claims, which expenses may be significant.

Bodman\_17531478\_11

# **EXHIBIT D**

LMI/Interstate Settlement Agreement

(see attached)